



Atlanta Corporate Headquarters 655 Engineering Drive Norcross, Georgia 30092 USA Phone +1-770-447-1144 Fax +1-770-447-1797 www.nacintl.com

May 23, 2000

U.S. Nuclear Regulatory Commission 11555 Rockville Pike Rockville, MD 20852-2738

Attn: Document Control Desk

Subject: Docket No. 72-1025

Submittal of NAC-MPC FSAR, Revision 0 (TAC No. L22907)

- Reference: 1. Submittal of Changed Pages for Revision 5 of the NAC-MPC SAR, NAC International, January 5, 2000
 - 2. Certificate of Compliance for the NAC International, Inc, Multi-Purpose Canister (MPC) System, United States Nuclear Regulatory Commission, March 17, 2000

In accordance with the requirements of 10 CFR 72.248(a)(1), NAC herewith provides the NAC-MPC Final Safety Analysis Report (FSAR), Revision 0, for the NAC-MPC System approved by Reference 2. This FSAR is based on the safety analysis report submitted with the initial application and includes all changes through Reference 1, which were incorporated during the review process. The FSAR reflects all of the requirements contained in Reference 2.

If you have any comments or questions, please contact me or Jim Ballowe at (770) 447-1144.

Sincerely,

.C. Shompson

Thomas C. Thompson Director, Licensing & Competitive Assessment Engineering & Design Services

cc: Joseph McCumber DE&S

前出任任正规制制



for the MAG Multi-Purpose Ganister System

ENTERNATIONAL.

Allanta Corporate Headquarters, 553 Engineering Drive, Norcross, Georgia 30.092, Phone 770-647 (194), Pax 770-457-179% www.andiail.com





Atlanta Corporate Headquarters, 655 Engineering Drive, Norcross, Georgia 30092, Phone 770-447-1144, Fax 770-447-1797, www.nacintl.com

__N 5 for the NAC Multi-Purpose Canister System

APRIL 2000

REVISION O



.

List of Effective Pages

1-1	Revision 0
1-2	Revision 0
1-3	Revision 0
1-4	Revision 0
1-5	Revision 0
1-б	Revision 0
1-7	Revision 0
1-8	Revision 0
1-9	Revision 0
1-10	Revision 0
1-11	Revision 0
1-12	Revision 0
1-13	Revision 0
1-14	Revision 0
1-15	Revision 0
1-16	Revision 0
1-17	Revision 0
1-18	Revision 0
1-19	Revision 0
1-20	Revision 0
1-21	Revision 0
1-22	Revision 0
1-23	Revision 0
1-24	Revision 0
1-25	Revision 0
1-26	Revision 0
1-27	Revision 0
1-28	Revision 0
1-29	Revision 0
1-30	Revision 0
1-31	Revision 0
1-32	Revision 0
1-33	Revision 0
1-34	Revision 0
1-35	Revision 0
1-36	Revision 0

List of Effective Pages	
1	Revision 0
2	Revision 0
3	Revision 0
4	Revision 0
5	Revision 0
6	Revision 0
7	Revision 0
8	Revision 0
9	Revision 0
10	Revision 0
11	Revision 0
12	Revision 0
13	Revision 0
14	Revision 0

Master Table of Contents

i	Revision 0
ii	Revision 0
iii	Revision 0
iv	Revision 0
v	Revision 0
vi	Revision 0
vii	Revision 0
viii	Revision 0
ix	Revision 0
x	Revision 0
xi	Revision 0
xii	Revision 0
xiii	Revision 0
xiv	Revision 0
xv	Revision 0

Chapter 1

1-i	Revision 0
1-ii	Revision 0

1

NAC-MPC FSAR Docket No. 72-1025

April 2000 Revision 0

List of Effective Pages (Continued)

1-37	Revision 0
1-38	Revision 0
1-39	Revision 0
1-40	Revision 0
1-41	Revision 0
1-42	Revision 0
1-43	Revision 0
1-44	Revision 0
1-45	Revision 0
1-46	Revision 0
1-47	Revision 0
1-48	Revision 0
1-49	Revision 0
1-50	Revision 0
1-51	Revision 0
1-52	Revision 0
1-53	Revision 0
1-54	Revision 0
1-55	Revision 0
1-56	Revision 0
1-57	Revision 0
1-58	Revision 0
1-59	Revision 0
1.1-1	Revision 0
1.1-2	Revision 0
1.1-3	Revision 0
1.1-4	Revision 0
1.1-5	Revision 0
1.1-6	Revision 0
1.2-1	Revision 0
1.2-2	Revision 0
1.2-3	Revision 0
1.2-4	Revision 0
1.2-5	Revision 0
1.2-6	Revision 0
1.2-7	Revision 0

1.2-8	Revision 0
1.2-9	Revision 0
1.2-10	Revision 0
1.2-11	Revision 0
1.2-12	Revision 0
1.2-13	Revision 0
1.2-14	Revision 0
1.2-15	Revision 0
1.2-16	Revision 0
1.2-17	Revision 0
1.2-18	Revision 0
1.2-19	Revision 0
1.2-20	Revision 0
1.2-21	Revision 0
1.2-22	Revision 0
1.2-23	Revision 0
1.2-24	Revision 0
1.2-25	Revision 0
1.3-1	Revision 0
1.3-2	Revision 0
1.4-1	Revision 0
1.4-2	Revision 0
1.5-1	Revision 0
1.5-2	Revision 0
26 drawings (see pages 1.5	-1 and 1.5-2)

Chapter 2

2-i	Revision 0
2-ii	Revision 0
2-iii	Revision 0
2-1	Revision 0
2-2	Revision 0
2-3	Revision 0
2.1-1	Revision 0
2.1-2	Revision 0
2.1-3	Revision 0

Docket No. 72-1025

April 2000 Revision 0

List of Effective Pages (Continued)

2.1-4	Revision 0
2.1-5	Revision 0
2.1-6	Revision 0
2.2-1	Revision 0
2.2-2	Revision 0
2.2-3	Revision 0
2.2-4	Revision 0
2.2-5	Revision 0
2.2-6	Revision 0
2.2-7	Revision 0
2.2-8	Revision 0
2.2-9	Revision 0
2.2-10	Revision 0
2.2-11	Revision 0
2.2-12	Revision 0
2.3-1	Revision 0
2.3-2	Revision 0
2.3-3	Revision 0
2.3-4	Revision 0
2.3-5	Revision 0
2.3-6	Revision 0
2.3-7	Revision 0
2.3-8	Revision 0
2.3-9	Revision 0
2.3-10	Revision 0
2.3-11	Revision 0
2.3-12	Revision 0
2.4-1	Revision 0
2.4-2	Revision 0
2.4-3	Revision 0
2.4-4	Revision 0

Chapter 3

3-i	Revision 0
3-ii	Revision 0
3-iii	Revision 0

3-iv	Revision 0
3-v	Revision 0
3.1-1	Revision 0
3.1-2	Revision 0
3.1-3	Revision 0
3.1-4	Revision 0
3.1-5	Revision 0
3.2-1	Revision 0
3.2-2	Revision 0
3.3-1	Revision 0
3.3-2	Revision 0
3.3-3	Revision 0
3.3-4	Revision 0
3.3-5	Revision 0
3.3-6	Revision 0
3.3-7	Revision 0
3.3-8	Revision 0
3.3-9	Revision 0
3.3-10	Revision 0
3.3-11	Revision 0
3.3-12	Revision 0
3.3-13	Revision 0
3.3-14	Revision 0
3.4-1	Revision 0
3.4-2	Revision 0
3.4-3	Revision 0
3.4-4	Revision 0
3.4-5	Revision 0
3.4-6	Revision 0
3.4-7	Revision 0
3.4-8	Revision 0
3.4-9	Revision 0
3.4-10	Revision 0
3.4-11	Revision 0
3.4-12	Revision 0
3.4-13	Revision 0

3

-

Docket No. 72-1025

April 2000 Revision 0

List of Effective Pages (Continued)

3.4-14	Revision 0
3.4-15	Revision 0
3.4-16	Revision 0
3.4-17	Revision 0
3.4-18	Revision 0
3.4-19	Revision 0
3.4-20	Revision 0
3.4-21	Revision 0
3.4-22	Revision 0
3.4-23	Revision 0
3.4-24	Revision 0
3.4-25	Revision 0
3.4-26	Revision 0
3.4-27	Revision 0
3.4-28	Revision 0
3.4-29	Revision 0
3.4-30	Revision 0
3.4-31	.Revision 0
3.4-32	Revision 0
3.4-33	Revision 0
3.4-34	Revision 0
3.4-35	Revision 0
3.4-36	Revision 0
3.4-37	Revision 0
3.4-38	Revision 0
3.4-39	Revision 0
3.4-40	Revision 0
3.4-41	Revision 0
3.4-42	Revision 0
3.4-43	Revision 0
3.4-44	Revision 0
3.4-45	Revision 0
3.4-46	Revision 0
3.4-47	Revision 0
3.4-48	Revision 0
3.4-49	Revision 0

3.4-50	Revision 0
3.4-51	Revision 0
3.4-52	Revision 0
3.4-53	Revision 0
3.4-54	Revision 0
3.4-55	Revision 0
3.4-56	Revision 0
3.4-57	Revision 0
3.4-58	Revision 0
3.4-59	Revision 0
3.4-60	Revision 0
3.4-61	Revision 0
3.4-62	Revision 0
3.4-63	Revision 0
3.4-64	Revision 0
3.4-65	Revision 0
3.4-66	Revision 0
3.4-67	Revision 0
3.4-68	Revision 0
3.4-69	Revision 0
3.4-70	Revision 0
3.4-71	Revision 0
3.4-72	Revision 0
3.4-73	Revision 0
3.4-74	Revision 0
3.4-75	Revision 0
3.4-76	Revision 0
3.4-77	Revision 0
3.4-78	Revision 0
3.4-79	Revision 0
3.4-80	Revision 0
3.4-81	Revision 0
3.4-82	Revision 0
3.4-83	Revision 0
3.4-84	Revision 0
3.4-85	Revision 0

Docket No. 72-1025

April 2000 Revision 0

List of Effective Pages (Continued)

3.4-86	Revision 0
3.4-87	Revision 0
3.4-88	Revision 0
3.4-89	Revision 0
3.4-90	Revision 0
3.4-91	Revision 0
3.5-1	Revision 0
3.6-1	Revision 0
3.6-2	Revision 0
3.7-1	Revision 0
3.7-2	Revision 0
3.8-1	Revision 0
3.8.1-1	Revision 0
3.8.1-2	Revision 0
3.8.1-3	Revision 0
3.8.1-4	Revision 0
3.8.2-1	Revision 0
3.8.2-2	Revision 0
3.8.2-3	Revision 0
3.8.2-4	Revision 0

Chapter 4

4-i	Revision 0
4-ii	Revision 0
4-iii	Revision 0
4.1-1	Revision 0
4.1-2	Revision 0
4.1-3	Revision 0
4.1-4	Revision 0
4.1-5	Revision 0
4.2-1	Revision 0
4.2-2	Revision 0
4.2-3	Revision 0
4.2-4	Revision 0
4.2-5	Revision 0
4.2-6	Revision 0

4.2-7	Revision 0
4.2-8	Revision 0
4.2-9	Revision 0
4.2-10	Revision 0
4.2-11	Revision 0
4.2-12	Revision 0
4.3-1	Revision 0
4.4-1	Revision 0
4.4-2	Revision 0
4.4-3	Revision 0
4.4-4	Revision 0
4.4-5	Revision 0
4.4-6	Revision 0
4.4-7	Revision 0
4.4-8	Revision 0
4.4-9	Revision 0
4.4-10	Revision 0
4.4-11	Revision 0
4.4-12	Revision 0
4.4-13	Revision 0
4.4-14	Revision 0
4.4-15	Revision 0
4.4-16	Revision 0
4.4-17	Revision 0
4.4-18	Revision 0
4.4-19	Revision 0
4.4-20	Revision 0
4.4-21	Revision 0
4.4-22	Revision 0
4.4-23	Revision 0
4.4-24	Revision 0
4.4-25	Revision 0
4.4-26	Revision 0
4.4-27	Revision 0
4.4-28	Revision 0
4.4-29	Revision 0

Docket No. 72-1025

April 2000 Revision 0

List of Effective Pages (Continued)

4.4-30	Revision 0
4.4-31	Revision 0
4.4-32	Revision 0
4.4-33	Revision 0
4.4-34	Revision 0
4.4-35	Revision 0
4.4-36	Revision 0
4.4-37	Revision 0
4.4-38	Revision 0
4.4-39	Revision 0
4.4-40	Revision 0
4.4-41	Revision 0
4.4-42	Revision 0
4.4-43	Revision 0
4.4-44	Revision 0
4.4-45	Revision 0
4.4-46	Revision 0
4.4-47	Revision 0
4.4-48	Revision 0
4.5-1	Revision 0
4.5-2	Revision 0
4.5-3	Revision 0

Chapter 5

5-i	Revision 0
5-ii	Revision 0
5-iii	Revision 0
5.1-1	Revision 0
5.1-2	Revision 0
5.1-3	Revision 0
5.1-4	Revision 0
5.1-5	Revision 0
5.2-1	Revision 0
5.2-2	Revision 0
5.2-3	Revision 0
5.2-4	Revision 0

5 2-5	Revision 0
5.2-5	Revision 0
5.2-0	Revision 0
5.2-8	Revision 0
5.2-9	Revision 0
5.2-10	Revision 0
5.2-11	Revision 0
5.2-12	 Revision 0
5.3-1	Revision 0
5.3-2	Revision 0
5.3-3	Revision 0
5.3-4	Revision 0
5.3-5	Revision 0
5.3-6	Revision 0
5.3-7	Revision 0
5.3-8	Revision 0
5.3-9	Revision 0
5.3-10	Revision 0
5.3-11	Revision 0
5.3-12	Revision 0
5.3-13	Revision 0
5.3-14	Revision 0
5.3-15	Revision 0
5.3-16	Revision 0
5.4-1	Revision 0
5.4-2	Revision 0
5.4-3	Revision 0
5.4-4	Revision 0
5.4-5	Revision 0
5.4-6	Revision 0
5.4-7	Revision 0
5.4-8	Revision 0
5.4-9	Revision 0
5.4-10	Revision 0
5.4-11	Revision 0
5.4-12	Revision 0

6

Docket No. 72-1025

April 2000 Revision 0

List of Effective Pages (Continued)

5.4-13	Revision 0
5.4-14	Revision 0
5.4-15	Revision 0
5.4-16	Revision 0
5.4-17	Revision 0
5.4-18	Revision 0
5.4-19	Revision 0
5.4-20	Revision 0
5.4-21	Revision 0
5.4-22	Revision 0
5.4-23	Revision 0
5.4-24	Revision 0
5.4-25	Revision 0
5.4-26	Revision 0
5.5-1	Revision 0
5.5-2	Revision 0

Chapter 6

б-і	Revision 0
6-ii	Revision 0
6-iii	Revision 0
6.1-1	Revision 0
6.1-2	Revision 0
6.1-3	Revision 0
6.2-1	Revision 0
6.2-2	Revision 0
6.2-3	Revision 0
6.2-4	Revision 0
6.2-5	Revision 0
6.2-6	Revision 0
6.2-7	Revision 0
6.3-1	Revision 0
6.3-2	Revision 0
6.3-3	Revision 0
6.3-4	Revision 0
6.3-5	Revision 0

6.3-6	Revision 0
6.3-7	Revision 0
6.3-8	Revision 0
6.4-1	Revision 0
6.4-2	Revision 0
6.4-3	Revision 0
6.4-4	Revision 0
6.4-5	Revision 0
6.4-6	Revision 0
6.4-7	Revision 0
6.4-8	Revision 0
6.4-9	Revision 0
6.4-10	Revision 0
6.5-1	Revision 0
6.5-2	Revision 0
6.5-3	Revision 0
6.5-4	Revision 0
6.5-5	Revision 0
6.5-6	Revision 0
6.5-6 6.5-7	Revision 0 Revision 0
6.5-6 6.5-7 6.5-8	Revision 0 Revision 0 Revision 0
6.5-6 6.5-7 6.5-8 6.5-9	Revision 0 Revision 0 Revision 0 Revision 0
6.5-6 6.5-7 6.5-8 6.5-9 6.5-10	Revision 0 Revision 0 Revision 0 Revision 0 Revision 0
6.5-6 6.5-7 6.5-8 6.5-9 6.5-10 6.5-11	Revision 0 Revision 0 Revision 0 Revision 0 Revision 0 Revision 0
6.5-6 6.5-7 6.5-8 6.5-9 6.5-10 6.5-11 6.5-12	Revision 0 Revision 0 Revision 0 Revision 0 Revision 0 Revision 0 Revision 0
6.5-6 6.5-7 6.5-8 6.5-9 6.5-10 6.5-11 6.5-12 6.5-13	Revision 0 Revision 0 Revision 0 Revision 0 Revision 0 Revision 0 Revision 0 Revision 0
6.5-6 6.5-7 6.5-8 6.5-9 6.5-10 6.5-11 6.5-12 6.5-13 6.5-14	Revision 0 Revision 0 Revision 0 Revision 0 Revision 0 Revision 0 Revision 0 Revision 0 Revision 0
6.5-6 6.5-7 6.5-8 6.5-9 6.5-10 6.5-11 6.5-12 6.5-13 6.5-14 6.5-15	Revision 0 Revision 0 Revision 0 Revision 0 Revision 0 Revision 0 Revision 0 Revision 0 Revision 0 Revision 0
6.5-6 6.5-7 6.5-8 6.5-9 6.5-10 6.5-11 6.5-12 6.5-13 6.5-14 6.5-15 6.5-16	Revision 0 Revision 0 Revision 0 Revision 0 Revision 0 Revision 0 Revision 0 Revision 0 Revision 0 Revision 0
6.5-6 6.5-7 6.5-8 6.5-9 6.5-10 6.5-11 6.5-12 6.5-13 6.5-14 6.5-15 6.5-16 6.5-17	Revision 0 Revision 0
6.5-6 6.5-7 6.5-8 6.5-9 6.5-10 6.5-11 6.5-12 6.5-13 6.5-14 6.5-15 6.5-16 6.5-18	Revision 0 Revision 0
6.5-6 6.5-7 6.5-8 6.5-9 6.5-10 6.5-11 6.5-12 6.5-13 6.5-14 6.5-15 6.5-16 6.5-17 6.5-18 6.5-19	Revision 0 Revision 0
6.5-6 6.5-7 6.5-8 6.5-9 6.5-10 6.5-11 6.5-12 6.5-13 6.5-14 6.5-15 6.5-16 6.5-17 6.5-18 6.5-20	Revision 0 Revision 0
6.5-6 6.5-7 6.5-8 6.5-9 6.5-10 6.5-11 6.5-12 6.5-13 6.5-14 6.5-15 6.5-16 6.5-17 6.5-18 6.5-20 6.5-21	Revision 0 Revision 0
6.5-6 6.5-7 6.5-8 6.5-9 6.5-10 6.5-10 6.5-12 6.5-13 6.5-13 6.5-14 6.5-15 6.5-16 6.5-16 6.5-17 6.5-18 6.5-19 6.5-20 6.5-22 6.5-22	Revision 0 Revision 0

Docket No. 72-1025

April 2000 Revision 0

List of Effective Pages (Continued)

6.6-2	Revision 0
6.7-1	Revision 0
6.7-2	Revision 0
6.7-3	Revision 0
6.7-4	Revision 0
6.7-5	Revision 0
6.7-6	Revision 0
6.7-7	Revision 0
6.7-8	Revision 0
6.7-9	Revision 0
6.7-10	Revision 0
6.7-11	Revision 0
6.7-12	Revision 0
6.7-13	Revision 0
6.7-14	Revision 0
6.7-15	Revision 0
6.7-16	Revision 0
6.7-17	Revision 0
6.7-18	Revision 0
6.7-19	Revision 0
6.7-20	Revision 0
6.7-21	Revision 0
6.7-22	Revision 0
6.7-23	Revision 0
6.7-24	Revision 0
6.7-25	Revision 0
6.7-26	Revision 0
6.7-27	Revision 0
6.7-28	Revision 0
6.7-29	Revision 0
6.7-30	Revision 0
6.7-31	Revision 0
6.7-32	Revision 0
6.7-33	Revision 0
6.7-34	Revision 0
6.7-35	Revision 0

6.7-36	Revision 0
6.7-37	Revision 0
6.7-38	Revision 0
6.7-39	Revision 0
6.7-40	Revision 0
6.7-41	Revision 0
6.7-42	Revision 0
6.7-43	Revision 0
6.7-44	Revision 0
6.7-45	Revision 0
6.7-46	Revision 0
6.7-47	Revision 0
6.7-48	Revision 0
6.7-49	Revision 0
6.7-50	Revision 0
6.7-51	Revision 0
6.7-52	Revision 0
6.7-53	Revision 0
6.7-54	Revision 0
6.7-55	Revision 0
6.7-56	Revision 0
6.7-57	Revision 0
6.7-58	Revision 0
6.7-59	Revision 0
6.7-60	Revision 0
6.7-61	Revision 0
6.7-62	Revision 0
6.7-63	Revision 0
6.7-64	Revision 0
6.7-65	Revision 0
6.7-66	Revision 0
6.7-67	Revision 0
6.7-68	Revision 0
6.7-69	Revision 0
6.7-70	Revision 0
6.7-71	Revision 0

8

Docket No. 72-1025

List of Effective Pages (Continued)

.1-5Revisio	on 0
.1-6Revisio	on 0
.1-7Revisic	on 0
7.1-8Revisio	on 0
7.1-9Revisio	on 0
2-1Revisio	on 0
2-2Revisio	on 0
V.3-1Revisio	on 0

Chapter 8

8-i	Revision 0
8-ii	Revision 0
8-1	Revision 0
8.1-1	Revision 0
8.1-2	Revision 0
8.1-3	Revision 0
8.1-4	Revision 0
8.1-5	Revision 0
8.1-6	Revision 0
8.1-7	Revision 0
8.1-8	Revision 0
8.1-9	Revision 0
8.1-10	Revision 0
8.2-1	Revision 0
8.2-2	Revision 0
8.3-1	Revision 0
8.3-2	Revision 0
8.3-3	Revision 0
8.3-4	Revision 0

Chapter 9

9-i	Revision 0
9-1	Revision 0
9.1-1	Revision 0
9.1-2	Revision 0
9.1-3	Revision 0

6.7-72	Revision 0
6.7-73	Revision 0
6.7-74	Revision 0
6.7-75	Revision 0
6.7-76	Revision 0
6.7-77	Revision 0
6.7-78	Revision 0
6.7-79	Revision 0
6.7-80	Revision 0
6.7-81	Revision 0
6.7-82	Revision 0
6.7-83	Revision 0
6.7-84	Revision 0
6.7-85	Revision 0
6.7-86	Revision 0
6.7-87	Revision 0
6.7-88	Revision 0
6.7-89	Revision 0
6.7-90	Revision 0
6.7-91	Revision 0
6.7-92	Revision 0
6.7-93	Revision 0
6.7-94	Revision 0
6.7-95	Revision 0
6.7-96	Revision 0
6.7-97	Revision 0
6.7-98	Revision 0

Chapter 7

7-i	Revision 0
7-ii	Revision 0
7-1	Revision 0
7.1-1	Revision 0
7.1-2	Revision 0
7.1-3	Revision 0
7.1-4	Revision 0

April 2000 Revision 0

List of Effective Pages (Continued)

9.1-4	Revision 0
9.1-5	Revision 0
9.1-6	Revision 0
9.1-7	Revision 0
9.1-8	Revision 0
9.2-1	Revision 0
9.2-2	Revision 0

Chapter 10

10-i	Revision 0
10-ii	Revision 0
10.1-1	Revision 0
10.1-2	Revision 0
10.2-1	Revision 0
10.2-2	Revision 0
10.3-1	Revision 0
10.3-2	Revision 0
10.3-3	Revision 0
10.3-4	Revision 0
10.3-5	Revision 0
10.3-6	Revision 0
10.3-7	Revision 0
10.3-8	Revision 0
10.4-1	Revision 0
10.4-2	Revision 0
10.4-3	Revision 0
10.4-4	Revision 0

Chapter 11

11-i	Revision 0
11-ii	Revision 0
11-iii	Revision 0
11-iv	Revision 0
11-v	Revision 0
11-vi	Revision 0
11-vii	Revision 0

11-viii	Revision 0
11-ix	Revision 0
11-1	Revision 0
11.1-1	Revision 0
11.1-2	Revision 0
11.1-3	Revision 0
11.1-4	Revision 0
11.1-5	Revision 0
11.1-6	Revision 0
11.1-7	Revision 0
11.1-8	Revision 0
11.1-9	Revision 0
11.1-10	Revision 0
11.1-11	Revision 0
11.1-12	Revision 0
11.1-13	Revision 0
11.1-14	Revision 0
11.1-15	Revision 0
11.1-16	Revision 0
11.2-1	Revision 0
11.2-2	Revision 0
11.2-3	Revision 0
11.2-4	Revision 0
11.2-5	Revision 0
11.2-6	Revision 0
11.2-7	Revision 0
11.2-8	Revision 0
11.2-9	Revision 0
11.2-10	Revision 0
11.2-11	Revision 0
11.2-12	Revision 0
11.2-13	Revision 0
11.2-14	Revision 0
11.2-15	Revision 0
11.2-16	Revision 0
11.2-17	Revision 0

Docket No. 72-1025

April 2000 Revision 0

.

List of Effective Pages (Continued)

11.2-18	Revision 0
11.2-19	Revision 0
11.2-20	Revision 0
11.2-21	Revision 0
11.2-22	Revision 0
11.2-23	Revision 0
11.2-24	Revision 0
11.2-25	Revision 0
11.2-26	Revision 0
11.2-27	Revision 0
11.2-28	Revision 0
11.2-29	Revision 0
11.2-30	Revision 0
11.2-31	Revision 0
11.2-32	Revision 0
11.2-33	Revision 0
11.2-34	Revision 0
11.2-35	Revision 0
11.2-36	Revision 0
11.2-37	Revision 0
11.2-38	Revision 0
11.2-39	Revision 0
11.2-40	Revision 0
11.2-41	Revision 0
11.2-42	Revision 0
11.2-43	Revision 0
11.2-44	Revision 0
11.2-45	Revision 0
11.2-46	Revision 0
11.2-47	Revision 0
11.2-48	Revision 0
11.2-49	Revision 0
11.2-50	Revision 0
11.2-51	Revision 0
11.2-52	Revision 0
11.2-53	Revision 0

11.2-54	Revision 0
11.2-55	Revision 0
11.2-56	Revision 0
11.2-57	Revision 0
11.2-58	Revision 0
11.2-59	Revision 0
11.2-60	Revision 0
11.2-61	Revision 0
11.2-62	Revision 0
11.2-63	Revision 0
11.2-64	Revision 0
11.2-65	Revision 0
11.2-66	Revision 0
11.2-67	Revision 0
11.2-68	Revision 0
11.2-69	Revision 0
11.2-70	Revision 0
11.2-71	Revision 0
11.2-72	Revision 0
11.2-73	Revision 0
11.2-74	Revision 0
11.2-75	Revision 0
11.2-76	Revision 0
11.2-77	Revision 0
11.2-78	Revision 0
11.2-79	Revision 0
11.2-80	Revision 0
11.2-81	Revision 0
11.2-82	Revision 0
11.2-83	Revision 0
11.2-84	Revision 0
11.2-85	Revision 0
11.2-86	Revision 0
11.2-87	Revision 0
11.2-88	Revision 0
11.2-89	Revision 0

April 2000 Revision 0

List of Effective Pages (Continued)

11.2-90	Revision 0
11.2-91	Revision 0
11.2-92	Revision 0
11.2-93	Revision 0
11.2-94	Revision 0
11.2-95	Revision 0
11.2-96	Revision 0
11.2-97	Revision 0
11.2-98	Revision 0
11.2-99	Revision 0
11.2-100	Revision 0
11.2-101	Revision 0
11.2-102	Revision 0
11.3-1	Revision 0
11.3-2	Revision 0
11.3-3	Revision 0
11.3-4	Revision 0
11.3-5	Revision 0
11.3-6	Revision 0
11.3-7	Revision 0
11.3-8	Revision 0
11.3-9	Revision 0
11.3-10	Revision 0
11.3-11	Revision 0
11.3-12	Revision 0
11.3-13	Revision 0
11.3-14	Revision 0
11.3-15	Revision 0
11.3-16	Revision 0
11.3-17	Revision 0
11.3-18	Revision 0
11.3-19	Revision 0
11.3-20	Revision 0
11.3-21	Revision 0
11.3-22	Revision 0
11.3-23	Revision 0

11 3-24	Revision 0
11.3-27	Revision 0
11.3-26	Revision 0
11.3-27	Revision 0
11.3-28	Revision 0
11.3-29	Revision 0
11.3-30	Revision 0
11.3-31	Revision 0
11.3-32	Revision 0
11.3-33	Revision 0
11.3-34	Revision 0
11.3-35	Revision 0
11.3-36	Revision 0
11.3-37	Revision 0
11.3-38	Revision 0
11.3-39	Revision 0
11.3-40	Revision 0
11.3-41	Revision 0
11.4-1	Revision 0
11.4-2	Revision 0
11.4-3	Revision 0
11.4-4	Revision 0
11.4-5	Revision 0
11.4-6	Revision 0
11.4-7	Revision 0
11.4-8	Revision 0
11.4-9	Revision 0
11.4-10	Revision 0
11.4-11	Revision 0
11.4-12	Revision 0
11.4-13	Revision 0
11.4-14	Revision 0
11.4-15	Revision 0
11.4-16	Revision 0
11.4-17	Revision 0
11.4-18	Revision 0

Docket No. 72-1025

List of Effective Pages (Continued)

11.4-19	Revision 0
11.5-1	Revision 0
11.5-2	Revision 0
11.6-1	Revision 0
11.6-2	Revision 0
11.6-3	Revision 0

Chapter 12

12-i	Revision 0
12-ii	Revision 0
12-1	Revision 0
12-2	Revision 0
12-3	Revision 0
12-4	Revision 0
12A-1	Revision 0
12A-2	Revision 0
12A-3	Revision 0
12A1-1	Revision 0
12A1-2	Revision 0
12A1-3	Revision 0
12A1-4	Revision 0
12A1-5	Revision 0
12A1-6	Revision 0
12A1-7	Revision 0
12A1-8	Revision 0
12A1-9	Revision 0
12A1-10	Revision 0
12A1-11	Revision 0
12A1-12	Revision 0
12A1-13	Revision 0
12A1-14	Revision 0
12A1-15	Revision 0
12A2-1	Revision 0
12A2-2	Revision 0
12A2-3	Revision 0
12A2-4	Revision 0

12A2-5	Revision 0
12A2-6	Revision 0
12A3-1	Revision 0
12A3-2	Revision 0
12A3-3	Revision 0
12A3-4	Revision 0
12A3-5	Revision 0
12A3-6	Revision 0
12A3-7	Revision 0
12A3-8	Revision 0
12A3-9	Revision 0
12A3-10	Revision 0
12A3-11	Revision 0
12A3-12	Revision 0
12A3-13	Revision 0
12A3-14	Revision 0
12A3-15	Revision 0
12A3-16	Revision 0
12A3-17	Revision 0
12A3-18	Revision 0
12A3-19	Revision 0
12A3-20	Revision 0
12A3-21	Revision 0
12A3-22	Revision 0
12A3-23	Revision 0
12A3-24	Revision 0
12A3-25	Revision 0
12A3-26	Revision 0
12A4-1	Revision 0
12A4-2	Revision 0
12A4-3	Revision 0
12A4-4	Revision 0
12A4-5	Revision 0
12A4-6	Revision 0
12A4-7	Revision 0
12A4-8	Revision 0

Docket No. 72-1025

List of Effective Pages (Continued)

12B3-27	Revision 0
12B3-28	Revision 0
12B3-29	Revision 0
12B3-30	Revision 0
12B3-31	Revision 0
12B3-32	Revision 0
12B3-33	Revision 0
12B3-34	Revision 0
12B3-35	Revision 0
12B3-36	Revision 0
12B3-37	Revision 0
12B3-38	Revision 0
12B3-39	Revision 0
12B3-40	Revision 0
12B3-41	Revision 0
12B3-42	Revision 0
12B3-43	Revision 0
12B3-44	Revision 0
12B3-45	Revision 0

Chapter 13

13-i	Revision 0
13.1-1	Revision 0
13.1-2	Revision 0
13.2-1	Revision 0
13.2-2	Revision 0
13.2-3	Revision 0
13.2-4	Revision 0
13.2-5	Revision 0
13.2-6	Revision 0
13.2-7	Revision 0
13.2-8	Revision 0
13.2-9	Revision 0

12A4-9	0
12A5-1 Revision	U
	-
12A5-2Revision	0
12A5-3Revision	0
12A5-4Revision	0
12B-1 Revision	0
12B-2Revision	0
12B1-1 Revision	0
12B2-1 Revision	0
12B2-2Revision	0
12B3-1 Revision	0
12B3-2Revision	0
12B3-3 Revision	0
12B3-4 Revision	0
12B3-5Revision	0
12B3-6Revision	0
12B3-7Revision	0
12B3-8 Revision	0
12B3-9Revision	0
12B3-10 Revision	0
12B3-11 Revision	0
12B3-12 Revision	0
12B3-13 Revision	0
12B3-14 Revision	0
12B3-15 Revision	0
12B3-16 Revision	0
12B3-17 Revision	0
12B3-18 Revision	0
12B3-19 Revision	0
12B3-20 Revision	0
12B3-21 Revision	0
12B3-22Revision	0
12B3-23Revision	0
12B3-24Revision	0
12B3-25 Revision	0
12B3-26 Revision	0

MASTER TABLE OF CONTENTS

1.0	GENERAL DESCRIPTION1-1				
1.1	Introduction				
1.2	The NA	AC-MPC System 1.2-1			
	1.2.1	NAC-MPC System Components 1.2-1			
		1.2.1.1 Transportable Storage Canister and Baskets 1.2-2			
		1.2.1.2 Vertical Concrete Cask			
		1.2.1.3 Transfer Cask 1.2-5			
		1.2.1.4 Auxiliary Equipment 1.2-6			
		1.2.1.5 Transport Cask 1.2-8			
	1.2.2	Operational Features			
	1.2.3	Cask Contents 1.2-11			
1.3	Agents and Contractors1.3-				
1.4	Generi	c Storage Cask Arrays 1.4-1			
1.5	Licens	e Drawings 1.5-1			
	1.5.1	NAC-MPC License Drawings 1.5-1			
	1.5.2	Yankee-Class Reconfigured Fuel Assembly License Drawings 1.5-2			
2.0	PRIN	CIPAL DESIGN CRITERIA			
2.1	Spent	Fuel To Be Stored 2.1-1			
	2.1.1	Bounding Fuel Evaluation			
	2.1.2	Reconfigured Fuel Assembly 2.1-2			
	2.1.3	Stainless Steel-Clad Fuel 2.1-2			

2.2	Design Criteria for Environmental Conditions and Natural Phenomena 2.2-1				
	2.2.1	Tornado and Wind Loadings 2.2-1			
		2.2.1.1	Applicable Design Parameters 2.2-1		
		2.2.1.2	Determination of Forces on Structures 2.2-1		
		2.2.1.3	Tornado Missiles 2.2-1		
	2.2.2	Water Le	Water Level (Flood) Design		
		2.2.2.1	Flood Elevations 2.2-2		
		2.2.2.2	Phenomena Considered in Design Load Calculations 2.2-3		
		2.2.2.3	Flood Force Application 2.2-3		
		2.2.2.4	Flood Protection		
	2.2.3	Seismic	Design		
		2.2.3.1	Input Criteria 2.2-4		
		2.2.3.2	Seismic - System Analyses 2.2-4		
	2.2.4	Snow an	d Ice Loadings 2.2-4		
	2.2.5	Combined Load Criteria			
		2.2.5.1	Load Combinations and Design Strength - Concrete Cask 2.2-6		
		2.2.5.2	Design Strength Reduction Factors - Concrete		
		2.2.5.3	Load Combinations and Design Strength – Canister		
			and Basket 2.2-6		
		2.2.5.4	Design Strength - Transfer Cask 2.2-7		
	2.2.6	Environmental Temperatures			
2.3	Safety	Protection	Systems		
	2.3.1	General.			
	2.3.2	Protectio	on by Multiple Confinement Barriers and Systems		
		2.3.2.1	Confinement Barriers and Systems		
	2.3.3	Protection by Equipment and Instrumentation Selection			
		2.3.3.1	Equipment		
		2.3.3.2	Instrumentation		
	2.3.4	Nuclear Criticality Safety			
		2.3.4.1	Error Contingency Criterion 2.3-5		

	2.3.5	Radiolog	ical Protection 2.3-5	
		2.3.5.1	Access Control 2.3-5	
		2.3.5.2	Shielding2.3-6	
		2.3.5.3	Ventilation Off-Gas 2.3-6	
		2.3.5.4	Radiological Alarm Systems 2.3-7	
	2.3.6	Fire and	Explosion Protection	
		2.3.6.1	Fire Protection	
		2.3.6.2	Explosion Protection 2.3-8	
2.4	Decom	missioning	g Considerations 2.4-1	
3.0	STRU	CTURAL	EVALUATION	
3.1	Structu	ral Design		
	3.1.1	Discussio	on 3.1-1	
	3.1.2	Design C	Criteria 3.1-4	
3.2	Weight	ghts and Centers of Gravity 3.2-1		
3.3	Mechanical Properties of Materials 3.3-1			
3.4	Genera	l Standard	s for Casks 3.4-1	
	3.4.1	Chemica	l and Galvanic Reactions 3.4-1	
		3.4.1.1	Component Operating Environment	
		3.4.1.2	Component Material Categories 3.4-2	
		3.4.1.3	General Effects of Identified Reactions	
		3.4.1.4	Adequacy of the Canister Operating Procedures	
		3.4.1.5	Effects of Reaction Products 3.4-7	
	3.4.2 Positive C		Closure 3.4-7	
	3.4.3	Lifting D	Devices	
		3.4.3.1	Storage Cask Bottom Lift 3.4-12	
		3.4.3.2	Canister Lift	
		3.4.3.3	Transfer Cask Lift	

	3.4.4	NAC-M	PC Components Under Normal Operating Loads
		3.4.4.1	Canister and Basket Analyses
		3.4.4.2	Vertical Concrete Storage Cask - Concrete Stress Analysis 3.4-81
	3.4.5	Cold	
3.5	Fuel R	ods	
3.6	Canist	er Closure	Weld Evaluation – Normal Conditions
	3.6.1	Stress E	valuation for the Canister Closure Weld
	3.6.2	Critical 1	Flaw Size for the Canister Closure Weld
3.7	Refere	nces	
	_		
3.8	Coatin	g Specific	ations
	3.8.1	Keeler a	nd Long E-Series Epoxy Enamel 3.8.1-1
	3.8.2	Ameron	PSX 738 Siloxane Coating 3.8.2-1
40	THER	MAT FV	
7.0	IIIII		ALCATION
4.1	Discus	sion	
4.2	Summa	ary of The	rmal Properties of Materials 4.2-1
	a 10		-
4.3	Specifi	cation of (Components 4.3-1
44	Therm	al Evaluati	ion for Normal Conditions of Storage 4.4-1
	4.4.1	Thermal	Models 44-1
		4.4.1.1	Two-Dimensional Axisymmetric Air Flow and
			Concrete Cask Model
		4.4.1.2	Three-Dimensional Canister Model
		4.4.1.3	Two-Dimensional Fuel Model
		4.4.1.4	Two-Dimensional Fuel Tube Model
		1115	Three-Dimensional Transfer Cask and Canister Model $14-25$
		т.т.т.J	Thee-Dimensional Transfer Cask and Camster Model

	4.4.2	Test Model 4.4-32	
	4.4.3	Maximum Temperatures for Normal Conditions 4.4-32	
		4.4.3.1 Maximum Temperatures in Long-Term Storage 4.4-32	
		4.4.3.2 Maximum Temperatures in Transient Operations 4.4-33	
	4.4.4	Minimum Temperatures	
	4.4.5	Maximum Internal Pressure for Normal Conditions	
	4.4.6	Maximum Thermal Stresses for Normal Conditions 4.4-48	
	4.4.7	Evaluation of Cask Performance for Normal Conditions of Storage 4.4-48	
4.5	Refere	nces	
5.0	SHIEI	DING EVALUATION 5.1-1	
5.1	Discus	sion and Results 5.1-1	
5.2	Source Specification		
	5.2.1	Gamma Source 5.2-2	
	5.2.2	Neutron Source	
	5.2.3	Source Axial Profile	
53	Model	Specification	
5.5	5.3.1	Description of the Radial and Axial Shielding Configurations	
	01012	5.3.1.1 One-Dimensional Radial and Axial Shielding Models	
		5.3.1.2 Three-Dimensional Top and Bottom Shielding Models 5.3-4	
	5.3.2	Shield Regional Densities 5.3-8	
5.4	Shield	ing Evaluation	
	5.4.1	Calculational Methods	
	5.4.2	Flux-to-Dose Rate Conversion Factors	
	5.4.3	Dose Rates	
		5.4.3.1 One-Dimensional Storage Cask Dose Rates	
		5.4.3.2 Three-Dimensional Storage Cask Dose Rates	
		5.4.3.3 One-Dimensional Transfer Cask Dose Rates	
		5.4.3.4 Three-Dimensional Transfer Cask Dose Rates	

	5.4.4	Storage	Cask Shielded Source Terms	5.4-7
5.5	Refere	nces		5.5-1
6.0	CRITI	CALITY	EVALUATION	6.1-1
6.1	Discus	sion and F	Results	6.1-1
6.2	Packag	e Fuel Lo	ading	6.2-1
6.3	Critica	lity Model	l Specification	6.3-1
	6.3.1	Descript	ion of Calculational Models	6.3-1
	6.3.2	Package	Regional Densities	6.3-2
		6.3.2.1	Fuel Region	6.3-3
		6.3.2.2	Cask Material	6.3-4
		6.3.2.3	Water Reflector Densities	6.3-5
6.4	Criticality Calculation			6.4-1
	6.4.1	Calculational Method 6.4		
	6.4.2	Fuel Loading Optimization 6.4-		
	6.4.3	Criticalit	ty Results	6.4-2
		6.4.3.1	Most Reactive Assembly	6.4-2
		6.4.3.2	Most Reactive Mechanical Configuration	6.4-2
		6.4.3.3	Transfer Cask Criticality Evaluation	6.4-5
		6.4.3.4	Storage Cask Criticality Evaluation	6.4-5
6.5	Critical Benchmark Experiments			6.5-1
	6.5.1	Benchmark Experiments and Applicability		6.5-3
		6.5.1.1	Description of Experiments	6.5-3
		6.5.1.2	Applicability of Experiments	6.5-3
	6.5.2	Results of	of Benchmark Calculations	6.5-4
		6.5.2.1	Trends	6.5-6
	6.5.3	Comparison of NAC Method to NUREG/CR-6361		

6.6	Refere	nces		6.6-1
6.7	Supple	emental Da	ta	6.7-1
7.0	CONF	INEMEN	Τ	
7.1	Confir	ement Bou	indary	
	7.1.1	Confiner	nent Vessel	
		7.1.1.1	Confinement Vessel - Canister	
		7.1.1.2	Design Documents, Codes, and Standards	
		7.1.1.3	Technical Requirements for the Canister	
		7.1.1.4	Release Rate	
	7.1.2	Confiner	nent Penetrations	
	7.1.3	Seals and	d Welds	
		7.1.3.1	Fabrication	
		7.1.3.2	Welding Specifications	
		7.1.3.3	Testing, Inspection, and Examination	
	7.1.4	Closure.		
7.2	Requir	ements for	Normal Conditions of Storage	
	7.2.1	Release	of Radioactive Material	7.2.1
	7.2.2	Pressuriz	zation of Confinement Vessel	
7.3	Confir	nement Rec	quirements for Hypothetical Accident Conditions	
8.0	OPER	RATING P	ROCEDURES	8-1
8.1	Loadin	ng the NAC	C-MPC Storage System	8.1-1
	8.1.1	Loading	and Closing the Transportable Storage Canister	
	8.1.2	Loading	the Vertical Concrete Cask	
	8.1.3	Transpor	rting the Vertical Concrete Cask	8.1-7

8.2	Remov	val of the 7	Fransportable Storage Canister from the	
	Vertica	al Concret	e Cask	8.2-1
8.3	Unload	ding the T	ransportable Storage Canister	8.3-1
9.0	ACCE	EPTANCE	E TESTS AND MAINTENANCE PROGRAM	
9.1	Accep	tance Test	S	9.1-1
	9.1.1	Visual a	nd Nondestructive Examination Inspection	9.1-1
		9.1.1.1	Nondestructive Weld Examination	9.1-2
		9.1.1.2	Fabrication Inspections	9.1-3
	9.1.2	Structur	al and Pressure Test	9.1-4
	9.1.3	Leak Te	sts	9.1-5
	9.1.4	Component Tests		
		9.1.4.1	Valves, Rupture Disks and Fluid Transport Devices	
		9.1.4.2	Gaskets	
	9.1.5	Shieldin	g Tests	
	9.1.6 Neutro		-Absorber Tests	
		9.1.6.1	General	
		9.1.6.2	Preparation of Samples	
		9.1.6.3	Wet Chemistry Test Performance	
		9.1.6.4	Neutron Absorption Test Performance	
		9.1.6.5	Acceptance Criteria	
	9.1.7	Thermal Tests.		
	9.1.8	Cask Ide	entification	
9.2	Maintenance Program			
	9.2.1	Continu	ing Maintenance Requirements	
	9.2.2	Required Maintenance of First Storage System Placed in Service		

10.0	RADIA	TION PR	OTECTION	10.1-1
10.1	Ensurin	g That Oc	cupational Radiation Exposures Are As Low As	
	Reasona	ably Achie	vable (ALARA)	10.1-1
	10.1.1	Policy Co	nsiderations	10.1-1
	10.1.2	Design Co	onsiderations	10.1-1
	10.1.3	Operation	al Considerations	10.1-2
10.2	Radiatio	on Protecti	on Design Features	10.2-1
	10.2.1	Design Ba	asis for Normal Storage Conditions	10.2-1
	10.2.2	Design B	asis for Accident Conditions	10.2-2
10.3	Estimat	ed On-Site	e Collective Dose Assessment	10.3-1
	10.3.1	Estimated	l Collective Dose for Loading a Single NAC-MPC	10.3-1
	10.3.2	Estimated	Annual Dose Due to Routine Operations	10.3-2
	10.3.3	Estimated	l Collective Dose for Unloading a Single NAC-MPC	10.3-3
10.4	Exposu	res to the l	Public	10.4-1
11.0	ACCIE	DENT AN	ALYSIS	11-1
11.1	Off-No:	rmal Even	ts	11.1-1
	11.1.1	Blockage	of Half of the Air Inlets	11.1-1
		11.1.1.1	Cause of Event	11.1-1
		11.1.1.2	Analysis of the Blockage Event	11.1-1
		11.1.1.3	Radiological Consequences	11.1-2
		11.1.1.4	NAC-MPC Performance	11.1-2
				11.1.0
		11.1.1.5	Recovery and/or Corrective Actions	11.1-2
	11.1.2	11.1.1.5 Canister	Recovery and/or Corrective Actions Off-Normal Handling Load	11.1-2
	11.1.2	11.1.1.5 Canister 11.1.2.1	Recovery and/or Corrective Actions Off-Normal Handling Load Cause of Event	11.1-2 11.1-3 11.1-3
	11.1.2	11.1.1.5 Canister (11.1.2.1 11.1.2.2	Recovery and/or Corrective Actions Off-Normal Handling Load Cause of Event Analysis of the Canister Off-Normal Handling Load F	11.1-2 11.1-3 11.1-3 Event 11.1-3
	11.1.2	11.1.1.5 Canister (11.1.2.1 11.1.2.2 11.1.2.3	Recovery and/or Corrective Actions Off-Normal Handling Load Cause of Event Analysis of the Canister Off-Normal Handling Load E Radiological Consequences	11.1-2 11.1-3 11.1-3 Event 11.1-3 11.1-6

		11.1.2.5	Recovery and/or Corrective Actions
	11.1.3	Failure of	f Instrumentation 11.1-6
		11.1.3.1	Cause of Accident 11.1-7
		11.1.3.2	Analysis of Instrumentation Failure
		11.1.3.3	Radiological Consequences for This Accident 11.1-7
		11.1.3.4	NAC-MPC Performance 11.1-8
		11.1.3.5	Recovery and Corrective Actions 11.1-8
	11.1.4	Severe E	nvironmental Conditions (100°F and -40°F) 11.1-8
		11.1.4.1	Cause of Event
		11.1.4.2	Analysis of the Off-Normal Ambient Temperature Event 11.1-9
		11.1.4.3	Radiological Consequences 11.1-10
		11.1.4.4	NAC-MPC Performance 11.1-10
		11.1.4.5	Corrective Actions
	11.1.5	Small Re	lease of Radioactive Particulate From the Canister Exterior 11.1-15
		11.1.5.1	Cause of Event 11.1-15
		11.1.5.2	Analysis 11.1-15
		11.1.5.3	Radiological Consequences 11.1-16
		11.1.5.4	NAC-MPC Performance 11.1-16
		11.1.5.5	Corrective Actions
11.2	Accide	nts	
	11.2.1	Accident	Pressurization
		11.2.1.1	Cause of Pressurization 11.2-1
		11.2.1.2	Analysis of Accident Pressurization 11.2-1
		11.2.1.3	Radiological Consequences 11.2-3
		11.2.1.4	NAC-MPC Performance 11.2-3
		11.2.1.5	Recovery and/or Corrective Actions 11.2-3
	11.2.2	Earthqual	ke Event 11.2-7
		11.2.2.1	Cause of Earthquake 11.2-7
		11.2.2.2	Earthquake Analysis 11.2-7
		11.2.2.3	Radiological Consequences 11.2-13
		11.2.2.4	NAC-MPC Performance 11.2-13
		11.2.2.5	Recovery and/or Corrective Actions

.

11.2.3	Explosion	1	11.2-14
	11.2.3.1	Cause of Accident	11.2-14
	11.2.3.2	Evaluation of the Explosion Event	11.2-14
	11.2.3.3	Radiological Consequences	11.2-15
	11.2.3.4	NAC-MPC Performance	11.2-15
	11.2.3.5	Recovery and/or Corrective Actions	11.2-15
11.2.4	Failure of	f All Fuel Rods With a Subsequent Ground Level Breach	h
	of the Ca	nister	11.2-15
11.2.5	Fire Acci	dent	11.2-16
	11.2.5.1	Cause of Accident	11.2-16
	11.2.5.2	Accident Analysis	11.2-16
	11.2.5.3	Radiological Consequences	11.2-19
	11.2.5.4	NAC-MPC Performance	11.2-19
	11.2.5.5	Recovery and/or Corrective Actions	11.2-19
11.2.6	Flood		11.2-20
	11.2.6.1	Cause of Flood	11.2-20
	11.2.6.2	Flood Analysis	11.2-20
	11.2.6.3	Radiological Consequences	11.2-24
	11.2.6.4	NAC-MPC Performance	11.2-24
	11.2.6.5	Recovery and/or Corrective Actions	11.2-24
11.2.7	Fresh Fu	el Loading in the Canister	11.2-24
	11.2.7.1	Cause of Accident	11.2-25
-	11.2.7.2	Analysis of Fresh Fuel Loading in the Canister	11.2-25
	11.2.7.3	Radiological Consequences For This Accident	11.2-25
	11.2.7.4	NAC-MPC Performance	11.2-25
	11.2.7.5	Recovery and Corrective Actions	11.2-25
11.2.8	Full Bloc	ckage of Air Inlets and Outlets	11.2-26
	11.2.8.1	Cause of Event	11.2-26
	11.2.8.2	Analysis of the Blockage Event	11.2-26
	11.2.8.3	Radiological Consequences	11.2-27
	11.2.8.4	NAC-MPC Performance	11.2-27
	11.2.8.5	Recovery and/or Corrective Actions	11.2-27
11.2.9	Lightnin	g	11.2-28

		11.2.9.1	Cause of Accident	8
		11.2.9.2	Analysis of the Lightning Strike Event 11.2-28	8
		11.2.9.3	Radiological Consequences 11.2-32	2
		11.2.9.4	NAC-MPC Performance	2
		11.2.9.5	Recovery and/or Corrective Actions 11.2-32	2
	11.2.10	Maximun	n Anticipated Heat Load (125°F Ambient Temperature) 11.2-32	2
		11.2.10.1	Cause of Accident 11.2-33	3
		11.2.10.2	Analysis of the 125°F Ambient Temperature Event 11.2-33	3
		11.2.10.3	Radiological Consequences	1
		11.2.10.4	NAC-MPC Performance 11.2-34	1
		11.2.10.5	Corrective Actions	1
	11.2.11	Storage C	ask 6-Inch Drop 11.2-34	1
		11.2.11.1	Cause of Accident	5
		11.2.11.2	Analysis of the 6-Inch Drop Event 11.2-35	5
		11.2.11.3	Radiological Consequences 11.2-38	3
		11.2.11.4	NAC-MPC Performance 11.2-38	3
		11.2.11.5	Recovery and/or Corrective Actions)
	11.2.12	Tip Over	of the Vertical Concrete Cask 11.2-41	L
		11.2.12.1	Cause of Tip Over 11.2-41	L
		11.2.12.2	Analysis of the Tip Over Event 11.2-41	L
		11.2.12.3	Analysis of the Canister and Basket 11.2-48	3
		11.2.12.4	Radiological Consequences	7
		11.2.12.5	NAC-MPC Performance	7
		11.2.12.6	Recovery and/or Corrective Actions	3
	11.2.13	Tornado a	nd Tornado Driven Missile11.2-88	3
		11.2.13.1	Cause of Tornado Event11.2-88	3
		11.2.13.2	Analysis of the Tornado Event11.2-88	3
		11.2.13.3	Radiological Consequences11.2-101	
		11.2.13.4	NAC-MPC Performance	
		11.2.13.5	Recovery and/or Corrective Actions	2
	_			
11.3	Design	Basis Loadi	ing of the Transportable Storage Canister11.3-1	•
	11.3.1	Canister In	npact Analysis11.3-1	
		11.3.1.1	Finite Element Model Description - Canister	2

<----

.

		11.3.1.2	Canister Bottom and Side Impact Analysis11.3-9
		11.3.1.3	Canister Buckling Evaluation for the Bottom End Impact11.3-16
	11.3.2	Canister	Fuel Basket End Impact Analysis11.3-17
		11.3.2.1	Stress Evaluation of Support Disk in the End Impact11.3-18
		11.3.2.2	Evaluation of Tie Rods and Spacers for an
			End Impact Condition11.3-20
		11.3.2.3	Fuel Tube Analysis11.3-32
		11.3.2.4	Fuel Basket Weldment Analysis for End Impact Conditions11.3-34
114	Reconfi	gured Fue	Assembly Evaluation
	11 4 1	Shell Cas	ing Weldment Evaluation 11.4-1
	11.1.1	11.4.1.1	Shell Casing Side Impact 11.4-1
		11.4.1.2	Shell Casing End Impact
		11 4 1 3	Lifting Tab Welds
	11.4.2	Basket A	ssembly and Fuel Tube Evaluation
		11.4.2.1	Corner Angle Side Impact 11.4-9
		11.4.2.2	Corner Angle End Impact 11.4-10
		11.4.2.3	Fuel Tube Side Impact 11.4-11
		11.4.2.4	Fuel Tube End Impact 11.4-12
		11.4.2.5	Tie Plate End Impact 11.4-14
		11.4.2.6	Tie Plate Side Impact 11.4-17
		11.4.2.7	Tie Plate Thermal Stress Analysis 11.4-19
11.5	Canist	er Closure	Weld Evaluation - Accident Conditions 11.5-1
	11.5.1	Stress E	valuation for the Canister Closure Weld 11.5-1
	11.5.2	Critical	Flaw Size for the Canister Closure Weld 11.5-1
11.6	Refere	nces	
11.0	Refere		

12.0	OPERA	TING CONTROLS AND LIMITS 12-1
12.1	Proposed	Operating Controls and Limits
12.2	Proposed	Training Topics for the NAC-MPC SYSTEM 12-2
12.3	Special F	Requirements for the First System Placed in Service
12.4	Surveilla	nce After a Natural Phenomena Off-Normal or Accident Event 12-2
12.5	Administ	trative Contols
Techr	nical Speci	ifications12A-1
Techr	nical Speci	ification Bases12B-1
13.0	QUALI	TY ASSURANCE 13.1-1
13.1	Introduc	tion 13.1-1
13.2	NAC Qu	ality Assurance Program Synopsis
	13.2.1	Organization
	13.2.2	Quality Assurance Program
	13.2.3	Design Control 13.2-2
	13.2.4	Procurement Document Control
	13.2.5	Procedures, Instructions, and Drawings 13.2-3
	13.2.6	Document Control
	13.2.7	Control of Purchased Items and Services
	13.2.8	Identification and Control of Material, Parts, and Components 13.2-4
	13.2.9	Control of Special Processes
	13.2.9 13.2.10	Control of Special Processes
	13.2.9 13.2.10 13.2.11	Control of Special Processes 13.2-4 Inspection 13.2-5 Test Control 13.2-5

13.2.13	Handling, Storage and Shipping	13.2-6
13.2.14	Inspection, Test and Operating Status	13.2-6
13.2.15	Control of Nonconforming Items	
13.2.16	Corrective Action	
13.2.17	Records	
13.2.18	Audits	

THIS PAGE INTENTIONALLY LEFT BLANK

Table of Contents

1.0 GENERAL DESCRIPTION
1.1 Introduction
1.2 The NAC-MPC System
1.2.1 NAC-MPC System Components
1.2.1.1 Transportable Storage Canister and Baskets
1.2.1.2 Vertical Concrete Cask1.2-4
1.2.1.3 Transfer Cask
1.2.1.4 Auxiliary Equipment1.2-6
1.2.1.5 Transport Cask1.2-8
1.2.2 Operational Features
1.2.3 Cask Contents
1.3 Agents and Contractors
1.4 Generic Storage Cask Arrays 1.4-1
1.5 License Drawings
1.5.1 NAC-MPC License Drawings
1.5.2 Yankee-Class Reconfigured Fuel Assembly License Drawings1.5-2

List of Figures

Figure 1.1-1	Major Components of the NAC-MPC System
Figure 1.1-2	Transportable Storage Canister Showing the Spent Fuel Basket
Figure 1.2-1	Vertical Concrete Storage Cask
Figure 1.2-2	Transfer Cask
Figure 1.2-3	NAC-STC Transport Configuration
Figure 1.2-4	Reconfigured Fuel Assembly
Figure 1.2-5	Transfer Cask and Canister Arrangement
Figure 1.2-6	Vertical Concrete Cask and Transfer Cask Arrangement
Figure 1.2-7	Major Component Configuration for Loading the Vertical Concrete Cask 1.2-19
Figure 1.4-1	Typical ISFSI Storage Pad Layout

List of Tables

Table 1-1	Terminology	1-2
Table 1-2	NUREG-1536 Compliance Matrix	1-6
Table 1.1-1	Design Characteristics of the NAC-MPC	1.1-4
Table 1.2-1	Major Physical Design Parameters for the Transportable	
	Storage Canister	1.2-20
Table 1.2-2	Transportable Storage Canister Fabrication Specification Summary	1.2-21
Table 1.2-3	Major Physical Design Parameters for the Vertical Concrete Cask	1.2-22
Table 1.2-4	Concrete Cask Fabrication Specification Summary	1.2-23
Table 1.2-5	Major Physical Design Parameters for the Transfer Cask	1.2-24
Table 1.2-6	NAC-MPC Design Basis Fuel Characteristics	1.2-25
1.0 GENERAL DESCRIPTION

NAC International Inc. (NAC) has designed a Multi-Purpose Canister system (NAC-MPC) for the long-term storage of spent nuclear fuel. The NAC-MPC system consists of a transportable storage canister, vertical concrete cask, and a transfer cask.

The transportable storage canister is designed and fabricated to meet the requirements for transport in the NAC Storable Transport Cask (NAC-STC) and to be compatible with the U.S. Department of Energy MPC Design Procurement Specification so as not to preclude the possibility of permanent disposal in a deep Mined Geological Disposal System.

In long-term storage, the transportable storage canister is installed in a vertical concrete cask, which provides passive radiation shielding and natural convection cooling. The vertical concrete storage cask also provides protection during storage for the transportable storage canister under adverse environmental conditions. The NAC-MPC employs a double-welded closure design to preclude loss of contents and to preserve the general health and safety of the public during long-term storage of spent fuel.

The transfer cask is used to move the transportable storage canister from the work stations where the canister is loaded and closed to the vertical concrete cask. It is also used to transfer the canister from the vertical concrete cask to the NAC-STC for transport.

This Safety Analysis Report demonstrates the ability of the NAC-MPC System to satisfy the Nuclear Regulatory Commission (NRC) requirements for the storage of spent fuel, as prescribed by 10 CFR 72.

This chapter provides a general description of the major components of the system and a description of the system operation. The terminology used throughout this report is summarized in Table 1-1. Table 1-2 provides a compliance matrix to the regulatory requirements and acceptance criteria specified in NUREG-1536. This matrix describes how the NAC-MPC Safety Analysis Report (SAR) complies with each requirement and criterion listed in NUREG-1536. Table 12A4-1 provides a list of the exceptions from the American Society of Mechanical Engineers Boiler and Pressure Vessel Code (ASME Code) for the canister.

Table 1-1 To	erminology
NAC-STC Cask	The licensed spent-fuel transport cask consisting of a spent fuel storable transport cask body with dual closure lids and energy-absorbing impact limiters (Certificate of Compliance No. 71-9235).
Confinement Sy	The components of the transportable storage canister intended to retain the radioactive material during storage.
Transportable S Canister (Canist	torage The stainless steel cylindrical shell, bottom end plate, shield lid, ander) structural lid that holds the spent fuel in the canister basket.
Contents	Up to 36 intact Yankee Class pressurized-water reactor (PWR) spent fuel assemblies and Reconfigured Fuel Assemblies to a maximum total contents weight of 30,600 pounds in the transportable storage canister.
Yankee Class Sp Fuel	ent Fuel that includes United Nuclear Type A and Type B, Combustion Engineering Type A and Type B, Exxon-ANF Type A and Type B, and Westinghouse Type A and Type B spent fuel assemblies.
Reconfigured Fu Assembly (RFA)	A stainless steel container having the same external dimensions as a standard Yankee Class spent fuel assembly that ensures criticality control geometry and which permits gaseous and liquid media to escape while minimizing dispersal of gross particulates. The reconfigured fuel assembly may contain a maximum of 64 intact fuel rod, damaged fuel rods or fuel debris from any type of Yankee Class spent fuel assembly.
Intact Fuel Asser	nbly A fuel assembly without known or suspected cladding defects greater than a pinhole leak or a hairline crack and which can be handled by normal means. A partial fuel assembly is a fuel assembly from which fuel rods are missing. A partial fuel assembly shall not be classified as an intact fuel assembly unless solid Zircaloy or stainless

1-2

displaced by the original fuel rod(s).

steel rods are used to displace an amount of water equal to that

Table 1-1Terminology (Continued)

Intact Fuel Rod	A fuel rod without known or suspected cladding defects greater than a pinhole leak or a hairline crack.
Damaged Fuel Assembly	A fuel assembly with known or suspected cladding defects greater than a hairline crack or a pinhole leak.
Damaged Fuel Rod	A fuel rod with known or suspected cladding defects greater than a hairline crack or a pinhole leak.
Fuel Debris	Fuel in the form of particles, loose pellets and fragmented rods or assemblies.
Canister Basket	The structure placed in the transportable storage canister to support the fuel assemblies (fuel basket).
- Support Disk	A circular stainless steel plate with square holes machined in a symmetrical pattern that provides the primary lateral load-bearing component of the canister basket.
- Heat Transfer Disk	A circular aluminum plate with square holes machined in a symmetrical pattern. The heat transfer disk enhances heat transfer in the fuel basket.
- Fuel Tube	A stainless steel tube having a square cross-section and BORAL neutron poison material on its exterior surfaces.
- Tie Rod	A stainless steel rod used to align the supports disks and heat transfer disks in the fuel basket structure.
- Split Spacer	Spacers installed on the tie rod between the support disks to properly position, and provide axial support for, the support disks and the heat transfer disks.

Shield Lid	The primary confinement boundary for the canister. It is located directly above the canister basket.		
- Drain Port	A penetration located in the shield lid to permit draining of the canister cavity.		
- Vent Port	A penetration located in the shield lid to aid in draining and backfilling the canister cavity.		
- Port Cover	The stainless steel covers that close the vent and drain ports, which are welded in place following draining, drying, and backfilling operations.		
- Quick Disconnect	The quick-disconnect valved nipple used in the vent and drain ports to facilitate operations.		
Structural Lid	The secondary confinement boundary for the canister. The structural lid provides the lifting point for the loaded canister.		
Vertical Concrete Cask (Concrete Cask) (Storage Cask)	A reinforced concrete cylinder closed at the top end by a shield plug and lid that holds the transportable storage canister during storage. The vertical concrete cask is formed around a steel inner liner and base.		
- Shield Plug	A thick carbon steel plug installed in the top end of the storage cask to reduce skyshine radiation. The shield plug contains a one-inch thick neutron shield.		
- Lid	A thick carbon steel bolted closure for the storage cask. The lid precludes access to the canister and provides additional radiation shielding.		

Table 1-1 Terminology (Continued)

- Liner A thick carbon steel shell that forms the annulus of the concrete storage cask. The liner serves as the inner form during concrete pouring and provides radiation shielding of the canister contents.

- Base A carbon steel weldment that contains the inlet air vents, the storage cask jacking points, and the pedestal that supports the canister inside the storage cask.
- Transfer Cask A shielded lifting device for the empty and loaded canister. It is used for the vertical transfer of the canister between work stations and the storage cask or the transport cask. The transfer cask incorporates bottom doors that permit the vertical loading of the storage and transport casks.
 - Lifting Trunnions Two carbon steel trunnions used to lift and move the transfer cask.

Adapter PlateA carbon steel plate that attaches to the top of the transport or storage
cask to facilitate the installation and alignment of the transfer cask. It
also provides the operating mechanism for the transfer cask bottom
doors.

Margin of Safety An analytically determined value defined as the "factor of safety" minus 1. Factor of safety is also analytically determined and is defined as the allowable stress of a material divided by its actual (calculated) stress.

Table 1-2 NUREG-1536 Compliance Matrix

Chapter 1 – General Description			
Area	Requirement	Acceptance Criteria	Description of Compliance
1. General Description and Operational Features	The application must present a general description and discussion of the DCSS, with special attention to design and operating characteristics, unusual or novel design features, and principal safety considerations. [10 CFR Part 72.24(b)]	The applicant should provide a broad overview and a general, non-proprietary description (including illustrations) of the DCSS, clearly identifying the functions of all components and providing a list of those components classified by the applicant as being "important to safety."	A general description of the system is provided in Section 1.2.
2. Drawings	Structures, systems, and components (SSCs) important to safety must be described in sufficient detail to enable reviewers to evaluate their effectiveness. [10 CFR Part 72.24(c)(3)]	The applicant should provide non- proprietary drawings of the storage system, of sufficient detail, that an interested party can ascertain its major design features and general operations.	Drawings of the system are provided in Section 1.5. Safety classifications are provided in Table 2.3-1
3. DCSS Contents	The applicant must provide specifications for the contents expected to be stored in the DCSS (normally spent fuel). These specifications may include, but not be limited to, type of spent fuel (i.e., boiling- water reactor [BWR], pressurized-water reactor [PWR], or both), maximum allowable enrichment of the fuel before any irradiation, burnup (i.e., megawatt- days/metric ton Uranium), minimum acceptable cooling time of the spent fuel before storage in the DCSS (aged at least 1 year), maximum heat designed to be dissipated, maximum spent fuel loading limit, condition of the spent fuel (i.e., intact assembly or consolidated fuel rods), weight and nature of non-spent fuel contents, and inert atmosphere requirements. [10 CFR Part 72.2(a)(1) and 10 CFR Part 72.236(a)]	The applicant should characterize the fuel and other radioactive wastes expected to be stored in the DCSS. If the potential exists that the DCSS will be used to store degraded fuel, the SAR should include a discussion of how the sub-criticality and retrievability requirements will be maintained.	A description of the contents to be stored is presented in Section 1.2.3.



	Chapter 1 – General Description			
Area	Requirement	Acceptance Criteria	Description of Compliance	
4. Qualifications of the Applicant	The application must include the technical qualifications of the applicant to engage in the proposed activities. Qualifications should include training and experience. [10 CFR Part 72.24(j), 10 CFR Part 72.28(a)]	The reviewer should ensure that the applicant has clearly identified the roles and responsibilities that the DCSS designer, vendor, and other agents, such as potential licensees, fabricators, and contractors will have in the review process. Verify that the applicant has provided clear evidence demonstrating that they are qualified to engage in the proposed activities. In addition, verify that the applicant has delineated the responsibilities for all those who will be involved in the construction and operation of the DCSS if known. The reviewer should ensure that the applicant has specifically defined activities, which they will not perform.	Applicant qualifications are discussed in Section 1.3.	
5. Quality Assurance	The safety analysis report (SAR) must include a description of the applicant's quality assurance (QA) program, with reference to implementing procedures. This description must satisfy the requirements of 10 CFR Part 72, Subpart G, and must be applied to DCSS SSC that are important to safety throughout all design, fabrication, construction, testing, operations, modifications and decommissioning activities. These implementing procedures need not be explicitly included in the application. [10 CFR Part 72.24(n)]	Verify that the applicant has described the proposed QA program, citing the applicable implementing procedures. This description should satisfy all requirements of 10 CFR Part 72, Subpart G, that apply to the design, fabrication, construction, testing, operation, modification, and decommissioning of the DCSS SSCs that are important to safety.	Applicant QA program is presented in Chapter 13.	
6. Consideration of 10 CFR Part 71 Requirements Regarding Transportation	If the DCSS under consideration has previously been reviewed and certified for use as a transportation cask, the application must include a copy of the Certificate of Compliance issued for the DCSS under 10 CFR Part 71, including drawings and other documents referenced in the certificate. [10 CFR 72.230(b)]	If the DCSS under review has previously been evaluated for use as a transportation cask, the submittal should include the Part 71 Certificate of Compliance and associated documents.	The transport SAR and Certificate of Compliance are discussed in Section 1.2.1.	

Chapter 2 – Principal Design Criteria			
Area	Requirement	Acceptance Criteria	Description of Compliance
1. Structures, Systems, and Components Important to Safety	The applicant must identify all SSC that are important to safety, and describe the relationships of non-important to safety SSC on overall DCSS performance. [10 CFR 72.24(c)(3) and 72.44(d)] The applicant must specify the design bases and criteria all SSC that are important to safety. [10 CFR 72.24(c)(1), 72.24(c)(2), 72.120(a), and 72.236(b)]	The applicant should discuss the general configuration of the DCSS, and should provide an overview of specific components and their intended functions. In addition, the applicant should identify those components deemed to be important to safety, and should address the safety functions of those components in terms of how they meet the general design criteria and regulatory requirements discussed above. Additional information concerning specific functional requirements for individual DCSS components are addressed in the subsequent chapters of this SRP	The safety classification of system components are described in Table 2.3-1. The design bases and criteria for the system are specified in Table 2-1. Detailed design criteria are presented in Section 2.2

Table 1-2 NUREG-1536 Compliance Matrix (Continued)

	Chapter 2 - Principal Design Criteria			
	Area	Requirement	Acceptance Criteria	Description of Compliance
2.	Design Bases for Structures, Systems, and Components Important to Safety	The applicant must provide the range of specifications for the spent fuel to be stored in the DCSS. These specifications should include, but are not to be limited to: the type	Detailed descriptions of each of the items listed below are generally found in specific sections of the SAR; however, a brief description of these areas, including a summary of the analytical	Specifications of the spent fuel contents are provided in Section 2.1. Specific physical parameters of the fuel are listed in Table 2.1-1.
a.	Components Important to Safety Spent Fuel Specifications	the DCSS. These specifications should include, but are not to be limited to: the type of spent fuel (i.e., boiling-water reactor [BWR], pressurized-water reactor [PWR], or both); content, weight, dimensions and configurations of the fuel; maximum allowable enrichment of the fuel before any irradiation; maximum fuel burnup (i.e., megawatt-days/MTU); minimum acceptable cooling time of the spent fuel before storage in the DCSS (aged at least 1 year); maximum heat load to be dissipated; maximum spent fuel elements to be loaded; spent fuel condition (i.e., intact assembly or consolidated fuel rods); and any inerting atmosphere requirements. [10 CFR 72.2(a)(1) and 72.236(a)]	the SAR; however, a brief description of these areas, including a summary of the analytical techniques used in the design process, should also be captured in Section 2 of the SAR. This description gives reviewers a perspective on how specific DCSS components interact to meet the regulatory requirements of 10 CFR Part 72. This discussion should be non-proprietary, since it may be used to familiarize interested persons with the design features and bounding conditions of operation of a given DCSS. The applicant should define the range and types of spent fuel or other radioactive materials that the DCSS is designed to store. In addition, these specifications should include, but are not to be limited to, the type of spent fuel (i.e., boiling-water reactor [(BWR], pressurized- water reactor [PWR], or both), weights of the stored materials, dimensions and configurations of the fuel, maximum allowable enrichment of the fuel before any irradiation, burnup (i.e., megawatt-days/MTU), minimum acceptable cooling time of the spent fuel before storage in the DCSS (aged at least 1 year), maximum heat designed to be dissipated, maximum number of spent fuel elements, condition of the spent fuel (i.e., intact assembly or consolidated fuel rods), inerting atmosphere requirements, and the maximum amount of fuel permitted for storage in the DCSS. For DCSSs that will be used to store radioactive materials other than spent fuel, that is, activated components associated with a spent fuel assembly (e.g., control rods, BWR fuel channels) the annicant should specify the	
			types and amounts of radionuclides, heat generation and the relevant source strengths and radiation energy spectra permitted for storage in the DCSS.	

.

L	Chapter 2 - Principal Design Criteria			
	Area	Requirement	Acceptance Criteria	Description of Compliance
2.	Design Bases for Structures, Systems, and Components Important to Safety	The design bases for SSC important to safety must reflect an appropriate consideration of environmental conditions associated with normal operations, as well	The SAR should define the bounding conditions under which the DCSS is expected to operate. Such conditions include both normal and off-normal	The environmental conditions and natural phenomena considered as design bases are described in Section 2.2.
b.	External Conditions	as design considerations for both normal and accident conditions and the effects of natural phenomena events. [10 CFR 72.122(b)]	environmental conditions, as well as accident conditions. In addition, the applicant should consider the effects of natural events, such as tornadoes, earthquakes, floods, and lightning strikes. The effects of such events are addressed in individual chapters of the SRP (e.g., the effects of an earthquake on the DCSS structural components are addressed in Chapter 3, "Structural Analysis").	
3. a.	Design Criteria for Safety Protection Systems General	The DCSS must be designed to safely store the spent fuel for a minimum of 20 years and to permit maintenance as required. [10 CFR 72.236(g)] SSC important to safety must be designed, fabricated, erected, and tested to quality standards commensurate with the importance to safety of the function to be performed. [10 CFR 72.122(a)] The applicant must identify all codes and standards applicable to the SSC. [10 CFR 72.24(c)(4)]	The SAR should define an expected lifetime for the cask design. The staff has accepted a minimum of 20 years as consistent with the licensing period. The applicant should also briefly describe the proposed quality assurance (QA) program, and applicable industry codes and standards that will be applied to the design, fabrication, construction, and operation of the DCSS. In establishing normal and off-normal conditions applicable to the design criteria for DCSS designs, applicants should account for actual facility operating conditions. Design considerations should therefore reflect normal operational ranges, including any seasonal variations or effects	The codes and standards of design and construction of the system are specified in Section 3.1.2.

April 2000 Revision 0

12	ible I-2 NUREG-1530	6 Compliance Matrix (Continued)		
		Chapter 2 - Pr	incipal Design Criteria	
	Area	Requirement	Acceptance Criteria	Description of Compliance
3. b.	Design Criteria for Safety Protection Systems Structural	SSC that are important to safety must be designed to accommodate the combined loads of normal operations, accidents, and natural phenomena events with an	The SAR should define how the DCSS structural components are designed to accommodate combined normal, off- normal, and accident loads, while	A discussion of the structural design criteria are presented in Section 2.2. Combined loadings are addressed specifically in Section 2.2.5 and in Tables
		adequate margin of safety. [10 CFR 72.24(c)(3), 72.122(b), and 72.122(c)] The design-basis earthquake must be equivalent to or exceed the safe shutdown earthquake of a nuclear plant at sites evaluated under 10 CFR Part 100. [10	protecting the DCSS contents from significant structural degradation, criticality, and loss of confinement, while preserving retrievability. This discussion is generally a summary of the analytical techniques and calculational results from the detailed analysis discussed in SAR	2.2-2 and 2.2-3. The design-basis earthquake is specified in Section 2.2.3 in accordance with 10 CFR 72.102 criteria.
		CFR 72.102(f)] The DCSS must maintain confinement of radioactive material within the limits of 10 CFR Part 72 and Part 20, under normal, off-normal, and credible accident conditions. [10 CFR 72.236(1)]	Section 3 and should be presented in a non-proprietary forum.	Because the system maintains adequate margins of safety during normal (Section 3.4.4.1), off-normal (Section 11.1) and accident condition (Section 11.2) events, confinement of the radioactive material is assured.
		The DCSS must be designed and fabricated so that the spent fuel is maintained in a subcritical condition all under all credible normal, off-normal, and accident conditions. [10 CFR 72.124(a) and 72.236(c)]		Because the system maintains adequate structural margins of safety during normal, off-normal and accident condition events, criticality control is assured based on the analyses presented in Chapter 6.
		The spent fuel cladding must be protected during storage against degradation that leads to gross ruptures, or the fuel must be otherwise confined such that degradation of the fuel during storage will not pose operational safety problems with respect to its removal from storage. [10 CFR 72, 122(h)(1)]		The maximum allowable cladding temperatures are specified in Tables 2-1 and 4.1-3. The temperature results for the fuel cladding listed in Table 4.1-4 show that the allowable cladding temperatures are not exceeded. Therefore, the fuel cladding is protected against degradation during storage.
		Storage systems must be designed to allow ready retrieval of spent fuel waste for further processing or disposal. [10 CFR 72.122(1)]		As described in Section 1.2, the system is designed to be readily transported as necessary for further processing or disposal.

<u> </u>	Chapter 2 - Principal Design Criteria			
	Area	Requirement	Acceptance Criteria	Description of Compliance
3. c.	Design Criteria for Safety Protection Systems Thermal	Each spent fuel storage or handling system must be designed with a heat removal capability having testability and reliability consistent with its importance to safety. [10 CFR 72.128(a)(4)] The DCSS must be designed to provide adequate heat removal capacity without active cooling systems. [10 CFR	Acceptance Criteria The applicant should provide a general discussion of the proposed heat removal mechanisms, including the reliability and verifiability of such mechanisms and any associated limitations. All heat removal mechanisms should be passive and independent of intervening actions under normal and off-normal conditions.	The testability of the heat removal capability of the storage system is described in Section 2.3.3.2. The reliability of the heat removal system is demonstrated in Chapter 4, and operating limits are established in Chapter 12 consistent with the temperature monitoring and routine surveillance described in Section 2.3.3.2 to ensure continued safe operation. As shown in Table 4.1-4, the storage system provides adequate heat removal
		72.236(f)]		through the passive cooling design features described in Section 1.2.1.2.

April 2000 Revision 0

	Chapter 2 - Principal Design Criteria			
	Area	Requirement	Acceptance Criteria	Description of Compliance
3. d.	Design Criteria for Safety Protection Systems Shielding/Confinement/ Radiation Protection	The proposed DCSS design must provide radiation shielding and confinement features that are sufficient to meet the requirements of 10 CFR 72.104 and 72.106. [10 CFR 72.126(a), 72.128(a)(2), 72.128(a)(3), and 72.236(d)]	The applicant should describe those features of the cask that protect occupational workers and members of the public against direct radiation dosages and releases of radioactive material, and minimize the dose after any off- normal or accident conditions	The confinement design features are described in Section 2.3.2.1, while the radiation shielding design features are described in Section 2.3.5.
		During normal operations and other anticipated occurrences, the annual dose equivalent to any real individual who is located beyond the controlled area must not exceed 25 mrem to the whole body, 75 mrem to the thyroid, and 25 mrem to any other organ as a result of exposure to (1) planned discharges to the general environment of radioactive materials except radon and its decay products, (2) direct radiation from operations of the ISFSI or monitored retrievable storage (MRS), and (3) any other radiation from uranium fuel cycle operations within the region. [10 CFR 72.24(d), 72.104(a), and 72.236(d)]		Section 10.4 presents the necessary minimum site boundary distances from an array of loaded storage systems to meet the controlled area dose limits.
		Any individual located at or beyond the nearest boundary of the controlled area shall not receive a dose greater than 5 rem to the whole body or any organ from any design-basis accident. The minimum distance from the spent fuel handling and storage facilities to the nearest boundary of the controlled area shall be 100 meters. [10 CFR 72.24(d), 72.24(m), 72.106(b), and 36(d)] The DCSS must be designed to provide redundant sealing of confinement systems. [10 CFR 72.236(e)]		As stated in Section 10.2.2, there is no postulated accident condition that would result in a release of radioactive materials. Therefore, the accident dose limit is met. The redundant sealing features of the confinement system are presented in Section 2.3.2.1.
		Storage confinement systems must have the capability for continuous monitoring in a manner such that the licensee will be able to determine when corrective action needs to be taken to maintain safe storage conditions. [10 CFR 72.122(h)(4) and 72.128(a)(1)] The DCSS design must include inspections, instrumentation and/or control (I&C) systems to monitor the SSC that are important to safety over anticipated ranges for normal and off-normal operation. In addition, the applicant must identify those control systems that must remain operational under accident conditions. [10 CFR 72.122(i)]		As described in Section 2.3.1, the system is passive and can operate through all postulated normal, off-normal, and accident events while maintaining safe storage conditions for the fuel. As specified in Appendix 12A, Section 3.1.7, temperature monitoring is utilized to verify that the systems operating conditions are maintained. Appendix 12A, Section 3.1.7 and Section 12.4 specify the surveillance requirements for the system under normal conditions and after an accident, respectively. These activities are specified to ensure that the system is operated within its design parameters at all times.

April 2000 Revision 0

	Chapter 2 - Principal Design Criteria					
	Area	Requirement	Acceptance Criteria	Description of Compliance		
3. e.	Design Criteria for Safety Protection Systems Criticality	Spent fuel transfer and storage systems must be designed to remain subcritical under all credible conditions. [10 CFR 72.124(a) and 72.236(c)] When practicable, the DCSS must be designed on the basis of favorable geometry, permanently fixed neutron- absorbing materials (poisons), or both. Where solid neutron-absorbing materials are used, the design shall allow for positive means to verify their continued efficacy. [10 CFR 72.124(b)]	The SAR should address the mechanisms and design features that enable the DCSS to maintain spent fuel in a subcritical condition under normal, off-normal, and accident conditions.	The criticality safety design criteria for the system are presented in Section 2.3.4.		

	Chapter 2 - Principal Design Criteria				
	Area	Requirement	Acceptance Criteria	Description of Compliance	
3. f.	Design Criteria for Safety Protection Systems Operating Procedures	The DCSS must be compatible with wet or dry spent fuel loading and unloading procedures. [10 CFR 72.236(h)]	The applicant should provide potential licensees with guidance regarding the content of normal, off-normal, and accident response procedures. Cautions	The operating procedures for the system are presented in Chapter 8 and include procedures for wet and dry loading and unloading operations.	
		Storage systems must be designed to allow ready retrieval of spent fuel for further processing or disposal. [10 CFR 72.122(1)]	The regarding both loading, unloading, and other important procedures should be retrimentioned here. Applicants may choose off-to provide model procedures to be used as fuel	The procedures include methods for retrieving the spent fuel after storage for off-site transport or for return to the spent fuel pool.	
		The DCSS must be designed to minimize the quantity of radioactive waste generated. [10 CFR 72.24(f) and 72.128(a)(5)]	an aid for preparing detailed site-specific procedures.	The decommissioning considerations of the system are described in Section 2.4. Operation of the system generates no radioactive waste, other than a limited amount of protective clothing and tools used during loading operations that could be easily disposed or decontaminated.	
		The applicant must describe equipment and processes proposed to maintain control of radioactive effluents. [10 CFR 72.24(1)(2)]		The radiation protection design features of the system are presented in Section 2.3.5. Operating procedures for the system include provisions for controlling potential effluents from the system.	
		To the extent practicable, the DCSS must be designed to facilitate decontamination. [10 CFR 72.236(I)]		The canister is designed to facilitate decontamination, as described in Section 2.3.5.3.	
		The applicant must establish operational restrictions to meet the limits defined in 10 CFR Part 20 and to ensure that radioactive materials in effluents and direct radiation levels associated with ISFSI operations will remain as low as is reasonably achievable (ALARA). [10 CFR 72.24(e) and 72.104(b)]		Fuel assembly specifications are provided in Appendix 12A, Section 2.1.1 to ensure that doses from effluents and direct radiation are maintained ALARA.	

	Chapter 2 - Principal Design Criteria				
	Area	Requirement	Acceptance Criteria	Description of Compliance	
3.	Design Criteria for Safety Protection Systems	The DCSS design must permit testing and maintenance as required. [10 CFR	The applicant should identify the general commitments and industry codes and	The acceptance tests and maintenance of the system are listed in Chapter 9, including the associated commitment	
g.	Acceptance Tests and Maintenance	SSC that are important to safety must be designed, fabricated, erected, tested, and maintained to quality standards commensurate with the importance to	maintenance, and periodic surveillance tests used to verify the capability of DCSS components to perform their designated functions. In addition, the applicant should discuss the methods used	industry standard or NRC regulation.	
		safety of the function to be performed. [10 CFR 72.24(c), 72.122(a), 72.122(f), and 72.128(a)(1)]	to assess the need for such tests with regard to specific components.		
3.	Design Criteria for Safety Protection Systems	The DCSS must be compatible with wet or dry unloading facilities. [10 CFR 72 236(b)]	Casks should be designed for ease of decontamination and eventual	Decommissioning of the system is discussed in Section 2.4.	
h.	Decommissioning	72.236(h)] The DCSS must be designed for decommissioning. Provisions must be made to facilitate decontamination of structures and equipment and to minimize the quantity of radioactive wastes, contaminated equipment, and contaminated materials at the time the ISFSI is permanently decommissioned. [10 CFR 72.24(f), 72.130, and 72.236(I)] The applicant must provide information concerning the proposed practices and procedures for decontaminating the site and facilities and for disposing of residual radioactive materials after all spent fuel has been removed. Such information must provide reasonable assurance that decontamination and decommissioning will adequately protect the health and safety of the public. [10 CFR 72.24(q) and 72.30(a)]	decommissioning. The applicant should describe the features of the design that support these two activities.		

Chapter 3 – Structural Evaluation				
	Area	F	Regulatory Requirement	Description of Compliance
1.	Structures, Systems, and Components Important to Safety	Structures, systems, a meet the regulatory re and (4), as well as 10	and components (SSC) important to safety must equirements established in 10 CFR 72.24(c)(3) CFR 72.122(a), (b), and (c).	
		10 CFR 72.24(c)(3)	Contents of Application: Descriptions of Components Important to Safety	Component descriptions are provided in Section 1.2. Description of the structural design is provided in Section 3.1.1.
		10 CFR 72.24(c)(4)	Contents of Application: Applicable Codes and Standards	The applicable codes and standards are specified in Sections 3.1.1 and 3.1.2.
		10 CFR 72.122(a)	Overall Requirements: Quality Standards	The quality standards of the system are provided in Table 2.3-1.
		10 CFR 72.122(b)	Overall Requirements: Protection Against Environmental Conditions and natural Phenomena	The system is evaluated structurally for normal operating loads in Sections 3.4.4 and 3.4.5. Off-normal and accident loads are evaluated in Sections 11.1 and 11.2, respectively.
		10 CFR 72.122(c)	Overall Requirements: Protection Against Fires and Explosions	The system is evaluated for fire and explosive loadings in Section 11.2.

	Chapter 3 – Structural Evaluation			
	Area		Regulatory Requirement	Description of Compliance
2.	Radiation, Shielding, Confinement, and Subcriticality	Radiation shielding, regulatory requiremed 72.124(a); and 10 Ci 10 CFR 72.24(d)	confinement, and subcriticality must meet the ents defined in 10 CFR 72.24(d); 10 CFR FR 72.236(c), (d), and (l). Contents of Application: Margins of Safety / Mitigation of Accident Consequences	The margins of safety for normal conditions are listed in Sections 3.4.4.1.5 and 3.4.4.2. Off-normal and accident condition margins of safety are presented in Sections 11.1 and 11.2, respectively. Adequate safety margins are maintained for all events, ensuring the mitigation of accident consequences, and the shielding, confinement, and criticality analyses presented in the SAR.
		10 CFR 72.124(a)	Criteria for Nuclear Criticality Safety: Design for Criticality Safety	The nuclear criticality safety design of the system is discussed in Sections 2.3.4 and 6.1.
		10 CFR 72.236(c)	Specific Requirements for Spent Fuel Storage Cask Approval: Maintain Subcritical Configuration	Subcriticality of the system is demonstrated in Section 6.4.3.
		10 CFR 72.236(d)	Specific Requirements for Spent Fuel Storage Cask Approval: Radiation Protection	Radiation protection of the system is demonstrated in Sections 6.4.3, 10.3 and 10.4.
		10 CFR 72.236(1)	Specific Requirements for Spent Fuel Storage Cask Approval: Maintain Confinement	Confinement of the spent fuel is discussed in Sections 7.2 and 7.3.
3.	Removal of Spent Fuel	As stated in 10 CFR must allow ready retr safety problems.	72.122(f) and (h)(l), the storage system design rieval of spent fuel without posing operational	The system is not adversely affected by normal, off-normal, or accident condition events as demonstrated in Sections 3.4.4.1.5, 3.4.4.2, 11.1 and 11.2. Operating procedures for removing spent fuel from the system are presented in Sections 8.2 and 8.3
4.	Design Basis Earthquake	As stated in 10 CFR must be equal to or g of nuclear plant sites in the case of sites lid Part 100, developed Program (SEP).	72.102(f), the design-basis earthquake (DBE) reater than the safe-shutdown earthquake (SSE) previously evaluated under 10 CFR Part 100 or, censed before the implementation of 10 CFR under Topic III-2 of the Systematic Evaluation	As described in Section 2.2.3.1, the system is designed for a seismic event that is greater than regulatory requirements.
5.	Minimum Lifetime	As stated in 10 CFR and evaluation of the demonstrate that the a minimum of 20 yea	72.24(c) and 10 CFR 72.236(g), the analysis structural design and performance must cask system will allow storage of spent fuel for us with an adequate margin of safety	Section 1.1 and Table 2-1 specify a 50-year design life for the system.

Tuble T 2 Tronds Tobe compriside Human (continued)					
	Area		Regulatory Requirement	Description of Compliance	
6.	Reinforced Concrete Structures	Reinforced concrete ventilation passages protection against na regulations include 1 10 CFR 72.24(c)	structures may have a role in shielding, form and weather enclosures, and providing atural phenomena and accidents. The pertinent 10 CFR 72.24(c) and 10 CFR 72.182(b) and (c). Contents of Application: Design Criteria, Design Bases, Component Descriptions, Codes and Standards	A general description of the Vertical Concrete Cask (VCC) is provided in Section 1.2.1.2. The design criteria for the VCC is presented in Table 2-1. The design bases considered in the structural evaluation of the VCC are presented in Section 2.2.5.1.	
		10 CFR 72.182(b)	Design for Physical Protection: Design Bases / Design Criteria	This requirement is applicable to the ISFSI, not the storage system.	
		10 CFR 72.182(c)	Design for Physical Protection: Security System Description	This requirement is applicable to the ISFSI, not the storage system.	

.

	Chapter 3 – Structural Evaluation	
Area	Acceptance Criteria	Description of Compliance
1. Confinement Cask a. Steel Confinement Cask	 The structural design, fabrication, and testing of the confinement system and its redundant sealing system should comply with an acceptable code or standard, such as Section III of the Boiler and Pressure Vessel Code (B&PV) promulgated by the American Society of Mechanical Engineers (ASME [The NRC has accepted use of either Subsection NB or Subsection NC of this code.]). Other design codes or standards may be acceptable depending on their application. i. The NRC staff evaluates the proposed limitations on allowable stresses and strains in the confinement cask, reinforced concrete components, system components important to safety, and other components subject to review, by comparison with those specified in applicable codes and standards. Where certain proposed load combinations will exceed the accepted limits for localized points on the structure, the applicant should provide adequate justification to show that the deviation will not affect the functional integrity of the structure. ii. The NRC has accepted the use of applicable subsections of the ASME B&PV Code, Division 1, for components used within the confinement cask but not integrated with it. This includes the "basket" structure used in casks to restrain and position multiple fuel elements. 	As specified in Section 3.1.2, the canister and basket structure are designed in accordance with the ASME Code, Section III, Division I, 1995 Edition. The canister is designed using Subsection NB from the code, while the basket structure is designed using Subsection NG criteria. A list of exceptions from the ASME code is provided in Appendix 12A, Table 12A4-1.

	Chapter 3 – Structural Evaluation				
	Area	Acceptance Criteria	Description of Compliance		
b.	Concrete Containments	i. ACI 359 (also designated as Section III, Division 2, of the ASME B&PV Code, Subsection CC) constitutes an acceptable standard for prestressed and reinforced concrete that is an integral component of a radioactive material containment vessel that must withstand internal pressure in operation or testing.	The NAC-MPC system does not utilize concrete containment vessels. Thus, ACI-359 is not applicable.		
		 ii. If ACI 359 pertains to a given ISFSI structure, it applies to all aspects of the design, material selection, fabrication, and construction of that structure. The NRC has not accepted the proposed substitution of elements from ACI 318 or ACI 349 for any portion of ACI 359 with regard to the structure of an ISFSI. ISFSI structures to which ACI 359 applies shall also meet the minimum functional requirements of ANSI/ANS-57.9 for subject areas not specifically addressed in ACI 359. 			
2.	Reinforced Concrete (RC) Structures Important to Safety, but not within the Scope of ACI 359	The NRC accepts the use of ACI 349 for the design, material selection and specification, and construction of all reinforced concrete structures that are not addressed within the scope of ACI 359. However, in such instances, the design, material selection and specification, and construction must also meet any additional or more stringent requirements given in ANSI/ANS-57.9, as incorporated by reference in NRC Regulatory Guide (RG) 3.60. Section V of this chapter provides additional guidance regarding specific review procedures.	As stated in Section 3.1.2, the VCC is designed in accordance with ACI-349 and ANSI/ANS-57.9.		
3.	Other Reinforced Concrete Structures Subject to Approval	The NRC accepts the use of either ACI 318 or ACI 349 for reinforced concrete structures that are subject to approval but are not important to safety. Section V of this chapter provides additional guidance regarding specific review procedures.	The NAC-MPC system has no concrete structures other than that addressed in #2 above.		

	Chapter 3 – Structural Evaluation				
	Area	Acceptance Criteria	Description of Compliance		
4.	Other System Components Important to Safety	The NRC accepts the use of ANSI/ANS-57.9 (together with the codes and standards cited therein) as the basic reference for ISFSI structures important to safety that are not designed in accordance with the Section III of the ASME B&PV Code. However, both the lifting equipment design and the devices for lifting system components that are important to safety must comply with American National Standards Institute (ANSI) Standard N14.6.	The lifting devices of the NAC-MPC system are evaluated in accordance with NUREG-0612 and ANSI N14.6, as specified in Section 3.1.2.		
		The NRC accepts the load combinations shown in Table 3-1 for structures not designed under either Section III of the ASME B&PV Code or ACI 359. These load combinations are based upon ANSI/ANS-57.9, with supplemental definition of terms and combinations.			
		The principal codes and standards include the following references that may apply to steel structures and components:			
		 American Institute of Steel Construction (AISC), "Specification for Structural Steel Buildings — Allowable Stress Design and Plastic Design" 			
		 AISC, "Load and Resistance Factor Design Specification for Structural Steel Buildings" 			
		 c. American Welding Society, "Structural Welding Code Steel," AWS D1.1 			
		 d. American Society of Civil Engineers, "Minimum Design Loads for Buildings and Other Structures," ASCE 7 9however, note that load combinations established on the basis of ANSI/ANS-57.9 [DCSS SRP Table 3-1] are to be used.). 			
		e. ACI 349-85, Appendix B, for embedments or 10.14 for composite compression sections, as applicable, when constructed of structural steel embedded in reinforced concrete.			

	Chapter 3 – Structural Evaluation					
	Area	Acceptance Criteria	Description of Compliance			
5.	Other Components Subject to NRC Approval	For structural design and construction of other components subject to NRC approval, the principal codes and standards include the following:	Not applicable. All components of the system subject to NRC approval are covered by the acceptance criteria specified in the previous sections.			
		a. ASCE 7				
		b. Uniform Building Code (UBC)				
		c. AISC, "Specification for Structural Steel Buildings—Allowable Stress Design and Plastic Design"				
		d. AISC "Code of Standard Practice for Steel Buildings and Bridges"				
		e. ASME B&PV Code, Section VIII				

April 2000 Revision 0

Area Regulatory Requirement Description of Compliance 1. Minimum Lifetime 10 CFR Part 72 requires an analysis and evaluation of DCSS thermal design and performance to demonstrate that the cask will permit cafe Section 1.1 and Table 2-1 specify a 50-year design	n life for
1. Minimum Lifetime 10 CFR Part 72 requires an analysis and evaluation of DCSS thermal design and performance to demonstrate that the cask will permit safe the system. Tables 4.1.2 and 4.1.4 and 4	n life for
storage of the spent fuel for a minimum of 20 years. In permit safe and 4.1-3 and 4.1-4 demonstrate that concrete temperatures are maintained within their a limits.	hat the allowable
2. Spent Fuel Cladding Protection The spent fuel cladding must be protected against degradation that may lead to gross ruptures. Tables 4.1-3 and 4.1-4 demonstrate that the fuel cla temperatures are maintained within allowable limits	ladding
3. Thermal Structures, systems, and components important to safety Thermal structures, systems, and components important to safety 10 CFR 72.24(c)(3) Contents of Application: Descriptions of Components Important to Safety The discussion of the thermal design features of the is presented in Section 4.1. 10 CFR 72.24(c)(3) Contents of Application: Descriptions of Components Important to Safety Tables 4.1-3 and 4.1-4 demonstrate that the temperare maintained within allowable limits for all compute system is not adversely affected by normal, off-norr accident consequences 10 CFR 72.122(h)(1) Overall Requirements: Confinement Barriers and Systems The temperatures of the system are maintained with allowable limits, and do not preclude retrieval of sp from the system. 10 CFR 72.122(h) Overall Requirements: Retrievability The temperature at the outlet vents is meas ensure proper operation of the passive heat removal Capacity 10 CFR 72.236(g) Specific Requirements for Spent Fuel Storage Cask Approval: Warproval: Minimum 20-year Lifetime Image and Unloading Compatibility 10 CFR 72.236(h) Specific Requirements for Spent Fuel Storage Cask Approval: Warproval: Warprovalis Warproval: Warproval: Warproval: Warpro	its. he system eratures iponents of e, the ormal, or ithin spent fuel , Section asured to val system. in life for hat the allowable d in ding and

	Chapter 4 – Thermal Evaluation				
	Area	Acceptance Criteria	Description of Compliance		
1.	Long-term Cladding Temperatures	 Fuel cladding (zircaloy) temperature at the beginning of dry cask storage should generally be below the anticipated damage-threshold temperatures for normal conditions and a minimum of 20 years of cask storage (Refs. 13 and 14). Ref 13: UCID-21181, "Spent Fuel Cladding Integrity During Dry Storage" Ref 14: PNL-6189, "Recommended Temperature Limits for Dry Storage of Spent Light-Water Zircalloy clad fuel Rods in Inert Gas" 	As shown in Table 4.1-4, the fuel cladding temperatures are maintained below 806°F (430°C) for both Zircaloy-clad and Stainless Steel-Clad fuel. This temperature is within the recommended temperature limits for Zircaloy-clad fuel (PNL-6189) and within the limits for Stainless Steel-clad fuel (EPRI TR-106440) for normal, off-normal and accident conditions.		
2.	Short Term Cladding Temperatures	Fuel cladding temperature should generally be maintained below 570 °C (1058 °F) for short-term accident conditions, short-term off- normal conditions, and fuel transfer operations (e.g., vacuum drying of the cask or dry transfer). (PNL-4835)	As shown in Table 4.1-4, the fuel cladding temperature for both Zircaloy and Stainless Steel are maintained below 806°F (430°C) for short term off-normal or accident condition events.		
3.	Maximum Internal Pressure	The maximum internal pressure of the cask should remain within its design pressures for normal, off-normal, and accident conditions assuming rupture of 1 percent, 10 percent, and 100 percent of the fuel rods, respectively. Assumptions for pressure calculations include release of 100 percent of the fill gas and 30 percent of the significant radioactive gases in the fuel rods.	The normal condition pressure calculation is presented in Section 4.4.5. The accident condition pressure calculation is presented in Section 11.2.1. The off-normal condition is bounded by the accident condition, which assumes 100% failure of the cladding.		
4.	Maximum Material Temperatures	Cask and fuel materials should be maintained within their minimum and maximum temperature criteria for normal, off-normal, and accident conditions in order to enable components to perform their intended safety functions.	Tables 4.1-3 and 4.1-4 demonstrate that the temperatures are maintained within allowable limits for all components of the system, including the fuel cladding. Therefore, the system is not adversely affected by normal, off-normal, or accident condition events.		
5.	Fuel Cladding Protection	For each fuel type proposed for storage, the DCSS should ensure a very low probability (e.g., 0.5 percent per fuel rod) of cladding breach during long-term storage.	As concluded in PNL-6189 (Zircaloy) and EPRI TR- 106449 (Stainless Steel), the probability of cladding breech is very low when the cladding temperature is maintained below allowable limits.		
6.	Long-Term Cladding Damage	Fuel cladding damage resulting from creep cavitation should be limited to 15 percent of the original cladding cross-sectional area during dry storage. (UCID-21181)	The maximum fuel cladding temperatures are determined in accordance with PNL-6189 (Zircaloy) and EPRI TR-106440 (Stainless Steel). Calculations were not performed using the methodology of UCID-21181.		
7.	Passive Cooling	The cask system should be passively cooled. [10 CFR 72.236(f)]	As stated in Sections 1.2 and 4.1, the system is passively cooled.		
8.	Thermal Operating Limits	The thermal performance of the cask should be within the allowable design criteria specified in SAR Section 2 (e.g., materials, decay heat specifications) and SAR Section 3 (e.g., thermal stress analysis) for normal, off-normal, and accident conditions.	The thermal stress analyses of the canister and VCC for normal conditions are provided in Sections 3.4.4.1.1 and 3.4.4.2.3, respectively. The canister is evaluated for off- normal thermal stresses in Section 11.1.4.2, and the VCC is analyzed for accident thermal stresses in Section 11.2.10.		

Table 1-2 NUREG-1536 Compliance Matrix (Continued)

April 2000 Revision 0

Chapter 5 – Shielding Evaluation				
Area	Regulatory Requirement	Description of Compliance		
1. Shielding System Description	10 CFR Part 72 requires that spent fuel radioactive waste storage and handling systems be designed with suitable shielding to provide adequate radiation protection under both normal and accident conditions. Consequently, the DCSS application must describe the shielding structures, systems, and components (SSCs) important to safety in sufficient detail to allow the NRC staff to thoroughly evaluate their effectiveness. It is the responsibility of the vendor, the facility owner, and the NRC staff to analyze such SSCs with the objective of assessing the impact of direct radiation doses on public health and safety.	A general description of the system is provided in Section 1.2. A detailed description of the shielding features of the system are provided in Section 5.1.		

	Chapter 5 – Shielding Evaluation			
	Area	R	egulatory Requirement	Description of Compliance
2.	Protection During Accidents	In addition, SSCs imp the effects of both cre without impairing the The applicable shield CFR 72.24(c)(3), 72.2 72.122(c), 72.128(a)(bortant to safety must be designed to withstand dible accidents and severe natural phenomena ir capability to perform their safety functions. ing requirements are identified, in part, in 10 24(d), 72.104(a), 72.106(b), 72.122(b), 2), and 72.236(d).	×
		10 CFR 72.24(c)(3)	Contents of Application: Descriptions of Components Important to Safety	A description of the shielding components of the system is provided in Section 5.1.
		10CFR 72.24(d)	Contents of Application: Margins of Safety / Mitigation of Accident Consequences	The design basis dose rates for accident conditions are listed in Section 10.2.2. Specific details of the dose rates due to the tip-over accident are presented in Section 11.2.12.3.
		10 CFR 72.104(a)	Criteria for Radioactive Materials in Effluents and Direct Radiation from an ISFSI or MRS: Annual Site Boundary Dose Limit	The controlled area boundary dose calculations and minimum site boundary distances are presented in Section 10.4.
		10 CFR 72.106(b)	Controlled Area of an ISFSI or MRS: Design Basis Accident Site Boundary Dose Limit	The accident condition dose rates are discussed in Section 10.2.2.
		10 CFR, 72.122(b)	Overall Requirements: Protection Against Environmental Conditions and Natural Phenomena	Evaluation of the system for off-normal and accident condition events is provided in Sections 11.1 and 11.2. The radiological consequences of each event are addressed.
		10 CFR 72.122(c)	Overall Requirements: Protection Against Fires and Explosions	The radiological consequences of a fire accident are provided in Section 11.2.5.3. The radiological consequences of an explosion are provided in Section 11.2.3.3.
		10 CFR 72.128(a)(2)	Criteria for Spent Fuel Storage and Handling: Radiation Protection	The dose rate results demonstrating the radiation protection of the system are presented in Section 5.1.
		10 CFR 72.236(d)	Specific Requirements for Spent Fuel Storage Cask Approval: Radiation Protection	As described above, the normal condition controlled area boundary dose rates are provided in Section 10.4. The accident condition doses are discussed in Section 10.2.2.

April 2000 Revision 0

	Chapter 5 – Shielding Evaluation				
	Area	Acceptance Criteria	Description of Compliance		
1.	Minimum Distance from Controlled Area Boundary	The minimum distance from each spent fuel handling and storage facility to the controlled area boundary must be at least 100 meters. The "controlled area" is defined in 10 CFR 72.3 as the area immediately surrounding an ISFSI or monitored retrievable storage (MRS) facility, for which the licensee exercises authority regarding its use and within which ISFSI operations are performed.	As described in Section 10.4, the minimum allowable controlled area boundary distance is 150 meters.		
2.	Controlled Area Boundary Dose Limits	The cask vendor must show that, during both normal operations and anticipated occurrences, the radiation shielding features of the proposed DCSS are sufficient to meet the radiation dose requirements in Sections 72.104(a). Specifically, the vendor must demonstrate this capability for a typical array of casks in the most bounding site configuration. For example, the most bounding configuration might be located at the minimum distance (100 meters) to the controlled area boundary, without any shielding from other structures or topography.	Section 10.4 presents the controlled area boundary dose rate evaluation for a typical array configuration. The minimum allowable controlled area boundary distance is 150 meters without taking credit for shielding provided by any intermediate structures or topography.		
3.	ALAKA	Dose rates from the cask must be consistent with a well-established "as low as reasonably achievable" (ALARA) program for activities in and around the storage site.	The dose rates for the system are presented in Section 5.1. These dose rates are within the allowables specified in Section 10.2.1, which are consistent with ALARA principles.		
4.	Maximum Accident Controlled Area Boundary Dose	After a design-basis accident, an individual at the boundary or outside the controlled area shall not receive a dose greater than 5 rem to the whole body or any organ.	Section 10.2.2 indicates that the controlled area boundary dose as a result of an accident will not exceed 5 rem.		
5.	Occupational Dose Limits	The proposed shielding features must ensure that the DCSS meets the regulatory requirements for occupational and radiation dose limits for individual members of the public, as prescribed in 10 CFR Part 20, Subparts C and D.	Occupational doses for typical loading operations are provided in Section 10.3. In practice, occupational doses would be controlled on a site-specific basis by the operator of the ISFSI.		

Ć.

NAC-MPC FSAR Docket No. 72-1025

April 2000 Revision 0

Chapter 6 – Criticality Evaluation			
Area	F	Regulatory Requirement	Description of Compliance
	Spent fuel storage sys unless at least two un spent fuel cask must l credible conditions. F of the cask system are Other pertinent regula and 72.236(g). Norm also identified in 10 (stems must be designed to remain subcritical likely independent events occur. Moreover, the be designed to remain subcritical under all Regulations specific to nuclear criticality safety e specified in 10 CFR 72.124 and 72.236(c). ations include 10 CFR 72.24(c)(3), 72.24(d), nal and accident conditions to be considered are CFR Part 72.	
	10 CFR 72.24(c)(3)	Contents of Application: Descriptions of Components Important to Safety	A general description of the system is provided in Section 1.2, with a detailed description of the criticality safety features of the system provided in Section 6.1.
	10 CFR 72.24(d)	Contents of Application: Margins of Safety / Mitigation of Accident Consequences	Section 6.4 presents the results of the criticality evaluation of the transfer cask and storage cask.
	10 CFR 72.124	Criteria for Nuclear Criticality Safety	The criteria for criticality safety are provided in Sections 2.3.4 and 6.1.
	10 CFR 72.236(c)	Specific Requirements for Spent Fuel Storage Cask Approval: Maintain Subcritical Configuration	Section 6.4 presents the results of the criticality evaluation of the storage cask for the most reactive credible conditions.
	10 CFR 72.236(g)	Specific Requirements for Spent Fuel Storage Cask Approval: Minimum 20-year Lifetime	the system.

		Chapter 6 – Criticality Evaluation	
	Area	Acceptance Criteria	Description of Compliance
1.	Subcriticality Margin	The multiplication factor (k_{eff}) , including all biases and uncertainties at a 95-percent confidence level, should not exceed 0.95 under all credible normal, off-normal, and accident conditions.	As stated in Section 6.1, the maximum allowable multiplication factor (k_s) for the system is 0.95, including adjustment for all biases and uncertainties, as calculated in Section 6.5.
2.	Double Contingency	At least two unlikely, independent, and concurrent or sequential changes to the conditions essential to criticality safety, under normal, off-normal, and accident conditions, should occur before an accidental criticality is deemed to be possible.	As stated in Section 6.1, the criticality analyses are performed for the most reactive credible configuration of the cask, at the highest enrichment, without credit for fuel burnup, and at the most reactive internal water moderator density, even though it is stated that water intrusion is not a credible event. Therefore, criticality cannot occur unless two separate events, such as (1) misloading a higher than design-basis enrichment, unirradiated fuel assembly and (2) water intrusion, occur.
	Criticality Design Features	When practicable, criticality safety of the design should be established on the basis of favorable geometry, permanent fixed neutron-absorbing materials (poisons), or both. Where solid neutron- absorbing materials are used, the design should provide for a positive means to verify their continued efficacy during the storage period.	As stated in Section 6.1, the criticality safety of the design is based on geometry and fixed neutron poisons. Recently proposed rule changes (Federal Register, June 9, 1998) include discussion clarifying the 10 CFR 72.124(b) requirement to verify the "continued efficacy" of neutron poisons as applicable only to wet storage systems, and not to dry, provided that the effectiveness of the poisons is demonstrated at the outset
4.	Conservative Assumptions	 Criticality safety of the cask system should not rely on use of the following credits: a. burnup of the fuel b. fuel-related burnable neutron absorbers c. more than 75 percent for fixed neutron absorbers when subject to standard acceptance tests. 	Section 6.1 provides a list of conservative assumptions that are used in the criticality safety evaluation. No fuel burnup is assumed, and only 75% of the minimum ¹⁰ B loading on the Boral plates is used. Also, no integral fuel burnable neutron absorbers, nor fission product neutron poisons, are considered in the analysis.

NAC-MPC FSAR Docket No. 72-1025

	Chapter 7 – Confinement Evaluation			
	Area	Regulatory Requirement	Description of Compliance	
1.	Description of Structures, Systems, and Components Important to Safety	The SAR must describe the confinement structures, systems, and components (SSCs) important to safety in sufficient detail to facilitate evaluation of their effectiveness. [10 CFR 72.24(c)(3) and 10 CFR 72.24(l)]	A general description of the system is provided in Section 1.2, with a detailed description of the confinement features of the system provided in Section 7.1.	
2.	Protection of Spent Fuel Cladding	The design must adequately protect the spent fuel cladding against degradation that might otherwise lead to gross ruptures during storage, or the fuel must be confined through other means such that fuel degradation during storage will not pose operational safety problems with respect to removal of the fuel from storage. [10 CFR $72.122(h)(1)$]	As described in Sections 7.2.1 and 7.3, the integrity of the canister is maintained under normal and accident conditions. Therefore, the inert helium atmosphere is maintained in the canister, protecting the fuel cladding against degradation.	
3.	Redundant Sealing	The cask design must provide redundant sealing of the confinement boundary. [10 CFR 72.236(e)]	As described in Section 7.1.3.2, the canister is sealed after loading by means of a redundant lid system.	
4.	Monitoring of Confinement System	Storage confinement systems must allow continuous monitoring, such that the licensee will be able to determine when to take corrective action to maintain safe storage conditions. [10 CFR 72.122(h)(4) and 10 CFR 72.128(a)(1)]	As described in Appendix 12A, Section 3.1.7, the system is designed for air outlet temperature monitoring. The canister is designed to withstand all normal, off-normal, and accident condition events, while maintaining the integrity of the confinement boundary.	
5.	Instrumentation	The design must provide instrumentation and controls to monitor systems that are important to safety over anticipated ranges for normal and off-normal operation. In addition, the applicant must identify those control systems that must remain operational under accident conditions. [10 CFR 72.122(i)]	As described in Appendix 12A, Section 3.1.7, the inlet and outlet vent air temperatures are monitored to ensure that temperatures within the confinement boundary are maintained within allowable limits.	
6.	Release of Nuclides to the Environment	The applicant must estimate the quantity of radionuclides expected to be released annually to the environment. [10 CFR 72.24(1)(1)]	As described in Sections 7.2.1 and 7.3, the leaktight integrity of the confinement boundary is maintained during all postulated normal and accident condition events. Therefore, there is no release of radionuclides to the environment.	

	Chapter 7 – Confinement Evaluation				
	Area	Regulatory Requirement	Description of Compliance		
7.	Evaluation of Confinement System	The applicant must evaluate the cask and its systems important to safety, using appropriate tests or other means acceptable to the Commission, to demonstrate that they will reasonably maintain confinement of radioactive material under normal, off-normal, and credible accident conditions. [10 CFR 72.236(l) and 10 CFR 72.24(d)] In addition, SSCs important to safety must be designed to withstand the effects of credible accidents and severe natural phenomena without impairing their capability to perform safety functions. [10 CFR 72.122(b)]	The confinement system is analyzed for normal conditions in Sections 3.4.3.2 and 3.4.4.1, and for off-normal, and accident conditions in Sections 11.1 and 11.2, respectively.		
8.	Annual Dose Limit in Effluents and Direct Radiation from an Independent Spent Fuel Storage Installation (ISFSI)	During normal operations and anticipated occurrences, the annual dose equivalent to any real individual who is located beyond the controlled area must not exceed 25 mrem to the whole body, 75 mrem to the thyroid, and 25 mrem to any other organ. [10 CFR 72.104(a)]	The site boundary dose calculations and minimum site boundary distances are presented in Section 10.4.		

Chapter 7 – Confinement Evaluation			
Area	Acceptance Criteria	Description of Compliance	
1. Redundant Sealing	The cask design must provide redundant sealing of the confinement boundary sealing surface. Typically, this means that field closures of the confinement boundary must either have double seal welds or double metallic o-ring seals.	As described in Section 7.1.3.2, the canister is sealed after loading by means of a redundant lid system.	
2. Code Compliance	The confinement design must be consistent with the regulatory requirements, as well as the applicant's "General Design Criteria" reviewed in Chapter 2 of this SRP. The NRC staff has accepted construction of the primary confinement barrier in conformance with Section III, Subsections NB or NC, of the Boiler and Pressure Vessel (B&PV) Code promulgated by the American Society of Mechanical Engineers (ASME [This code defines the standards for all aspects of construction, including materials, design, fabrication, examination, testing, inspection, and certification required in the manufacture and installation of components.]). In such instances, the staff has relied upon Section III to define the minimum acceptable margin of safety; therefore, the applicant must fully document and completely justify any deviations from the specifications of Section III. In some cases after careful and deliberate consideration, the staff has made exceptions to this requirement.	The codes and standards utilized for the confinement system design are specified in Section 7.1.1. ASME Section III, Subsection NB is utilized for the design of the canister.	
3. Maximum Allowable Leakage Rates	The applicant must specify the maximum allowed leakage rates for the total primary confinement boundary and redundant seals (Applicants frequently display this information in tabular form, including the leakage rate of each seal.). In addition, the applicant's leakage analysis should be consistent with the principles specified in the "American National Standard for Leakage Tests on Packages for Shipment of Radioactive Materials" (ANSI N14.5). Generally, the allowable leakage rate must be evaluated for its radiological consequences and its effect on maintaining the necessary inert atmosphere within the cask.	As specified in Sections 7.2.1 and 7.3, leakage from the confinement system under normal, off-normal, and accident conditions is not credible because the canister is demonstrated to be leaktight.	

	Chapter 7 – Confinement Evaluation			
	Area	Acceptance Criteria	Description of Compliance	
4.	Monitoring and Surveillance	The applicant should describe the proposed monitoring capability and/or surveillance plans for mechanical closure seals. In instances involving welded closures, the staff has previously accepted that no closure monitoring system is required. This practice is consistent with the fact that other welded joints in the confinement system are not monitored. However, the lack of a closure monitoring system has typically been coupled with a periodic surveillance program that would enable the licensee to take timely and appropriate corrective actions to maintain safe storage conditions after closure degradation. The discussion in (a) below taken from Chapter 2 of this SRP expands on the requirement for continuous monitoring.	The system utilizes welded closures, as specified in Section 7.1. Therefore, no monitoring system is required. Routine surveillance is required as specified in Appendix 12A of Chapter 12.	
		(a) Continuous Monitoring		
		The Office of the General Counsel (OGC) has developed an opinion as to what constitutes "continuous monitoring" as required in 10 CFR Part 72.122(h)(4). The staff, in accordance with that opinion has concluded that both routine surveillance programs and active instrumentation meets the intent of "continuous monitoring." Cask vendors may propose, as part of the SAR, either active instrumentation and/or surveillance to show compliance with 10 CFR Part 72.122(h)(4).		
		The reviewer should note that some DCSS designs may contain a component or feature whose continued performance over the licensing period has not been demonstrated to staff with a sufficient level of confidence. Therefore the staff may determine that active monitoring instrumentation is required to provide for the detection of component degradation or failure. This particularly applies to components whose failure immediately affects or threatens public health and safety. In some cases the vendor or staff in order to demonstrate compliance with 10 CFR Part 72.122(h)(4), may propose a technical specification requiring such instrumentation as part of the initial use of a cask system. After initial use, and if warranted and approved by staff, such instrumentation may be discontinued or modified.		

Chapter 7 – Confinement Evaluation				
Area	Acceptance Criteria	Description of Compliance		
5. Non-Reactive Environment	The cask must provide a non-reactive environment to protect fuel assemblies against fuel cladding degradation, which might otherwise lead to gross rupture. Measures for providing a non-reactive environment within the confinement cask typically include drying, evacuating air and water vapor, and backfilling with a non-reactive cover gas (such as helium). For dry storage conditions, experimental data have not demonstrated an acceptably low oxidation rate for UO ₂ spent fuel, over the 20-year licensing period, to permit safe storage in an air atmosphere. Therefore, to reduce the potential for fuel oxidation and subsequent cladding failure, an inert atmosphere (e.g., helium cover gas) has been used for storing UO ₂ spent fuel in a dry environment (See Chapter 8 of this SRP for more detailed information on the cover gas filling process.). Note that other fuel types, such as graphite fuels for the high-temperature gas-cooled reactors (HTGRs), may not exhibit the same oxidation reactions as UO ₂ fuels and, therefore, may not require an inert atmosphere. Applicants proposing to use atmospheres other than inert gas should discuss how the fuel and cladding will be protected from oxidation	As described in Sections 7.1.1.1 and 7.1.2, the confinement system is vacuum dried and backfilled with inert helium gas during loading operations.		

Chapter 8 – Operating Procedures			
Area	Regulatory Requirement	Description of Compliance	
Health and Safety	 The applicant must develop operating procedures that adequately protect health and minimize danger to life or property. [10 CFR 72.40(a)(5)] 	Operating procedures are provided in Chapter 8. Notes and Cautions are listed among the steps provided to emphasize steps important to maintaining health and safety.	
ALAKA	2. The applicant must establish operational restrictions to meet the regulatory requirements of 10 CFR Part 20 and objective limits that are as low as is reasonably achievable (ALARA) for radioactive materials in effluents and direct radiation levels associated with ISFSI operations. [10 CFR 72.104(b) and 10 CFR 72.24(e)]	Section 8.0 specifies that the procedures are developed to maintain occupational dose ALARA. Automated welding systems and temporary shielding are utilized to minimize worker dose during canister loading operations. Appendix 12A, Section 3.2.1 specifies maximum external dose rates to maintain reasonable dose level within a cask array for routine surveillance and inspection activities.	
Control of Radioactive Effluents	3. The applicant must describe all equipment and processes used to maintain control of radioactive effluents. [10 CFR 72.24(l)(2)]	As described in Sections 7.1.1.4, 7.2.1, and 7.3, there are no radioactive effluents in routine operations other than pool water and helium gas that are removed from the canister. These effluents are routinely handled in Licensee operations.	
Written Procedures	 The general licensee shall conduct activities related to storage of spent fuel in accordance with written procedures. [10 CFR 72.212(b)(9)] Vendors seeking approval of a cask design shall ensure that written procedures and appropriate tests are established before initial use of the casks. In addition, the vendor must provide a copy of these procedures and tests to each prospective cask user. [10 CFR 72.234(f)] 	Written procedures for the system are provided in Chapter 8. These procedures are intended to provide general operational guidance for use of the system. These procedures would be used by an ISFSI operator to develop detailed, site specific procedures for use of the system.	
Wet or Dry Loading and Unloading Facilities	6. The cask must be compatible with wet or dry spent fuel loading and unloading facilities. [10 CFR 72.236(h)]	The operating procedures provided in Chapter 8 include procedures for wet and dry loading and unloading operations.	
Decontamination Features	 To the extent practicable, the design of the cask must facilitate decontamination. [10 CFR 72.236(i)] 	The canister is designed to facilitate decontamination as described in Section 2.3.5.3. As described in Section 8.1.1, the annulus between the canister and transfer cask is filled with clean water prior to placement in the fuel pool to minimize the potential for contamination of the surface of the canister.	
4

Chapter 8 – Operating Procedures			
Area	Regulatory Requirement	Description of Compliance	
Ready Retrieval of Spent Fuel8.The design of storage systems must allow ready retrieval of spent fuel for further processing or disposal. [10 CFR 72.122(1)]		The procedure provided in Sections 8.2 and 8.3 specify the steps necessary for retrieval of the spent fuel from the system for further processing or disposal.	
Radioactive Waste Generation	9. The design of the cask must minimize the quantity of radioactive waste generated. [10 CFR 72.128(a)(5) and 10 CFR 72.24(f)]	Operation of the system generates no radioactive waste, other than a limited amount of protective clothing and tools used during loading operations that could be easily disposed or decontaminated.	
Inspection, Maintenance, and Testing	10. The design of structures, systems, and components (SSCs) that are important to safety must permit inspection, maintenance, and testing. [10 CFR 72.122(f)]	Section 9.2 specifies the inspection and maintenance activities required for the system.	

Chapter 8 – Operating Procedures			
Area	Acceptance Criteria	Description of Compliance	
Scope of Application	1. Major operating procedures apply to the principal activities expected to occur during dry cask storage. The expected scope of activities for the SAR operating procedure descriptions is described in Section II, "Areas of Review" (of the SRP), as well as Section 8 of Regulatory Guide 3.61. Operating procedure descriptions should be submitted to address the cask design features and planned operations.	The operating procedures provided in Chapter 8 cover all planned operations of the system, including loading of spent fuel, placement of the system at the site, and unloading of the system.	
Process Control and Hazard Mitigation	 Operating procedure descriptions should identify measures to control processes and mitigate potential hazards that may be present during planned normal operations. Section V, "Review Procedures" (of the SRP), discusses previously identified processes and potential hazards. 	The operating procedures provided in Chapter 8 include Notes and Cautions to indicate steps important in maintaining safety.	
Operating Controls and Limits	3. Operating procedure descriptions should ensure conformance with the applicable operating controls and limits described in the technical specifications provided in SAR Section 12.	The operating controls and limits specified in Section 12.1 are included with the appropriate procedures in Chapter 8 and the Technical Specifications in Appendix 12A. of Chapter 12.	

Chapter 8 – Operating Procedures			
Area	Acceptance Criteria	Description of Compliance	
Operational Planning	4. Operating procedure descriptions should reflect planning to ensure that operations will fulfill the following acceptance criteria:		
	a. Occupational radiation exposures will remain ALARA	As stated in Section 8.0, the operating procedures are developed to support maintaining occupational doses ALARA.	
	b. Effective measures will be taken to preclude potential unplanned and uncontrolled releases of radioactive materials	Sections 8.1.1 and 8.3 include steps to preclude releases of radioactive material during loading and unloading operations. As stated in Sections 7.2.1 and 7.3, release of radioactive material from the system under normal, offnormal and accident conditions is not a credible event.	
	 c. Offsite dose rates will be maintained within the limits of 10 CFR Part 20 and 10 CFR 72.104 for normal operations, and 10 CFR 72.106 for accident conditions. 	Section 10.4 presents the site boundary dose rate evaluation, including the minimum controlled area boundary distance needed to meet an annual dose limit of 25 mrem for normal conditions. Section 10.2.2 indicates that the accident condition controlled area boundary dose will not exceed 5 rem to any organ.	
	In addition, the operating procedure descriptions should support and be consistent with the bases used to estimate radiation exposures and total doses (Refer to Chapter 10 of this SRP).	The operating procedures specified in Chapter 8 and the previous cask loading and unloading experience of NAC support the calculation of occupational dose rates presented in Section 10.3.	

	Chapter 8 – Operating Procedures			
Area	Acceptance Criteria	Description of Compliance		
Surveillance, Maintenance, and Contingency Plans	5. Operating procedure descriptions should include provisions for the following activities:			
	a. testing, surveillance, and monitoring of the stored material and casks during storage and loading and unloading operations	Section 9.2 specifies the inspection and maintenance activities required for the system during storage. The limits established in Appendix 12A, Sections 2.0 and 3.0, are provided to ensure that the spent fuel is protected during loading and unloading operations.		
	b. maintenance of casks and cask functions during storage	Normal operational maintenance and surveillance activities are specified in Section 9.2.		
	c. contingency actions triggered by inspections, checks, observations, instrument readings, and so forth (Some of these may involve off-normal conditions addressed in SAR Section 11.).	These activities include contingency actions that may be required as a result of the inspection.		
Cladding Protection	6. As required by 10 CFR 72.122(h)(1), the operating procedure descriptions should facilitate reducing the amount of water vapor and oxidizing material within the confinement cask to an acceptable level to protect the spent fuel cladding against degradation that might otherwise lead to gross ruptures.	As specified in Appendix 12A, Sections 3.1.2 and 3.1.3, the canister is vacuum dried to eliminate water and backfilled with inert helium gas during fuel loading operations to protect the fuel cladding against oxidation.		

NAC-MPC FSAR

April 2000 Revision 0

Docket	No.	72-1025	

Chapter 9 – Acceptance Test and Maintenance Program			
Area Regulatory Requirement Description of Compliance			Description of Compliance
1. Testing and Maintena	ance a.	The SAR must describe the applicant's program for preoperational testing and initial operations. [10 CFR 72.24(p)]	Section 9.1 presents the acceptance testing for the system.
	b.	The cask design must permit maintenance as required. [10 CFR 72.236(g)]	Section 9.2 presents the maintenance activities for the system.
	c.	Structures, systems, and components (SSCs) important to safety must be designed, fabricated, erected, tested, and maintained to quality standards commensurate with the importance to safety of the function they are intended to perform. [10 CFR 72.122(a), 10 CFR 72.122(f), 10 CFR 72.128(a)(1), and 10 CFR 72.24(c)]	The acceptance tests and maintenance activities presented in Sections 9.1 and 9.2 are performed to verify compliance with the design bases and criteria, and that the system continues to perform as designed.
	d.	The applicant or licensee must establish a test program to ensure that all required testing is performed to meet applicable requirements and acceptance criteria. In addition, at least 30 days before the receipt of spent fuel, the licensee must submit to the NRC a report concerning the pre-operational test acceptance criteria and test results. [10 CFR 72.162 and 10 CFR 72.82(e)]	The testing and maintenance provided in Sections 9.1 and 9.2 are intended to be used by an ISFSI user in the development of site-specific programs.
	e.	The applicant or licensee must evaluate the cask and its systems important to safety, using appropriate tests or other means acceptable to the Commission, to demonstrate that they will reasonably maintain confinement of radioactive material under normal, off-normal, and credible accident conditions. [10 CFR 72.236(1)]	The acceptance tests presented in Section 9.1 demonstrate that the system will maintain confinement of the spent fuel under normal, off-normal, and accident conditions.
	f.	The applicant or licensee must inspect the cask to ascertain that there are no cracks, pinholes, uncontrolled voids, or other defects that could significantly reduce confinement effectiveness. [10 CFR 72.236(j)]	As described in Section 9.1.1, the canister is visually and non-destructively examined prior to use.
	g.	The applicant must perform, and make provisions that permit the Commission to perform, tests that the Commission deems necessary or appropriate. [10 CFR 72.232(b)]	As described in Section 9.2.2, a thermal test is to be performed on the first in-service system. This testing is performed in accordance with a historical NRC certification requirement. Provisions shall be made, as necessary, to
	h.	The general licensee must accurately maintain the record provided by the cask supplier showing any maintenance performed on each cask. This record must include evidence that any maintenance and testing have been conducted under an NRC-approved quality assurance (QA) program. [10 CFR 72.212(b)(8)]	Records of maintenance activities would be maintained by the ISFSI user, and thus are not applicable.
		The applicant or licensee must assure that the casks are conspicuously and durably marked with a model number, unique identification number, and the empty weight [10 CFR 72.236(k)]	As specified in Section 9.1.8, each system is to be marked with the model number, unique cask number, empty weight, and additional information.

	Chapter 9 – Acceptance Test and Maintenance Program			
	Area	Regulatory Requirement	Description of Compliance	
2.	Resolution of Issues Concerning Adequacy or Reliability	 The SAR must identify all SSCs important to safety for which the applicant cannot demonstrate functional adequacy and reliability through previous acceptable evidence. For this purpose, acceptable evidence may be established in any of the following ways: prior use for the intended purpose reference to widely accepted engineering principles reference to performance data in related applications In addition, the SAR should include a schedule showing how the applicant or licensee will resolve any associated safety questions before the initial receipt of spent fuel [10 CFR 72 24(i)]	As described in Sections 3.1 and 3.3, the design of the system is based on industry standard codes and standards for materials and margins of safety. The acceptance tests specified in Section 9.1 are performed to demonstrate the adequacy of each fabricated system in accordance with applied Codes and Standards. The system does not rely on any materials or design standards that lack acceptable evidence of functional adequacy.	
3.	Cask Identification	The applicant or licensee must conspicuously and durably mark the	As specified in Section 9.1.8, each system is to be marked	
		cask with a model number, unique identification number, and empty weight. [10 CFR 72.236(k)]	with the model number, unique cask number, empty weight, and additional information.	

Chapter 9 – Acceptance Tests and Maintenance Program			
Area Acceptance Criteria		Description of Compliance	
Confinement System	American Society of Mechanical Engineers (ASME), "Boiler and Pressure Vessel (B&PV) Code", Section III, Subsection NB or NC "American National Standard for Radioactive Materials— Leakage Tests on Packages for Shipment" (ANSI N14.5-1987)	As specified in Section 3.1.2, the canister is designed in accordance with the ASME Code, Section III, Subsection NB. Exceptions to the Code are provided in Appendix 12A, Table 12A4-1. The confinement system is leak tested in accordance with ANSI N14.5 following shield lid welding as specified in Appendix 12A, Section 3.1.4.	
Confinement Internals (e.g., basket)	ASME B&PV Code, Section III, Subsection NG	As specified in Section 3.1.2, the basket structure is designed in accordance with the ASME Code, Section III, Subsection NG.	
Metal Cask Overpack	ASME B&PV Code, Section VIII	Not applicable.	
Concrete Cask Overpack	American Concrete Institute (ACI) Standards 318 and 349, as appropriate	As stated in Section 3.1.2, the VCC is designed in accordance with ACI-349 and ANSI/ANS-57.9.	
Other Metal Structures	ASME B&PV Code, Section III, Subsection NF American Institute of Steel Construction (AISC), "Manual of Steel Construction"	Not applicable.	

April 2000 Revision 0

Chapter 10 – Radiation Protection			
Area		Regulatory Requirement	Description of Compliance
1. Effluent and Direct Radiation	Criteria for radioacti radiation from an IS 10 CFR 72.104 Cr Di	ive material released due to effluents and direct FSI or MRS are contained 10 CFR 72.104. iteria for Radioactive Materials in Effluents and rect Radiation from an ISFSI or MRS	The controlled area boundary dose calculations and minimum controlled area boundary distances are presented in Section 10.4.
2. Occupational Exposures	Criteria for Occupati 20.1201, 10 CFR 20 10 CFR 20.1201 10 CFR 20.1207 10 CFR 20.1208 10 CFR 20.1301	ional Exposures are contained in 10 CFR .1207, 10 CFR 20.1208, and 10 CFR 20.1301 Occupational Dose Limits for Adults Occupational Dose Limits for Minors Dose to an Embryo/Fetus Dose Limits for Individual Members of the Public	Occupational doses for typical loading operations are provided in Section 10.3. In practice, occupational doses would be controlled on a site-specific basis by the operator of the ISFSI.
3. Public Exposures	Criteria for public ex are contained within 10 CFR 72.104 10 CFR 72.106	posures under normal and accident conditions [10 CFR 72.104 and 10 CFR 72.106] Criteria for Radioactive Materials in Effluents and Direct Radiation from an ISFSI or MRS Controlled Area of an ISFSI or MRS	The controlled area boundary dose calculations and minimum site boundary distances are presented in Section 10.4. Section 10.2.2 indicates that the controlled area boundary dose as a result of an accident will not exceed 5 rem to any organ exclusive of skin

Table 1-2 NUREG-1536 Compliance Matrix (Continued)

1-44

Chapter 10 – Radiation Protection			
Area		Regulatory Requirement	Description of Compliance
4. ALARA	Criteria for ALARA are contained within 10 CFR 20.1101, 10 CFR 72.24(e), 10 CFR 72.104(b), and 10 CFR 72.126(a)		
	10 CFR 20.1101	Radiation Protection Programs	The description of the radiation protection and ALARA considerations of the system are provided in Section 10.1.
	10 CFR 72.24(e)	Contents of Application: ALARA Features	
	10 CFR 72.104(b)	Criteria for Radioactive Materials in Effluents and Direct Radiation from an ISFSI or MRS: Operational Restrictions	The design basis for radiation protection is presented in Section 10.2.
	10 CFR 72.126(a)	Criteria for Radiological Protection: Exposure Control	Operational methods utilized to provide radiation protection are discussed in Section 10.1.3.

		Chapter 10 – Radiation Protection	
ļ	Area	Acceptance Criteria	Description of Compliance
1.	Design Criteria	Limitations on dose rates associated with direct radiation from the cask are established on the basis of the shielding and confinement evaluations in order to satisfy the regulatory requirements for public dose limits. As stated in 10 CFR Part 72.104, during normal operations and anticipated occurrences, the annual dose equivalent to a real individual located beyond the controlled area, must not exceed the limits discussed below.	The dose rate design criteria are specified in Section 10.2.1.
2.	Occupational Exposures	a. dose limits for adults: 5 rem/yr (total effective dose equivalent) b. dose limits for minors: 0.5 rem/yr c. dose to an embryo or fetus (declared pregnant woman): 0.5 rem during entire pregnancy	Occupational doses for typical loading operations are provided in Section 10.3. In practice, occupational doses would be controlled on a site-specific basis by the operator of the ISFSI.
3.	Public Exposures	a. Normal Conditions: whole body: 25 mrem/yr thyroid: 75 mrem/yr other organ: 25 mrem/yr	The controlled area boundary dose calculations and minimum controlled area boundary distances under normal conditions are presented in Section 10.4.
		These doses include the cumulative effects of other nuclear fuel cycle facilities that may be at the same location as the storage system (i.e., the nuclear power plant) and apply to the limiting real individual of the general public residing at a permanent location nearest the facility.	Contribution to the controlled area boundary dose rate from other facilities co-located with the ISFSI are beyond the scope of the SAR, and are addressed on a site-specific basis by the ISFSI operator.
		 b. Accident Conditions and Natural Phenomenon Events 5 rem to the whole body or any organ of any individual located at or beyond the nearest boundary of the controlled area. 	Section 10.2.2 indicates that the controlled area boundary dose as a result of an accident will not exceed 5 rem to any organ, exclusive of skin.

Chapter 10 – Radiation Protection				
Area	Regulatory Requirement	Description of Compliance		
4. ALARA	As a minimum, the proposed ALARA policy must fulfill the following criteria:	The description of the ALARA considerations of the system are provided in Section 10.1.		
	a. To the extent practicable, the applicant should employ procedures and engineering controls that are founded upon sound radiation protection principles.	The operating procedures provided in Chapter 8 are developed to keep occupational doses ALARA.		
	b. Any design change should account for radiation protection, technological, and economical considerations.			
	c. The applicant should have a written policy statement reflecting management commitment to maintain occupational and public exposures to radiation and radioactive material ALARA.			

	Chapter 11 – Accident Analysis			
Area		Pegulatory Poguiroment		
1	Credible Accident and	negulatory nequirement	Description of Compliance	
	Natural Phenomena	Structures, systems, and components (SSC) important to safety must be designed to withstand credible accidents and natural phenomena without impairing their ability to perform safety functions. [10 CFR 72.24(d)(2); 10 CFR 72.122(b)(2), (3), (d), and (g)]	Analyses of the system for a variety of postulated off- normal and accident conditions are presented in Sections 11.1 and 11.2, respectively.	
2.	Controlled Area Boundary Dose	During normal operations and anticipated occurrences, the annual dose equivalent to any real individual who is located beyond the controlled area must not exceed 25 mrem to the whole body, 75 mrem to the thyroid and 25 mrem to any other organ as a result of exposure to the sources listed in the regulations. [10 CFR 72.104(a); 10 CFR 72.236(d); and 10 CFR 72.24(d)]	The controlled area boundary dose calculations and minimum controlled area boundary distances under normal conditions are presented in Section 10.4.	
3.	Design Basis Accident Dose	Dose Limits for Design-Basis Accidents require that any individual located on or beyond the nearest boundary of the controlled area shall not receive a dose greater than 5 rem to the whole body or any organ from any design basis accident. [10 CFR 72.106(b); 10 CFR 72.24(m); and 10 CFR 72.24(d)(2)]	Section 10.2.2 indicates that the controlled area boundary dose as a result of an accident will not exceed 5 rem to any organ, exclusive of skin.	
4.		The spent fuel must be maintained in a subcritical condition under credible conditions. [10 CFR 72.236(c) and 10 CFR 72.124(a)]	Section 6.4 presents the results of the criticality evaluation of the storage cask for the most reactive credible conditions, including the consequences of the off-normal and accident condition events evaluated in Sections 11.1 and 11.2, respectively.	
3.	Continement Control	The cask and its systems important to safety must be evaluated, using appropriate tests or by other means acceptable to the Commission, to demonstrate that they will reasonably maintain confinement of radioactive material under credible accident conditions. [10 CFR 72.236(1)]	As stated in Section 7.3, the confinement system maintains its integrity for all credible off-normal and accident conditions.	
0.	Ready Retrieval of Spent Fuel	Storage systems must allow ready retrieval of spent fuel for further processing or disposal. [10 CFR 72.122(1)]	The off-normal and accident condition analyses presented in Sections 11.1 and 11.2 demonstrate that the spent fuel contents are protected during off-normal and accident conditions. Therefore, retrieval of the spent fuel from the system is not impacted by these postulated events.	

	Chapter 11 – Accident Analysis			
	Area	Regulatory Requirement	Description of Compliance	
7.	Monitoring Systems	Instrumentation and control systems must be provided to monitor systems that are important to safety over anticipated ranges for normal operation and off-normal operation. Those instruments and control systems that must remain operational under accident conditions must be identified in the Safety Analysis Report. [10 CFR 72.122(i)]	The system utilizes temperature monitoring instrumentation but utilizes routine inspection and surveillance to verify proper thermal operation of the system. The confinement system is fully welded and is leak tested to leaktight criteria as specified in Appendix 12A, Section 3.1.4. No seal monitoring is required.	
8.	Surveillance	Where instrumentation and control systems are not appropriate, storage confinement systems must have the capability for continuous monitoring in a manner such that the licensee will be able to determine when corrective action needs to be taken to maintain safe storage conditions. $[72.122(h)(4)]$	No active, continuous monitoring systems are required. Licensee radiological monitoring programs assure ISFSI operations meet 10 CFR 72.104 and 72.106 requirements.	

	Chapter 11 – Accident Analysis		
	Area	Acceptance Criteria	Description of Compliance
1.	Dose Limits for Off- Normal Events	During normal operations and anticipated occurrences, the requirements specified in 10 CFR Part 20 must be met. In addition the annual dose equivalent to any individual located beyond the controlled area must not exceed 25 mrem to the whole body, 75 mrem to the thyroid, and 25 mrem to any other organ as a result of exposure to the following sources:	The controlled area boundary dose calculations and minimum controlled area boundary distances under normal conditions are presented in Section 10.4. No off-normal events are postulated that would result in a controlled area boundary dose in excess of the normal condition analysis.
		 a. planned discharges to the general environment of radioactive materials (with the exception of radon and its decay products) b. direct radiation from operations of the independent spent fuel storage installation (ISFSI) c. any other cumulative radiation from uranium fuel cycle operations (i.e., nuclear power plant) in the affected area 	
2.	Dose Limit for Design- Basis Accidents	Any individual located at or beyond the nearest controlled area boundary must not receive a dose greater than 5 rem to the whole body or any organ from any design-basis accident.	Section 10.2.2 indicates that the controlled area boundary dose as a result of an accident will not exceed 5 rem to any organ, exclusive of skin.
3.	Criticality	The spent fuel must be maintained in a subcritical condition under credible conditions (i.e., k_{eff} equal to or less than 0.95). At least two unlikely, independent, and concurrent or sequential changes must be postulated to occur in the conditions essential to nuclear criticality safety before a nuclear criticality accident is possible (double contingency).	Section 6.4 presents the results of the criticality evaluation of the storage cask for the most reactive credible conditions, including the consequences of the off-normal and accident condition events evaluated in Sections 11.1 and 11.2, respectively. As stated in Section 6.1, the criticality analyses are performed for the most reactive credible configuration of the cask, at the highest enrichment, without credit for fuel burnup, and at the most reactive internal water moderator density, even though it is stated that water intrusion is not a credible event. Therefore, criticality cannot occur unless two separate events, such as (1) misloading a higher than design-basis enrichment, unirradiated fuel assembly and (2) water intrusion, occur.

	Chapter 11 – Accident Analysis			
	Area	Acceptance Criteria	Description of Compliance	
4.	Confinement	The cask and its systems important to safety must be evaluated, using appropriate tests or by other means acceptable to the Commission, to demonstrate that they will reasonably maintain confinement of radioactive material under credible accident conditions.	As stated in Section 7.3, the confinement system maintains its integrity for all credible off-normal and accident conditions.	
5.	Retrievability	Retrievability is the capability to return the stored radioactive material to a safe condition without endangering public health and safety. This generally means ensuring that any potential release of radioactive materials to the environment or radiation exposures is not in excess of the limits in 10 CFR 20 or 10 CFR 72.122(h)(5). ISFSI and MRS storage systems must be designed to allow ready retrieval of the stored spent fuel or high level waste (MRS only) for compliance with 10 CFR 72.122(l).	The off-normal and accident condition analyses presented in Sections 11.1 and 11.2 demonstrate that the spent fuel contents are protected during off-normal and accident conditions. Therefore, retrieval of the spent fuel from the system is not impacted by these postulated events.	
6.	Instrumentation	The SAR must identify all instruments and control systems that must remain operational under accident conditions.	The system does not utilize instrumentation and control systems, but utilizes routine inspection and surveillance to verify proper operation of the system.	

Chapter 12 – Operating Controls and Limits			
Begulatory Requirement			
ridguidadly riequirement			
1. General Requirement for	Fechnical Specifications		
The applicant shall propose technical specifications (complete with acceptable bases and adequate justification). These specifications must include the following five areas [10 CFR 72.44(c), 10 CFR 72.24(g), and 10 CFR 72.26]:			
a. functional/operating limits, monitoring instruments, and limiting controls	Functional and operating limits are specified in Appendix 12A,		
b. limiting conditions	Section 2.0.		
c. surveillance requirements	Limiting conditions for operation are specified in Appendix 12A, Section 3.0.		
d. design features	Surveillance requirements are specified in Appendix 12A, Section 3.0.		
e. administrative controls	Design features are specified in Appendix 12A, Section 4.0.		
Subpart E, "Siting Evaluation Factors," and Subpart F, "General Design Criteria," to 10 CFR Part 72, provide the bases for the cask system design and, hence, are applicable as bases for appropriate technical specifications.	Administrative controls are specified in Section 12.5.		
2. Specific Requirements for Technical Specifications — Storage Cask Approval			
As a condition of approval, the design, fabrication, testing, and maintenance of a spent fuel DCSS must comply with the requirements of 10 CFR 72.236. [10 CFR 72.234(a)]	The operating controls, limits, and surveillance activities specified in Chapter 12 are intended to ensure that the system is maintained within its design basis		
10 CFR 72.236 Specific Requirements for Spent Fuel Storage Cask Approval	anough all normal, off-normal, and accident conditions.		

NAC-MPC FSAR

Docket No. 72-1025

Table 1-2 NUREG-1536 Compliance Matrix (Continued)

Chapter 12 – Operating Controls and Limits			
Regulatory Requirement	Description of Compliance		
The applicant must provide specifications for the spent fuel to be stored in the DCSS. At a minimum, these specifications should include, but not be limited to the following details [10 CFR 72.236(a)]:	Specifications for the spent fuel contents are provided in Appendix 12A, Tables 12A2-1 and 12A2-2.		
 a. type of spent fuel (i.e., BWR, PWR, or both) b. maximum allowable enrichment of the fuel prior to any irradiation c. burn-up (i.e., megawatt-days/MTU) d. minimum acceptable cooling time of the spent fuel prior to storage in the DCSS (minimum 1 year) e. maximum heat that the DCSS system is designed to dissipate f. maximum spent fuel loading limit weights and dimensions h. condition of the spent fuel (i.e., intact assembly or consolidated fuel rods) i. inerting atmosphere requirements 	As specified in Appendix 12A, Section 3.1.3, the canister is backfilled with helium gas to maintain an inert atmosphere for the spent fuel.		
The applicant must provide design bases and design criteria for structures, systems, and components (SSCs) important to safety. [10 CFR 72.236(b)]	The design bases and criteria for the system are specified in Section 2.2.		
The applicant must design and fabricate the DCSS so that the spent fuel will be maintained in a subcritical condition under credible conditions. [10 CFR 72.236(c)]	As shown in Section 6.4, the spent fuel is maintained in a subcritical configuration under all credible configurations.		
The applicant must provide radiation shielding and confinement features that are sufficient to meet the requirements in 10 CFR 72.104 and 72.106 regarding radioactive material in effluents, direct radiation, and area control. [10 CFR 72.236(d) and 10 CFR Part 20]	The maximum external dose rates for the system are specified in Appendix 12A, Section 3.2.1. These limits are established to ensure that, for the minimum controlled area boundary distance presented in Section 10.4, the controlled area boundary annual dose will be maintained within allowable		
10 CFR 72.104Criteria for Radioactive Materials in Effluents and Direct Radiation from an ISFSI or MRS	limits.		
10 CFR 72.106 Controlled Area of an ISFSI or MRS			

•

Chapter 12 – Operating Controls and Limits			
Regulatory Requirement	Description of Compliance		
The applicant must design the DCSS to meet the following criteria:			
• Provide redundant sealing of confinement systems. [10 CFR 72.236(e)]	The redundant sealing features of the confinement system are presented in Section 2.3.2.1 and Chapter 7.		
• Provide adequate heat removal capacity without active cooling systems. [10 CFR 72.236(f)]	As shown in Table 4.1-4, the system provides adequate heat removal through the passive cooling design features described in Section 4.1.		
• Safely store the spent fuel for a minimum of 20 years and permit maintenance as required. [10 CFR 72.236(g)]	Section 1.1 and Table 2-1 specify a 50-year design life for the system. Routine maintenance is permitted as specified by Section 9.2.		
Facilitate decontamination to the extent practicable. [10 CFR 72.236(i)]	Decommissioning of the system is discussed in Section 2.4.		
The DCSS must be compatible with wet or dry spent fuel loading and unloading facilities. [10 CFR 72. 236(h)]	The operating procedures for the system are presented in Chapter 8, and include procedures for wet and dry loading and unloading operations.		
The applicant must inspect the DCSS to ascertain that there are no cracks, pinholes, uncontrolled voids, or other defects that could significantly reduce its confinement effectiveness. [10 CFR 72.236(j)]	As described in Section 9.1.1, the canister is visually and non-destructively examined prior to use.		
The applicant must evaluate the DCSS, and its systems important to safety, using appropriate tests or other means acceptable to the Commission, to demonstrate that they will reasonably maintain confinement of radioactive material under normal, off-normal, and credible accident conditions. [10 CFR 72.236(1)]	The canister is analyzed for normal conditions in Section 3.4.4.1, and for off- normal and accident conditions in Sections 11.1 and 11.2, respectively. Because the canister maintains adequate positive margins of safety, the system will reasonably maintain confinement under all credible conditions.		

Chapter 13 – Quality Assurance		
Regulatory Requirement	Description of Compliance	
According to 10 CFR 72.24, "Contents of Application: Technical Information," the application must include, at a minimum, a description that satisfies the requirements of 10 CFR Part 72, Subpart G, "Quality Assurance," with regard to the QA program to be applied to the design, fabrication, construction, testing, and operation of the DCSS SSCs important to safety. Moreover, Subpart G states that the licensee shall establish the QA program at the earliest practicable time consistent with the schedule for accomplishing the activities.	A synopsis of the NAC Quality Assurance Program is presented in Section 13.2. This program description is consistent with the 18 criteria specified in Subpart G. The Quality Assurance Program is approved by the NRC under 10 CFR 71, Subpart H.	

April 2000 Revision 0

Chapter 13 – Quality Assurance			
	Area	Acceptance Criteria	Description of Compliance
1.	Quality Assurance Organization	The SAR should describe (and illustrate in an appropriate chart) the organizational structure, interrelationships, and areas of functional responsibility and authority for all organizations performing quality- and safety-related activities, including both the applicant's organization and principal contractors, if applicable. Persons or organizations responsible for ensuring that an appropriate QA program has been established and verifying that activities affecting quality have been correctly performed should have sufficient authority, access to work areas, and organizational freedom to carry out that responsibility.	The QA organization is described in Section 13.2.1. An organizational chart is provided in Figure 13.2-1.
2.	Quality Assurance Program	The SAR should provide acceptable evidence that the applicant's proposed QA program will be well-documented, planned, implemented, and maintained to provide the appropriate level of control over activities and SSCs, consistent with their relative importance to safety.	The implementation of the QA program is described in Section 13.2.2.
3.	Design Control	The SAR should describe the approach that the applicant will use to define, control, and verify the design and development of the DCSS. An effective design control program will provide assurance that the proposed DCSS will be appropriately designed and tested and will perform its intended function.	Design control is described in Section 13.2.3.
4.	Procurement Document Control	Documents used to procure SSCs or services should include or reference applicable design bases and other requirements necessary to ensure adequate quality. To the extent necessary, these procurement documents should require that suppliers have a QA program consistent with the quality level of the SSCs or services to be procured.	Procurement document control is described in Section 13.2.4.
5.	Instructions, Procedures, and Drawings	The SAR should define the applicant's proposed procedures for ensuring that activities affecting quality will be prescribed by, and performed in accordance with, documented instructions, procedures, or drawings of a type appropriate for the circumstances.	Procedures, instructions and drawings are described in Section 13.2.5.
6.	Document Control	The SAR should define the applicant's proposed procedures for preparing, issuing, and revising documents that specify quality requirements or prescribe activities affecting quality. These procedures should provide adequate control to ensure that only the latest documents are used. In addition, the applicant's authorized personnel should carefully review and approve the accuracy of all documents and associated revisions before they are released for use	Document control is described in Section 13.2.6.

Table 1-2 NOREO-1550 Compliance Matrix (Continued)				
Chapter 13 – Quality Assurance				
Area	Acceptance Criteria	Description of Compliance		
7. Control of Purchased	The SAR should define the applicant's proposed procedures for	Control of purchased items and services is described in		
Material, Equipment, and	controlling purchased material, equipment, and services to ensure	Section 13.2.7.		
Services	conformance with specified requirements.			
8. Identification and Control	The SAR should define the applicant's proposed provisions for	Identification and control of material, parts and components		
of Materials, Parts, and	identifying and controlling materials, parts, and components to	are described in Section 13.2.8.		
Components	ensure that incorrect or defective SSCs are not used.			
9. Control of Special	The SAR should describe the controls that the applicant will	Control of special processes is described in Section 13.2.9.		
Processes	establish to ensure the acceptability of special processes (such as			
· · · ·	welding, heat treatment, nondestructive testing, and chemical			
	cleaning) and that they are performed by qualified personnel using			
	qualified procedures and equipment.			
10. Licensee Inspection	The SAR should define the applicant's proposed provisions for	Inspection is described in Section 13.2.10.		
	inspection of activities affecting quality to verify conformance with			
	instructions, procedures, and drawings.			
11. Test Control	The SAR should define the applicant's proposed provisions for tests	Test control is described in Section 13.2.11.		
	to verify that SSCs conform to specified requirements and will			
	perform satisfactorily in service. The applicant should specify test			
	requirements in written procedures, including provisions for			
	documenting and evaluating test results. In addition, the applicant			
	should establish qualification programs for test personnel.			
12. Control of Measuring and	The SAR should define the applicant's proposed provisions to ensure	Control of measuring and test equipment is described in		
Test Equipment	that tools, gauges, instruments, and other measuring and testing	Section 13.2.12.		
	devices are properly identified, controlled, calibrated, and adjusted at			
	specified intervals.			
13. Handling, Storage, and	The SAR should define the applicant's proposed provisions to	Handling, storage and shipping are described in Section		
Shipping Control	control the handling, storage, shipping, cleaning, and preservation of	13.2.13.		
	SSCs in accordance with work and inspection instructions to prevent			
	damage, loss, and deterioration.			

.

April 2000 Revision 0

Area	Acceptance Criteria	Description of Compliance
14. Inspection, Test, and	The SAR should define the applicant's proposed provisions to	Inspection, test, and operating status are described in
Operating Status	control the inspection, test, and operating status of SSCs to prevent	Section 13.2.14.
	inadvertent use or bypassing of inspections and tests.	
15. Nonconforming Materials,	The SAR should define the applicant's proposed provisions to	Control of nonconforming items is described in Section
Parts, or Components	control the use or disposition of nonconforming materials, parts, or	13.2.15.
	components.	
16. Corrective Action	The SAR should define the applicant's proposed provisions to ensure	Corrective action is described in Section 13.2.16
	that conditions adverse to quality are promptly identified and	
	corrected and that measures are taken to preclude recurrence.	
17. Quality Assurance	The SAR should define the applicant's proposed provisions for	Records are described in Section 13.2.17
Records	identifying, retaining, retrieving, and maintaining records that	
	document evidence of the control of quality for activities and SSCs	
	important to safety.	
18. Audits	The SAR should define the applicant's proposed provisions for	Audits are described in Section 13.2.18
	planning, scheduling, and conducting audits to verify compliance	
	with all aspects of the QA program, and to determine the	
	effectiveness of the overall program. The SAR should clearly	
	identify responsibilities and procedures for conducting audits,	
	documenting and reviewing audit results, and designating	
	management levels to review and assess audit results. In addition, the	
	SAR should describe the applicant's provisions for incorporating the	
L	status of audit recommendations in management reports.	

	Chapter 14 – Decommissioning			
	Area	Regulatory Requirement	Description of Compliance	
1.	Facility Design Features	The ISFSI or MRS must be designed for decommissioning. Provisions must be made to facilitate decontamination of structures and equipment, minimize the quantity of radioactive wastes and	The design of the ISFSI facility is site-specific, and thus not applicable to a DCSS.	
		contaminated equipment, and facilitate the removal of radioactive wastes and contaminated materials at the time the ISFSI or MRS is permanently decommissioned. [10 CFR 72.130.]	Decommissioning considerations are discussed in Section 2.4.	
2.	Cask Design Features	The cask must be designed to facilitate decontamination to the extent practicable. [10 CFR 72.236(i).]	The decontamination features of the system are discussed in Section 2.4.	
3.	Financial / Records	The requirements for financial assurance and record keeping associated with decommissioning are found in 10 CFR 72.30.10 CFR 72.30Financial Assurance and Record keeping for Decommissioning	Financial and record keeping issues are site-specific, and thus not applicable to a DCSS.	
4.	License Termination	The requirements for terminating an ISFSI license and decommissioning ISFSI sites and buildings are found in 10 CFR 72.54, including the requirements for submitting the final decommissioning plan.	ISFSI license termination is a site-specific issue, and thus not applicable to a DCSS.	

	Chapter 14 – Decommissioning		
Acceptance Criteria Description of Compliance			
1.	Decontamination of buildings and equipment, as specified in RG 1.86.	The decontamination features of the system are discussed in Section 2.4.	
2.	Classification and disposal of wastes, as contained in 10 CFR 61.55.	Not applicable.	

THIS PAGE INTENTIONALLY LEFT BLANK

1.1 Introduction

The NAC-MPC system is a transport compatible dry storage system that uses a vertical concrete storage cask and a stainless steel transportable storage canister (canister) with a welded closure to safely store irradiated nuclear fuel (spent fuel). The canister is stored in the central cavity of the concrete cask and is compatible with the NAC-STC transport cask for future off-site shipment. The concrete storage cask provides radiation shielding and contains internal air flow paths that allow the decay heat from the canister contents to be removed by natural air circulation around the canister wall. The system is designed to meet the requirements for storage of Yankee Class spent fuel. The NAC-MPC is designed and analyzed for a 50-year life.

The principal components of the NAC-MPC system are the canister, the vertical concrete cask and the transfer cask. The loaded canister is moved to and from the concrete cask with the transfer cask. The transfer cask provides radiation shielding while the canister is being closed and sealed and while the canister is being transferred. The canister is placed in the concrete cask by positioning the transfer cask with the loaded canister on top of the concrete cask and lowering the canister into the concrete cask. Figure 1.1-1 depicts the major components of the NAC-MPC system and shows the transfer cask positioned on the top of the concrete cask.

The NAC-MPC is designed to safely store up to 36 intact Yankee Class spent fuel and reconfigured fuel assemblies. The fuel is initially loaded into a canister containing a fuel basket. Figure 1.1-2 depicts the canister and the spent fuel basket. The design characteristics of the NAC-MPC are shown in Table 1.1-1.

Yankee Class fuel includes United Nuclear, Combustion Engineering, Exxon-ANF, and Westinghouse Type A and Type B fuel designs. The Type A and Type B fuel designs are complementary configurations that accommodate the use of a cruciform control blade in reactor operations. The fuel specifications that serve as the design basis for the NAC-MPC are presented in Section 2.1.

The system design and analyses were performed in accordance with Title 10, Code of Federal Regulations, Part 72 (10 CFR 72), ANSI/ANS 57.9-1984 and the applicable sections of the ASME Boiler and Pressure Vessel Code and the American Concrete Institute Code.

1.1-1





NAC-MPC FSAR	April 2000
Docket No. 72-1025	Revision 0





Table 1.1-1	Design Characteristics of the NAC-MPC
-------------	---------------------------------------

Dimension ¹	Material
5/8 thick Plate	Type 304L Stainless Steel
1.0 thick Plate	Type 304L Stainless Steel
5.0 thick Plate	Type 304 Stainless Steel
3.0 thick Plate	Type 304L Stainless Steel
0.5 x 68.98 dia.	Type 304 Stainless Steel
0.5 x 69.15 dia.	Type 17-4 PH Stainless Steel
0.5 x 68.70 dia.	Type 6061-T6 Aluminum Alloy
7.80 x 7.80 x 0.048	Type 304 Stainless Steel encasing BORAL
2.5 diameter	Type 304 Stainless Steel
1-1/8 diameter	Type 304 Stainless Steel
	Dimension ¹ 5/8 thick Plate 1.0 thick Plate 5.0 thick Plate 3.0 thick Plate 0.5 x 68.98 dia. 0.5 x 69.15 dia. 0.5 x 68.70 dia. 7.80 x 7.80 x 0.048 2.5 diameter 1-1/8 diameter

1. Dimensions in inches unless otherwise noted.

Design Characteristic	Dimension ¹	Material
Transfer Cask		
- Outer Shell	1.5 x 86.5 dia.	ASTM A588 Low Alloy Steel
- Inner Shell	0.75 x 73.0 dia.	ASTM A588 Low Alloy Steel
- Retaining Ring	0.75 x 80.8 dia.	ASTM A588 Low Alloy Steel
- Trunnions	10.0 dia.	A350 LF2 Carbon Steel
- Bottom Plate	1.0 thick Plate	ASTM A588 Low Alloy Steel
- Top Plate	2.0 thick Plate	ASTM A588 Low Alloy Steel
-Shield Doors	9 1/2 thick Plate	A350 LF2 Carbon Steel
- Door Rails	9.88 x 6.5	A350 LF2 Carbon Steel
- Gamma Shield	3.5 thick	ASTM B29, Chem. Grade Lead
- Neutron Shield	2.0 thick	NS-4-FR, Solid Synthetic Polymer
Transfer Adapter		
- Base Plate	2 thick Plate	ASTM A36 Carbon Steel
- Locating Ring	2.0 wide x 78.25 dia.	ASTM A36 Carbon Steel

Table 1.1-1 Design Characteristics of the NAC-MPC (Continued)

1. Dimensions in inches unless otherwise noted.

Table 1.1-1 Design Characteristics of the NAC-MPC (Continued)

Design Characteristic	Dimension ¹	Material
Vertical Concrete Cask		
Weldment Structure		
- Shell	3.5 thick x 86.0 dia.	ASTM A36 Carbon Steel
- Top Flange	2.0 thick x 97.9 dia.	ASTM A36 Carbon Steel
- Support Ring	2.5 thick x 79.0 dia.	ASTM A36 Carbon Steel
- Base Plate	2.0 thick x 72.0 dia.	ASTM A36 Carbon Steel
Concrete Cask		
- Concrete Shell	21.0 thick x 128 dia.	Type II Portland Cement
- Shield Plug	5.13 x 78.5 dia.	ASTM A36, Carbon Steel NS-4-FR
- Top Plate	1.5 thick x 92.1 dia.	ASTM A36, Carbon Steel
- Rebar	Various	ASTM A615, GR 60, Carbon Steel

1. Dimensions in inches unless otherwise noted.

1.2 The NAC-MPC System

The NAC-MPC system provides long-term storage and subsequent transport of Yankee Class spent fuel using the certified NAC-STC. During long-term storage, the system provides an inert environment; passive shielding, cooling, and criticality control; and, a confinement boundary closed by welding. The structural integrity of the system precludes the release of contents in any of the design basis normal conditions and off-normal or accident events, thereby assuring public health and safety during use of the system.

1.2.1 NAC-MPC System Components

The NAC-MPC system consists of three principal components:

- Transportable storage canister (canister),
- Vertical concrete cask, and
- Transfer cask.

Ancillary equipment needed to use the NAC-MPC System is:

- Automated or manual welding equipment;
- An air pallet or hydraulic roller skid (used to move the storage cask on and off the heavy haul transfer trailer and to position the storage cask on the storage pad);
- Suction pump, vacuum drying, helium backfill and leak detection equipment;
- A heavy haul trailer or cask transporter (for storage cask transport to the storage pad);
- Adapter plate and hardware to position the transfer cask with respect to the storage or transport cask; and,
- A lifting yoke for the transfer cask and lifting slings for the canister and canister lids.

In addition to these items, the system requires utility services (electric, air and water), common tools and fittings, and miscellaneous hardware.

The transportable storage canister is designed to be transported in the NAC-STC Storable Transport Cask (Certificate of Compliance No. 71-9235). The transport load conditions produce higher stresses in the canister than would be produced by the storage load conditions alone.

NAC-MPC FSAR	April 2000
Docket No. 72-1025	Revision 0

Consequently, the canister design is conservative with respect to storage conditions. The evaluation of the canister for transport conditions is found in the Safety Analysis Report for the NAC-STC Storable Transport Cask, Docket No. 71-9235.

1.2.1.1 Transportable Storage Canister and Baskets

The transportable storage canister contains a basket that accommodates up to 36 intact Yankee Class spent fuel assemblies and reconfigured fuel assemblies (RFA) up to a total contents weight of 30,600 pounds.

The canister assembly consists of a right circular cylindrical shell with a welded bottom plate, a fuel basket, a shield lid, two penetration port covers, and a structural lid. The cylindrical shell, plus the bottom plate and lids, constitutes the confinement boundaries. The fuel basket is based on the directly loaded fuel basket design used in the certified NAC-STC. This basket features the NAC-patented poison tubes and stacked disk design with heat transfer disks. The basket was analyzed using the ANSYS computer code to demonstrate that it can withstand the horizontal drop loads without deforming in a way that damages or constrains a fuel assembly. This tube and disk design has been accepted and approved by the NRC, pursuant to 10 CFR 71 and 10 CFR 72. Table 1.2-1 summarizes the major physical design parameters of the canister.

The fuel basket design is a right-circular cylinder configuration with 36 fuel tubes laterally supported by a series of support disks, which are retained by spacers on eight radially located tie rods. The support disks are stainless steel (17-4 PH) with holes for the poison fuel tubes. The basket top and bottom weldments are fabricated from Type 304 stainless steel. The tie rods and spacer sleeves are also fabricated from Type 304 stainless steel. The fuel assemblies are contained in fuel tubes. The fuel tubes are fabricated from Type 304 stainless steel with encased BORAL sheets on all four outside surfaces of the fuel tube. The BORAL provides criticality control in the basket.

The heat transfer disks are aluminum with holes for the fuel tubes. The heat transfer disks are spaced midway between the support disks and are the primary path for conducting the heat from the fuel assemblies to the canister wall. Holes in the heat transfer disks for the tubes and tie rods are sized to accommodate thermal expansion occurring after the fuel is placed into the basket.

NAC-MPC FSAR	April 2000
Docket No. 72-1025	Revision 0

The fuel basket tube and disk design provides the structural integrity to maintain the spent fuel in a subcritical configuration during normal operations and the hypothetical accident events, even if optimum moderator condition and fresh fuel are assumed. With the most reactive fuel, the fuel basket maintains $K_{eff} \leq 0.95$. Subcriticality is assured assuming fresh fuel loading and no soluble boron in the spent fuel pool water during fuel loading operations.

The transportable storage canister assembly is designed to facilitate filling with water and subsequent draining and drying. A rounded notch is located at the bottom of each fuel tube, ensuring free flow of water between the inner tube regions and the disk regions. Each of the disks also has three holes to supplement the flow of water between disks. In addition, the bottom plate is positioned by supports above the bottom of the canister to facilitate water flow to the drain line.

The canister is fabricated from 5/8-inch thick Type 304L stainless steel rolled plate, joined at its edges by a full penetration weld, which is radiographed. The bottom closure is a 1-inch thick Type 304L stainless steel plate joined to the canister shell by a full penetration weld, which is ultrasonically examined. The stainless steel material was selected to be compatible with the DOE MPC program guidelines for future disposal and to minimize the potential for any adverse chemical reactions in the spent fuel pool. The design of the shield lid and structural lid allows a redundant confinement seal at the top of the canister. Each lid weld is liquid penetrant examined.

The vent and drain ports through the shield lid allow the inner cavity to be drained, evacuated, and backfilled with helium to provide an inert atmosphere for long-term dry storage. The drain port is equipped with a quick disconnect fitting and a drain tube that extends nearly to the bottom of the canister. The vent port extends to the underside of the shield lid and is equipped with a quick disconnect fitting used for vacuum drying and helium backfilling. After draining, drying, backfilling, and testing operations are complete, port covers are installed and welded to the shield lid to seal the penetration.

A summary of the canister fabrication specifications is presented in Table 1.2-2.

1.2.1.2 Vertical Concrete Cask

The vertical concrete cask (storage cask) is the storage overpack for the transportable storage canister. It provides structural support, shielding, protection from environmental conditions, and natural convection cooling of the canister during long-term storage. Table 1.2-3 lists the principal physical design parameters of the storage cask. The storage cask is a reinforced concrete (Type II Portland cement) structure with a structural steel inner liner. The concrete wall and steel liner provide the neutron and gamma radiation shielding to reduce the average contact dose rate to less than 50 millirem per hour for design basis fuel. Inner and outer reinforcing steel (rebar) assemblies are contained within the concrete. The reinforced concrete wall provides the structural strength to protect the canister and its contents in natural phenomena events such as tornado wind loading and wind driven missiles. The storage cask incorporates reinforced chamfered corners at the edges to facilitate construction. "Fire block," an insulating material provided by BISCO, is placed on the base of the cavity to prevent contact between the stainless steel canister and the carbon steel pedestal. The storage cask is shown in Figure 1.2-1.

The storage cask has an annular air passage to allow the natural circulation of air around the canister to remove the decay heat from the spent fuel. The air inlet and outlet vents are steellined penetrations that take nonplanar paths to the concrete cask cavity to minimize radiation streaming. The decay heat is transferred from the fuel assemblies to the tubes in the fuel basket and through the heat transfer disks to the canister wall. Heat flows by radiation and convection from the canister wall to the air circulating through the concrete cask annular air passage and is exhausted through the air outlet vents. This passive cooling system is designed to maintain the peak cladding temperature of both stainless steel and zircaloy clad fuel well below acceptable limits during long-term storage. This design also maintains the bulk concrete temperature below 150°F and localized concrete temperatures below 200°F in normal operating conditions.

The top of the storage cask is closed by a shield plug and lid. The shield plug is approximately 5 inches thick and incorporates carbon steel plate as gamma radiation shielding and NS-4-FR as neutron radiation shielding. A carbon steel lid that provides additional gamma radiation shielding is installed above the shield plug. The shield plug and lid reduce skyshine radiation and provide a cover and seal to protect the canister from the environment and postulated tornado

1.2-4

missiles. The lid is bolted in place and has tamper indicating seals on two of the installation bolts.

Fabrication of the storage cask involves no unique or unusual forming, concrete placement, or reinforcement requirements. The concrete portion of the storage cask is constructed by placing concrete between a reusable, exterior form and the inner metal liner. Reinforcing bars are used at the inner and outer concrete surfaces to provide structural integrity. The inner liner and base of the storage cask are shop fabricated.

The principal fabrication specifications for the storage cask are shown in Table 1.2-4.

1.2.1.3 Transfer Cask

The transfer cask, with its lifting yoke, is primarily a lifting device used to move the canister assembly. It provides biological shielding when it contains a loaded canister. The transfer cask is used for the vertical transfer of the canister between work stations and the storage cask, or transport cask. The general arrangement of the transfer cask and canister is shown in Figure 1.2-5, and the arrangement of the transfer cask and concrete cask is shown in Figure 1.2-6. The configuration of the transfer cask, canister and concrete cask during loading of the concrete cask is shown in Figure 1.2-7.

Table 1.2-5 shows the principal design parameters of the transfer cask.

The transfer cask is a multiwall (steel/lead/NS-4-FR neutron shield/steel) design, which limits the average contact radiation dose rate to less than 200 mrem/hr. The transfer cask design incorporates a top retaining ring, which is bolted in place, that prevents a loaded canister from being inadvertently removed through the top of the transfer cask. The transfer cask has retractable bottom shield doors. During loading operations, the doors are closed and secured by pins, so they cannot inadvertently open. During unloading, the doors are retracted using hydraulic cylinders to allow the canister to be lowered into the storage or transport casks. The transfer cask is shown in Figure 1.2-2.

To qualify the transfer cask as a heavy lifting device, it is designed, fabricated, and load tested to the requirements of NUREG-0612 and ANSI N14.6.

To minimize potential contamination during loading operations in the spent fuel pool, clean water is circulated in the gap between the transfer cask interior surface and the canister exterior surface. The transfer cask has two supply and two discharge lines passing through its wall.

These lines allow hoses to be connected and clean water to be pumped into and through the annular gap to preclude the intrusion of pool water when the canister is being loaded.

Exposed surfaces of the transfer cask, other than the load-bearing surfaces of the trunnions and the bottom door rails, are coated with Keeler & Long E-Series epoxy enamel to protect the carbon steel and to provide a smooth surface to facilitate decontamination.

1.2.1.4 <u>Auxiliary Equipment</u>

This section presents a brief description of the principal auxiliary equipment needed to operate the NAC-MPC in accordance with its design.

1.2.1.4.1 Adapter Plate

The adapter plate is a carbon steel table that bolts to the top of the vertical concrete (storage) cask or the NAC-STC and mates the transfer cask to either of those casks. It has a large center hole that allows the transportable storage canister to be raised or lowered through the plate into or out of the transfer cask. Rails are incorporated in the adapter plate to guide and support the bottom shield doors of the transfer cask when they are in the open position. The adapter plate also supports the hydraulic system and the actuators that open and close the transfer cask bottom doors.

1.2.1.4.2 Air Pad Rig Set

The air pad rig set (air pad set) is a commercially available device, sometimes referred to as an air pallet. When inflated, the air pad set lifts the concrete cask by using high volume air. The air pads employ a continuous, regulated air flow and a control system that equalizes lifting heights of the four air pads by regulating compressed air flow to each of the air pads. The compressed air supply creates an air filler between the inflated air cushion and the supporting surface. The thin film of air allows the concrete cask to be lifted and moved. Once lifted, a suitable towing vehicle, such as a commercial tug or forklift can move the storage cask.

1.2.1.4.3 <u>Automatic Welding System</u>

The automatic welding system consists of commercially available components with a customized weld head. The components include: a welding machine, a remote pendant, a carriage, a drive motor and welding wire motor, and the weld head. The system is designed to make at least one weld pass automatically around the canister after its weld tip is manually positioned at the proper location. As a result, radiation exposure during canister closure is much less than would be incurred from manual welding.
1.2.1.4.4 Draining and Drying System

The draining and drying system consists of a suction pump and a vacuum pump. The suction pump is used to remove free water from the canister cavity. The vacuum pump is a two-stage unit for drying the interior of the canister. The first stage is a large capacity or "roughing" pump intended to remove free water not removed by the suction pump. The second stage is a vacuum pump used to evacuate the canister interior of the small amounts of remaining moisture and establish the vacuum condition.

1.2.1.4.5 Helium Leak Test Equipment

A helium leak detector is required to verify the integrity of the welds of the canister shield lid. The helium leak detector is the mass spectrometer type.

1.2.1.4.6 <u>Heavy-Haul Trailer</u>

The heavy haul trailer is used to move the vertical concrete storage cask. A special trailer has been designed for transport of the empty or loaded storage cask. The design incorporates a builtin jacking system that facilitates raising the storage cask to allow installation of the air pad set used to move the cask onto the storage pad. The trailer incorporates both reinforcing to increase the trailer load-bearing area and design features that reduce its turning radius. However, any commercial double-drop-frame trailer having a deck height approximately matching that of the storage pad could be used.

1.2.1.4.7 Lifting Jacks

Hydraulic jacks are installed at jacking pads in the bottom air ducts to lift the storage cask so that the air pad set can be installed or removed. Four hydraulic pad jacks are provided, along with a control panel, an electric hydraulic oil pump, an oil reservoir tank and all hydraulic lines and fittings. The jacks are used to lift the cask approximately three inches. This permits installation of four air pads under the concrete cask.

1.2.1.4.8 Rigging and Slings

Load rated rigging attachments and slings are provided for major components. The rigging attachments are swivel hoist rings that allow attachment of the slings to the hook. All slings are commercially purchased to have adequate safety margin to meet the requirements of ANSI N14.6 and NUREG 0612. The slings include a concrete cask lid sling, concrete cask shield plug sling, and canister shield lid sling, loaded canister transfer sling (also used to handle the structural lid), canister retaining ring sling. The appropriate rings or eye bolts are provided to accommodate each sling and component.

The transfer cask lifting yoke is specially designed and fabricated for lifting the transfer cask. It is designed to meet the requirements of ANSI N14.6 and NUREG 0612. It is single-failure-proof by design. The transfer cask lifting yoke is initially load tested to 300 percent of the design load.

1.2.1.4.9 <u>Temperature Instrumentation</u>

The vertical concrete cask has four outlet vents near the top of the cask and four inlet vents at the bottom. Each outlet is equipped with a permanent remote temperature detector mounted in the outlet air plenum. The detector is used to measure the outlet air temperature, which can be read at a display device located on the outside surface of the concrete cask or at a remote location. The detectors are installed on all of the storage casks at the ISFSI. One inlet of one concrete cask, or the ambient temperature of the ISFSI site, is also monitored.

1.2.1.5 Transport Cask

The transportable storage canister is designed to be transported in the NAC-STC. The canister is positioned in the NAC-STC cavity with two axial spacers. The spacers are required because the transport cask cavity length is 165 inches, while the length of the canister is 122.5 inches.

The NAC-STC is licensed by the NRC pursuant to 10 CFR 71 (Certificate of Compliance No. 71-9235). A request for an amendment to the NAC-STC Certificate of Compliance to incorporate transport of the canister was submitted to the NRC on December 30, 1996. The NAC-STC is designed for free interchange/rail shipment.

The transport configuration of the NAC-STC is shown in Figure 1.2-3.

1.2.2 <u>Operational Features</u>

This section outlines the principal handling activities of the NAC-MPC storage system. The system provides passive long-term storage of spent fuel in an inert environment. In storage, the only active system is for temperature monitoring of the outlet air. This temperature is recorded daily as a check of the thermal performance of the storage system design. This system does not penetrate the confinement boundary and is not essential to the safe operation of the NAC-MPC.

The principal activities associated with the use of the system are closing the canister and loading the canister in the storage cask. The transfer cask is designed to meet the requirements of these operations. The transfer cask holds the canister during loading with fuel; provides biological shielding during closing of the canister; and provides the means by which the loaded canister is moved to, and installed in, the storage cask. The canister assembly consists of five principal components: the canister shell (side wall and bottom), the shield lid, the vent port, the drain port (together with the vent and drain port covers), and the structural lid. A drain tube extends from the shield lid drain port to the bottom of the canister. The location of the drain and vent ports is shown in Figure 8.1-1.

The vent and drain ports allow the draining, vacuum drying, and backfilling with helium necessary to provide a dry, inert atmosphere for the contents. The vent and drain port covers, the shield lid, the canister shell, and the joining welds form the primary confinement boundary. This boundary is shown in Figure 7.1-1. A secondary confinement boundary is formed over the shield lid by the structural lid and the weld that joins it to the canister shell. This boundary is shown in Figure 7.1-2.

The structural lid contains the drilled and tapped holes for attachment of the swivel hoist rings used to lift the loaded canister. The drilled and tapped holes are filled with bolts or plugs to avoid collecting debris, and to preclude the possibility of radiation streaming from the holes, when the hoist rings are not installed.

NAC-MPC FSAR	April 2000
Docket No. 72-1025	Revision 0

The step-by-step procedures for use of the NAC-MPC system are presented in Chapter 8. The following list presents a brief description of the principal activities. This list assumes that the empty canister is installed in the transfer cask for spent fuel pool loading.

- Lift the transfer cask over the pool and start the flow of water to the transfer cask annulus and canister. After the annulus and canister are filled, lower the cask to the bottom of the pool.
- Load the selected spent fuel assemblies into the canister and set the shield lid.
- Raise the transfer cask from the pool. Decontaminate the transfer cask exterior as it clears the pool surface. Drain the annulus. Place the transfer cask in the decontamination area.
- Weld the shield lid to the canister shell. Pressure test the weld. Drain the pool water from the canister. Attach the vacuum system to the drain line, and operate the system to achieve a vacuum.
- Hold the vacuum and backfill with helium to 1 atmosphere. Restart the vacuum system and remove the helium. After achieving vacuum, backfill and pressurize the canister with helium.
- Helium leak check the shield lid weld. Vent the helium pressure to 1 atmosphere (absolute). Install the vent and drain port covers and weld them to the shield lid.
- Install the structural lid and weld it to the canister shell. Install the hoist rings, and attach the canister lifting sling. Install the adapter plate on the storage cask.
- Lift the transfer cask to the top of the storage cask and set it on the adapter plate, ensuring that the bottom door hydraulic actuators are engaged.
- Attach the canister lifting slings to the crane hook and lift the canister.
- Open the bottom doors of the transfer cask.
- Lower the canister into the storage cask. Detach the canister slings from the hook.
- Remove the transfer cask and adapter plate. Remove the canister lifting slings.
- Install the shield plug and lid on the concrete cask.
- Move the loaded storage cask to the storage pad.
- Using the air pad rig set and a towing vehicle, move the storage cask to its designated location on the storage pad.

The removal operations are essentially the reverse of these steps, except that weld removal and cool down of the contents are required.

The special equipment needed to operate the NAC-MPC system has been described in Section 1.2.1.4. Other items required are miscellaneous hardware, connection hose and fittings, and hand tools typically found at a reactor site.

1.2.3 Cask Contents

The NAC-MPC is designed to store up to 36 Yankee Class spent fuel assemblies. The Yankee Class fuel consists of fuel assemblies manufactured by Westinghouse, United Nuclear, Exxon, and Combustion Engineering. The assemblies vary in initial enrichment from 3.5 to 4.94 wt % ²³⁵U. Each manufacturer's types of assemblies include two configurations identified as Type A and Type B. The arrangement of fuel rods differ in each types to allow the fuel assembly to accept a segment of a control blade used for criticality control. The characteristics of the Yankee Class spent fuels are presented in Table 1.2-6. Unenriched fuel assemblies are not evaluated and are not included as a proposed contents.

A canister may contain one or more Reconfigured Fuel Assemblies (RFA) designed to confine Yankee Class intact or damaged spent fuel rods and fuel debris. The RFA is designed to confine its contents during all storage and transport conditions. The assembly can accept up to 64 full length spent fuel rods in an eight by eight array of tubes. A sketch of the assembly is provided in Figure 1.2-4.

The Reconfigured Fuel Assembly consists of a shell (square tube with end fittings), a basket assembly and 64 fuel tubes. The external dimensions of the shell are the same as those of a standard Yankee Class spent fuel assembly and all materials are stainless steel. It is designed such that it can be handled in the same manner as a standard Yankee Class spent fuel assembly. The spent fuel is confined in the fuel tubes. The tubes are supported by a basket assembly within the shell and have end plugs with drilled holes to permit draining drying and inerting with helium. The shell has holes in the top and bottom fittings to permit draining, drying and inerting of the assembly.

NAC-MPC FSAR	April 2000
Docket No. 72-1025	Revision 0

The total number of full length rods that can be placed in the Reconfigured Fuel Assembly is less than the number that are in the Yankee Class fuel assemblies (maximum of 64 versus 231 rods of the most reactive fuel). Consequently, the effects of a Reconfigured Fuel Assembly placed in a canister (e.g., criticality, thermal output, source term) are significantly less than the effects of a design basis Yankee Class spent fuel assembly. These effects are evaluated in the appropriate chapters that follow.





Figure 1.2-2 Transfer Cask



(NAC-MPC FSAR Docket No. 72-1025

April 2000 Revision 0

Figure 1.2-3 NAC-STC Transport Configuration



April 2000 Revision 0

Figure 1.2-4 Reconfigured Fuel Assembly



April 2000 Revision 0





Figure 1.2-6 Vertical Concrete Cask and Transfer Cask Arrangement







Table 1.2-1 Major Physical Design Parameters for the Transportable Storage Canister

Parameter	Design Value
Outside Diameter	70.64 in.
Length	122.5 in.
Capacity	36 Yankee Class spent fuel assemblies
XX7_1_1_4	54.700 H (
Weight	54,730 lbs. (nominal) w/ fuel
Marine hast land	
Maximum neat load	12.5 kW (tuel)
Maximum fuel clad temperature	Zircaloy or Stainless Clad
Normal conditions	340°C
Off-normal and accident	430°C
Internal atmosphere	Helium

 Table 1.2-2
 Transportable Storage Canister Fabrication Specification Summary

Materials

• All material shall be in accordance with the referenced drawings and meet the applicable ASME standard.

Welding

- All welds shall be in accordance with the referenced drawings.
- All filler metals shall be appropriate ASME material.
- All welders and welding operators shall be qualified in accordance with ASME Section IX.
- All welding procedures shall be written and qualified in accordance with ASME Section IX.
- All welds specified to be visually examined shall be examined as specified in ASME Section V, Article 9 with acceptance per ASME Section III, NB-4424 and NB-4427.
- All welds specified to be liquid penetrant examined shall be examined in accordance with the requirements of ASME Section V, Article 6, with acceptance in accordance with ASME Section III, NB-5350.
- All personnel performing examinations shall be qualified in accordance with the NAC International quality assurance program and SNT-TC-1A.
- All welds specified to be radiographed shall be examined in accordance with the requirements of ASME Section V, Article 2, with acceptance per ASME Section III, NB 5320.
- All welds specified to be ultrasonically examined shall be examined in accordance with ASME Section V, Article 5, with acceptance in accordance with ASME Section III, NB-5330.

Fabrication

- All cutting, welding, and forming shall be in accordance with ASME, Section III, NB-4000 unless otherwise specified. Code stamping is not required.
- All surfaces shall be cleaned to a surface cleanness classification C or better as defined in ANSI N45.2.1, Section 2.
- All fabrication tolerances shall meet the requirements of the referenced drawings after fabrication.

Packaging

• Packaging and shipping shall be in accordance with ANSI N45.2.2.

Quality Assurance

- The canister shall be fabricated under a quality assurance program that meets 10 CFR 72 Subpart G and 10 CFR 71 Subpart H.
- The supplier's quality assurance program must be accepted by the licensee prior to initiation of work.
- Hold points for inspection of a completed basket assembly are verification of the basket assembly diameter and length, insertion of a "dummy" fuel assembly into each fuel tube, and insertion of the basket into the canister shell.

A Certificate of Compliance shall be issued by the fabricator stating that the canister meets the specifications and drawings.

Parameter	Value					
Height	160 in.					
Outside diameter	128 in.					
Shielding (side wall)						
Concrete thickness	21 in.					
Steel thickness	3.50 in.					
Radiation dose rate (average):						
Side surface	\leq 50 mrem/hr					
Top surface	\leq 35 mrem/hr					
Air inlet/ outlet vents	≤ 100 mrem/hr					
Weight	155,000 lbs. (nominal)					
Air flow at design heat load	1 (lbsm)/sec					
Material of construction						
Concrete	Type II Portland Cement					
Reinforcing steel	A615 Grade 60					
Steel liner	A36 Carbon Steel					
Service life	50 years					
Maximum concrete temperatures for normal	150°F bulk					
operation	200°F local					

Table 1.2-3 Major Physical Design Parameters for the Vertical Concrete Cask

 Table 1.2-4
 Concrete Cask Fabrication Specification Summary

Materials

- Concrete mix shall be in accordance with the requirements of ACI 318 and ASTM C94.
- Type II Portland Cement, ASTM C150.
- Fine aggregate ASTM C33 and C637.
- Coarse aggregate ASTM C33 and C637.
- Admixtures

Water Reducing ASTM C494.

Pozzolanic Admixture ASTM C618.

- Compressive Strength 4000 psi at 28 days.
- Specified Air Entrainment 3% 6%.
- All steel components shall be of material as specified in the referenced drawings.

Welding

• Visual inspection of all welds shall be performed to the requirements of AWS D1.1, Section 8.15.

Construction

- Specimens shall be obtained or prepared for each batch or truck load of concrete per ASTM C172 and ASTM C192.
- Test specimens shall be tested in accordance with ASTM C39.
- Formwork shall be in accordance with ACI 318.
- All sidewall formwork and shoring shall remain in place for at least 24 hours.
- All bottom formwork and shoring shall remain in place for 14 days.
- Grade, type, and details of all reinforcing steel shall be in accordance with the referenced drawings.
- Embedded items shall conform to ACI 318 and the referenced drawings.
- The placement of concrete shall be in accordance with ACI 318.
- Surface finish shall be in accordance with ACI 318.

Quality Assurance

• The concrete cask shall be constructed under a quality assurance program that meets 10 CFR 72 Subpart G. The quality assurance program must be accepted by NAC International and the licensee prior to initiation of the work.

April 2000 Revision 0

Table 1.2-5 Major Physical Design Parameters for the Transfer Cask

Parameter	Value
Inside Diameter	71.5 in.
Outside Diameter	86.5 in.
Height	133.38 in.
Empty Weight (nominal)	80,800 lbs.
Side Wall Dose Rate (average)	≤ 200 mrem/hr

	Yankee	Class Spent Fuel ^{1, 2}	2, 3,4
	United Nuclear	Combustion	Westinghouse
Parameter	Туре А	Type A	Type B
Number of Assemblies per Canister ⁶	36	36	34
Assembly Weight, lbs.	850	850	900
Assembly Length, in.	111.25	111.79	111.25
Active Fuel Length, in.	91	91	92
Fuel Rod Cladding	Zircaloy	Zircaloy	Stainless Steel
Maximum Uranium, kgU	245.6	239.4	286.9
Maximum Initial ²³⁵ U, wt %	4.0	3.9	4.94
Minimum Initial ²³⁵ U, wt %	4.0	3.7 ⁵	4.94
Maximum Burnup, MWD/MTU	32,000	36,000 ⁵	32,000
Maximum Assembly Decay Heat, kW	0.347	0.347	0.347
Maximum Decay Heat, kW	9.3	12.5	9.0
Minimum Cool Time, yr	13.0	8.1 ⁵	21.0

Table 1.2-6NAC-MPC Design Basis Fuel Characteristics

- 1. The Yankee Class spent fuel includes United Nuclear Type A and Type B, Combustion Engineering Type A and Type B, Exxon-ANF Type A and Type B, Westinghouse Type A and Type B. The United Nuclear Type A is the most reactive assembly and is used as the design basis fuel for criticality analyses. The Combustion Type A is the design basis fuel for shielding and thermal evaluations. The Westinghouse Type B fuel is the heaviest assembly and is the design basis fuel for structural considerations.
- 2. The Exxon-ANF fuel was provided by Exxon after Exxon was acquired by Advanced Nuclear Fuel (ANF). Fuel provided by ANF was designated Exxon-ANF. This fuel is considered to be Exxon fuel throughout this report. Except for lower enrichment and fuel hardware, Exxon fuel (and therefore, Exxon-ANF fuel) has the same characteristics as Combustion Engineering fuel.
- 3. The NAC-MPC can accommodate one or more Reconfigured Fuel Assemblies containing up to 64 fuel rods or rod segments classified as failed.
- 4. Exxon fuel assemblies with 3.5 wt % ²³⁵U and burnups of 36,000 MWD/MTU require a minimum cool time of 9 or 16 years for Zircaloy or stainless steel fuel assemblies, respectively.
- 5. Combustion Engineering fuel assemblies with an initial minimum enrichment of 3.5 wt %, burnups up to 32,000 MWD/MTU require minimum cooling times of 8.0 years.
- 6. Up to 36 intact fuel assemblies of any type not exceeding 30,600 pounds total weight are authorized.

THIS PAGE INTENTIONALLY LEFT BLANK

1.3 Agents and Contractors

The prime contractor for the NAC-MPC design is NAC International Inc. (NAC). All design and specification activities are performed by NAC. Fabrication of the steel components will be by qualified vendors. A qualified concrete contractor will perform fabrication of the vertical concrete storage cask. All fabrication activities will be performed in accordance with quality assurance programs meeting the requirements of 10 CFR 71 and 10 CFR 72.

NAC is a private corporation founded in 1968, whose primary focus is the tracking, inspection, handling, storage, and transportation of spent nuclear fuel. NAC is recognized in the industry as expert in all aspects of the design, licensing, and operation of spent fuel handling, inspection, storage, and transport equipment, as well as in the management of spent fuel inventories.

NAC is the leading U.S. company in the transport and storage of spent nuclear fuel, owning and operating the largest fleet of commercial spent fuel transport casks in the United States. This fleet includes the following casks:

- 5 NLI-1/2 (truck) 1 PWR/2 BWR Approved for the transport of LWR and metallic fuel and for high level waste.
- 5 NAC LWT 1 PWR/2 BWR Approved for the transport of LWR fuel, metallic fuel, research reactor fuel and high level waste.
- 2 NLI-10/24 (Rail) 10 PWR/24 BWR Approved for the transport of LWR fuel.

These casks are approved by the U.S. NRC under 10 CFR 71 and have successfully and safely completed more than 1,000 shipments of spent fuel and high level waste for more than 40 nuclear facilities in the last 15 years. NAC has also designed and licensed the NAC-STC rail cask for the storage and transport of spent fuel. The NAC-MPC canister is designed to be transported in the NAC-STC.

NAC has designed, analyzed, and obtained NRC approval for the following dry storage casks:

- NAC-I26 S/T Certificate of Compliance approval for storage of 26 PWR fuel assemblies (Docket 72-1002)
- NAC-I28 S/T Site specific approval for the storage of 28 PWR fuel assemblies (Docket 72-1020)
- NAC-C28 S/T Certificate of Compliance approval for the storage of 28 consolidated PWR fuel canisters (56 assemblies) (Docket 72-1003)

- 14 Marc ------

Within the last 10 years, NAC contractors have completed the fabrication of two NAC-I28 S/T and one NAC-I26 S/T storage casks, and five NAC-LWT transport casks.

NAC has also designed the NAC-STC rail cask for the storage and transport of directly loaded spent fuel. The NAC-STC is approved for use in site specific applications (Docket 72-1013) for storage and for transport (Certificate of Compliance Number 71-9235). NAC has also prepared, and submitted, a Safety Analysis Report for the transport of canistered Yankee Class spent fuel (Docket 71-9235).

Safety Analysis Reports have also been prepared and submitted for the NAC Universal MPC System (NAC-UMSTM) for the transport (Docket 71-9270) and long-term storage of several classes of spent fuel (Docket 72-1015).

1.4 Generic Storage Cask Arrays

A typical ISFSI storage pad layout for 16 storage casks is provided in Figure 1.4-1. As shown in this figure, roads parallel the sides of the pad to facilitate transfer of the storage cask from the transporter to the designated storage position on the pad. Loaded storage casks are placed in the vertical position on the pad in a linear array. Array sizes could accommodate from 1 to more than 200 casks. Figure 1.4-1 shows typical spacing and representative site dimensions. However, these are dependent on the general site layout, access roads, site boundaries, and transfer equipment selection.

The reinforced concrete foundation is capable of sustaining the transient loads from the air pad and the general loads of the stored casks. If necessary, the pad can be constructed in phases to specifically meet utility-required expansions.

Figure 1.4-1 Typical ISFSI Storage Pad Layout



NAC-MPC FSAR Docket No. 72-1025

1.5 License Drawings

This section presents the License Drawings for the NAC-MPC System.

1.5.1 NAC-MPC License Drawings

Drawing		Revision	No. of
Number	Title	No.	Sheets
455-821	Adapter Ring, Transfer Adapter to NAC-STC MPC-Yankee	0	1
455-856	Name Plate - NAC-VCC Cask	0	1
455-859	Assembly, Transfer Adapter, MPC-Yankee	1	3
455-860	Assembly, Transfer Cask (TFR), MPC - Yankee	4	4
455-861	Weldment, Structure, Vertical Concrete Cask (VCC),	4	2
	MPC - Yankee		
455-862	Loaded Vertical Concrete Cask (VCC), MPC - Yankee	2	1
455-863	Lid, Vertical Concrete Cask (VCC), MPC- Yankee	2	1
455-864	Shield Plug, Vertical Concrete Cask (VCC), MPC- Yankee	1	1
455-866	Reinforcing Bar and Concrete Placement, Vertical Concrete	0	3
	Cask (VCC), MPC- Yankee		
455-870	Canister Shell, MPC- Yankee	3	1
455-871	Details, Canister, MPC- Yankee	4	2
455-872	Assembly, Transportable Storage Canister (TSC),	6	2
	MPC- Yankee		
455-873	Assembly, Drain Tube, Canister, MPC- Yankee	2	1
455-881	PWR Fuel Tube, MPC- Yankee	3	1
455-891	Bottom Weldment, Fuel Basket, MPC- Yankee	0	1
455-892	Top Weldment, Fuel Basket, MPC- Yankee	1	1
455-893	Support Disk and Misc. Basket Details, MPC- Yankee	3	1
455-894	Heat Transfer Disk, Fuel Basket, MPC- Yankee	1	1
455-895	Fuel Basket Assembly, MPC- Yankee	2	1

NAC-MPC FSAR Docket No. 72-1025 April 2000 Revision 0

Drawing		Revision	No. of
Number	Title	No.	Sheets
YR-00-060	Yankee-Class Reconfigured Fuel Assembly	1	1
YR-00-061	Yankee-Class Reconfigured Fuel Assembly,	1	1
	Shell Weldment		
YR-00-062	Yankee-Class Reconfigured Fuel Assembly,	1	1
	Top End Fitting Assembly		
YR-00-063	Yankee-Class Reconfigured Fuel Assembly,	1	1
	Bottom End Fitting Assembly		
YR-00-064	Yankee-Class Reconfigured Fuel Assembly,	1	1
	Nozzle Bolt and Alignment Pin		
YR-00-065	Yankee-Class Reconfigured Fuel Assembly,	1	1
	Fuel Basket Assembly		
YR-00-066	Yankee-Class Reconfigured Fuel Assembly,	1	1
	Fuel Tube Assembly		

1.5.2 <u>Yankee Class Reconfigured Fuel Assembly License Drawings</u>

ASS	ASSY ASSY						
Γ	QUANTITY						
570	GEOWETRY	UNLESS	TOLE	ANCES INSE SPICE	ø	CROUP	
P	FLATHESS	J PLACE DEC.	10L.	2 PLACE DE	c to.		1.14
	STRAIGH THESS	UNDER 3		UNDER 6	± 04	00000	
H		3-12		6-15	\$ 00	MOLEI	L.L
\leq	AHQULARITY	OVER 12		OVER 18	1 00	*/oet	57
	PERPENDICULARITY	1 PLACE DEC.	2.1	FUES	.03		- 14
7	PERPENDICULARIT	FM9H (+)-M-125	1000	ANGLES	±.5'	MORET	1.10
//	PARALLELISU	HEADS-UNFED	124-20	CORNERS	R.03		110-11
0	CONCENTRICITY	NEXT ASSEMBL	Y:			QUALITY .	6 f a
\$	TRUE POSITION	DRAWING TYPE	:	LICENS	E	191 V .	L'had
			_				





ASS	Y ASSY	ASSY					
Γ	QUANTIT	,					
579	0.04	EURY	UNLES	TOLE	ANCES	ED	ø
P	FLATHES	s	S PLACE DEC.	TOL.	2 PLACE DEC	ta.	
	STRAICH	THESS	UNDER 3		UNDER 6	t.94	~
		_	3-12		6-18	2.04	
2	ANGULA	HIY .	OVER 12		OVER 18	±.06	
Π	PERPENC	CULARITY	1 PLACE DEC.	±.1	FILLETS .	03	
F 77			M91 (1)-M-125		ANGLES I	1.5'	
//	PARALLE	USM	INFELOS-UNITED	0.2×-28	CORNERS	R.03	12
Õ	CONCENT	ROTY	HEAT ASSEMBL	.¥;	45536	6	•
¢	TRUE PO	IS THOM	ORAWHG TYPE	:	LICENSE		
			3				

			ſ	NAC INTERN/	ATIONAL							
	rimon and a		NAMEPLATE- NAC-VCC CASK									
	Selener I	124/97	455	DESIGN PACKAGE	ORAWAG 85	6	λų Ο					
· • •	Asterna 4	SCALE	FULL	EST.WT.	SH 1 0/1	11,38	AU 97					
	2				4							

ASST	ASSY	ASST Y									A	NAC INTERI	NATION	AL		
SYNI.	GEON	ETRY	UNLESS OTHERWISE SPECIFIED 3 PLACE DEC. TOL. 2 PLACE DEC TOL			GROUP	HAME	DATE	1		ACCEN					
	FLATNES	is					K. Yellen	1.4.97	ASSEMBLY,							
_	STRAICH	THESS	UNDER 3		UNDER 6	1.04	CHEDUR	lac	VKAR	TRANSFER ADAPTER						
_			3-12		6-10	4.06	MAD ECT	KLAST	Viles							
\leq	ANGULA	Refy	OVER 12		OVER 18	2.00	*/047	Thereas	1/1/04	1		мрс-тл	ANKEE			
1.	PERPEN	DICULARIT	I PLACE DEC.	‡. 1	FILLETS .	03			<u> </u>							
11	PARALL	1104	116H (J)-AA-125 1	800 8	ANGLES 1	.5'	UCDOW		17-10							
"			HEADS-DIGED C	124-28	CORNERS	R.03	anning a	Se Hillion	17/77	PROJECT A	55	DESIGN PACKAGE	DRAWIN	6 O	50	REV
Q	CONCEN	TRICITY	NEXT ASSEMBLY:		MUMU	REAL	17/99	4	25			0	29	1		
\$	TRUE P	OSITION	DRAWING TYPE:		LICENSE		390 149	NA		SCALE 1/10)	EST.WT. NOTE	D 9+1	of 3	9.1 12-1	644 10-10
			7					0					4			

Ĥ	NAC INTERNAT	IONAL				
ASSEMBLY, TRANSFER ADAPTER, MPC-YANKEE						
PROJECT 455	DESIGN PACKAGE		59 1			
SCALE 1/10	EST.WT. NOTED	s+2 or 3	9:17AN 12-30-90			
[-				



	A	NAC INTERNAT	IONAL	
	TRAI	ASSEMBL NSFER AD MPCYANI	.Y,)APTER, KEE	
PROJECT	455	DESIGN PACKAGE	DRAWING 85	59 1
SCALE 1	/5	EST.WT. NOTED	SH3 0∓3	9:17AH 12-30-90

A\$\$	Y ASSY ASSY A	SY ASSY					
	QUANTITY						
5	UNSPECIFIED OMEN	NERANCING SH ISIONS AND TO INCHES, FRAC	ALL BE ERANC	PER ANSE THE	1.5-82 LOW, ±1/8	GROUP	
STN.	CEONE 18Y	XXX,	10L	. #	TOL	MONEL	
		UNCER 3		UNDER &	8.04		_ (
$ \Delta $	223HFAJ7	3-12		6-10	8.04	~~~~	
_	STRAIGHTMESS	OVER 12		OVER 14	4.00		
-		. X.	£.I	MOLES I	.5	PROJECT	
~	ANOLLANITY	ALL UNSPEC		M			
L	PEPPENDICULARITY	BREAK ALL	9444	CORNERS .01	03		41
11	PARALITUS	ALL U	ALL UNSPECIFIED WACHINED SUNFACES SHALL BE VOR BETTER				
-							
O	CONCENTING TY	NEXT ASSEMB	MEXT ASSEMBLY:				77
ቅ	THUE POSITION	DRAWING TYPE	6	LICENSE			K
				ž			T





A NAC INTER	NATIONAL
ASSEI TRANSFER MPC-Y	MBLY, CASK (TFR) 'ANKEE
PROJECT 455	D*AMHG 860 4

	NAC INTERI	NATION	4 <i>L</i>	
TRA	ASSEN NSFER (MPC-Y	IBLY, CASK (ANKEE	TFR)
200 ECT 45	5	Date of the C	86	50 4
serve 1/10	ESF.WT,	913	or 4	2 39PU 11-34-99

A	NAC INTERNA	TIONAL	
TRAN	ASSEMBI SFER CAS MPC-YAN	LY, SK (TFR) IKEE	
^{PROJECT} 455	· · · · · · · · · · · · · · · · · · ·	DRAWING 860	
scale 1/8	E\$T.wt.	SH 4 07 4 11-23-	

A55	USI V ASST ASSY ASSY OUANIRTY								R NAC INTERNATIONAL				
	DADINGONANG ANG TOLEDANCING SULL BE FOR ANSI V14.5-82 UNIPECIFIC DANCINGANG AND TOLERANCES SHOWN BELOW. BADING NIS ANY IN NOVEL TRACTIONAL TOLERANCE: \$1/0					CROUP	NAME	DATE	WELDMENT, STRUCTURE,				
574	CEONE IRY	JUDI. E RECENU	IOL .	.201 8 A3dHu	10L.	NOW	R. mellen	4-23-11	VERTICAL CONCRETE CASK (VCC MPC-YANKEE				
É	TLAINESS STRAIGH INESS	3-12 0401 12		6-18 OVER 16	1.00 1.00	****	la	uktr					
Z	ANOULARITY	JI AL UNSTON	11 TOOL	RADA .01	03	SHE	It and	1/2 /m	-				
Ĺ	POPPHORULANIN	BARCAN ALL S	HARP CO	44(JRS .81	63	海豚	Theladan	"/22/40					
//	PARALLELISM	SURFACES SHALL BE TOR BETTER					10. Manual and	The					
0	CONCENTING TY	HEAT ASSOME	n 8:	55-86	6				455	861 4			
¢	MUE POSINON	DRANNE TYPE	L	ICENSE		SULUTI	WIDENITY	1/24/20	SCALE 1/16 EST #1	H 1 07 2 11-13-00			
ſ	NAC INTERNAT	TONAL											
--	-----------------	---------------	---------	--	--	--	--						
WELDMENT, STRUCTURE, VERTICAL CONCRETE CASK (VCC) MPC-YANKEE													
Prodet 455	<u>,</u>	10F + 40Hrt 8	51 7										
scale 1/16	CST WT.	≫2 ∞ 2	11 1340										

ASS	ASST	ASSY					
	QUANTI	Υ					
SYN.	CE ON	IE TRY	1040 5 45	TOLEA	ANCES		CRO
Ŋ	FLATHE	22	3 PLACE DEC.	TOL	2 PLACE DEC	TOL	-
	STRAICH	ITNESS	UNDER 3		UNDER 6	2.04	000
			3-12		6-18	±.06	1 200
\sim	ANCULA	ALL N	OVER 12		OVER 18	3.08	1 9/
	PERFE		ITY I PLACE DEC.	1.1	AULTS	1	
-,,			F1851 (/)-A4-125	HOR N	ANGLES :	PROJ	
//	PARALL	ELISM	HEADS-UNITED	0.24-28	CORNERS	R.03	1000
0	CONCEN	TRICITY	NEXT ASSEMBL	Yî.			-
¢	TRUE P	OSITION	DRAWING TYPE	:	LICENSE		- 10
			.3				



			A	NAC INTERNA	TIONA			[
	RAUSA ALANA June NA	DATE 1-6-99 1/6/99 1/6/99	LO CONC I	ADED VEF RETE CAS MPC-YAN	RTICA SK (N IKEE	L /CC)		4
а, N	Rupper	/ <i>7/1</i> 7 V 199	PROJECT 455	DESIGN PACKAGE	DRAWING	86	2	PEV 2	
*	NA		scale 1/8	EST.WT.	SH 1	or 1	9 ! -	2AU - 19	
	2				1				

22A	Y ASSY ASST						
	GUANTITY						
SYN.	GEOMETRY		TOLE	UNCES	•••••	CROUP	NA
	T ADVICE	UNILES	1 On U	KINSE SPECIAL	D	HEPMER	2.2
$\mathbf{\mathbf{E}}$	TUAINESS	3 PLACE DEC.	TOL	2 PLACE DEC	TÓL.		
	STRAIGHTNESS	UNDER 3		UNDER 6	±.04	ORORDA	LL.
		3-12	_	6-18	±.06	MOLECT	IL.
~	ANGULARITY	OVER 12		OVER 18	\$.08	W/080	10.0
1	PERPENDICULARITY	I PLACE DEC.	±.1	FILLETS .	03	Citing I	4
F;;		THEH (+)-N-125	NOIC IN	ANGLES ±	5'	PROECTS	
//	PARALLELISM	INCLOS-UNFED	0.2A-28	CORNERS	R.03		11 11
Ø	CONCENTRICITY	NEXT ASSEMBL	Y:	455-862	2		Rul
\$	TRUE POSITION	DRAWING TYPE	:	LICENSE		201 14	N



ASST	ASSY ASSY QUANTITY				L		I				AC ITERNA	TIONA	 L		
SYN,	GEOME TRY		TOLE LESS OTH	RANCES	9	CROUP	NAME	DATE		CU					1
Q	FLAINESS	3 PLACE E	EC. TOL	2 PLACE DE	TOL		R. W.LA	1491		28	IELU P	LUG,			
	STRAICHTNESS	UNDER 3		UNDER 6	±.04		1ac	16/11	VERTIC	CAL CO	NCRE T	E CAS	к (VCC)	
	ANCLE ADITY	3-12	_	6-18	*.04		VELAL	1/1/19			C VAN			,	
1		DWLR 12		OVER 18	11.00	0000	70.000	14/19		MP	C-TAN	INEE			
1	PLIPERDICITA		-125 1000 1	ANGLES	±.5'	PROJECTOR OF	NA								
//	PARALLEUSH	THE AUS-IN	TD 0.24-2	CORMERS	R.03	LICO-COM	Ha Vin Au	1/0/00							1
0	CONCENTRICITY	NEXT ASS	DMBLY:	455-86	2	QUAUTY	Rippy	1/7/92	PROJECT 4	55 °	GN PACKAGE	DRAWING	86	4 ^{rev}	
\$	TRUE POSITION	DRAWING 1	MPE:	LICENS	Ε	-44 URE	NA	12	scale 1/5	EST.	ит.	541 (× 1	1:3044	1
		_													

~~~	Y ASSY ASSY					
	QUANTITY					
SYN:	CEONE INV			ANCES		· n
$\square$	FLATHESS	3 PLACE DEC.	TOL.	PLACE	0.0	10
-	STRACHTNESS	UNDER 3		UNCER	6	±.04
Z	ANGULARITY	J-12		6-1	_	1.06
1	PERPENDICULARIT	I PLACE DEC.	±.1	(ALL	15	03
11	PARALLELISM	HISH (+)-AA-125	4000 B	ANG	(5 ±	.5" R.0.1
0	CONCENTRICITY	NEXT ASSEMBL	¥:	455	86:	2
₽	TRUE POSITION	ORAWING TYPE		LICEN	SE	
		7			П	











23	T ASST ASST						
	QUANTITY						
SYN.	GEOMETRY		TOLD	RANCES		CROUP	Ì.
J		UNILIS	5 01140	KANZE GHEONI	80	PERMIT	К
$\sim$	FLATNESS	3 PLACE DEC.	ŤDL.	2 PLACE DEC	TÓL.	1	Ψ
	STRAICHINESS	UNDER 3		UNDER 6	2.04	06040	k
		3-12		6-18	4.04	PROJECT	L
2	ANCULARITY	OVER 12	_	OVER 10	1.00	W/040	ť
Γ	PERSONAL ARCT	I PLACE DEC.	2.1	DUCTS	03	000	1
┶		THEN LA LAND	-	ANGLE		PROFESSION	U.
//	PARALLELISM	HALADS-UNFILD	114-20	CONVES	R.03	UCDER	Ŀ
0	CONCENTRICITY	NEXT ASSEMBL	Y:	455-87	2	GUALITY	f
$\overline{\Phi}$	TRUE POSITION	DRAWING TYPE		LICENSE		-	t
		3	-				









	HANDRAL	94C		DRAWING No.	0	CSCHP RON			ł	-
					AC TERNAT	TONA	L			
ROUP	NAME	DATE	AS	SEMBLY.	TRANS	SPOR	TA	BLE	:	
	R.grallin	1+23.95	S	TORAGE	CANIST	ER (	TS	C),	-	
1000	la	1/actor	MPC-YANKEE							A
	the t. al	4 ·/syke	ļ							-
	W. Lehol	10 1/24	9							
	+ Handson	U 1124/49	PROJECT	455		DR A WING	87	2	REV	
	Berend	a 1/24/9	scale 1	/8 [51.#	1.	<b>эн</b> 1 о	2	<u>د</u> ۱۱-1	0 74	
									_	

3







45	004011117	SY A35Y			A NAC INTERNATIONAL								
D	MACHEMINE AND FOLDIANCHIG SMILL BE FEL AND THAS-02 UNEFEITED DATE AND RELEASED SHOW BELOW SACTORIS ME IN NOMES TRACTIONAL TOLENCES ST/S					GROUP	NAME	DATE	PWR FUEL TUBE.				
\$7¥	CLOWE TRY	.DOI	TOL.	.0	10. 1.44		R millin	11-23-94	MPC-YA	NKFF			
	FLAMESS	3-13		4-14 0400 14	2.06	-	le	alesta					
Ζ	AHOLLARTY		1.1 100 10			25	KANK	Vista	1				
I	TUPUGGLANT			CONNERS .PI	63	125	And the Aland	1/40/101					
//	PARALITIN	ALL U SUFFACES	944L	K ∯or K			JA. Mariles	1/2 1/2					
0	CONCENTINGTY	Ext Associa	11.	455-89	5		permanenter	1	455	881 3			
•	THUE POSITION	DRANNIS THPE	1	LICENSE		\$	Blanto	141/1	scale 1/1 est.nt.	91 1 0 1 11 11 11 11 11 11 11 11 11 11 11			

A55	Y ASSY	ASSY					
	QUANTIT	,					
5YQ.	6100	E INY	UNITS	10.0		~	GROUP
$\square$	FLATHES	5	S PLACE DEC.	TOL.	2 PLACE DEC	TOL	-
	STRAICH	THESS	UNDER 3		UNOCH 6	±.04	D-EOK
			3-12		6-18	±.06	ANALYSS
$\leq$	AHOULA	H L Y	OVER 12		OVER, 18	\$.09	
Π	PERPEN	CULARITY	1 PLACE DEC.	2.1	FILLETS .	oj.	
÷;;			MSH (v)-M-125	-	ANGES 1	5	MOLETS
//	PARALLE	U\$¥	HEADS-UNFID	124-28			Dunuty
0	CONCENT	RIDTY	RAD	VS CO	RHERS		
ě	TRUE PO	ISI TICIN	TYPE OF .	l	ICENSE		NONET ASSESS
<b>1</b>							



A\$51	ASSY QUANTIT	ASSY Y									A	NAC INTERNA	TIONA	L		
5 m		CIRV.	UNLES		ANCES AMSE SECO	•	CROUP MEPAEA			2	T.					
$\leq$	STRACH	ne st	J PLACE DEC.	TOL.	2 PLACE DE	C TOL	04040	Me.		6	2					A
$\geq$	AHOLEAN	6FY	3-12		6-18	2.08	PROJECT I	to Alt	Alle le		۲ ۱	UEL DAS	INCT			
I	PERPEN	KULARI	TH PLACE DEC.	±.1	FELLTS	.03	Currant Ca	127	1/2.4	7	I		NKEE			
$\overline{\prime}$	PARALLE	11 <b>5</b> 4	HELOS-UPED	124-3	CORNERS	±.5' A.03	UCURNE A	SA TI MAL	11.20	2						
0	CONCEN	NO IV	HEAT ASSCHE	۴.	455-89	)5	QUALITY	A anti	They	MOJECT 2	455	DESCH PACKAGE	DRAWNC	892	) nev 1	
<del>\$</del>	TRUE PC	S TON	DRAMHG TYPE:		LICENS	E	W *	Sint -	24/1	scale 1/	<u>′5</u>	EST.wt,	9×1 0	1	317	

455	T ASST ASST			
L	QUAN BITY			
STV.	GEOMETRY	1001555.01	DRANCES	CHO
$\rho$	FLAINESS	S PLACE DEC TOL	& PLACE DECT TOL	1
	STRAIGHTMESS	UNDER 3	UNDER & ±.04	000
		J-12	6-18 \$.06	
$\leq$	ANCULARITY	OVER 12	OVER 18 1.00	*/0
	PERPENDICULARITY	1 PLACE DEC. 1.1	FILLETS .03	
11		THEN (/)-44-125 18040	# ANGLES ±.5'	PROF
//	PARALLEUSH	MELOS-UNITE Q2A-	CORNERS R.03	
Q	CONCENTRICITY	HEAT ASSEMBLY:	455-895	-
¢	TRUE POSITION	DRAWING TYPE:	LICENSE	=
		-		



	_					
45	ASST	ASST				
	QUANTT	۲				
2 MU.	CILCON	E INV	UNLESS	TOLE	ANCES RWSE SPECIFIC	D
$\square$	FLATHES	55	J PLACE DEC.	TOL.	2 PLACE DEC	TOL.
	STRAIGH	THESS	UNDER 3		UNDER &	2.04
-			3-12		4~18	2.04
~	ANGULA	<b>N</b> TY	OVER 12		OVER 18	1.00
	юнен	NOULAR	TY I PLACE DEC.	2.1	FILLETS .	03
77		_	FILEN (v)-M-125		ANGLES ±	5
//	PARALLE	1134	MEADS-WIED C	124-20		
Öl	CONCEN	TRICITY	RAD	us co	MERS .03	
¢	TRUE PO	SITION	TYPE OF .	L	ICENSE	
					_	



ASSY	ASSY					
QUANTITY						
GEOWE	îRY.	UNILES	TOLES S OTHES	ANCES	60	8
FLA THE SS	i	3 PLACE DEC.	TOL.	2 PLACE DEC	TOL	
STRAICHT	WESS	UNDER 3		UNDER 6	2 04	04
	•••••	3-12		6-18	1.06	
ANDAAN	TY .	OVER 12		OVER 18	1.09	-
PERPENDI		I PLACE DEC.	2.1	FILLETS .	03	1 <u></u>
		FNI91 (v )-AA-125	-	ANGLES 1	.5	
PARALLEL		HAEAOS-UNITED	Q.24-29	CORNERS	R.03	UQ1
CONCENTR	HC: TY	NEXT ASSEMBL	.¥:	455-87	2	-
TRUE POS	TICH	DRAWING TYPE	:	LICENSE		-
	Y ASSY QUANTITY GEOWE FLATHESS STRAIGHT ANGULAR PERPENDI PARALLEL CONCENTR	Y         ASSY         ASSY           QUANTITY         CEONETRY           CEONETRY         FLATHESS           STRAIGHTNESS         ANQULARITY           PERPENDICULARITY         PARALLEUSM           CONCENTRICTY         TIME POSITION	V         ASSV         ASSV           QUANTITY         UNLES         PLACE DEL           CEOWETRY         UNLES         SPLACE DEL           STRAIGHTHESS         SPLACE DEL         3-12           ANGOLARITY         OVER 13         SPLACE DEL           PARALELISN         MENGS-UNITO         DERECO-UNITO           PARALELISN         MEXT ASSEMP         TIME POSITION	Y         ASSY           QUANTITY         UNLESS OTHER           CEONCTRY         UNLESS OTHER           FLATHESS         3 PLACE DEC.           3TRAICHTNESS         UNDER 3           ANGULANITY         OVER 12           ANGULANITY         OVER 12           PERFENDICULARITY         PLACE DEC.           PARALLEUSN         MEDIG-UNED DIA-15 MODIN           PARALLEUSN         MEDIG-UNED DIA-15 MODIN           CONCENTRICTY         NEXT ASSEMBLY:           TIME POSITION         DRAWING TYPE:	V         ASSV         ASSV           QUANTITY         UMLESS         ODERMISE         SECON           CEONETRY         UMLESS         ODERMISE         SECON           TLATHESS         3 PLACE DEC.         TOL.         FLACE DEC.           STRAGENESS         UMDER 3         UMDER 3         UMDER 5           ANGLANTY         OKER 12         DVEP 10           PERFENDICULARITY         IMACE DEC.         2.1         FULTS           PARALELISM         MEXAS: UMED QA-28         CORCERS         CONCENTRO           CONCENTRICTY         NEXAS: SEMBLY:         4.5587         TICE NSEE	V         ASSV         ASSV           QUANTITY         UNLESS         TOLEPRINCES           CEOWETRY         UNLESS         OTERMISE           TLATMESS         3 PLACE DEC.         TOL           STRACHTHESS         3 PLACE DEC.         TOL           STRACHTHESS         3-12         6-10           ANDOLARITY         OVER 13         OVER 18         600           PERFENDICILARITY         TOLER DEC.         2.1         FULTS 0.3           PARALLELISM         MOLISS UNICD D.14-18         CONCRES R.03         CONCRES R.03           CONCENTRICTY         WEXT ASSEMELT:         4.55872         TTME POSITION



ASST A	197 ASS7			
	TIT			
\$116.	COLE TRT		official	
	PLATHESS	I MAR MC.	18.	1 1.10
	STRAIGHTHESE	UNCER 3	5-963	UNCLA
7		111	1.005	8-18
		ENGT 12	1.018	-
	PENPERDICALARITY	I PUR ME.	\$.1	FILLETS
7	Parat In 194	FBC58-7-44-120	<b>U CR</b> 14	NIL
4		ROLAS-COT D	<b>111-00</b>	
0	CONCENTRICITY	RADINS CORNER	.018-	,034
•	TRUE PESETION	SCALE	AS NO	TED
	1			



			1 1 1 1 1					
ASSY	ABST .	A <b>78</b> 7	$\bigcap$					
			-					
STN.		<b>di</b> ta	IC TRY	_			1 / St	TOLETA
	n	ATHE	11		57	m	BET.	10.
	67	-	-			100	3	1.000
1	-			_		3-12		L.008
			1177				1	2.010
	PE	<b>MPE I</b>	ICHLAR	TT	1.	, ICE	HEC.	2.1
Π					7 mils	1-1/1	4125	III CH9 14
-4-			L ( 34		net/	il - 94	17 MP (	1.24-90
Q	C0	HCEN1	HICITY		RAD		me.	5 .010
•	TR	A N	61710H		S	AL		45 N
							-	



ASST A	ATT ATT				
	1117				
51%L	GEOMETRY	Γ	-		
	PLADES	1	LACE MEL.	10.	I PLACE
	STRAIGHTHESE		1 (014	L 145	INCER
4	MOUNTTY	H	3-12 MER 12	1.005	6-18 OVER 1
	PENPENBICALARITY	1	LAS MC.	4.1	FILLET
7	PARALLELISE		SH-7-14-125	1000 in	ANDLES
Ö	CONCERTRICITY		JUE CORNEL	1.01	.030
•	THE POSITION	5	CALE:		





	102 1 4417 AU	01 1 00 1 mm			
	\$11J.	BENETRY			υ
		PLATHESS	I AUE Mt.	14.	1 1.4
1		STRAIGHTNESS	unnel 3	1.945	3
			7-12	1.005	•
	4	AND ANITY	<b>BVCI 12</b>	1.010	ÔW
	1	PERPENDICULARITY	I PLACE MEL.	1.1	FILL
		PARAL BUTT	FIM \$1-4-12		200
	-4		100.45 -011 50	0.11.49	
	Ų	CONCENTRICITY	RADIES CORES	S .015-	,630
	<b></b>	THUE POSITION	SCALEI		
		3			





ASST	ASST	ASET			
(as	ANTET				
511		-	7	mitt	
	14	LATHESS		S PLACE SEC.	10.
		TRAI ON THE	H	#8C1 3	1.00
-	-			3-12	6.00
	^	HER MITT		MER 12	-
<u> </u>			71184	I PLACE MEL.	1,1
77	-			FIRSH-1-14-12	
1				heuds anf ith	11. ÷
0	0	CHICEN TR ICI	TT	MATTINE CORNER	1 . Ô1
•	1	-		SCALE	





102	01 1 00	1124	TWE		
ASST	ISST ASST		1		
QUA	ATIT				
STIL		NE TRY	-	7	
	PLATHE	33	I RACE MC.	RL.	1 24
-	-	HTHE SE	INDER 3	1.005	UHC
	1		3-12	1.005	
4	ANDRA	ALTY	ENER 12	LMI	01
	PEAPER	DIGULARITY	I PLACE OUC.	2,1	FILL
77	+		FIR. 300-71-16-125	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	a dec
14	PROLL			6.H.#	
0	CONCEN	TRICITY	INGINE COME	.015	. #36
•	-	011110N	SCALE		
	_				



A9417 A1	HIT AFET		
	TITY		-
STH.	BEAME TRY	-	
$\square$	PLAINESS	J PLACE MEC.	m.
	STRAIGHTHESS	10001 3	1.90
		3-12	6.90
~	AMOULARITY	PHOR 12	-
L	PERPENDICULARITY	I ALACE MEC.	8+1
77		11K91-1-14-12	
-4	PAGELELIS	THE WE WIT TO	<b>1</b>
0	CONCENTRICITY	INFINE COME	.01
•	TRUE POSITION	SCALEI	
	۲		





#### **Table Of Contents**

2.0	PRIN	CIPAL DESIGN CRITERIA
2.1	Spent	Fuel To Be Stored 21-1
2.1	3 pem	Dounding Fuel Evaluation 21-1
	2.1.1	Bounding Fuel Evaluation
	2.1.2	Reconfigured Fuel Assembly
	2.1.3	Stainless Steel-Clad Fuel 2.1-2
2.2	Desig	n Criteria for Environmental Conditions and Natural Phenomena
	2.2.1	Tornado and Wind Loadings
		2.2.1.1 Applicable Design Parameters
		2.2.1.2 Determination of Forces on Structures
		2.2.1.3 Tornado Missiles
	2.2.2	Water Level (Flood) Design
		2.2.2.1 Flood Elevations
		2.2.2.2 Phenomena Considered in Design Load Calculations 2.2-3
		2.2.2.3 Flood Force Application
		2.2.2.4 Flood Protection
	2.2.3	Seismic Design
		2.2.3.1 Input Criteria
		2.2.3.2 Seismic - System Analyses
	2.2.4	Snow and Ice Loadings
	2.2.5	Combined Load Criteria
		2.2.5.1 Load Combinations and Design Strength - Concrete Cask 2.2-6
		2.2.5.2 Design Strength Reduction Factors - Concrete
		2.2.5.3 Load Combinations and Design Strength - Canister
		and Basket
		2.2.5.4 Design Strength - Transfer Cask
	2.2.6	Environmental Temperatures

#### NAC-MPC FSAR Docket No. 72-1025

2.3	Safet	y Protection Systems	2.3-1
	2.3.1	General	2.3-1
	2.3.2	Protection by Multiple Confinement Barriers and Systems	2.3-2
		2.3.2.1 Confinement Barriers and Systems	2.3-2
	2.3.3	Protection by Equipment and Instrumentation Selection	2.3-3
		2.3.3.1 Equipment	2.3-3
		2.3.3.2 Instrumentation	2.3-4
	2.3.4	Nuclear Criticality Safety	2.3-4
		2.3.4.1 Error Contingency Criterion	2.3-5
	2.3.5	Radiological Protection	2.3-5
		2.3.5.1 Access Control	2.3-5
		2.3.5.2 Shielding	2.3-6
		2.3.5.3 Ventilation Off-Gas	2.3-6
		2.3.5.4 Radiological Alarm Systems	2.3-7
	2.3.6	Fire and Explosion Protection	2.3-7
		2.3.6.1 Fire Protection	2.3-7
		2.3.6.2 Explosion Protection	2.3-8
2.4	Decor	mmissioning Considerations	2.4-1

#### List Of Tables

Table 2-1	Summary of the NAC-MPC Design Criteria	2-2
Table 2.1-1	Yankee Class Fuel Parameters	2.1-4
Table 2.1-2	Yankee Class Reconfigured Fuel Assembly Parameters	2.1-6
Table 2.2-1	Tornado and Wind Loading Criteria	2.2-9
Table 2.2-2	Load Combinations for the NAC-MPC Vertical Concrete Cask	2.2-10
Table 2.2-3	Load Combinations for the Transportable Storage Canister	2.2-11
Table 2.2-4	Structural Design Criteria for Components Used in the Transportable	
	Storage Canister	2.2-12
Table 2.3-1	Safety Classification of NAC-MPC Components	2.3-9
Table 2.4-1	Activity Concentration Summary for the NAC-MPC Vertical	
	Concrete Cask and Transportable Storage Canister	2.4-4

THIS PAGE INTENTIONALLY LEFT BLANK

____

#### 2.0 PRINCIPAL DESIGN CRITERIA

The NAC-MPC is a canister-based dry storage cask system that is designed to be transported in the NAC-STC licensed transport cask. It is designed to store Yankee Class spent fuel.

This chapter presents the design basis, including the principal design criteria, limiting load conditions, and operational parameters of the NAC-MPC dry storage system. The principal design criteria are summarized in Table 2-1.

#### Table 2-1Summary of the NAC-MPC Design Criteria

Design Criteria	
Design Life	50 years
Design Code - Confinement	ASME Code, Section III, Subsection NB for
	confinement boundary
Design Code - Nonconfinement	
Basket	ASME Code, Section III, Subsection NG and
	NUREG/CR-6322
Vertical Concrete Cask	ACI-349, ACI-318, ANSI/ANS 57.9
Transfer Cask	ANSI N14.6 and NUREG-0612
Design Weight:	
Canister Assembly w/fuel	54,730 lbs.
Transfer Cask	80,743 lbs.
Vertical Concrete Cask	151,364 lbs.
Thermal:	
Maximum Temperature,	340°C for 10-yr. Cooled
Zircaloy Cladding	380°C for 5-yr. Cooled
	430°C Off-Normal/Accident/Transfer
Maximum Temperature,	340°C for 10-yr. Cooled
Stainless Steel Cladding	430°C Off-Normal/Accident/Transfer
Ambient Temperature Range	-40° to 125°F
Average Annual Ambient Temperature	75°F
Concrete Temperature:	
Normal Conditions	$\leq 150^{\circ}$ F; $\leq 200^{\circ}$ F local
Off-Normal/Accident Conditions	$\leq$ 350°F local/ surface
Canister Cavity Atmosphere	Helium

#### Table 2-1 Summary of the NAC-MPC Design Criteria (Continued)

Design Criteria	
<b>RADIATION PROTECTION/SHIELDING</b>	
Concrete Cask Side Wall Contact Dose Rate	< 50 mrem/hr.
Concrete Cask Top Lid Contact Dose Rate	< 35 mrem/hr.
Concrete Cask Air Inlet/Outlet	< 100 mrem/hr.
Owner Controlled Area Boundary	
Normal/Off-Normal	
Annual Whole Body Dose	25 mrem/yr.
Accident Whole Body Dose	5 rem
SPENT FUEL SPECIFICATIONS	
Spent Fuel Type	Yankee Class
Fuel Configuration/Vendor	Westinghouse 18 x 18, 4.94 wt $\%$ ²³⁵ U
	United Nuclear 16 x 16, 4.0 wt % ²³⁵ U
	Combustion Engineering 16 x 16,
	3.5 to 3.9 wt $\%^{235}$ U
	Exxon 16 x 16, 3.5 to 4.0 wt % ²³⁵ U
Fuel Cladding	Stainless Steel - Westinghouse
	Zircaloy - All others
Spent Fuel Capacity – Intact Fuel	36 United Nuclear Assemblies
Assemblies	36 Combustion Engineering Assemblies
(May include one or more Reconfigured	36 Exxon Assemblies, or
Fuel Assemblies)	34 Westinghouse Assemblies
	Up to 36 Fuel Assemblies of any Type Not
	Exceeding 30,600 pounds Total Weight
Spent Fuel Assembly Burnup (max)	36,000 MWD/MTU [*]
Decay Heat/Fuel Assembly or Reconfigured Fuel Assembly	
Zircalov Clad Fuel	0.247 1/30
Stainless Steel Clad Fuel	0.347 KW
Reconfigured Fuel Assembly	0.102 kW

1. Based on the design basis, Combustion Engineering fuel at 36,000 MWD/MTU cooled 8.1 years. Exxon fuel is limited to 34,000 MWD/MTU and 10 years minimum cool time. The maximum burnup of all other fuel types is 32,000 MWD/MTU.

THIS PAGE INTENTIONALLY LEFT BLANK

#### 2.1 Spent Fuel To Be Stored

The NAC-MPC has been designed to safely store up to 36 Yankee Class spent fuel assemblies. The spent fuel designs are delineated by various factors including manufacturer, type, enrichment, burnup, cool time, and cladding material. The design basis consists of two types, designated A and B. The Type A assembly incorporates a protruding corner of fuel pins while the Type B assembly omits one corner of the fuel rods. These fuel types, as well as minor differences among manufacturers, are illustrated in Figures 6.2-1 through 6.2-3. During reactor operations, the symmetric stacking of the alternating assemblies permitted the insertion of cruciform control blades between the assemblies. Table 2.1-1 lists the nominal design parameters of each fuel design type. These parameters, including the minimum enrichment limit, exclude the loading of unenriched fuel assemblies in the transportable storage canister.

#### 2.1.1 Bounding Fuel Evaluation

The criticality evaluations show that the United Nuclear Type A 16 x 16 fuel assembly is the most reactive fuel type, even though the stainless steel clad Westinghouse fuel has a higher enrichment (4.94 wt % ²³⁵U). The criticality evaluation considers a complete assembly fuel rod matrix. Consequently, solid rods fabricated from Zircaloy or Type 304 stainless steel must replace any fuel rod that is removed.

The shielding evaluations show that the Combustion Engineering Type A has the largest dose rates. The United Nuclear assemblies are evaluated for a source term based on an initial enrichment of 4.0 weight percent, a maximum burnup of 32,000 megawatt days per metric ton of uranium, and a minimum cool time of 13 years after reactor discharge. Exxon fuel with Zircaloy or stainless steel hardware is evaluated at 36,000 MWD/MTU and 9 or 16-years cool time, respectively, with an initial enrichment of 3.5 wt %. Westinghouse fuel is evaluated at 32,000 MWD/MTU and 21-years cool time with an initial minimum enrichment of 4.94 wt %. Combustion Engineering fuel is evaluated for a source term based on an initial minimum enrichment of 3.7 wt %, a maximum burnup of 36,000 megawatt days per metric ton of uranium, and a minimum cool time of 8.1 years after reactor discharge. For Combustion Engineering assemblies at a maximum burnup of 32,000 MWD/MTU and initial minimum enrichment of 3.5 wt %, a minimum cool time of 8.1 years after reactor discharge.

The NAC-MPC maximum decay heat load is 12.5 kilowatts. This results in a maximum decay heat load for the design basis fuel assemblies of 0.347 kilowatt per assembly, based on 36 fuel assemblies per canister.

#### 2.1.2 <u>Reconfigured Fuel Assembly</u>

One or more transportable storage canisters may hold Reconfigured Fuel Assemblies containing intact or damaged spent fuel rods and fuel debris. The Reconfigured Fuel Assembly may consist of up to 64 rod segments or whole rods. The rods, or rod segments, are held in individual tubes in an 8 by 8 array. The array of tubes is positioned in a stainless steel container having the same external dimensions as a standard fuel assembly. It has a top end fitting that has the same configuration as a standard fuel assembly. The container is closed on the top and bottom ends by perforated plates, which act as a barrier to the release of gross particles to the canister, but allow the draining and drying of the container. The tubes are stainless steel and are closed on each end by a plug. Each plug has a small hole drilled through it. The perforated plate screens the drilled hole. The hole allows the draining and drying of the release of gross particles to the canister. The effects of the RFA container are evaluated in the appropriate sections. The structural, thermal, shielding, confinement, and criticality effects of the Reconfigured Fuel Assembly are bounded by those of an intact fuel assembly.

The physical parameters of the Reconfigured Fuel Assembly are provided in Table 2.1-2.

#### 2.1.3 <u>Stainless Steel-Clad Fuel</u>

The short-term and long-term temperature limits for stainless steel-clad fuel are derived based on the limits presented in EPRI report TR-106440, "Evaluation of Expected Behavior of LWR Stainless Steel-Clad Fuel in Long-Term Dry Storage," April 1996. In this report, the potential failure modes in both wet and dry storage environments were assessed to develop the bounding conditions for the prevention of any potential cladding degradation phenomena and cladding failure modes of stainless steel-clad fuel. The potential cladding degradation mechanisms
NAC-MPC FSAR	April 2000
Docket No. 72-1025	Revision 0

evaluated include: general corrosion, stress corrosion cracking, localized corrosion, mechanical failures and chemical/metallurgical-based failure mechanisms. The EPRI report is based on several types of stainless steel cladding, including 304, modified 348, 348 and 348H, which includes Yankee Class fuel.

The long-term temperature limit is conservatively established as 340°C for the NAC-MPC System for both stainless steel and Zircaloy-clad fuel. This value is significantly lower than the 430°C temperature limit defined by the EPRI report for stainless steel cladding. The short-term temperature limit for stainless steel-clad fuel is 430°C for the NAC-MPC System. This short-term limit is also conservatively used for Zircaloy-clad fuel.

### Table 2.1-1 Yankee Class Fuel Parameters

	Combustion	Combustion							United	United	
	Engineering	Engineering	Exxon	Exxon	Exxon	Exxon	Westinghouse	Westinghouse	Nuclear	Nuclear	
	Туре А	Туре В	Туре А	Туре В	Type A	Туре В	Туре А	Туре В	Туре А	Туре В	
			ASSI	EMBLY CO	NFIGURA	TION					
Assembly Length (cm)	283.9	283.9	283.3	283.3	283.9	283.9	282.6	282.6	282.4	282.4	
Assembly Width (cm)	19.2	19.2	19.3	19.3	19.3	19.3	19.3	19.3	19.4	19.4	
Assembly Cross Section									1		
(cm)	18.1	18.1	18.2	18.2	18.2	18.2	18.2	18.2	18.2	18.2	
Assembly Array	16x16	16x16	16x16	16x16	16x16	16x16	18x18	18x18	16x16	16x16	
Assembly Weight (kg)	352	350.6	372	372	372	372	408.2	408.2	385.5	385.5	
Enrichment-wt. % ²³⁵ U											
Maximum	3.90	3.90	4.00	4.00	4.00	4.00	4.94	4.94	4.00	4.00	
Minimum	3.70	3.70	3.50	3.50	3.50	3.50	4.94	4.94	4.00	4.00	
Initial Fuel Weight											
(KgUO ₂ /Assembly)	264.8	264.1	268.3	266.6	266.2	265.0	311	310	273.8	272.6	
Initial Heavy Metal											
(KgU/Assembly)	233.4	232.8	236.5	235	234.5	233.6	274.1	273.2	241.3	240.3	
Max. Burnup		1									
(MWD/MTU)	36,000'	36,000'	36,000	36,000	36,000	36,000	32,000	32,000	32,000 32,000		
Min. Cool Time (yr)	8.1	8.1'	16.0	16.0	9.0	9.0	21.0	21.0	13.0	13.0	
Max. Decay Heat (kW)	0.3471	0.347 ¹	0.269	0.269	0.331	0.331	0.264	0.264	0.257	0.257	
		_	FUEI	ROD CO	FIGURAT	ION					
Fuel Rod Pitch (cm)	1.2	1.2	1.2	1.2	1.2	1.2	1.1	1.1	1.2	1.2	
Rod Length (cm)	242.3	242.3	242.2	242.2	242.2	242.2	237.7	237.7	242.1	242.1	
Active Fuel Length (cm)	231.1	231.1	231.1	231.1	231.1	231.1	234.0	234.0	231.1	231.1	
Rod OD (cm)	0.9	0.9	0.9	0.9	0.9	0.9	0.86	0.86	0.9	0.9	
Clad ID (cm)	0.8	0.8	0.8	0.8	0.8	0.8	0.76	0.76	0.8	0.8	
Pellet OD (cm)	0.79	0.79	0.79	0.79	0.79	0.79	0.75	0.75	0.79	0.79	
Rods per Assembly	231	230	231	230	231	230	305	304	237	236	
Fuel Material	UO2	UO2	UO2	UO2	UO2	UO2	UO2	UO2	UO2	UO2	
Clad Material	Zircaloy	Zircaloy	Zircaloy	Zircaloy	Zircaloy	Zircaloy	SS 348	SS 348	Zircaloy	Zircaloy	
Fill Gas	Helium	Helium	Helium	Helium	Helium	Helium	Air	Air	Helium	Helium	
Fill Gas Pressure (psi)	315	315	125	125	125	125	0.0	0.0	140	140	
		]	DISPLACE	MENT ROI	CONFIGU	JRATION					
Displacement Rod	N/A	N/A	N/A	N/A	N/A	N/A	N/A	N/A	Zircalov -	Zircalov -	
Material								A 17 8 A	4	4	
Displacement Rod	N/A	N/A	N/A	N/A	N/A	N/A	N/A	N/A	0.9	0.9	
Diameter (cm)										0.12	
Displacement Rod Length	N/A	N/A	N/A	N/A	N/A	N/A	N/A	N/A	242.1	242.1	
(cm)											
Number Per Assembly	N/A	N/A	N/A	N/A	N/A	N/A	N/A	N/A	2	2	

1. Combustion Engineering fuel may be loaded at a cool time of 8.0 years with a maximum burnup of 32,000

MWD/MTU and a minimum enrichment of 3.5 wt%. The maximum decay heat for this assembly is 0.304 kW.

#### NAC-MPC FSAR Docket No. 72-1025

Table 2.1-1	Yankee Class	Fuel Parameters	(Continued)
-------------	--------------	-----------------	-------------

	Combustion	Combustion	·						United	United
	Engineering	Engineering	Exxon	Exxon	Exxon	Exxon	Westinghouse	Westinghouse	Nuclear	Nuclear
	Type A	Туре В	Туре А	Туре В	Туре А	Туре В	Туре А	Туре В	Туре А	Туре В
		···	GUID	E BAR CO	ONFIGURA	TION				L
Guide Bar Material	Zircaloy - 4	Zircaloy - 4	SS 304L	SS 304L	Zircaloy	Zircaloy			N/A ²	N/A ²
Guide Bar Width (cm)	1.1	1.1	1.1	1.1	1.1	1.1			N/A ²	N/A ²
Guide Bar Length (cm)	245.2	245.2	244.6	244.6	244.6	244.6			N/A ²	N/A ²
Assembly										
Configuration	Туре А	Туре В	Type A	Туре В	Type A	Туре В	Туре А	Туре В	Туре А	Type B
Top Nozzle Material	SS 304	SS 304	SS 304	SS 304	SS 304	SS 304	SS 304	SS 304	SS 304	SS 304
Bottom Nozzle Material	SS 304	SS 304	SS 304	SS 304	SS 304	SS 304	SS 304	SS 304	SS 304	SS 304
Upper Plenum Spring		·····	Inconel	Inconel	Inconel	Inconel			Inconel	Inconel
Material	SS 302	SS 302	X 750	X 750	X 750	X 750	N/A	N/A	X 750	X 750
Lower Plenum Spacer										
Material	N/A	N/A	SS 304	SS 304	SS 304	SS 304	N/A	N/A	SS 304	SS 304
Shroud Material									SS 304	SS 304
Top Nozzle Length										
(cm)	20.0	20.0	19.8	19.8	20.4	20.4	22.2	22.2	18.9	18.9
Bottom Nozzle Length										
(cm)	18.3	18.3	18.9	18.9	18.9	18.9	22.2	22.2	18.9	18.9
Upper Plenum Length										
(cm)	4.9	4.9	4.9	4.9	4.9	4.9	4.6	4.6	4.8	4.8
Lower Plenum Length										
(cm)	N/A	N/A	3.2	3.2	3.2	3.2	N/A	N/A	3.1	3.1
Shroud Length (cm)	N/A	N/A	N/A	N/A	N/A	N/A	N/A	N/A	246.9	246.9
Shroud Thickness (cm)	N/A	N/A	N/A	N/A	N/A	N/A	N/A	N/A	0.09	0.09
Top Nozzle Weight (kg)	5.50	5.50	6.70	6.70	6.70	6.70	6.70	6.70	17.024	17.02 ⁴
Bottom Nozzle Weight			5.18							
- (kg)	9.10	9.10		5.18	5.18	5.18	5.20	5.20	13.21 ⁴	13.21 ⁴
Upper Plenum Spring										
Weight (g)	3.3	3.3	3.3	3.3	3.3	3.3	N/A	N/A	3	3
Lower Plenum Spring										
Weight (g)	N/A	N/A	4.5	4.5	4.5	4.5	N/A	N/A	7.5	7.5
Grid Spacer Material	Zirc-	Zirc-								
	4/Inconel	4/Inconel			Zircaloy	Zircaloy			Inconel	Inconel
	625 ³	625 ³	SS 304L	SS 304L	-4	-4	N/A	N/A	718	718
Grid Spacer Weight (g)	590/960	590/960	622.3	622.3	634.3	648.4	N/A	N/A	0	0
Number of Grid Spacers	6	6	6	6	6	6	N/A	N/A	6	6

2. United Nuclear assemblies fabricated from Zircaloy incorporate displacement rods 95.32-inches long and 0.365-inch in outside diameter.

3. Five grid spacers are Zircaloy 4. The bottom spacer is Inconel 625.

4. Estimated weight.

	T T		1		United
	CE	Exxon	Exxon	West.	Nuclear
Parameter	Type A/B				
	ASSEMBLY	CONFIGURATION	J		
Assembly Array	8x8	8x8	8x8	8x8	8x8
Max. Enrichment (wt % ²³⁵ U)	3.90	4.00	3.70	4.94	4.00
Max. kgU*	66.33	66.33	66.33	60.21	66.33
FUEL ROD CONFIGUE	RATION (EACH RO	DD PLACED WITT	HIN ENCAPS	ULATING RO	D)
Rod Pitch (cm)	1.905	1.905	1.905	1.905	1.905
Active Fuel Length (cm)	231.1400	231.1400	231.1400	233.9975	231.1400
Rod OD (cm)	0.9271	0.9271	0.9271	0.8636	0.9271
Clad ID (cm)	0.8052	0.8052	0.8052	0.7569	0.8052
Pellet OD (cm)	0.7887	0.7887	0.7887	0.7468	0.7887
Diametrical Gap (cm)	0.0165	0.0165	0.0165	0.0102	0.0165
Max Rods per Assembly	64	64	64	64	64
Fuel Material	UO ₂				
Clad Material	Zircaloy	Zircaloy	Zircaloy	SS 348	Zircaloy
	Encapsu	ILATING ROD			
Rod OD (cm)	1.27	1.27	1.27	1.27	1.27
Rod ID (cm)	1.1278	1.1278	1.1278	1.1278	1.1278
Rod Material	SS-304	SS-304	SS-304	SS-304	SS-304

## Table 2.1-2Yankee Class Reconfigured Fuel Assembly Parameters

* Maximum kgU based on 95% of UO₂ theoretical density for the fuel pellet stack density.

Note: Intact or broken fuel rods may be contained in the reconfigured fuel assembly.

#### 2.2 Design Criteria for Environmental Conditions and Natural Phenomena

The design criteria defined in this section identifies the site environmental conditions and natural phenomena to which the storage system could reasonably be exposed during the period of storage. Analyses to demonstrate that the NAC-MPC design meets the design criteria defined in this chapter are presented in later chapters of this report.

#### 2.2.1 Tornado and Wind Loadings

The NAC-MPC may be stored on an unsheltered reinforced concrete storage pad at an ISFSI site. This storage configuration exposes the NAC-MPC to tornado and wind loading.

#### 2.2.1.1 Applicable Design Parameters

The design basis tornado and wind loading is defined based on Regulatory Guide 1.76 Region 1 and NUREG-0800. The tornado and wind loading criteria are presented in Table 2.2-1.

#### 2.2.1.2 Determination of Forces on Structures

Tornado wind forces on the NAC-MPC are calculated by multiplying the dynamic wind pressure by the frontal area of the cask normal to the wind direction. Wind forces are applied to the cask in the wind direction. No streamlining is assumed. The evaluation of wind loading and tornado missile effects on the NAC-MPC is presented in Section 11.2.13.

#### 2.2.1.3 <u>Tornado Missiles</u>

The design basis tornado missile impacts are defined in Paragraph 4, Subsection III, Section 3.5.1.4 of NUREG-0800. The design basis tornado is considered to generate three types of missiles that impact the cask at normal incidence:

1.	Massive Missile -	Weight = 1800 kg (3960 pounds)
	(Deformable w/high	Frontal Area = 20 sqft.
	kinetic energy)	
_		
2.	Penetration Missile -	Weight = $125 \text{ kg} (275 \text{ pounds})$
	(Rigid hardened steel)	Diameter = 8.0 inches
3.	Protective Barrier Missile -	Weight = $0.068 \text{ kg} (0.15 \text{ pounds})$
	(Solid steel sphere)	Diameter = $1.0$ inch

Each missile is assumed to impact the cask at a velocity of 126 miles per hour, horizontal to the ground, which is 35 percent of the maximum wind speed of 360 miles per hour. For missile impacts in the vertical direction, the assumed missile velocity is: (0.7)(126) = 88.2 miles per hour.

The detailed analysis of the NAC-MPC for the missile impacts applies the laws of conservation of momentum and conservation of energy to estimate the impact force on the cask as a function of the time of impact and the amount of missile deformation. Each missile impact is evaluated, and all missiles are assumed to impact in a manner that produces the maximum damage to the NAC-MPC.

#### 2.2.2 Water Level (Flood) Design

The NAC-MPC may be exposed to a flood during storage on an unsheltered concrete storage pad at an ISFSI site. The source and magnitude of the probable maximum flood depend on several variables.

#### 2.2.2.1 Flood Elevations

The NAC-MPC is evaluated for a maximum flood water depth of 50 feet above the base of the storage cask. The flood water velocity is considered to be 15 feet per second. Under these conditions, the NAC-MPC will not float, tip or significantly slide on the storage pad, and the confinement function will be maintained.

2.2-2

#### 2.2.2.2 Phenomena Considered in Design Load Calculations

The occurrence of flooding at an ISFSI site is dependent upon the specific site location and the surrounding geographical features, natural and man-made. Flooding of an ISFSI site is highly improbable because of the extensive environmental impact studies that are performed during the selection of a site for a nuclear facility. Some possible sources of a flood at an ISFSI site are: (1) overflow from a river or stream, due to unusually heavy rain, snow-melt runoff, a dam or major water supply line break caused by a seismic event (earthquake); (2) high tides produced by a hurricane; and (3) a tsunami (tidal wave) caused by an underwater earthquake or volcanic eruption.

#### 2.2.2.3 Flood Force Application

The evaluation of the NAC-MPC for a flood condition determines a maximum allowable flood water current velocity and a maximum allowable flood water depth. The criteria employed in the determination of the maximum allowable values are that cask sliding or tip-over will not occur, and that the canister material yield strength is not exceeded. The evaluation of the effects of flood conditions on the NAC-MPC is presented in Section 11.2.6.

The force of the flood water current on the NAC-MPC is calculated as a function of the current velocity by multiplying the dynamic water pressure by the frontal area of the cask that is normal to the current direction. The dynamic water pressure is calculated using Bernoulli's equation relating fluid velocity and pressure. The force of the flood water current is limited such that the overturning moment on the cask will be less than that required to tip the cask over.

#### 2.2.2.4 Flood Protection

The inherent strength of the reinforced concrete cask component of the NAC-MPC provides a substantial margin of safety against any permanent deformation of the cask for a credible flood event at an ISFSI site. Therefore, no special flood protection measures for the NAC-MPC are necessary. The evaluation presented in Section 11.2.6 shows that for the design basis flood, the allowable stresses in the canister are not exceeded.

#### 2.2.3 <u>Seismic Design</u>

The NAC-MPC may be exposed to a seismic event (earthquake) during storage on an unsheltered concrete pad at an ISFSI site. The seismic response spectra experienced by the cask will depend upon the geographical location of the specific site and the distance from the epicenter of the earthquake. The only significant effect of a seismic event on the NAC-MPC would be a possible tip-over; however, tip-over does not occur in the evaluated design basis earthquake. Seismic response of the NAC-MPC is presented in Section 11.2.2.

#### 2.2.3.1 Input Criteria

The magnitude of the maximum seismic accelerations to which the NAC-MPC may be subjected to are site specific. 10 CFR 72.102 defines a 0.10 g horizontal ground motion design earthquake as the minimum allowable seismic design criteria, and 0.25 g is suggested for sites east of the Rocky Mountain front. The NAC-MPC is designed to 0.25 g horizontal and 0.167 g vertical seismic acceleration. This acceleration provides seismic qualification for a predominant number of nuclear facilities within the United States.

#### 2.2.3.2 <u>Seismic - System Analyses</u>

The seismic ground acceleration that will cause the NAC-MPC to tip over is calculated in Section 11.2.2 using quasi-static analysis methods. Both horizontal and vertical acceleration components are considered in the analyses. These components are calculated and combined according to Section 3.7.1 of NUREG-0800. Evaluation of the consequences of a tip over event is provided in Section 11.2.12.

#### 2.2.4 <u>Snow and Ice Loadings</u>

The criterion for determining design snow loads is based on ANSI/ASCE 7-93, Section 7.0. Flat roof snow loads apply and are calculated from the following formula:

$$p_f = 0.7C_eC_tIp_g$$

where:

 $p_{f} = \text{flat roof snow load (psf)}$   $C_{e} = \text{Exposure factor} = 1.0$   $C_{t} = \text{Thermal factor} = 1.2$  I = Importance factor = 1.2  $p_{g} = \text{ground snow load, (psf)} = 100$ 

The numerical values of  $C_e$ ,  $C_t$ , I and  $p_g$  are obtained from Tables 18, 19, 20 and Figure 7, respectively, of ANSI/ASCE 7-93.

The exposure factor accounts for wind effects. The NAC-MPC is assumed to have a site location typical for siting Category C, which is defined to be "locations in which snow removal by wind cannot be relied on to reduce roof loads because of terrain, higher structures, or several trees nearby."

The thermal factor accounts for the importance of buildings and structures in relation to public health and safety. The NAC-MPC is conservatively classified as Category III.

Ground snow loads for the contiguous United States are given in Figures 5, 6 and 7 of ANSI/ ASCE 7-93. A worst case value of 100 pounds per square foot (psf) was assumed.

Based on the above, the design criterion for snow and ice loads is:

Flat Roof Snow Load,  $p_f = (0.7) (1.0) (1.2) (1.2) (100) = 100.8 \text{ psf}$ 

This load is bounded by the weight of the loaded transfer cask. The snow load is considered in the load combinations described in Section 3.4.4.2.2.

#### 2.2.5 <u>Combined Load Criteria</u>

Each normal, off-normal and accident condition has a combination of load cases that defines the total combined loading for that condition. The individual load cases considered include thermal, seismic, external and internal pressure, missile impacts, drops, snow and ice loads, and/or flood water forces.

The load conditions to be evaluated for storage casks are identified in 10 CFR 72 and in the "Design Criteria for an Independent Spent Fuel Storage Installation (Dry Storage Type)" (ANSI/ANS 57.9 - 1992).

#### 2.2.5.1 Load Combinations and Design Strength - Concrete Cask

The load combinations specified in ANSI/ANS 57.9 - 1992 for concrete structures are applied to the concrete casks as shown in Table 2.2-2. The live loads are considered to vary from 0 percent to 100 percent to ensure that the worst-case condition is evaluated. In each case, use of 100 percent of the live load produces the maximum load condition. The steel liner of the concrete cask is a stay-in-place form and it provides radiation shielding. The concrete cask is designed to the requirements of ACI 349.

#### 2.2.5.2 Design Strength Reduction Factors - Concrete

In calculating the design strength of the NAC-MPC concrete body, nominal strength values are multiplied by a strength reduction factor in accordance with Section 9.3 of ACI 349.

#### 2.2.5.3 Load Combinations and Design Strength - Canister and Basket

The canister is designed in accordance with the 1995 edition of the ASME Code, Section III, Subsection NB for Class 1 components. The basket structure is designed per ASME Code, Section III, Subsection NG, and structural buckling of the basket is evaluated per NUREG/CR-6322.

The load combinations for all normal, off-normal, and accident conditions and corresponding service levels are shown in Table 2.2-3. Levels A and D service limits are used for normal and accident conditions, respectively. Levels B and C service limits are used for off-normal conditions. The analysis methods allowed by the ASME Code are employed. Stress intensities caused by pressure, temperature, and mechanical loads are combined before comparing to ASME code allowables, which are listed in Table 2.2-4.

#### 2.2.5.4 Design Strength - Transfer Cask

The transfer cask is a special lifting device and is designed and fabricated to the requirements of ANSI N14.6 and NUREG 0612 for the lifting trunnions and supports. The criteria are:

The combined shear stress or maximum tensile stress during the lift (with 10 percent dynamic load factor) shall be  $\leq S_y/6$  and  $S_u/10$  for a nonredundant load path, or shall be  $\leq S_y/3$  and  $S_u/5$  for redundant load paths.

The ferritic steel material used for the load bearing members of the transfer cask shall satisfy the material toughness requirements of ANSI N14.6, paragraph 4.2.6.

#### 2.2.6 Environmental Temperatures

A temperature of 75°F was selected to bound all annual average temperatures in the United States, except the Florida Keys and Hawaii.

The 75°F normal temperature was used as the base for thermal evaluations. The evaluation of this environmental condition is discussed along with the thermal analysis models in Chapter 4.0. The thermal stress evaluation for the normal operating conditions is provided in Section 3.4.4. Normal temperature fluctuations are bounded by the severe ambient temperature cases that are evaluated as off-normal and accident conditions.

Off-normal, severe environmental conditions were defined as -40°F with no solar loads and 100°F with solar loads. An extreme environmental condition of 125°F with maximum solar loads is evaluated as an accident case to show compliance with the maximum heat load case required by ANSI-57.9 (Section 11.2.10). Thermal performance was also evaluated for the cases of: (1) half the air inlets blocked; and (2) all air inlets and outlets blocked. Thermal analyses for

NAC-MPC FSAR	April 2000
Docket No. 72-1025	Revision 0

these cases are presented in Sections 11.1.1 and 11.2.8. The evaluation based on ambient temperature conditions is presented in Section 4.4.

The design basis temperatures used in the NAC-MPC analysis are shown below. Solar insolance is as specified in 10 CFR 71.71 and Regulatory Guide 7.8.

<u>Condition</u>	Ambient Temperature	Solar Insolance
Normal	75°F	yes
Off-Normal - Severe Heat	100°F	yes
Off-Normal - Severe Cold	-40°F	no
Accident - Extreme Heat	125°F	yes

/

# Table 2.2-1 Tornado and Wind Loading Criteria

Environmental Condition	Limit
Rotational Wind Speed, mph	290
Translational Wind Speed, mph	70
Maximum Wind Speed, mph	360
Radius of Max. Wind Speed, ft.	150
Pressure Drop, psi	3.0
Rate of Pressure Drop, psi/sec	2.0

Load Combination	Condition	Dead	Live	Wind	Thermal	Seismic	Tornado/ Missile	Drop/ Impact	Flood
1	Normal	1.4D	1.7L						
2	Normal	1.05D	1.275L		1.275T _o				
3	Normal	1.05D	1.275L	1.275W	1.275T _o				
4	Off-Normal & Accident	D	L		T _a				
5	Accident	D	L		T _o	E _{ss}			
6	Accident	D	L		To			Α	
7	Accident	D	L		Τ₀	·····			F
8	Accident	D	L		To		W _t		

## Table 2.2-2 Load Combinations for the NAC-MPC Vertical Concrete Cask

Load Combinations are from ANSI 57.9 and ACI 349.

D	=	Dead Load	Ta	=	Off- Normal or Accident Temperature
L	=	Live Load	E _{ss}	=	Design Basis Earthquake
W	=	Wind	Wt	=	Tornado/Tornado Missile
To	=	Normal Temperature	А	=	Drop/Impact
F	=	Flood			

# Table 2.2-3 Load Combinations for the Transportable Storage Canister

	LOAD		ORN	<b>IAL</b>	OFF-NORMAL				ACCIDENT						
ASME Service Lev	el	A				B C				D					
Load Combinations		1	2	3	1	2	3	4	5	1	2	3	4	5	6
DEAD WEIGHT	Canister with fuel	X	Χ	X	X	X	X	Х	Х	X	X	X	Х	х	Х
Thermal	In Storage Cask 75° F Ambient	x		X				X		x	x	x	X	X	
	In Transfer Cask 75° F Ambient		X		x		x								x
	In Storage Cask -40°F or 100°F Ambient					X			X						
INTERNAL Pressure	Normal	X	х	Х				Х		x	X	х	х		
	Off-Normal Accident				X	X	X		X					x	X
HANDLING LOAD	Normal Off-Normal		X	X	X		x	X	X						
DROP/IMPACT	Accident				T					X					
Seismic	Accident										X				
FLOOD	Accident											Χ			
Tornado	Accident												Χ		

# Table 2.2-4 Structural Design Criteria for Components Used in the Transportable Storage Canister

	Component	Criteria
1.	Normal Operations: Service Level A	$P_m \leq S_m$
	Canister: ASME Section III, Subsection NB	$P_L + P_b \le 1.5 S_m$
	Basket: ASME Section III, Subsection NG	$P_L + P_b + Q \le 3S_m$
	Conjeter Lifting Devices	
	ANSI N14 6 and NUREC 0612	
	AINSI IN14.0 and INUREG 0012	Redundant load path: combined shear or max. tensile stress $\leq S_u/5$ and $S_y/3$
2.	Off-Normal Operations: Service Level B	$P_{\rm m} < 1.1 \ {\rm S}_{\rm m}$
	Canister: ASME Section III, Subsection NB	$P_{L} + P_{b} < 1.65 S_{m}$
2	Off Normal Operational Service Level C	
5.	Consistent ASME South HI S. Level C	Subsection NB Allowables:
	Canister: ASME Section III, Subsection NB	$P_m < 1.2 S_m$ or $S_y$ (whichever is greater)
	Basket: ASME Section III, Subsection NG	$P_L + P_b < 1.8 S_m \text{ or } 1.5 S_y \text{ (whichever is less)}$
		Note: Level C allowables for Subsection NG
		are larger than those for Level C per Subsection
		NB. Therefore, it is conservative to employ
		Subsection NB allowables for the basket.
4	Accident Conditions Service Level D	$\mathbf{P} = 245 \times 0.75$
7.	Canister: ASME Section III Subsection ND	$\Gamma_{\rm m} \leq 2.4  {\rm S_m}  {\rm OI}  {\rm O}.7  {\rm S_u}$
	Baskett ASME Section III, Subsection INB	(whichever is less)
	Basket: ASME Section III, Subsection NG	$P_{\rm L} + P_{\rm b} \le 3.6  \rm S_m  or  1.05  \rm S_u$
		(whichever is less)
5.	Basket Structural Buckling	NUREG/CR-6322

#### 2.3 <u>Safety Protection Systems</u>

The NAC-MPC relies upon passive systems to ensure the protection of public health and safety, except in the case of fire or explosion. As discussed in Section 2.3.6, fire and explosion events are effectively precluded by site administrative controls that prevent the introduction of flammable and explosive materials. The use of passive systems provides protection from mechanical or equipment failure.

#### 2.3.1 <u>General</u>

The NAC-MPC is designed for safe, long-term storage of spent nuclear fuel. The NAC-MPC will survive all of the evaluated normal, off-normal, and postulated accident conditions without release of radioactive material or excessive radiation exposure to workers or the general public. The major design considerations that have been incorporated in the NAC-MPC system to assure safe long-term fuel storage are:

- 1. Continued confinement in postulated accidents.
- 2. Thick concrete and steel biological shield.
- 3. Passive systems that ensure reliability.
- 4. Inert atmosphere to provide corrosion protection for stored fuel cladding.

Each NAC-MPC system storage component is classified with respect to its function and corresponding effect on public safety. In accordance with Regulatory Guide 7.10, each system component is assigned safety classification into Category A, B or C, as shown in Table 2.3-1. The safety classification is based on review of each component's function and the assessment of the consequences of component failure following the guidelines of NUREG/CR-6407, "Classification of Transportation Packaging and Dry Spent Fuel Storage System Components According to Importance to Safety."

Category A - Components critical to safe operations whose failure or malfunction could directly result in conditions adverse to safe operations, integrity of spent fuel or public health and safety.

Category B - Components with major impact on safe operations whose failure or malfunction could indirectly result in conditions adverse to safe operations, integrity of spent fuel or public health and safety.

Category C - Components whose failure would not significantly reduce the packaging effectiveness and would not likely result in conditions adverse to safe operations, integrity of spent fuel, or public health and safety.

As discussed in the following sections, the NAC-MPC design incorporates features addressing the above design considerations to assure safe operation during fuel loading, handling, and storage.

#### 2.3.2 <u>Protection by Multiple Confinement Barriers and Systems</u>

#### 2.3.2.1 Confinement Barriers and Systems

The radioactivity that the NAC-MPC must confine originates from the spent fuel assemblies to be stored and residual contamination that may remain inside the canister as a result of contact with the water in the fuel pool where the canister loading is conducted.

The NAC-MPC is designed to confine the radioactive fuel. The canister is closed by welding. The shield lid weld is pressure tested. All of the field-installed welds are liquid penetrant examined following the root and final weld passes. The shield lid weld is leak tested to  $1.0 \times 10^{-7}$  cubic centimeters per second (air). The closure of the canister structural lid, which provides a redundant closure over the shield lid and port covers, is accomplished by multi-pass welding that is tested by liquid penetrant examination on the root, intermediate and final passes. The longitudinal and girth welds and bottom welds of the canister shell are full penetration welds that are radiographed or ultrasonically inspected during fabrication.

The canister welds are an impenetrable boundary to the release of fission gas products during the period of storage. There are no evaluated normal, off-normal, or accident conditions that result in the breach of the canister and the subsequent release of fission products. The canister is designed to withstand a postulated drop accident in a transportation cask without precluding the subsequent removal of the fuel (i.e., the fuel tubes do not deform such that they bind the fuel).

Personnel radiation exposure during handling and closure of the canister is minimized by the following steps:

- 1. Placing the shield lid on the canister while the transfer cask and canister are under water in the fuel pool.
- 2. Decontaminating the exterior of the transfer cask prior to draining the canister to preserve the shielding benefit of the water.
- 3. Using temporary shielding.
- 4. Using a retaining ring on the transfer cask to ensure that the canister is not raised out of the shield provided by the transfer cask.
- 5. Placing a shielding ring over the annular gap between the transfer cask and the canister.

#### 2.3.3 Protection by Equipment and Instrumentation Selection

The NAC-MPC is a passive storage system that does not rely on equipment or instruments to preserve public health or safety and to meet its safety functions in long-term storage. The system employs support equipment and instrumentation to facilitate operations. These items and the actions taken to assure performance are described below.

#### 2.3.3.1 Equipment

The only important-to-safety equipment employed in the use and operation of the NAC-MPC is the lifting yoke used to lift the transfer cask. The transfer cask lifting yoke is designed to meet the requirements of ANSI N14.6 and NUREG-0612. It is single failure-proof by design. The lifting yoke is proof load tested to 300 percent of design load when fabricated. The lifting yoke is inspected for visible defects prior to each use and is inspected annually.

Additional handling equipment (such as trailers, skids, air pads, portable cranes, or cask transporters) are not important to safety as the NAC-MPC system is designed to withstand the failure of any of these components.

#### 2.3.3.2 Instrumentation

A remote temperature measuring system is employed to measure the outlet air temperature of the NAC-MPC in long-term storage. The outlet temperature is recorded daily as a check of the thermal performance of the heat rejection capability of the storage cask. The outlet temperature is expected to increase in the unlikely event that one or more inlet or outlet ventilation ports become blocked.

The inlet and outlet ports are visually inspected each day during the same walk-through in which the temperatures are recorded. This visual inspection assists in ensuring that the temperature measuring system is continuing to perform as expected.

The canister shield lid weld is helium leak tested during closure. The leak detector is checked against a known helium source immediately prior to, and after, use to preclude unknown leak detector failure.

#### 2.3.4 <u>Nuclear Criticality Safety</u>

The primary nuclear criticality safety design criterion of the NAC-MPC is to provide features that ensure that the cask remains subcritical under normal, off-normal, and accident conditions. Neutron poison sheets (BORAL) are employed in the basket design to capture thermalized neutrons, and preclude uncontrolled fission events. BORAL sheets are attached to each side of each fuel tube. These sheets are mechanically supported by the fuel tube structure to ensure that the poison sheets remain in place during the design basis normal, off-normal, and accident events.

The efficiency of the BORAL sheets in preserving nuclear criticality safety is demonstrated by the Criticality Evaluation presented in Chapter 6.

#### 2.3.4.1 Error Contingency Criterion

The design of the canister and fuel basket is such that, under all conditions, the highest neutron multiplication factor ( $k_{eff}$ ) will be less than 0.95. The criticality evaluation for the design basis fuel is presented in Section 6.4. Assumptions made in the analyses used to demonstrate conformance to this criterion include:

- 1. Most reactive Yankee Class fuel assembly type with maximum ²³⁵U loading;
- 2. 75 percent of the nominal ¹⁰B loading in the BORAL;
- 3. Infinite array of casks in the X-Y (horizontal) plane;
- 4. Infinite fuel length with no inclusion of end leakage effects;
- 5. No structural material present in the assembly;
- 6. No credit taken for boron in the cask cavity or surrounding loading or storage area; and
- 7. No credit taken for fuel burnup or for the buildup of fission product neutron poisons.

These assumptions demonstrate adequate controls to assure subcriticality in the use of the NAC-MPC system.

#### 2.3.5 Radiological Protection

The NAC-MPC system, in keeping with the As Low As Reasonably Achievable (ALARA) philosophy, is designed to minimize, to the extent practicable, operator radiological exposure.

#### 2.3.5.1 <u>Access Control</u>

Access to an NAC-MPC ISFSI site is controlled by a peripheral fence to meet the requirements of 10 CFR 72 and 10 CFR 20. Access to the storage area, and its designation as to the level of radiation protection required, is established by site procedure. The storage area will be surrounded by a fence, having lockable truck and personnel access gates. The fence will have intrusion-detection features as determined by the site procedure.

#### 2.3.5.2 Shielding

The NAC-MPC is designed to provide an external side surface dose (gamma and neutron) of less than 50 mrem/hr (average) on the storage cask sides, 35 mrem/hr (average) top, and 100 mrem/hr at the air vent inlets. The transfer cask side wall contact dose rate limit is 200 mrem/hr (average). The design maximum dose rate at the top of the canister structural lid, with supplemental shielding, is 200 mrem/hr (average) to limit personnel exposure during canister closure operations.

Sections 72.104 and 72.106 of 10 CFR 72 set whole body dose limits for an individual located beyond the controlled area at 25 millirems per year (whole body) during normal operations and 5 rems (5,000 millirems) from any design basis accident. The analyses showing the actual NAC-MPC doses are included in Sections 5.0 and 11.0.

#### 2.3.5.3 Ventilation Off-Gas

The NAC-MPC is passively cooled by radiant and natural convection heat transfer at the outer surface of the canister and natural convective heat transfer in the canister-concrete cask annulus. The bottom of the cask is conservatively assumed to be an adiabatic surface. The design criterion for the air-flow in the annulus is that the pressure difference, due to the buoyancy effect created by the heating of the air, is equal to the flow pressure drop. The details of the passive ventilation system design are provided in Section 4.0.

There are no radioactive releases during normal operations. Also, there are no credible accidents that cause significant releases of radioactivity from the NAC-MPC and, hence, there are no off-gas system requirements for the NAC-MPC during normal storage operation. The only time an off-gas system is required is during the canister drying phase. During this operation, the reactor off-gas system or a HEPA filter system will be used.

The surface of the canister is exposed to cooling air when the canister is placed in the storage cask. If the surface is contaminated, the possibility exists that contamination could be carried aloft by the cooling air stream. To ensure that the canister surface is free of contamination, pool water is prevented from contacting the canister exterior by filling the transfer cask/canister annular gap with clean water as the transfer cask is being lowered into the fuel pool.

2.3-6

NAC-MPC FSAR	April 2000
Docket No. 72-1025	Revision 0

Clean water is injected into the gap during the entire time the transfer cask is submerged. These steps preclude the intrusion of contaminated water into the canister annular gap.

Once the transfer cask is removed from the pool, a smear survey is taken of the exterior surface of the canister near the top. While no contamination is expected to be found, it is possible that the surface could be contaminated. The allowable upper limit on surface contamination is calculated in Section 12.2.1.4. If this limit is exceeded, then steps to decontaminate the canister surface must be taken and continued until the contamination is less than the allowable limit.

To facilitate decontamination, the canister is fabricated so that its exterior surface is smooth. There are no corners or pockets that could trap and hold contamination.

#### 2.3.5.4 <u>Radiological Alarm Systems</u>

There are no radiological alarms required on the NAC-MPC. Justification for this is provided in analysis in Sections 5.0 (Shielding), 10.0 (Radiological Protection), and 11.0 (Accident Analysis).

Typically, total radiation exposure due to the ISFSI installation is determined by the use of Thermo-Luminescent Detectors (TLDs) mounted at convenient locations on the ISFSI fence. The TLDs are read quarterly to provide a record of boundary dose.

#### 2.3.6 Fire and Explosion Protection

Fire and explosion protection of the NAC-MPC is primarily provided by administrative controls applied at the site, which preclude the introduction of any explosive and any excessive flammable materials into the ISFSI area.

#### 2.3.6.1 Fire Protection

A major ISFSI fire is not considered credible, since there is very little material near the casks that could contribute to a fire. The concrete cask is largely impervious to incidental thermal events. Administrative controls will be put in place to ensure that the presence of combustibles is minimized. A hypothetical fire event is evaluated as an accident condition in Section 11.2.5.

#### 2.3.6.2 <u>Explosion Protection</u>

The cask and associated systems are analyzed to ensure their proper function under an overpressure condition. As described in Section 11.2.3, in the evaluated 22 psig over pressure condition, stresses in the canister remain below allowable limits and there is no loss of confinement. These results are conservative as the canister is protected from direct over-pressure conditions by the concrete storage cask.

For the same reasons as the fire condition, a severe explosion on an ISFSI site is not considered credible. The evaluated over-pressure is considered to bound any explosive over pressure resulting from an industrial explosion at the boundary of the owner-controlled area.

# Table 2.3-1 Safety Classification of NAC-MPC Components

				Item			Safety
Drawing	Ch	Title	Rev	No.	Component	Function	Class
Number	51	Woldmont Structure	3	26	Baffle	Heat Transfer	В
455-801	1	Weldment, Structure	5		Outlet (4)	Heat Transfer	В
	Ì			20	Shield Plate	Shielding	B
2 2				17	Nelson Stud	Structural	С
				16	Base Plate	Structural	B
				15	Stand	Structural	B
					Inlet (4)	Heat Transfer	B
				12	Bottom	Structural	B
	]			11	Shield Ring	Shielding	B
	1		1	10	Cover	Operations	В
			i	-	Jack (Leveling)	Operations	<u> </u>
				3	Support Ring	Structural	B
				2	Top Flange	Structural	В
		1		1	Shell	Structural	B
155 967	+	Loaded Vertical Concrete Cask	2	9	Cover	Operations	В
455-802	1	Loaded Vertical Concrete Cash		8	Insulation	Operations	В
				7	Washer (Lid Bolt)	Operations	C
				6	Lid Bolt	Operations	B
				5	Lid	Operations	B
				4	Shield Plug	Shielding	B
455-866	1	Reinforcing Bar and Concrete	0	18	Name Plate	Operations	C
		r lacement		17	Outlet Screen	Operations	C
				16	Inlet Screen	Operations	C
				15	Concrete	Shielding/ Structural	B
				13	Liner Weldment	Shielding/ Structural	B

Drawing Number	Sh	Title	Dou	Item	Component	Electron d'automation and a second and a second	Safety
155-866	1	Deinforcing Der and Conserve	<u>Rev</u>	110.		Function	Class
(continued)		Placement	0	-	Reinforcing Bar	Structural	B
455 970	+	Conjeten Chall					
433-870	1	Canister, Snell	3	3	Location Nut	Operations	В
				2	Bottom	Structural/ Confinement	A
			<u> </u>	1	Shell	Structural/ Confinement	Α
455-871	1	Details, Canister	2	8	Кеу	Operations	С
				7	Port Cover (Two)	Confinement/	В
						Operations	
				6	Valved Nipple (Two)	Operations	С
				5	Structural Lid	Structural	Α
				3	Shield Lid	Shielding	В
				2	Backing Ring	Structural	С
·				1	Lid Support Ring	Structural	В
455-872	1	Assembly, Transportable Storage Canister	5	3	Drain Tube Assembly	Operations	С
455-881	1	PWR Fuel Tube	2	4	Flange	Structural	A
				3	Cladding	Criticality Control	A
				2	BORAL	Criticality Control	Α
				1	Tubing	Structural/ Criticality	A
455-887		Drawing Deleted					
455-888		Drawing Deleted					
455-891	1	Bottom Weldment, Fuel Basket	0	-	Support	Structural	В
				2	Circular Pad	Structural	C
				1	Plate	Structural	Α

# Table 2.3-1 Safety Classification of NAC-MPC Components (Continued)

NAC-MPC FSAR	April 2000
Docket No. 72-1025	Revision 0

# Table 2.3-1Safety Classification of NAC-MPC Components (Continued)

.

Drawing				Item			Safety
Number	Sh	Title	Rev	No.	Component	Function	Class
455-892	1	Top Weldment, Fuel Basket	1	4	Baffle	Structural	Α
				3	Support	Structural	A
				2	Support Ring	Structural	В
				1	Plate	Structural	A
455-893	1	Support Disk and Misc. Basket Details	3	7	Top Spacer	Structural	В
				6	Split Spacer	Structural	B
				5	Tie Rod	Structural	A
				4	Top Nut	Structural	Α
				3	Bottom Spacer	Structural	В
				2	Spacer	Structural	A
				1	Plate	Structural	A
455-894	1	Heat Transfer Disk	1	1	Heat Transfer Disk	Heat Transfer	A
455-895	1	Fuel Basket Assembly	2	13	Flat Washer	Structural	В
				4	Drain Tube Sleeve	Operations	C
455-860	1	Assembly, Transfer Cask	3	17	Connector	Operations	С
				16	Fill/ Drain Line	Operations	C
				15	Retaining Ring Bolt	Operations	В
				14	Retaining Ring	Operations	В
				13	Door Lock Bolt	Operations	С
				11-12	Shield Door	Structural/Shielding	B
				10	Door Rail	Operations	В
				9	Top Plate	Structural	В
				8	Neutron Shield	Shielding	В
				7	Scuff Plate	Operations	C
				6	Trunnion Cap	Operations	C
				5	Trunnion	Structural	B

Table 2.3-1	Safety Classification of NAC-MPC Components (Continued)	
-------------	---------------------------------------------------------	--

Drawing				Item			Safety
Number	Sh	Title	Rev	No.	Component	Function	Class
455-860 (continued)		Assembly, Transfer Cask	3	4	Outer Shell	Structural	В
				3	Gamma Shield Brick	Shielding	В
				2	Inner Shell	Structural	В
				1	Bottom Plate	Structural	В
YR-00- 060	1	Yankee – Class Reconfigured Fuel Assembly	1	1	Shell Weldment	Structural	A
				2	Top End Fitting Assembly	Structural/ Criticality	A
				3	Bottom End Fitting Assembly	Structural/ Criticality	A
				4	Basket Assembly	Structural/ Criticality	A
				5	Top Nozzle Bolts	Structural/ Criticality	Α
				6	Alignment Pin	Operations	C

#### 2.4 Decommissioning Considerations

The principal elements of the NAC-MPC storage system are the vertical concrete cask (storage cask) and the transportable storage canister (canister).

The storage cask provides biological shielding and physical protection for the contents of the The storage cask is not expected to become surface canister during long-term storage. contaminated during use, except through incidental contact with other contaminated surfaces. Incidental contact could occur at the interior surface (liner) of the storage cask, the top surface that supports the transfer cask during loading and unloading operations, and the floor of the storage cask that supports the canister. All of these surfaces are carbon steel, and it is anticipated that these surfaces could be decontaminated as necessary for decommissioning. A layer of insulation and stainless steel is placed on the floor of the storage cask in order to separate the stainless steel canister bottom from the carbon steel storage cask bottom plate. The insulation rests on the storage cask carbon steel pedestal. The insulation is covered by a sheet of stainless steel. Contamination of these surfaces is expected to be minimal, since the canister is isolated from spent fuel pool water during loading in the pool and the transfer cask is decontaminated prior to transfer of the canister to the storage cask. In the unlikely event that the insulation became contaminated, it is not reasonable to expect that it could be decontaminated. Consequently, the insulation would have to be disposed of as surface-contaminated material.

The concrete that provides biological shielding is not expected to become contaminated during the period of use, as it does not come into contact with other contaminated objects or surfaces.

Activation of the carbon steel liner, concrete, support plates, and reinforcing bar could occur due to neutron flux from the stored fuel. Since the neutron flux rate is low, only minimal activation of carbon steel in the storage cask is expected to occur. The activity concentrations from activation of storage cask components are listed in Table 2.4-1. Table 2.4-1 includes the radiological significant isotopes, together with a total concentration of all activated nuclides in the respective component. The total concentrations listed include activities of radionuclides, which do not have any substantial contribution to radiation dose and are not specifically identified by 10 CFR 61 waste classification. In particular the isotope contributing the majority of the carbon steel total curie activity is ⁵⁵Fe, which decays by electron capture and is not of radiological concern.

Decommissioning of the storage cask would involve the removal of the canister and the subsequent disassembly of the storage cask. It is expected that the concrete would be broken up, and steel components segmented to reduce volume. Any contaminated or activated items are expected to qualify for near-surface disposal as low specific activity material.

The transportable storage canister is designed and fabricated to be suitable for use as a waste package for permanent disposal in a deep Mined Geological Disposal System, in that it meets the requirements of the DOE MPC Design Procurement Specification. The canister is fabricated from materials having high long-term corrosion resistance, and the canister contains no paints or coatings that could adversely affect the permanent disposal of the canister. Consequently, decommissioning of the canister would occur only if the fuel contained in the canister had to be removed, or if current requirements for disposal were to change. Decommissioning would require that the closure welds at the canister structural lid, shield lid and shield lid port covers be cut, so that the spent fuel could be removed. Removal of the contents of the canister would require that the canister be returned to a spent fuel pool or dry unloading facility, such as a hot cell. Closure welds can be cut either manually or with automated equipment, with the procedure being essentially the reverse of that used to initially close the canister.

Following removal of the contents, the canister could have significant internal contamination due to the contents, and may contain "crud" or other residual material in the bottom of the canister. Some effort may be required to remove the surface contamination prior to disposal; however, in practice, it would not be absolutely necessary to decontaminate the canister internals. Any contaminated canister and internal components are expected to qualify for near-surface disposal as low specific activity waste without internal contamination, as the internal contamination would consist only of by-product materials. Should internal decontamination be necessary, the canister and basket surfaces are smooth, and the design precludes the presence of crud traps, thus facilitating any required decontamination. Since the neutron flux rate from the stored fuel is low, only minimal activation of the canister is expected to occur. The activity concentrations from activation of canister components are listed in Table 2.4-1.

The unloaded canister could also qualify as a strong, tight container for other waste. In this case, the canister could be filled, within weight limits, with other qualified waste, closed and transported whole and complete to a near-surface disposal site. Use of the canister for this purpose could reduce decommissioning costs by avoiding decontamination, segmenting and repackaging.

NAC-MPC FSAR	April 2000
Docket No. 72-1025	Revision 0
DUCKULIU. 72 1025	

The storage pad, fence and supporting utility fixtures are not expected to require decontamination as a result of use of the NAC-MPC system. The design of the cask and canister precludes the release of contamination from the contents over the period of use of the system. Consequently, these items may be reused or disposed of as locally generated clean waste.

# Table 2.4-1Activity Concentration Summary for the NAC-MPC Vertical Concrete<br/>Cask and Transportable Storage Canister

		Activity C	oncentration ¹	(Ci/m ³ )	
	Verti	ical Concrete Ca	sk	Transpor	table Storage nister
Isotope	Shell ²	Shield Plug	Lid	Shell	Lids
⁵⁴ Mn	4.2E-03	8.77E-04	1.16E-04	6.0E-05	1.52E-05
⁶⁰ Co	2.7E-05	1.40E-05	3.05E-06	2.3E-04	8.44E-05
⁵⁹ Ni				2.6E-07	
⁶³ Ni				3.0E-01	1.10E-01
Total	8.9E-02	4.22E-02	1.19E-02	3.1E-01	1.13E-01

1. Seven days after removal of spent fuel.

2. Includes liner and concrete.

# **Table of Contents**

3.0	STRUCTURAL EVALUATION	3.1.1
0.1		211
3.1	Structural Design	3.1-1
	3.1.1 Discussion	
	3.1.2 Design Criteria	3.1-4
3.2	Weights and Centers of Gravity	3.2-1
3.3	Mechanical Properties of Materials	3.3-1
3.4	General Standards for Casks	3.4-1
	3.4.1 Chemical and Galvanic Reactions	3.4-1
	3.4.1.1 Component Operating Environment	3.4-1
	3.4.1.2 Component Material Categories	3.4-2
	3.4.1.3 General Effects of Identified Reactions	3.4-7
	3.4.1.4 Adequacy of the Canister Operating Procedures	3.4-7
	3.4.1.5 Effects of Reaction Products	3.4-7
	3.4.2 Positive Closure	3.4-7
	3.4.3 Lifting Devices	3.4-9
	3.4.3.1 Storage Cask Bottom Lift	3.4-12
	3.4.3.2 Canister Lift	3.4-19
	3.4.3.3 Transfer Cask Lift	3.4-27
	3.4.4 NAC-MPC Components Under Normal Operating Loads	3.4-47
	3.4.4.1 Canister and Basket Analyses	3.4-47
	3.4.4.2 Vertical Concrete Storage Cask - Concrete Stress Analysis	3.4-81
	3.4.5 Cold	3.4-91
3.5	Fuel Rods	3.5-1

NAC-MPC FSAR	•
Docket No. 72-10	25

____

.

3.6	Canister Closure Weld Evaluation - Normal Conditions	
	3.6.1 Stress Evaluation for the Canister Closure Weld	
	3.6.2 Critical Flaw Size for the Canister Closure Weld	
3.7	References	
3.8	Coating Specifications	
	3.8.1 Keeler and Long E-Series Epoxy Enamel	
	3.8.2 Ameron PSX 738 Siloxane Coating	

1

# List of Figures

Figure 3.1-1	Principal Components of the NAC-MPC System 3.1-5
Figure 3.4.2-1	NAC-MPC Welded Closure System 3.4-8
Figure 3.4.3-1	Transfer Cask Lifting Trunnion Design 3.4-10
Figure 3.4.3-2	Canister Hoist Ring Design
Figure 3.4.3.2-1	Canister Lift Finite Element Model 3.4-25
Figure 3.4.3.2-2	Canister Lift Model Stresses Intensity Contours (psi) 3.4-26
Figure 3.4.3.3-1	Finite Element Model for Transfer Cask Trunnion and Shells 3.4-40
Figure 3.4.3.3-2	Node Locations for Transfer Cask Outer Shell Adjacent to Trunnion 3.4-41
Figure 3.4.3.3-3	Node Locations for Transfer Cask Inner Shell Adjacent to Trunnion 3.4-42
Figure 3.4.3.3-4	Stress Contours for Transfer Cask Outer Shell 3.4-43
Figure 3.4.3.3-5	Stress Contours for Transfer Cask Inner Shell 3.4-44
Figure 3.4.4.1-1	Canister ANSYS Finite Element Model 3.4-62
Figure 3.4.4.1-2	Weld Regions of Canister ANSYS Finite Element Model at
	Structural and Shield Lids 3.4-63
Figure 3.4.4.1-3	Bottom Plate of the Canister ANSYS Finite Element Model 3.4-64
Figure 3.4.4.1-4	Locations for Section Stresses in the Canister ANSYS
	Finite Element Model
Figure 3.4.4.1-5	Fuel Basket Support Disk ANSYS Finite Element Model 3.4-66
Figure 3.4.4.1-6	Fuel Basket Top Weldment ANSYS Finite Element Model 3.4-67
Figure 3.4.4.1-7	Fuel Basket Bottom Weldment ANSYS Finite Element Model 3.4-68
Figure 3.4.4.1-8	Fuel Tube Configuration
Figure 3.4.4.2-1	Concrete Cask Axisymmetric Thermal Stress Model

# List Of Tables

Table 3.2-1	NAC-MPC System Weights and Centers of Gravity	
Table 3.3-1	Mechanical Properties of SA 240, A 479 Type 304 Stainless Steel 3.3-2	
Table 3.3-2	Mechanical Properties of SA 336, Type 304 Stainless Steel 3.3-3	
Table 3.3-3	Mechanical Properties of SA 240, Type 304L Stainless Steel 3.3-4	
Table 3.3-4	Mechanical Properties of SA 705, SA 693 and SA 564, Type 630,	
	H1150, 17-4 PH Stainless Steel	
Table 3.3-5	Mechanical Properties of A36 Carbon Steel	
Table 3.3-6	Mechanical Properties of A615, GR 60, Reinforcing Steel	
Table 3.3-7	Mechanical Properties of A500 Carbon Steel	
Table 3.3-8	Mechanical Properties of A588, Type A, B Low Alloy Steel 3.3-9	
Table 3.3-9	Mechanical Properties of 6061-T6 Aluminum Alloy	
Table 3.3-10	Mechanical Properties of Concrete	
Table 3.3-11	Mechanical Properties of NS-4-FR 3.3-12	
Table 3.3-12	Mechanical Properties of SA193, Grade B6,	
	High Alloy Steel Bolting Material 3.3-13	
Table 3.3-13	Mechanical Properties of SA-350/A-350, Grade LF 2,	
	Class 1 Low Alloy Steel 3.3-14	
Table 3.4.3.3-1	Top 30 Stresses for Transfer Cask Outer Shell	
Table 3.4.3.3-2	Top 30 Stresses for Transfer Cask Inner Shell 3.4-46	
Table 3.4.4.1-1	Summary of Maximum Canister Thermal Stresses (ksi)	
Table 3.4.4.1-2	Summary of Maximum Canister Dead Load Primary Membrane (Pm)	
	Stresses (ksi)	
Table 3.4.4.1-3	Summary of Maximum Canister Dead Load Primary Membrane Plus	
	Primary Bending (P _m + P _b ) Stresses (ksi)	
Table 3.4.4.1-4	Summary of Maximum Canister Internal Pressure Load Primary	
	Membrane (P _m ) Stresses (ksi)	
Table 3.4.4.1-5	Summary of Maximum Canister Internal Pressure Load Primary	
	Membrane Plus Primary Bending (P _m + P _b ) Stresses (ksi)	
Table 3.4.4.1-6	Summary of Maximum Canister Dead Load + Handling Load	
	Primary Membrane (P _m ) Stresses (ksi)	
Table 3.4.4.1-7	Summary of Maximum Canister Dood Lood + Handling Lood Primary	
	Summary of Maximum Canister Dead Load + Handing Load Primary	
NAC-MPC FSA Docket No. 72-1	R April 20 025 Revisio	000 on 0
--------------------------------	-----------------------------------------------------------------------------------------------	-------------
Table 3.4.4.1-8	Summary of Maximum Canister Combined Load Primary	
	Membrane (P _m ) Stresses (ksi)	-77
Table 3.4.4.1-9	Summary of Maximum Canister Combined Load Primary	
	Membrane Plus Primary Bending (P _m + P _b ) Stresses (ksi)	-78
Table 3.4.4.1-10	Summary of Maximum Canister Combined Load Primary Membrane	
	Plus Primary Bending Plus Secondary (P _m + P _b + Q) Stresses (ksi) 3.4-	.79
Table 3.4.4.1-11	Summary of Maximum Stresses for the Fuel Basket Weldments	
	and Support Disks 3.4-	·80
Table 3.4.4.2-1	Stress Summary for Concrete Cask Load Combinations 3.4-	89
Table 3.4.4.2-2	Maximum Concrete Stress and Reinforcing Bar Forces	.90

THIS PAGE INTENTIONALLY LEFT BLANK

#### 3.0 STRUCTURAL EVALUATION

This section describes the design and analyses of the principal structural components of the NAC-MPC System under normal operating conditions. It demonstrates that the NAC-MPC System meets the requirements to assure confinement of contents, criticality control, radiological shielding, and contents retrievability as required by 10CFR72 for the design basis operating conditions. Off-normal and accident conditions are evaluated in Chapter 11.

#### 3.1 <u>Structural Design</u>

#### 3.1.1 Discussion

The NAC-MPC System consists of three major components: 1) the vertical concrete cask (storage cask); 2) the transportable storage canister (canister); and 3) the transfer cask. These components are shown in Figure 3.1-1. The principal structural member of the vertical concrete cask is the reinforced concrete shell. The principal structural members of the canister are the shell, structural lid, bottom plate, the welds joining these components, and the basket assembly. The primary structural components of the transfer cask are its trunnions, inner and outer steel walls and the bottom doors and their support rails. All of the components are shown on the license drawings provided in Section 1.5.

The concrete cask is a reinforced concrete cylinder with an outside diameter of 128 inches and an overall height of 160 inches. The internal cavity of the concrete cask is formed by a 3.5-inch thick cylindrical carbon steel liner having an inside diameter of 79 inches. The liner is a stay-inplace form. Its thickness is primarily determined by shielding requirements, but is related to the need to establish a practical limit to the diameter of the concrete shell. The concrete is Type II Portland Cement, having a nominal density of 140 lbs/ft³, and a nominal compressive strength of 4000 psi. The inner and outer reinforcing bar assemblies are formed by vertical hook bars and horizontal hoop bars. The air flow path is formed by channels at the bottom that provide the entrance for cooling air, the air inlet ducts that admit the air to the storage cask liner interior surface, and the air outlet ducts.). A 5-inch thick carbon steel shield plug, that encloses a 1-inch thick layer of NS-4-FR neutron shield material, is installed in the concrete cask cavity above the canister. The plug is supported by a support ring welded to the liner. A 1.5-inch thick carbon steel lid provides a cover to protect the canister from adverse environmental conditions and

postulated tornado driven missiles. The shield plug and lid provide shielding to reduce the skyshine radiation. The lid is bolted in place.

The canister consists of a cylindrical shell assembly closed at its top end by an inner shield lid and an outer structural lid. The canister contains a basket assembly that holds the spent fuel. The canister shell is 122.5 inches long and is fabricated from 304L stainless steel plate. The canister shield lid is 5-inch thick Type 304 stainless steel, and the structural lid is 3.0-inch thick Type 304L stainless steel. Both lids are welded to the canister shell to close the canister. The shield lid is supported from below, prior to welding, by a support ring. The structural lid is supported, prior to welding, by the shield lid. The bottom of the canister is a 1-inch thick, Type 304L stainless steel plate that is welded to the canister shell.

The basket assembly is designed to hold up to 36 Yankee Class fuel assemblies. It incorporates 22 Type 17-4 PH stainless steel support disks and 14 Type 6061-T6 aluminum alloy heat transfer disks. The remaining components of the basket assembly are Type 304 stainless steel. These disks, together with the top and bottom weldments, are positioned by tie rods (with spacers and washers) that extend the length of the basket and clamp the components together. The support disks provide heat removal and support the fuel tubes that pass through the disks. The heat transfer disks provide the heat removal capability, but are not considered to be structural components. The fuel tubes have an inside square dimension of 7.8 inches and a composite wall thickness of 0.14 inches. All walls of each fuel tube contain a sheet of BORAL neutron poison material. No structural credit is taken for the BORAL sheet.

A transportable storage canister containing spent fuel may also contain one or more Reconfigured Fuel Assemblies. The Reconfigured Fuel Assembly is designed to contain Yankee Class spent fuel rods, or portions thereof, which are classified as failed, and to maintain the geometric positions of the rods. The assembly has a capacity of 64 full length spent fuel rods in an eight by eight array of tubes. As shown in Figure 1.2-5, the reconfigured fuel assembly consists of a shell (square tube with end fittings), a basket assembly, and 64 fuel tubes. All of the materials are stainless steel.

The Yankee Class Reconfigured Fuel Assembly is designed to contain failed fuel rods, in fuel tubes, during all storage and transport conditions. The Reconfigured Fuel Assembly is designed to the requirements of ASME Boiler and Pressure Vessel Code, Section III, Article NG-3000 and

NUREG/CR-6322, "Buckling Analysis of Spent Fuel Baskets" and using the additional guidance contained in ASME Section III, Article NF-3000 and in ASME Section III, Appendix F. The structural evaluation of the Reconfigured Fuel Assembly is presented in Section 11.4.

The external dimensions of the Reconfigured Fuel Assembly are the same as those of other Yankee Class fuel assemblies. The weight of a loaded reconfigured fuel assembly (approximately 550 pounds) is less than the weight of other Yankee Class fuel assemblies (approximately 850 pounds). The maximum temperature of the Reconfigured Fuel Assembly components is determined by the thermal analyses presented in Section 4.4.

The Reconfigured Fuel Assembly has been evaluated and is capable of withstanding, within code allowable limits (Service Level A/B), a postulated end impact resulting in a deceleration of 20 g. It is also, when located in a fuel slot in the transportable storage container, capable of withstanding, within code allowable limits (Service Level A/B), a postulated side impact resulting in a deceleration of 20 g. This analysis bounds the design conditions of the Reconfigured Fuel Assembly for normal conditions of storage.

The Reconfigured Fuel Assembly has also been evaluated for accident conditions and is capable of withstanding, within code allowable limits (Service Level D), a postulated end impact resulting in a deceleration of 57 g. It is also, when located in a fuel slot in the transportable storage container, capable of withstanding, within code allowable limits (Service Level D), a postulated side impact resulting in a deceleration of 55 g. This analysis bounds the design conditions of the Reconfigured Fuel Assembly for accident conditions of storage.

Therefore, the structural evaluations of the NAC-MPC System containing other Yankee Class fuel assemblies (Chapters 3.0 and 11.0) bound those of the NAC-MPC System containing one or more Yankee Class Reconfigured Fuel Assemblies.

The following components are evaluated in this chapter:

- Canister lifting devices
- Canister shell, bottom, and structural lid
- Canister shield lid support ring
- Basket assembly
- Transfer cask trunnions, shells, retaining ring, bottom doors, and support rails
- Vertical concrete cask body
- Concrete cask steel components (reinforcement, liner, lid, bottom plate, bottom, etc.)

All other NAC-MPC system components shown on the drawings presented in Section 1.5 are either nonstructural or not classified as important to safety. They are appropriately included as loads in the evaluation of the components listed above.

The structural evaluations demonstrate that all of the NAC-MPC components meet their structural design criteria and are capable of safely storing the design basis spent fuel.

# 3.1.2 Design Criteria

The NAC-MPC structural design criteria is specified in Section 2.2. The load combinations of normal, off-normal, and accident loadings have been evaluated in accordance with ANSI 57.9 and ACI-349 for the concrete cask (see Table 2.2-2), and in accordance with the 1995 edition of the ASME Code, Section III, Division I, Subsection NB for Class 1 components for the canister (see Table 2.2-3). The basket is evaluated in accordance with ASME Code, Section III, Subsection NG, and NUREG-6322. The transfer cask and the lifting yoke are lifting devices that are designed to NUREG-0612 and ANSI N14.6.

NAC-MPC FSAR	April 2000
Docket No. 72-1025	Revision 0





THIS PAGE INTENTIONALLY LEFT BLANK

-----

# 3.2 Weights and Centers of Gravity

The component weights and centers of gravity for the NAC-MPC system are summarized in Table 3.2-1.

# Table 3.2-1 NAC-MPC System Weights and Centers of Gravity

Item Description	Calculated Weight (lbs)	Center of Gravity (inches above bottom of item)
Concrete Cask Lid	2,838	160.7
Concrete Cask Shield Plug	5,490	153.3
Canister Structural Lid	3,230	121.0
Canister Shield Lid	5,390	116.8
Transfer Adapter Plate	12,659	N/A
Canister (empty, without lid)	15,510	49.7
Canister (loaded with fuel with water and shield lid)	62,270	61.2
Canister (loaded with fuel with lid)	54,730	65.0
Concrete Cask (empty, with shield plug, and without lid)	148,526	80.5
Concrete Cask and Canister (loaded with fuel with lids)	206,094	83.2
Transfer Cask (empty)	80,743	57.0
Transfer Cask and Canister (empty, without lids)	96,253	58.0
Transfer Cask and Canister (loaded with fuel, with water and shield lid)	143,013	63
Transfer Cask and Canister (loaded with fuel, dry with lids)	135,473	64.1
Water in Canister (36 assemblies)	10,618	N/A
Fuel	30,600	55.7

3.3

## Mechanical Properties of Materials

The mechanical properties of steels used in the fabrication of the NAC-MPC components are presented in Tables 3.3-1 through 3.3-8 and in Tables 3.3-12 and 3.3-13. The primary steels are Type 304 and Type 304L stainless steel, selected because of their high strength, ductility, resistance to corrosion and brittle fracture, and metallurgical stability for long-term storage. The mechanical properties for the 6061-T6 aluminum heat transfer disks in the fuel basket are provided in Table 3.3-9. The mechanical properties of the concrete are presented in Table 3.3-10. The mechanical properties of the neutron shield material, NS-4-FR, are presented in Table 3.3-11.

		Temperature (°F)									
Property (units)	-40	-20	+70	+200	+300	+400	+500	+750			
Ultimate Strength ¹ (ksi)	75.0	75.0	75.0	71.0	66.0	64.4	63.5	63.1			
Yield Strength ² (ksi)	30.0	30.0	30.0	25.0	22.5	20.7	19.4	17.3			
Design Stress Intensity ³ (ksi)	20.0	20.0	20.0	20.0	20.0	18.7	17.5	15.6			
Modulus of Elasticity ⁴ (ksi)	28.7E+3	28.7E+3	28.3E+3	27.6E+3	27.0E+3	26.5E+3	25.8E+3	24.4E+3			
Alternating Stress ⁵ @ 10 cycles (ksi)	718.0	718.0	708.0	690.5	675.5	663.0	645.5	610.4			
Alternating Stress ⁵ @ 10 ⁶ cycles (ksi)	28.7	28.7	28.3	27.6	27.0	26.5	25.8	24.4			
Coefficient of Thermal Expansion ⁶ (in/in/°F)	8.13E-6	8.19E-6	8.46E-6	8.79E-6	9.00E-6	9.19E-6	9.37E-6	9.76E-6			
Poisson's Ratio ⁷	0.275										
Density ⁸ (lbm/in ³ )				0.2	88						

# Table 3.3-1 Mechanical Properties of SA 240, A 479 Type 304 Stainless Steel

¹ ASME Code, Section II, Part D, Table U.

- ² ASME Code, Section II, Part D, Table Y-1.
- ³ ASME Code, Section II, Part D, Table 2A.
- ⁴ ASME Code, Section II, Part D, Table TM-1.
- ⁵ ASME Code, Section III, Appendix I, Table I-9.1.
- ⁶ ASME Code, Section II, Part D, Table TE-1.
- ⁷ Hanford, Volume 1, Design Data, Property Code 2110.
- ⁸ Hanford, Volume 1, Design Data, Property Code 3304.

Table 3.3-2 Mecha	inical Properties	of SA 336,	, Type 304 Stainless	Steel
-------------------	-------------------	------------	----------------------	-------

		Temperature (°F)									
Property (units)	-40	-20	+70	+200	+300	+400	+500	+750			
Ultimate Strength ¹ (ksi)	. 70.0	70.0	70.0	66.2	61.5	60.0	59.3	58.9			
Yield Strength ² (ksi)	30.0	30.0	30.0	25.0	22.5	20.7	19.4	17.3			
Design Stress Intensity ³ (ksi)	20.0	20.0	20.0	20.0	20.0	18.7	17.5	15.6			
Modulus of Elasticity ⁴ (ksi)	28.7E+3	28.7E+3	28.3E+3	27.6E+3	27.0E+3	26.5E+3	25.8E+3	24.4E+3			
Alternating Stress ⁵ @ 10 cycles (ksi)	718.0	718.0	708.0	690.5	675.5	663.0	645.5	610.4			
Alternating Stress ⁵ @ 10 ⁶ cycles (ksi)	28.7	28.7	28.3	27.6	27.0	26.5	25.8	24.4			
Coefficient of Thermal Expansion ⁶ (in/in/°F)	8.13E-6	8.19E-6	8.46E-6	8.79E-6	9.00E-6	9.19E-6	9.37E-6	9.76E-6			
Poisson's Ratio ⁷	0.275										
Density ⁸ (lbm/in ³ )				0.2	88						

- ² ASME Code, Section II, Part D, Table Y-1.
- ³ ASME Code, Section II, Part D, Table 2A.
- ⁴ ASME Code, Section II, Part D, Table TM-1.
- ⁵ ASME Code, Section III, Appendix I, Table I-9.1.
- ⁶ ASME Code, Section II, Part D, Table TE-1.
- ⁷ Hanford, Volume 1, Design Data, Property Code 2110.
- ⁸ Hanford, Volume 1, Design Data, Property Code 3304.

¹ ASME Code, Section II, Part D, Table U.

				Tempera	ture (°F)				
Property (units)	-40	-20	+70	+200	+300	+400	+500	+750	
Ultimate Strength ¹ (ksi)	70.0	70.0	70.0	66.2	60.9	58.5	57.8	55.9	
Yield Strength ² (ksi)	25.0	25.0	25.0	21.4	19.2	17.5	16.4	14.7	
Design Stress Intensity ³ (ksi)	16.7	16.7	16.7	16.7	16.7	15.8	14.8	13.3	
Modulus of Elasticity ⁴ (ksi)	28.7E+3	28.7E+3	28.3E+3	27.6E+3	27.0E+3	26.5E+3	25.8E+3	24.4E+3	
Alternating Stress ⁵ @ 10 cycles (ksi)	718.0	718.0	708.0	690.5	675.5	663.0	645.5	610.4	
Alternating Stress ⁵ @ 10 ⁶ cycles (ksi)	28.7	28.7	28.3	27.6	27.0	26.5	25.8	24.4	
Coefficient of Thermal Expansion ⁶ (in/in/°F)	8.13E-6	8.19E-6	8.46E-6	8.79E-6	9.00E-6	9.19E-6	9.37E-6	9.76E-6	
Poisson's Ratio ⁷	0.275								
Density ⁸ (lbm/in ³ )				0.2	88				

# Table 3.3-3Mechanical Properties of SA 240, Type 304L Stainless Steel

¹ ASME Code, Section II, Part D, Table U.

- ² ASME Code, Section II, Part D, Table Y-1.
- ³ ASME Code, Section II, Part D, Table 2A.
- ⁴ ASME Code, Section II, Part D, Table TM-1.
- ⁵ ASME Code, Section III, Appendix I, Table I-9.1.
- ⁶ ASME Code, Section II, Part D, Table TE-1.
- ⁷ Hanford, Volume 1, Design Data, Property Code 2110.
- ⁸ Hanford, Volume 1, Design Data, Property Code 3304.

April 2000 Revision 0

		Temperature (°F)										
Property (units)	-40	-20	+70	+200	+300	+400	+500	+650				
Ultimate Strength ^{1, 2} (ksi)	135.0	135.0	135.0	135.0	135.0	131.4	128.5	125.7 ²				
Yield Strength ³ (ksi)	105.0	105.0	105.0	97.1	93.0	89.5	87.0	83.6				
Design Stress Intensity⁴ (ksi)	45.0	45.0	45.0	45.0	45.0	43.8	42.8	41.9				
Modulus of Elasticity ⁵ (ksi)	28.7E+3	28.7E+3	28.3E+3	27.6E+3	27.0E+3	26.5E+3	25.8E+3	25.1E+3				
Alternating Stress ⁶ @ 10 cycles (ksi)	401.8	401.8	396.2	386.4	378.0	371.0	361.2	341.6				
Alternating Stress ⁶ @ 10 ⁶ cycles (ksi)	19.1	19.1	18.9	18.4	18.0	17.7	17.2	16.3				
Coefficient of Thermal Expansion ⁷ (in/in/°F)	5.88E-6	5.88E-6	5.89E-6	5.90E-6	5.90E-6	5.91E-6	5.91E-6	5.93E-6				
Poisson's Ratio ⁸	0.291											
Density ⁹ (lbm/in ³ )				0.2	.84							

Table 3.3-4	Mechanical Properties of SA 705, SA 693 and SA 564, Type 630,
	H1150, 17-4 PH Stainless Steel

¹ ASME Code, Section II, Part D, Table U.

- ² Tabulated value is calculated by ratioing from the Design Stress Intensity.
- ³ ASME Code, Section II, Part D, Table Y-1.
- ⁴ ASME Code, Section II, Part D, Table 2A.
- ⁵ ASME Code, Section II, Part D, Table TM-1.
- ⁶ ASME Code, Section III, Appendix I, Table I-9.1.
- ⁷ ASME Code, Section II, Part D, Table TE-1.
- ⁸ ARMCO, Table 7.
- ⁹ ARMCO, Table 16.

	Temperature (°F)											
Property (units)	100	200	300	400	500	600	650	700				
S _u (ksi) ¹	58.0	58.0	58.0	58.0	_	_	_	-				
S _y (ksi) ¹	36.0	32.8	31.9	30.8	29.1	26.6	26.1	25.9				
$S_{m}$ (ksi) ²	19.3	19.3	19.3	19.3	19.3	17.7	17.4	17.3				
Modulus of Elasticity ³ (ksi)	29.0E+3	28.8E+3	28.3E+3	27.7E+3	27.3E+3	26.7E+3	26.1E+3	25.5E+3				
$\alpha$ (in/in/ °F) ⁴	5.53E-6	5.89E-6	6.26E-6	6.61E-6	6.91E-6	7.17E-6	7.30E-6	7.41E-6				
Poisson's Ratio ⁵		0.31										
Density ⁶ (lbm/in ³ )				0.2	.84							

# Table 3.3-5 Mechanical Properties of A36 Carbon Steel

¹ Cases of ASME Boiler and Pressure Vessel Code, Case N-71-17, Table 1, 2, 3, 4, 5.

- ² ASME Code, Section II, Part D, Table 2A.
- ³ ASME Code, Section II, Part D, Table TM-1.
- ⁴ ASME Code, Section II, Part D, Group C in Table TE-1.
- ⁵ ASME Code, Section II, Part D, Table NF-1.
- ⁶ Ross.

# Table 3.3-6Mechanical Properties of A615, GR 60, Reinforcing Steel

		Tempera	ture (°F)							
Property (units)	148	328	508	688						
S _u (ksi) ¹	58.0	65.3	69.6	53.7						
$S_y (ksi)^2$		60	0							
Modulus of Easticity ¹ (ksi)		29.88	3E+3							
$\alpha$ (in/in/ °F) ¹		6.1E-6								
Density ¹ (lbm/in ³ )		0.2	84							

¹ Ross.

² Cases of ASME Code, Case N-71-17, Table 1, 2, 3, 4, 5.

		Temperature (°F)									
Property (units)	100	200	300	400	500	600	650	700			
S _u (ksi) ¹	58.0	58.0	58.0	58.0	_	_	_	_			
S _y (ksi) ¹	42.0	38.3	37.2	35.9	33.9	31.0	30.4	30.2			
Modulus of Elasticity ² (ksi)	29.0E+3	28.8E+3	28.3E+3	27.7E+3	27.3E+3	26.7E+3	26.1E+3	25.5E+3			
$\alpha$ (in/in/ °F) ³	5.53E-6	5.89E-6	6.26E-6	6.61E-6	6.91E-6	7.17E-6	7.30E-6	7.41E-6			
Poisson's Ratio ⁴		0.31									
Density ⁵ (lbm/in ³ )				0.2	284						

# Table 3.3-7Mechanical Properties of A500 Carbon Steel

² ASME Code, Section II, Part D, Table TM-1.

¹ Cases of ASME Boiler and Pressure Vessel Code, Case N-71-17, Table 1, 2, 3, 4, 5.

³ ASME Code, Section II, Part D, Group C in Table TE-1.

⁴ ASME Code, Section II, Part D, Table NF-1.

⁵ Ross.

		Temperature (°F)							
Property (units)	100	200	300	400	500	600	650	700	
$S_u (ksi)^1$	70.0	70.0	70.0	70.0	70.0	70.0	70.0	70.0	
S _y (ksi) ¹	50.0	47.5	45.6	43.0	41.8	39.9	38.9	37.9	
S _m (ksi) ¹	23.3	23.3	23.3	23.3	23.3	23.3	23.3	23.3	
Modulus of Elasticity ² (ksi)	29.0E+3	28.8E+3	28.3E+3	27.7E+3	27.3E+3	26.7E+3	26.1E+3	25.5E+3	
$\alpha$ (in/in/ °F) ³	5.53E-6	5.89E-6	6.26E-6	6.61E-6	6.91E-6	7.17E-6	7.30E-6	7.41E-6	
Poisson's Ratio ⁴	0.31								
Density ⁵ (lbm/in ³ )		0.284							

# Table 3.3-8Mechanical Properties of A588, Type A and B Low Alloy Steel

¹ Cases of ASME Boiler and Pressure Vessel Code, Case N-71-17, Table 1, 2, 3, 4, 5.

² ASME Code, Section II, Part D, Table TM-1.

³ ASME Code, Section II, Part D, Group C in Table TE-1.

⁴ ASME Code, Section II, Part D, Table NF-1.

⁵ Ross.

<u></u>		Temperature (°F)							
Property (units)	70	100	200	300	400	500	600	700	
Ultimate Strength ^{1, 2} (ksi)	42.0	40.7	38.2	31.5	17.2	6.7	3.4	2.1	
Yield Strength (ksi) ^{1, 2}	35.0	33.9	32.2	26.9	14.0	5.3	2.5	1.4	
Design Stress ¹ (ksi)	10.5	10.5	10.5	8.4	4.4	N/A	N/A_	N/A	
Modulus of Elasticity ³ (ksi)	10.0E+3	9.9E+3	9.6E+3	9.2E+3	8.7E+3	8.1E+3	7.0E+3 ⁵	N/A	
Coefficient of Thermal Expansion ⁴ (in/in/°F)		12.6E-6	12.9E-6	13.22E-6	13.52E-6	13.7E-6 ⁶	14.3E-6 ⁶	N/A	
Poisson's Ratio ⁷	0.33								
Density ⁸ (lbm/in ³ )				0.0	98	_			

# Table 3.3-9Mechanical Properties of 6061-T6 Aluminum Alloy

¹ ASME Code, Section II, Part D, Table 1-B.

² Strength at elevated temperatures calculated using the following relationships from MIL-HDBK-5G, Figures 3.6.2.2.1(a) and 3.6.2.2.1(b):

 $S_{u \text{ @temp}} = (\% \text{Value}) (S_{u \text{ @70}})$ 

 $S_{y @temp} = (\%Value) (S_{y @70})$ 

#### % Value @ Room Temperature

Temp°F	<u>100</u>	<u>200</u>	<u>300</u>	<u>400</u>	<u>500</u>	<u>600</u>	<u>700</u>
$S_u$	97	91	75	41	16	8	5
Sy	97	92	77	40	15	7	4

- ³ ASME Code, Section II, Part D, Table TM-2.
- ⁴ ASME Code, Section II, Part D, Table TE-2.
- ⁵ MIL-HDBK-5G, Figure 3.6.2.2.4.
- ⁶ MIL-HDBK-5G, Figure 3.6.2.0.
- ⁷ ASME Code, Section II, Part D, Table NF-1.
- ⁸ ASME Code, Section II, Part D, Table NF-2.

# Table 3.3-10Mechanical Properties of Concrete

	Temperature (°F)						
Property (units)	70	100	200	300	400	500	
Density ¹ (lbs/ft ³ )	140	140	140	140	140	140	
Compressive Strength ¹ (psi)	4,000	4,000	4,000	3,800	3,600	3,400	
Coefficient of Thermal Expansion ¹ (in/in/°F)	5.5x10 ⁻⁶						
Modulus of Elasticity ¹ (ksi)		3.64x10 ⁶	3.38x10 ⁶	3.09x10 ⁶	3.73x10 ⁶	3.43x10 ⁶	

¹ Fintel.

# Table 3.3-11 Mechanical Properties of NS-4-FR

Property (units)	86	158	212	302			
Compressive Modulus of Elasticity ¹ (ksi)	561						
Coefficient of Thermal Expansion ¹ (in/in/°F)	2.22E-5	4.72E-5	5.88E-5	5.74E-5			
Density ¹ (lbm/in ³ )	0.0607						

¹ GESC Product Data.

	Temperature (°F)							
Property (units)	-40	-20	70	200	300	400	500	750
S _u (ksi)	110.0 ¹	110.0 ¹	110.0 ¹	104.9 ²	101.5 ²	98.3 ²	95.6 ²	89.4 ²
S _y (ksi)	85.0 ¹	$85.0^{1}$	85.0 ¹	81.2 ²	78.4 ²	76.0 ²	73.9 ²	69.1 ²
$S_m (ksi)^{2.3}$	28.3	28.3	28.3	27.0	26.1	25.3	24.6	23.0 ²
Modulus of Elasticity ⁴ (ksi)	30.1E+3	30.1E+3	29.2E+3	28.5E+3	27.9E+3	27.3E+3	26.7E+3	25.2E+3
Alternating Stress ⁵ @ 10 cycles (ksi)	1,104.4	1,100.0	1,085.0	1,058.0	1,035.0	1,015.0	989.0	935.3
Alternating Stress ⁵ @ 10 ⁶ cycles (ksi)	13.0	12.9	12.7	12.4	12.2	11.9	11.6	11.0
α (in/in/ °F) ⁶	5.73E-6	5.76E-6	5.92E-6	6.15E-6	6.30E-6	6.40E-6	6.48E-6	6.64E-6
Density ⁷ (lbm/in ³ )		0.283						

Table 3.3-12 N	Mechanical Propertie	s of SA193,	Grade B6,	High Alloy	Steel Bolting Material
----------------	----------------------	-------------	-----------	------------	------------------------

¹ ASME Code, Section III, Appendix 1, Table I-1.3.

² Tabulated values are calculated by ratioing from the Design Stress Intensity.

³ ASME Code, Section II, Part D, Table 4.

- ⁴ ASME Code, Section II, Part D, Table TM-1.
- ⁵ ASME Code, Section III, Appendix 1, Table I-9.1.
- ⁶ ASME Code, Section II, Part D, Table TE-1.

⁷ Baumeister and Marks, "Standard Handbook for Mechanical Engineers," 1966.

-----

I <del></del>	T					
Property			V	alue		
Temperature (°F)	70	200	300	400	500	700
Ultimate strength, ¹	70.0	70.0	70.0	70.0	70.0	70.0
S _u , (ksi)						
Yield strength, ¹	36.0	32.8	31.9	30.8	29.1	25.9
S _y (ksi)						
Design Stress Intensity, ¹	23.3	21.9	21.3	20.6	19.4	17.3
S _m (ksi)						
Modulus of Elasticity, ¹	29.2	28.5	28.0	27.4	27.0	25.3
E, (× 10 ³ ksi)						
Coefficient of Thermal		5.89	6.26	6.61	6.91	7.41
Expansion, ¹						
$\alpha$ (× 10 ⁻⁶ in/in/°F)						
Alternating Stress ²	12.5	12.2	11.9	11.7	11.5	10.8
at 10 ⁶ cycles (ksi)						
Alternating Stress ²	580.0	566.0	556.1	544.2	536.3	502.5
at 10 cycles (ksi)						
Poisson's Ratio ¹			0	0.31		
Density ¹			0.279	lbm/in ³		

Table 3.3-13	Mechanical Properties	of SA-350/A-350,	Grade LF 2, C	Class 1 Low Alloy Steel
--------------	-----------------------	------------------	---------------	-------------------------

1 ASME Code, Section II, Part D.

2 ASME Code, Appendix I

#### 3.4 General Standards for Casks

#### 3.4.1 Chemical and Galvanic Reactions

The materials used in the fabrication and operation of the NAC-MPC system have been evaluated to determine whether chemical, galvanic, or other reactions among the materials, contents, and environments can occur. All phases of operation—loading, unloading, handling, and storage—have been considered for the environments that may be encountered under normal, off-normal, or accident conditions. Based on the evaluation, there is one potential reaction that could adversely affect the overall integrity of the storage cask, the fuel basket, the transportable storage canister, or the structural integrity and retrievability of the fuel from the canister. That potential reaction, between aluminum and spent fuel pool water, which may produce hydrogen is mitigated by the specific canister loading procedures presented in Section 8.1.1.

#### 3.4.1.1 Component Operating Environment

Most of the component materials of the NAC-MPC are exposed to two typical operating environments: 1) an open canister containing pool water or borated water with a pH of 4.5 and spent fuel or other radioactive material; or 2) a sealed canister containing helium, but with the canister in environs that include air, rain water/snow/ice, and marine (salty) water/air. The spent fuel assemblies consist of zircaloy or stainless steel clad fuel and other fuel assembly components of stainless steel.

Each category of canister component materials is evaluated for potential reactions in each of the operating environments to which those materials are exposed. These environments may occur during fuel loading or unloading, handling or storage, and include normal, off-normal, and accident conditions.

One of the operating environments to which the canister internal component materials are exposed does not provide the conditions necessary for a reaction (corrosion); i.e., both moisture and oxygen must be present for corrosion to occur. This long-term environment is the sealed canister, backfilled with helium. Helium displaces the oxygen in the canister effectively precluding corrosion. Galvanic corrosion (i.e., between dissimilar metals that are in contact) could occur, but only if there is water present at the point of contact and the metals are in

3.4-1

electrical contact with each other (i.e., mechanically held together). NAC's operating procedures provide two helium backfill cycles in series separated by a vacuum-drying cycle for the canister during the preparation of the canister for storage. Therefore, the canister cavity is effectively dry and galvanic corrosion is precluded.

#### 3.4.1.2 <u>Component Material Categories</u>

The component materials evaluated are categorized based on similarity of physical and chemical properties and/or on similarity of component functions. The categories of materials that are considered are stainless/nickel alloy steels, nonferrous metals, and criticality control materials. These categories are evaluated based on the environment to which they could be exposed during operation or use of the canister.

The canister component materials are not reactive among themselves, with the canister's contents, nor with the canister's operating environments, except aluminum, during any phase of normal, off-normal, or accident condition loading, unloading, handling, or storage operations. Therefore, only the potential aluminum reaction with spent fuel pool water is evaluated.

#### 3.4.1.2.1 <u>Stainless Steels</u>

No reaction of the canister component stainless steel is expected in any environment except for the marine environment, where chloride-containing salt spray might initiate pitting of the steels if the chlorides are allowed to concentrate and stay wet for extended periods of time (weeks). Only the external canister surface could be so exposed. The corrosion rate will, however, be so low that no detectable corrosion products or gases will be generated. The NAC-MPC has smooth external surfaces to minimize the collection of such materials as salts.

There is a significant electrochemical potential difference between austenitic (300 series) stainless steel and aluminum. If aluminum is in electrical contact with the austenitic stainless steel, the aluminum could be expected to exhibit corrosion driven by electrochemical EMF when immersed in water. Pressurized water reactor (PWR) pool water does provide a conductive potential. The only aluminum components that will be in contact with stainless steel and exposed to the pool water are the heat transfer disks in the fuel basket. Since the fuel basket is not welded or bolted to the canisters, poor, if any, electrical contact with the stainless steels is

present. Therefore, in the absence of electrical contact between the aluminum and the stainless steel, galvanically driven corrosion does not occur.

No coatings are applied to the stainless steels.

Based on the foregoing discussion, there are no potential reactions associated with the stainless steel canister components.

#### 3.4.1.2.2 <u>Nonferrous Metals</u>

Heat transfer disks fabricated from 6061-T6 aluminum alloy are used in the NAC-MPC fuel basket to augment heat transfer from the spent fuel through the basket structure to the canister exterior. Vendor and Nuclear Regulatory Commission safety evaluations of the NUHOMS Dry Spent Fuel Storage System (Docket No. 72-1004) have concluded that combustible gases, primarily hydrogen, may be produced by a chemical reaction and/or radiolysis when aluminum or aluminum flame-sprayed components are immersed in spent fuel pool water. The evaluations further concluded that it is possible, at higher temperatures (above 150°F to 160 °F), for the aluminum/water reaction to produce a hydrogen concentration in the cask or canister that approaches or exceeds the Lower Flammability Limit (LFL) for hydrogen of 4 percent. The NRC Inspection Reports No. 50-266/96005 and 50-301/96005 dated July 01, 1996, for the Point Beach Nuclear Plant concluded that hydrogen generation by radiolysis was insignificant relative to other sources.

Thus, it is reasonable to conclude that small amounts of combustible gases, primarily hydrogen, may be produced during NAC-MPC canister loading or unloading operations as a result of a chemical reaction between the 6061-T6 aluminum heat transfer disks in the fuel basket and the spent fuel pool water. The generation of combustible gases stops when the water is removed from the cask or canister and the aluminum surfaces are dry.

A galvanic reaction may occur at the contact surfaces between the aluminum disks and the stainless steel tie rods and spacers in the presence of an electrolyte, like the pool water. The galvanic reaction ceases when the electrolyte is removed. Each metal has some tendency to ionize, or release electrons. A voltage, or electromotive force (emf), associated with this release of electrons is generated between two dissimilar metals in an electrolytic solution. The emf between aluminum and stainless steel is small and the amount of corrosion is directly proportional to the emf. Loading operations generally take less than 24 hours, a large portion of which has the canister immersed in and open to the pool water after which the electrolyte (water)

3.4-3

is drained and the cask or canister is dried and back-filled with helium, effectively halting any galvanic reaction.

The potential chemical or galvanic reactions do not have a significant detrimental effect on the ability of the aluminum heat transfer disks to perform their function for all normal and accident conditions associated with dry storage.

#### Loading Operations

After the canister is removed from the pool and during canister closure operations, an air space is created inside the canister beneath the shield lid by the drain-down of 50 gallons of water so that the shield-lid-to-canister-shell can be performed. The resulting air space is approximately 70 inches in diameter and 3 inches deep. Although there is some clearance between the inside diameter of the canister shell and the outside diameter of the shield lid, it is possible that gases released from a chemical reaction inside the canister could accumulate beneath the shield lid. A bare aluminum surface oxidizes when exposed to air, reacts chemically in an aqueous solution and may react galvanically when in contact with stainless steel in the presence of an aqueous solution.

The reaction of aluminum in water, which results in hydrogen generation, proceeds as:

 $2 \text{ Al} + 3 \text{ H}_20 \Rightarrow \text{Al}_2\text{O}_3 + 3 \text{ H}_2$ 

The aluminum oxide  $(Al_2O_3)$  produces the dull, light gray film that is present on the surface of bare aluminum when it reacts with the oxygen in air or water. The formation of the thin oxide film is a self limiting reaction as the film isolates the aluminum metal from the oxygen source acting as a barrier to further oxidation. The oxide film is stable in pH neutral (passive) solutions, but is soluble in borated PWR spent fuel pool water. The oxide film dissolves at a rate dependent upon the pH of the water, the exposure time of the aluminum in the water, and the temperatures of the aluminum and water.

PWR spent fuel pool water is a boric acid and demineralized water solution. The pH, water chemistry, and water temperature vary from pool to pool. Since the reaction rate is largely dependent upon these variables, it may vary considerably from pool to pool. Thus, the generation rate of combustible gas (hydrogen) that could be considered representative of spent fuel pools in general is very difficult to accurately calculate.

NAC-MPC FSAR	April 2000
Docket No. 72-1025	Revision 0

To ensure safe loading and/or unloading of the NAC-MPC transportable storage canister, the loading and unloading procedures defined in Chapter 8 provide for the monitoring of hydrogen gas before and during the welding operations joining the shield lid to the canister shell and joining the vent and drain port coverplates to the shield lid. The monitoring system shall be capable of detecting hydrogen at 60% of the lower flammability limit for hydrogen (i.e.,  $0.6 \times 4.0 = 2.4\%$ ). The hydrogen detector shall be mounted so as to detect hydrogen prior to initiation of the weld, and continuously at a point ahead of the weld head. Detection of hydrogen in a concentration exceeding 2.4% shall be cause for the welding operation to stop. If hydrogen gas is detected at concentrations above 2.4% at any time, the hydrogen gas shall be removed by flushing ambient air into the region below the shield lid or port coverplate. To remove hydrogen from below the shield lid, the vacuum pump is attached to the vent port and operated for a sufficient period of time to remove at least five times the air volume of the space below the lid by drawing ambient air through the gap between the shield lid and the canister shell, thus removing or diluting any combustible gas concentrations.

The vacuum pump shall exhaust to a system or area where hydrogen flammability is not an issue. If hydrogen gas is detected at the port coverplates, the coverplate is removed and service air is used to flush combustible gases from the port. Once the root pass weld is completed, there is no further likelihood of a combustible gas burn because the ignition source is isolated from the combustible gas. Once welding of the shield lid has been completed, the canister is drained, vacuum dried and back-filled with helium.

No hydrogen is expected to be detected prior to, or during, the welding operations. The vent port in the shield lid remains open from the time that the loaded canister is removed from the spent fuel pool until the time that the vent port coverplate is ready to be welded to the shield lid. Since the postulated combustible gases are very light, the open vent port provides an escape path for any gases that are generated prior to the time that the canister is vacuum dried. Once the canister is dry, no combustible gases form within the canister. The mating surfaces of the support ring and inner lid are machined to provide a good level fitup, but are not machined to provide a metal to metal seal. Consequently, additional exit paths for the combustible gases exist at the circumference of the shield lid.

#### **Unloading Operations**

It is not expected that the canister will contain a measurable quantity of combustible gases during the time period of storage. The canister is vacuum dried and backfilled with helium immediately

prior to being welded closed. There are only minor mechanisms by which hydrogen is generated after the canister is dried and sealed.

As shown in Section 8.3, the principal steps in opening the canister are the removal of the structural lid, the removal of the vent and drain port coverplates, and the removal of the shield lid. These steps are performed by cutting or grinding. The design of the canister precludes monitoring for the presence of combustible gases prior to the removal of the structural lid and the vent and drain port coverplates. Following removal of the vent port coverplate, a vent line is connected to the vent port quick disconnect. The vent line incorporates a hydrogen gas detector, which is capable of detecting hydrogen at a concentration of 2.4% (60% of its lower flammability limit of 4%). The pressurized gases (expected to be greater than 96% helium) in the canister are expected to carry combustible gases out of the vent port. If the exiting gases in the vent line contain no hydrogen at concentrations above 2.4%, the drain port coverplate weld is cut and the coverplate removed. If levels of hydrogen gas above 2.4% concentration are detected in the vent line, then the vacuum system is used to remove all residual gas prior to removal of the drain port coverplate. During the removal of the drain port coverplate, the hydrogen gas detector is attached to the vent port to ensure that the hydrogen gas concentrations remains below 2.4%. Following removal of the drain port coverplate, the canister is filled with water using the vent and drain ports. Prior to cutting the shield lid weld, 50 gallons of water are removed from the canister to permit the removal of the shield lid. Monitoring for hydrogen would then proceed as described for the loading operations.

## 3.4.1.2.3 Criticality Control Material

The criticality control material is a sheet consisting of boron carbide mixed in an aluminum alloy. This material is effectively a sheet of aluminum that is in contact with the stainless steel fuel tubes and is completely enclosed by a welded stainless steel cover. This "aluminum" is protected by an oxide layer that forms shortly after fabrication and precludes further oxidation of the aluminum or interaction with the stainless steel. Consequently, there are no potential reactions associated with the aluminum-based criticality control material.

# 3.4.1.2.4 <u>Coatings</u>

The exposed carbon steel surfaces of the transfer cask are coated with Keeler & Long E-Series epoxy enamel. Exposed carbon steel in the vertical concrete cask and transfer cask adapter plate may be coated with either E-Series epoxy enamel or Ameron PSX 738 Siloxane coating. E-Series epoxy enamel is approved for use in Coating Service Level 1 areas of nuclear power plants, which include the spent fuel pool and reactor containment areas. These are chemically

NAC-MPC FSAR	April 2000
Docket No. 72-1025	Revision 0

resistant, shop-applied coatings that are designed for elevated service temperatures. Neither of the coatings contains Zinc, nor do they offgas when exposed to borated water. The coatings also serve to chemically isolate the carbon steel from stainless steel and aluminum components used in the NAC-MPC. The technical specifications for Keeler & Long E-Series epoxy enamel and for Ameron PSX 738 Siloxane coating are provided in Sections 3.8.1 and 3.8.2, respectively.

#### 3.4.1.3 <u>General Effects of Identified Reactions</u>

One potential chemical, galvanic, or other reaction has been identified for the NAC-MPC system. A condition, such as the generation of flammable or explosive quantities of combustible gases does not result in an adverse event during any phase of canister operations for normal, off-normal, or accident conditions because specific detection procedures and responses are defined in Section 8.1.1.

#### 3.4.1.4 Adequacy of the Canister Operating Procedures

Based on this evaluation, which resulted in one identified reaction, it is concluded that the NAC-MPC operating controls and procedures presented in Chapter 8.0 are adequate to minimize the occurrence of hazardous conditions.

#### 3.4.1.5 Effects of Reaction Products

One potential chemical, galvanic, or other reaction has been identified for the NAC-MPC. The effects of that reaction are mitigated by the operating procedures described in Section 8.1.1. The overall integrity of the canister and the structural integrity and retrievability of the spent fuel are not adversely affected for the design basis life of the canister. Based on the evaluation, there will be no change in the canister or fuel cladding thermal properties, there will be no binding of mechanical surfaces, no change in basket clearances, and no degradation of any safety components either directly or indirectly.

#### 3.4.2 <u>Positive Closure</u>

The NAC-MPC employs a positive closure system that is composed of multipass welds to join the canister shield lid to the shell, and to join the canister structural lid to the shell. Port covers that are welded to the shield lid (see Figure 3.4.2-1) close the penetrations to the canister cavity through the shield lid. The welds employed for closing the NAC-MPC canister preclude inadvertent opening of the canister. A bolted lid closes the top of the concrete storage cask. The lid weighs approximately 3,000 pounds. The weight of the lid, its inaccessibility and the presence of the bolts effectively preclude inadvertent opening of the lid.

# Figure 3.4.2-1 NAC-MPC Welded Closure System



3.4-8

## 3.4.3 Lifting Devices

To provide more efficient handling of the components of the NAC-MPC system, different methods of lifting, i.e., trunnions, hoist rings, and jacks and air pads, have been designed for each of the components—the transfer cask, the transportable storage canister, and the storage cask, respectively.

The transfer cask is lifted by two trunnions located near the top of the cask. The 10-inch diameter trunnions extend through the multiwall body to 5.25 inches beyond the outer shell and are full-penetration welded to both the inner and the outer shells (Figure 3.4.3-1). The transfer cask is designed as a heavy-lifting device that satisfies the requirements of NUREG-0612 and ANSI N14.6 for lifting the combined weight of the transfer cask and a fully loaded canister of fuel and water (approximately 143,000 pounds). This is the maximum weight for the transfer cask during a lifting operation.

The transportable storage canister is lifted and handled, while supported on the shield doors within the transfer cask during all preparation, loading and cask closure operations and is then moved to the top of the storage cask. Six hoist rings that are threaded into the structural lid are used to lift the loaded and closed canister just off the shield doors of the transfer cask and to lower the canister into the storage cask after the shield doors are opened. The hoist rings are also used for any subsequent lifting of the loaded canister whose weight is approximately 54,730 pounds (Figure 3.4.3-2).

The vertical concrete cask is raised approximately 3 inches by four lifting jacks placed at the jacking pads located near the end of each air inlet. A system of air pads consisting of 4 units is then inserted under the concrete cask. The cask is lowered onto the air pads (uninflated) and the jacks are removed. The air pads are inflated to lift the concrete cask and position it as required on the storage pad or on the transport vehicle. When positioning is complete, the jacks are again used to raise the cask and remove the air pads.

SCUFF PLATE

<mp

10.00

- (.38)





Figure 3.4.3-2

Canister Hoist Ring Design



### 3.4.3.1 Storage Cask Bottom Lift

The concrete cask is lifted from the bottom via an air pad system. Insertion of the air pad system is made possible by the use of lifting jacks. The current design utilizes a Synchronous Lifting System with four (4) hydraulic jack cylinders. The system is designed to equally distributes hydraulic pressure among four cylinders.

# 3.4.3.1.1 Bottom Lift By Hydraulic Jack

To ensure the concrete bearing stresses at the jack locations do exceed the allowable stress, the required diameter of the jack piston rod is determined. The concrete allowable bearing capacity (in pounds) at each jack is:

$$U_{b} = \phi f_{c}'A = \frac{(0.7)(4,000)\pi d^{2}}{4} = 2,199.1 d^{2},$$

where:

 $\phi = 0.7$  strength reduction factor for bearing,

 $f_c' = 4,000$  psi concrete compressive strength,

A = 
$$\frac{\pi d^2}{4}$$
, concrete bearing area (d = bearing area diameter).

For dead load conditions, the concrete bearing strength must be greater than the storage cask weight times a load factor,  $L_f$ , of 1.4:

2,199.1 d² > 
$$\frac{L_f \times W}{n} = \frac{1.4(206,100)}{4} \Rightarrow d > 5.73$$
 in.,
where:

- n = the number of jacks, 4
- W = the weight of the VCC, 206,100 lbs

Lf = the load factor, 1.4

The diameter obtained in the above equation is the maximum permissible area at the surface of the concrete over which the load must be distributed. The force exerted by the jack to lift the storage cask is applied to the bottom surface of the top plate of the air inlet, which is separated from the concrete surface by a one inch thick steel plate. The force exerted by the jack will be distributed over a larger area on the concrete surface. The effective diameter of the load acting on the concrete surface is increased by 2 x tan ( $45^\circ$ ) x thickness of the steel plate. Therefore, the required hydraulic jack piston diameter is:

5.73 in. - 2 in. = 3.73 in.

The actual hydraulic cylinder to be used has a piston rod diameter of 4 1/8 inch, which is greater than the required 3.73 inch.

#### Nelson Studs

During the bottom lift of the storage cask with hydraulic jacks, the weight of the loaded canister, pedestal and air inlet vent system are transferred to the baseplate of the storage cask (total weight = 63,230 lbs). As the baseplate is loaded, the plate tends to flex, thus separating the concrete from the bottom plate. To prevent this separation, Nelson studs are utilized to bond the concrete to the bottom plate.

Use of the Nelson studs requires proper spacing to prevent overlapping of the shear cones of adjacent Nelson studs (TRW). The term "shear cone" refers to the geometry that a failed concrete section takes when a Nelson stud pulls out of concrete. Overlapping of adjacent shear cones tends to reduce the effectiveness of the Nelson stud. In the case of the storage cask, thirty-two (32),  $3/4 \times 6$  3/16 inch Nelson studs are used. For 4,000 psi strength concrete, the minimum

NAC-MPC FSAR	April 2000
Docket No. 72-1025	Revision 0

spacing between two studs is 7.984 inches (2 x 3.992 inch, where, 3.992 inch is surface area diameter of the full concrete shear cone for  $3/4 \ge 6-3/16$  inch Nelson studs.). The spacing of the Nelson studs on the storage cask bottom exceeds requirements. The total capacity is:

Capacity = 32 Anchors 
$$\times$$
 23,860 lb/Anchor = 763,500 lb

The allowable load, P_u, with a load factor of 2.0, as specified in the TRW design data, is:

$$P_{\rm u} = \frac{763,500}{2.0} = 381,750\,\rm{lb}$$

The total load applied to the storage cask bottom plate (including a 10% dynamic load factor) is:

$$63,230 \times 1.1 = 69,553$$
 lbs

Therefore, the margin of safety is:

$$\text{M.S.} = \frac{381,750}{69,553} - 1 = +4.49$$

### Pedestal - Horizontal Plate

The canister weight (54,730 pounds) is uniformly distributed over the 2-inch thick horizontal circular plate of the pedestal. The self-weight of the plate is 2,313 pounds ( $0.284 \text{ lbs/in}^3 \text{ x } \pi \text{ x } 36^2 \text{ x } 2$ ). The equivalent pressure is:



NAC-MPC FSAR	Ap	oril 2000
Docket No. 72-1025	Re	vision 0

The horizontal plate is supported by the pedestal ring (vertical plate support) at four locations (each has a circumferential length of 16.85 inch). The bending moment at the cross-section of the plate at support location is:

M = (LF)(q)(A)(P_c) = (1.1)(14.01) 
$$\left(\frac{\pi(36^2 - 25^2)}{4}\right)$$
(5.5) = 44,670 in.-lb,

where:

A = the area of the pedestal from the plate support to the edge of the circular plate,

 $P_c = 5.5$  in., the location of the resultant force,

LF = 1.1, load factor to account for 10% dynamic load factor

The bending stress is:

$$f_{b} = \frac{6M}{bt^{2}} = \frac{(6)(44,670)}{(16.85)(2)^{2}} = 3,977 \text{ psi}$$

The material of the pedestal horizontal plate is ASTM A36 with a yield stress (Fy) of 36,000 psi. The allowable stress for flexural members per the Manual of Steel Construction (AISC) is:

$$F_b = 0.66 F_y = 23,760 \text{ psi},$$

and the resulting margin of safety is:

$$M.S. = \frac{23,760}{3,977} - 1 = +5.0$$

The maximum shear stress at the support location is:

$$f_v = \frac{W}{L} = \frac{(54,730+2,313)(1.1)}{4(16.85)(2)} = 465.5 \,\text{psi}$$

The allowable shear stress is 14,400 psi (0.4  $F_y = 0.4 \times 36,000$ ) and the margin of safety is:

$$M.S. = \frac{14,400}{465.5} - 1 = +29.9$$

#### Pedestal Ring (Vertical Plate)

The pedestal ring is subjected to an axial compressive force and bending moments, due to weight of the canister (54,730 lbs), weight of the pedestal horizontal plate (2,313 lbs) and self-weight of the pedestal ring (0.284 lbs/in.  $\times$  16.85  $\times$  2  $\times$  6 = 230 lbs). The maximum compressive stress at the critical cross-section (2 inch  $\times$  1.5 inch, 8 locations) is:

$$f_a = \frac{(54,730 + 2,313 + 230)(1.1)}{8(1.5)(2)} = 2,625 \text{ psi.}$$

The allowable stress,  $F_a$ , for compression member is determined per Chapter E of the Manual of Steel Construction (AISC):

$$\frac{\mathrm{KL}}{\mathrm{r}} = \frac{0.65 \times 6}{0.433} = 9,$$

where:

K = 0.65, effective-length factor for the end conditions (rotation and translation fixed),

L = 6.0 in., height of pedestal ring (unbraced),

$$r = \frac{1.5}{\sqrt{12}} = 0.433$$
, radius of gyration.

Using Chapter E and Tables 3 through 5 of Section 5 of the AISC for A36 material ( $F_y = 36,000$  psi), the allowable stress,  $F_a$ , for compression is determined to be 21,200 psi.

NAC-MPC FSAR	April 2000
Docket No. 72-1025	Revision 0

The bending stress at the same cross-section is conservatively calculated below:

 $P_t$  = one-fourth of the total load = (54,730 + 2,313) / 4 = 14,261 lbs  $\approx$  14,300 lbs



The pedestal is represented as a combination of beams to describe the load path.

 $P_1 = 14,300 \times (15.35 / 37.7) = 5,822$  lbs

 $P_2 = 14,300 \times (22.35 / 37.7) = 8,478$  lbs

 $M_1$  and  $M_2$  are conservatively considered to be the fixed-end moments of beams with a concentrated load at mid-span (with a 10% dynamic load factor).  $L_1$  (15.35 inch) and  $L_2$  (22.35 inch) are the length of the beams.  $M_3$  (the moment at the 2 inch by 1.5 inch cross-section) is considered to be the difference of  $M_1$  and  $M_2$ .

$$M_1 = \frac{1.1(P_1L_1)}{8} = \frac{(1.1)(5,822)(15.35)}{8} = 12,288 \text{ in.} - 1\text{lbs}$$

3.4-17

$$M_2 = \frac{1.1(P_2L_2)}{8} = \frac{(1.1)(8,478)(22.35)}{8} = 26,054$$
in. – lbs

 $M_3 = M_2 - M_1 = 13,766$  in-lbs

The maximum bending stress f_b is computed as:

$$f_b = \frac{6M_3}{bt^2} = \frac{(6)(13,766)}{(2)(1.5)^2} = 18,355 \,\text{psi}$$

The allowable stress for bending ( $F_b$ ) is 23,760 psi (0.66  $F_y$ ). Since  $f_a / F_a$  is less than 0.15, Equation (H1-3) in the Manual of Steel Construction (AISC), Chapter H, is used to evaluate combined stress:

$$\frac{f_a}{F_a} + \frac{f_b}{F_b} = \frac{2,625}{21,200} + \frac{18,355}{23,760} = 0.90 < 1.0$$

Therefore, the pedestal is structurally adequate to support the weight of the loaded canister.

## 3.4.3.1.2 Bottom Support by Air Pads

The storage cask is supported by air pads in each of 4 quadrants during transport. The layout of the air pads (four 48 in.  $\times$  48 in. square pads) must clear the air inlet locations by approximately 3 inches to allow for hydraulic jack access.

The air pad system maximum lift elevation to the storage cask bottom surface is 5.69 inches (3-inch lift, maximum, plus 2-11/16-inch overall height, when deflated) above the transport surface. The air pad system has a vendor rated lift capacity of 360,000 pounds.

The air pad system must supply sufficient force to overcome the weight of the storage cask under full load with a lift load factor, L.F of 1.1. Assuming a fully loaded weight of 206,100 pounds: the required lift load is 1.1(206,100) = 226,710 pounds. Since the available lift force is greater than the required lift force, the air pads are adequate to lift the storage cask. The lifting force margin of safety = (360,000 / 226,700) -1 = +0.59.

## **Operational/Handling Requirements**

The handling force required to control the storage cask on a sloped surface, assuming minimal friction ( $F_{\text{friction}}$ ), is:

- $F = W_{VCC} \sin \alpha$ 
  - = 206,100 sin  $\alpha$



where  $W_{VCC}$  is the weight of the storage cask.

The amount of handling force, F, required on slanted surfaces can be very high. Therefore, the storage cask, while supported by air pads, will be transferred from trailer deck to storage pad site over essentially flat horizontal surfaces.

Once positioned on the storage pad, the storage cask is lifted from the air pads by the jacks for air pad removal. The maximum air pad deflated height plus lift height during all handling and transport is 5.69 inches above the transporter or storage pad surface. Therefore, the storage pad handling method limits the potential drop height to 5.69 inches.

## 3.4.3.2 Canister Lift

The adequacy of the canister lifting devices is demonstrated by considering each of the hoist rings, the canister structural lid, and the weld that joins the structural lid to the canister shell. Lifting of the canister employs redundant three-legged lifting slings for single failure proof lifting in accordance with NUREG-0612.

When considering a three-point lifting configuration, the load-bearing members of the canister maintain a factor of safety greater than three based on material yield strength. Additionally, the load-bearing members of the canister maintain a factor of safety greater than five based on material ultimate strength. The hoist rings are designed with a 5 to 1 safety factor based on ultimate. Therefore, the lifting requirements are satisfied.

Each lifting device in a three-point lift would experience the following load, F, when lifting the canister (the total load is based on the dead weight of the loaded canister with a dynamic load factor of 10 %):

$$F = \frac{54,730 \times 1.1}{3} = 20,068 \text{ lbs}$$

The hoist rings used to lift the canister have a rated load capacity of 24,000 pounds with a 5 to 1 safety factor based on material ultimate strength. The length of the hoist ring bolt is 2.50 inches. The following calculation (using a formula from Machinery's Handbook) demonstrates that a thread engagement length of 2.50 inches is satisfactory.

Definition of Terms

- D = basic major diameter of bolt threads = 1.5 inches
- n = number of threads per inch = 6
- $A_t$  = tensile area of the bolt thread (in²)
- L_e = minimum thread engagement length for mating materials with equal tensile strengths (in.)

 $K_nmax = maximum minor diameter of lid (internal) thread = 1.35 inches$ 

 $E_smin = minimum pitch diameter of bolt (external) threads = 1.3812 inches$ 

 $A_s$  = shear area of bolt threads (in²)

$$A_n$$
 = shear area of lid threads (in²)

 $D_smin = minimum major diameter of bolt (external) threads = 1.4794 inches$ 

 $E_n max = maximum pitch diameter of lid (internal) threads = 1.402 inches$ 

Q = length of thread engagement required to prevent shearing when the mating thread materials are different tensile strengths (in.)

J = scale factor for calculation of Q

H = height of sharp "V" thread = 0.1443 inches

Hoist ring material= 4140 High strength alloy steelTensile strength of 4140= 180,000 psiHoist ring thread= 1-1/2 - 6 UNCStructural lid material= Type 304 L stainless steelTemperature of structural lid=  $250^{\circ}$ FTapped hole in structural lid= 1-1/2 - 6 UNC

For steels up to 100,000-psi tensile strength, the tensile area of the bolt thread is given by:

$$A_t = 0.7854 \left( D - \frac{0.9743}{n} \right)^2$$
  
 $A_t = 1.405 \text{ in}^2$ 

For mating materials having equal tensile strengths, the minimum length of thread engagement is given by:

$$L_{e} = \frac{2A_{t}}{3.1416 K_{n} \max\left[\frac{1}{2} + 0.57735 n (E_{s} \min - K_{n} \max)\right]}$$

 $L_e = 1.0898$  inches

For mating materials of differing tensile strengths:

$$J = \frac{A_s x \text{ tensile strength of bolt thread material}}{A_n x \text{ tensile strength of lid thread material}}$$

The shear areas are calculated as follows:

$$A_{s} = 3.1416 \text{ n } L_{e} \text{ K}_{n} \max \left[ \frac{1}{2n} + 0.57735 (\text{E}_{s} \min - \text{K}_{n} \max) \right]$$

$$A_{s} = 2.8106 \text{ in}^{2}$$

$$A_{n} = 3.1416 \text{ n } L_{e} \text{ D}_{s} \min \left[ \frac{1}{2n} + 0.57735 (\text{D}_{s} \min - \text{E}_{n} \max) \right]$$

$$A_{n} = 3.8871 \text{ in}^{2}$$

At a temperature of 250°F, the ultimate strength of Type 304L stainless steel is 63,550 psi.

$$J = \frac{A_s x (S_u)_{bolt}}{A_n x (S_u)_{lid}}$$

= 2.048

The length of thread engagement necessary to prevent shearing is:

$$Q = JL_e$$

Q = (2,048)(1.073) = 2.20 in. < 2.5 in.

Therefore, the required length of thread engagement is 2.23 inches, which is less than the hoist ring thread length of 2.5 inches.

The hoist rings are rated at 24,000 lbs with a safety factor of 5. The ultimate capacity of the rings is, therefore, 24,000 lbs x 5 = 120,000 lbs (120 kip).

Because the hoist rings and sling legs must demonstrate a safety factor of 10 when compared to ultimate strength, a design load,  $F_D = 11.0$  kips, is chosen.

 $\frac{\text{Ring Ultimate Capacity}}{\text{Design Load}} = \frac{120 \text{ kip}}{11.0 \text{ kip}} = 10.9 > 10$ 

The design load,  $F_D$ , is:

$$F_{\rm D} = \sqrt{F_{\rm H}^2 + F_{\rm V}^2}$$

where:

 $F_{\rm H} = (F_{\rm V}) (\tan \theta)$  horizontal load on the hoist ring, where  $\theta$  is the angle of the sling with respect to the vertical axis of the canister

 $F_V = 10.0$  kip vertical load on the hoist ring

Given a six-point lift, and applying a dynamic load factor of 10%, each hoist ring sees a vertical load,  $F_V = 1.1 (54,730 \text{ lbs})/6 = 10.0 \text{ kip}$ . Since

$$F_{\rm D} = \sqrt{F_{\rm H}^{2} + F_{\rm V}^{2}} = F_{\rm D} = \sqrt{(F_{\rm V} \tan \theta)^{2} + F_{\rm V}^{2}}$$

Solving for  $\theta$ :

 $\theta = \tan^{-1} \frac{\sqrt{(11.0 \text{kips})^2 - (10.0 \text{kips})^2}}{10.0 \text{kip}} = 24.6 \text{ degrees}$ 

The distance from the center line of the canister to the hoist ring bolt circle is 30.25 inches. The minimum allowable distance, Y, from the top of the canister is:

$$Y = \frac{X}{\tan \theta} = \frac{30.25}{\tan 24.6} = 66.1$$
 inches

Therefore, the minimum allowable distance from the master link of the sling to the top of the canister is 66.1 inches. A minimum distance of 67 inches is specified in Sections 8.1.2, 8.2 and Appendix 12A, Section 4.5.4.

The structural adequacy of the canister structural lid and weld was evaluated using a finite element representation of the upper portion of the canister using the ANSYS program. As shown in Figure 3.4.3.2-1, the model represents one-half (180° section) of the upper 50-in. of the canister (including the structural and shield lids). The lids and shell in the model are comprised of SOLID45 elements. CONTACT52 elements are used to model the interaction between the structural lid and the canister shell and between the shield lid and the canister shell, just below the respective lid weld joints. The size of the CONTACT52 gaps was determined from nominal dimensions of contacting components. COMBIN40 elements are used between the structural and shield lids in the axial direction and between the shield lid and the backing ring. These gaps are assigned small gap sizes of 1E-8 inches. All gap/spring elements are assigned a stiffness of 1E8 lbs/in.

To enforce symmetry at the boundary of the model (in the x-y plane), all nodes on the x-y symmetry plane were restrained perpendicular to the symmetry plane (UZ). In addition, the nodes in the x-z plane at the bottom of the model were restrained in the axial direction.

Load-bearing members of a lifting device must be capable of lifting three (3) times the combined weight of the shipping container with which it will be used, plus the weight of intervening components of the lifting device without exceeding the tensile yield strength of their materials of construction. In addition, the lifting components must be capable of lifting five (5) times that combined weight without exceeding the ultimate tensile strength of the materials. NUREG 0612 also requires that the lifting loads must be based on the combined maximum static and dynamic loads that could be imparted on the handling device based on characteristics of the crane that will be used. A dynamic load factor of 10% has been applied.

To simulate the lifting of the canister by a three-point lifting device (the lifting configuration is actually two independent three-point lifting devices), point loads equal to one-third of the total canister and contents weight plus a dynamic loading factor of 10% were applied to the model. Because the model represents a half section of the canister, only two point loads were applied 120° apart as shown in Figure 3.4.3.2-1. Because of the symmetry conditions of the model, the force applied to the node on the symmetry plane was one-half of the value applied at the other location. A uniform temperature of 200°F is used in the model for modulus of elasticity. A temperature of 250°F is conservatively used to determine allowable stresses.

The maximum stress intensity generated in the canister model from the applied lifting forces was 3,753 psi, which occurs in the structural lid where the lifting loads were applied (see Figure 3.4.3.2-2).

As stated previously, the maximum stress intensity in the canister lifting model occurs in the structural lid that is constructed of Type 304L stainless steel. The yield strength  $(S_y)$  of Type 304L stainless steel at 250°F is 20,300 psi. The ultimate strength  $(S_u)$  of Type 304L stainless steel at 250°F is 63,550 psi.

The factor of safety (FS) for the canister lift based on yield strength is:

$$FS = \frac{S_y}{S_{INT}} = \frac{20,300 \, \text{psi}}{3,753 \, \text{psi}} = 5.40 > 3$$

The factor of safety for the canister lift based on ultimate strength is:

$$FS = \frac{S_u}{S_{INT}} = \frac{63,550 \text{ psi}}{3,753 \text{ psi}} = 16.93 > 5$$

# Figure 3.4.3.2-1 Canister Lift Finite Element Model





# Figure 3.4.3.2-2 Canister Lift Model Stresses Intensity Contours (psi)



## 3.4.3.3 <u>Transfer Cask Lift</u>

The evaluation of the transfer cask presented here shows that the design meets ANSI N14.6 and NUREG 0612 for heavy lifts. The adequacy of the transfer cask is shown by evaluating the stress levels in all of the load-path components. The maximum weight of the loaded transfer cask is calculated to be 143,013 pounds (Table 3.2-1). For this analysis, the transfer cask is conservatively assumed to weigh 150,000 pounds. A dynamic load factor of 10 percent is applied to establish a design basis loaded canister weight of 165,000 pounds.

### 3.4.3.3.1 Transfer Cask Shell and Trunnion

A structural evaluation was prepared for the transfer cask to evaluate the structural adequacy of the cask shell and trunnion during lifting conditions, in accordance with ANSI N14.6 and NUREG 0612.

An ANSYS 3-D model is used to evaluate the lifting of a fully loaded transfer cask. Because of symmetry, the 3D ANSYS model considers one quarter of the transfer cask. The model contains only the upper portion of the transfer cask since the purpose of this calculation is to evaluate the shells at the trunnion region. The lead and NS-4-FR between the inner and outer shells of the transfer cask are also neglected since they are not structural components. SOLID95 and SHELL93 elements are used to model the trunnion and shells, respectively. BEAM4 elements are used at the interface of the trunnion and the inner and outer shells to transfer bending moment from the SOLID95 elements, which do not have a rotational degree of freedom, to the SHELL93 elements. The ANSYS model is shown in Figure 3.4.3.3-1.

The total (design) load for the transfer cask is conservatively assumed to be 165,000 pounds, including a 10% dynamic load factor. The loading applied to the model is (165,000) / 4 = 41,250 pounds. The load is applied upward at the trunnion as a "surface load." The lifting yoke dimensions determine the location of the load. The maximum temperature in the transfer cask shell/trunnion region is 300°F. A uniform temperature of 300°F is used in the model for the modulus of elasticity.

Per ANSI N14.6 and NUREG 0612, factors of safety of 6 on material yield strength and 10 on material ultimate strength are required. For the ASTM A-588 shell material, the yield strength,  $S_y$ , is 45.6 ksi, and the ultimate strength,  $S_u$ , is 70 ksi at 300°F. The trunnions are constructed of

3.4-27

NAC-MPC FSAR	April 2000
Docket No. 72-1025	Revision 0

ASTM A-350 carbon steel, Grade LF2. From the ASME Code, Section II, Part D, Tables U ( $S_u$ ) and Y-1 ( $S_y$ ), the yield stress,  $S_y$  is found to be 31.9 ksi, and the ultimate stress,  $S_u$ , is found to be 70 ksi at 300°F.

Tables 3.4.3.3-1 and 3.4.3.3-2 provide a summary of the top 30 maximum stresses for the outer shell and inner shell, respectively (see Figures 3.4.3.3-2 and 3.4.3.3-3 for node locations for outer shell and inner shell, respectively). Stress contour plots for the outer shell and inner shell are shown in Figures 3.4.3.3-4 and 3.4.3.3-5, respectively. As shown in Table 3.4.3.3-1 and 3.4.3.3-2, all stresses, except the local stresses, meet the requirement of ANSI N14.6 and NUREG 0612 with a factor of safety of 6 on material yield strength and 10 on material ultimate strength. Per ANSI N14.6, Section 4.2.1.2, the high local stresses, as categorized in accordance with ASME Code, Section III, NB-3213.10, are relieved by slight local material yielding and the stress design factors are not applicable.

The localized stresses occur at the interface of the trunnion with the inner and outer shells. The meridional direction lengths of the high stress areas are 2.8 inches and 2.0 inches for the inner shell and outer shell, respectively. In accordance with ASME NB-3213.10, the area of localized stresses is limited in the meridional direction to:

1.0√Rt

where:

R is the minimum midsurface radius, and t is the minimum thickness in the region considered.

Based on this formula, the meridional direction length limitations for local stress regions are 5.2 inch (>2.8 inch) and 7.3 inch (>2.0 inch) for the inner and outer shells, respectively.

For the trunnion, the maximum bending and shear stress occur at the interface with the outer shell. The linearized stresses through the trunnion is 3,287 psi in bending and 1,221 psi in shear. Comparing to the material yield stress and ultimate stress (A350 carbon steel), the F.S. on yield and ultimate strength are 9.7 (> 6) and 21.3 (> 10), respectively.

# 3.4.3.3.2 <u>Retaining Ring and Bolts</u>

A retaining ring is bolted on the top of the transfer cask to prevent the inadvertent lifting of the canister out of the transfer cask, which could result in increased radiation exposure to nearby

NAC-MPC FSAR	April 2000
Docket No. 72-1025	Revision 0

workers. In the event that the loaded transfer cask is inadvertently lifted during handling of the canister, the retaining ring and bolts must have sufficient strength to support the weight of the transfer cask. This event is evaluated in this section as an off-normal condition.

#### Retaining Ring

To qualify the retaining ring, the equations presented in Roark for annular rings are utilized. The retaining ring is represented as shown in the sketch below (Roark, Table 24, Case 1e). The following sketch assists in defining the variables used to calculate the stress in the retaining ring and bolts. The model assumes a uniform annular line load w applied at radius  $r_0$ .



The boundary conditions for the model are outer edge fixed, inner edge free.

The material properties and parameters for the analysis are:

Plate dimensions:	Weight of transfer cask (with 10% dynamic load factor):	Number of bolts:	
thickness:	wt = 81,000 lbs × 1.1	Nb = 24	
t = 0.75 in	Radial location of applied load: (canister outer radius)	Radial length of applied load:	
bolt circle:	r _o = 35.3 in	$L_{\tau} = 2\pi r_{o}$	
a = 39.2 in	Ring material:	$L_{\tau} = 221.8$ in	
outer radius (outer edge)	ASTM A588	Applied unit load:	
c = 40.4 in	Modulus of elasticity:	wt	
inner radius:	$E = 28.3 \times 10^6  \text{psi}$	w≡ <u> </u>	
b = 34.3 in	Poisson's ratio:	w = 401.7 psi	
	v = 0.31	•	

____

April 2000 Revision 0

The shear modulus is:

$$G = \frac{E}{2 \cdot (1 + v)}$$
$$= 1.08 \times 10^7 \text{ psi}$$

A plate constant, D, is used in determining boundary values and in the general equations for deflection, slope, moment and shear.

$$D = \frac{E \cdot t^3}{12 \cdot (1 - v^2)}$$
$$D = 1.101 \times 10^6 \text{ lbs-in}$$

Tangential shear constants, K_{sb} and K_{sro}, are used in determining the deflection due to shear:

$$K_{\rm sro} = 1.2 \cdot \frac{r_{\rm o}}{a} \cdot \ln \left(\frac{a}{r_{\rm o}}\right)$$
$$= -0.113$$
$$K_{\rm sb} = K_{\rm sro}$$

The calculated shear stresses at points b and a (inner and outer radius) are:

$$\tau_b = 0$$
 psi  
 $\tau_a = 482.3$  psi

The calculated radial bending stresses at points b and a (inner and outer radius) are:

$$\sigma_r$$
 (b) = 0 psi  
 $\sigma_r$  (a) = 15,220 psi

The calculated tangential bending stresses at points b and a (inner and outer radius) are:

$$\sigma_{t}$$
 (b) = 828.0 psi

$$\sigma_t$$
 (a) = 4,717.9 psi

The principal stresses at the outer radius are:

$$\sigma_{1a} = \left(\frac{\sigma_{r}(a) + \sigma_{t}(a)}{2}\right) + \sqrt{\tau_{a}^{2} + \left(\frac{\sigma_{r}(a) - \sigma_{t}(a)}{2}\right)^{2}}$$

 $\sigma_{la} = 15,240 \text{ psi}$ 

$$\sigma_{2a} = \left(\frac{\sigma_{r}(a) + \sigma_{t}(a)}{2}\right) - \sqrt{\tau_{a}^{2} + \left(\frac{\sigma_{r}(a) - \sigma_{t}(a)}{2}\right)^{2}}$$

 $\sigma_{2a} = 4,695.8 \text{ psi}$ 

 $\sigma_{3a} = 0 \text{ psi}$ 

The stress intensity at the outer radius  $(P_m + P_b)$  is:

$$SI_a = \sigma_{1a} - \sigma_{3a}$$
$$SI_a = 15,240 \text{ psi}$$

The principal stresses at the inner radius are:

$$\sigma_{1b} = \left(\frac{\sigma_r(b) + \sigma_t(b)}{2}\right) + \sqrt{\varpi^2 + \left(\frac{\sigma_r(b) - \sigma_t(b)}{2}\right)^2}$$
  
$$\sigma_{1b} = 0 \text{ psi}$$
  
$$\sigma_{2b} = \left(\frac{\sigma_r(b) + \sigma_t(b)}{2}\right) - \sqrt{\pi^2 + \left(\frac{\sigma_r(b) - \sigma_t(b)}{2}\right)^2}$$

 $\sigma_{2b}$  = - 828.0 psi

 $\sigma_{3b} = 0 \text{ psi}$ 

The stress intensity at the inner radius  $(P_m + P_b)$  is:

 $SI_b = \sigma_{1b} - \sigma_{2b}$ SI b = 828.0 psi

The maximum stress intensity occurs at the outer radius of the retaining ring. For off-normal conditions, the allowable stress is equal to the lesser of 1.8  $S_m$  and 1.5  $S_u$  (Service Level C, ASME Code, Subsection NB). For ASTM A588, the allowable stress at 300°F is 1.8  $S_m = 1.8(23.3) = 41.94$  ksi. The calculated stress intensity of 15.24 ksi is less than the allowable stress and the margin of safety is +1.75.

The bearing stress between the retaining ring and canister is calculated below:

Weight of Transfer Cask = 81,000 x 1.1 = 89,100 lbs

Area of contact between retaining ring and canister:

$$A = \pi \left( 35.3^2 - 34.2^2 \right) = 240 \text{ in }^2$$

$$S_{brg} = \frac{89,100}{240} = 371 \, \text{psi}$$

Bearing stress allowable is  $S_y$ . For ASTM A588, the allowable stress at 300°F is 45.6 ksi. The calculated bearing stress is well below the allowable stress with a large margin of safety.

Shearing stress of Retaining Plate under the bolt head:

Outside diameter of bolt head = 1.125 in.

Total shearing area under the bolt head =  $\pi(1.125)(24)(0.75)=63.6 \text{ in}^2$ 

Shearing stress under the bolt head = 89,100/63.6 = 1,401 psi

Allowable Stress =  $0.60 \times S_m = 0.60 \times 23.3 = 13.98 \text{ ksi}$ 

Margin of Safety = (13,980/1401) - 1 = + large

Bolt Edge Distance:

Using Table J3.5 "Minimum Edge Distance, in." of Section J3 from "Manual of Steel Construction Allowable Stress Design," the required edge distance for a 0.75-inch bolt is 1.00 inch for a saw-cut edge. The actual edge distance for the bolts is:

$$\frac{80.8 - 78.31}{2} = 1.25 \text{ inch} \rangle 1.0 \text{ inch}$$

## **Retaining Ring Bolts**

The load on a single bolt,  $F_F$ , due to the reactive force caused by inadvertently lifting the transfer cask, is:

$$F_{\rm F} = \frac{\rm wt}{\rm Nb} = 3,713 \rm ~lbs$$

The load on each bolt,  $F_M$ , due to the bending moment, is:

$$F_{M} = \left(\frac{2 \cdot \pi \cdot a}{Nb}\right) \cdot \left(\frac{\sigma \cdot t^{2}}{6 \cdot L}\right)$$

 $F_M = 12,200 \text{ lbs}$ 

where:

 $\sigma$  = the radial bending stress at point a (15,220 psi)

L = the distance between the bolt center line and ring outer edge (c - a = 1.2 inches)

The total tension, F, on each bolt is:

 $F = F_F + F_M = 15,910$  lbs

The bolt cross-sectional area,  $A_b$ , is 0.4418 in². The bolt tensile stress is:

$$\sigma_{t} = \frac{F}{A_{b}} = 36,012 \text{ psi}$$

For off-normal conditions, the allowable stress for primary membrane stress in a bolt is  $2S_m$ . The allowable stress for SA193 Grade B6 high alloy steel bolts is  $2 \times 27.0 = 54.0$  ksi at 200°F. The margin of safety for the bolts is +0.50.

The top plate internal threads are evaluated for resistance to shear pull out. The thread specifications are 3/4-10 UNC (Class 2A external thread and class 2B internal thread):

D = 0.7482 in. (basic major diameter of the bolt threads) n = 10 (number of bolt threads per inch)  $D_{s}min = 0.7353 \text{ in.(minimum major diameter of bolt threads)}$   $E_{n}max = 0.6927 \text{ in. (maximum pitch diameter of lid threads)}$   $L_{e} = 2.25 - 0.75 = 1.5 \text{ in. (minimum thread engagement)}$ 

For the top plate (ASTM 588) at a temperature of 200°F, the yield and ultimate stresses are:

$$S_y = 47.5$$
 ksi  
 $S_u = 70.0$  ksi  
 $S_m = 23.3$  ksi

The shear area for the internal (top plate) threads (A_n) is:

A_n = 3.1416nL_eD_smin 
$$\left[\frac{1}{2n} + 0.57735(D_s \min - E_n \max)\right] = 2.584 \text{ in}^2$$

The shear stress  $(\tau_n)$  in the top plate is:

$$\tau_{\rm n} = \frac{F_{\rm y}}{A_{\rm n}} = \frac{15,910}{2.584} = 6,157 \; \rm psi$$

where:

$$F_{\rm v} = F = F_{\rm F} + F_{\rm M}$$

Conservatively, the shear allowable for normal conditions is used.

$$\tau_{\text{allowable}} = (S_{\text{m}}) (0.6) = (23.3 \text{ ksi})(0.6) = 13.98 \text{ ksi}$$

The Margin of Safety is: (13,980/6,157) -1 = +1.27

## 3.4.3.3.3 Rails and Welds – Shield Door

This section demonstrates the adequacy of the shield doors, door rails, and welds in accordance with ANSI N14.6, which requires safety factors of 6 and 10 on material yield strength and ultimate strength, respectively.

The shield door rails support the weight of a wet, fully loaded canister and the weight of the shielding doors themselves. The shield doors are 9.5-inch thick plates resting on top of the bottom section of the door rails. The door rails are 9.88 inches deep by 6.5 inches thick and are welded to the bottom plate of the transfer cask. Both the doors and the rails are constructed of A-350, Grade LF2 carbon steel. The design load for the rails (considering 10% dynamic factor) is conservatively assumed to be:

 $W = 150,000 \times 1.1 = 165,000 \text{ lbs}$ 

This evaluation shows that the shield doors, door rail structures, and welds are adequate to support a wet, fully loaded canister.

#### **Operating Conditions**

Based on the thermal analysis as presented in Chapter 4, the maximum calculated temperature for the transfer cask doors is 257°F. The material allowables are conservatively taken at 300°F.

#### Material Properties

The material properties for the ASTM A-350, Grade LF2, carbon steel are taken from the ASME Code, Division II, Part D, Tables U ( $S_u$ ) and Y-1 ( $S_y$ ).

Yield Strength ( $S_y$ ) = 31.9 ksi at 300° F (ASTM A-350) Ultimate Strength ( $S_u$ ) = 70.0 ksi at 300° F (ASTM A-350)

#### Stress Evaluation for Door Rail

The shear stress in each door rail bottom plate due to the applied load of W is:

$$\tau = \frac{W}{2 \times A_s} = \frac{165}{2 \times 129.95} = 0.64 \text{ ksi}$$

## NAC-MPC FSAR Docket No. 72-1025



and,

- L = rail length supporting doors =  $2 \times (45.38-17.39-4/2) = 51.98$  inch,
- t = bottom plate thickness = 2.5 inch.

The bending stress in each rail bottom section due to the applied load of W is:

$$\sigma_{\rm b} = \frac{M}{S} = \frac{90.8}{54.15} = 1.68 \text{ ksi},$$

where:

M = moment at the bottom section,

$$=\frac{W}{2} \times e = \frac{165}{2} \times 1.1$$

$$= 90.8$$
 in-kips,



and,

e = 
$$\frac{2-0.19}{2} + 0.19 = 1.1$$
 inch applied load moment arm.  
S =  $\frac{b \times d^2}{6} = \frac{51.98 \times 2.5^2}{6}$   
= 54.15 in³.

Per ANSI N14.6, Section 4.2.1, shear stress or maximum tensile stress are to be compared with material yield and ultimate strength. The resulting factors of safety are:

$$\varphi_y = \frac{31.9}{1.68} = 19.0 > 6$$
 (For yield strength criteria)  
 $\varphi_u = \frac{70}{1.68} = 41.7 > 10$  (For ultimate strength criteria)

#### Stress Evaluation for the Shield Doors

The shield doors consist of two 9.5-inch thick plates that rest on top of the rails. Stresses can be approximated by modeling the shield door as a simply supported beam with a concentrated load at the center.



= 0.31 ksi

April 2000 Revision 0

and the bending stress in each shield door is

$$\sigma_{\rm b} = \frac{M}{S} = \frac{1549.35}{421} = 3.68 \text{ ksi},$$

where:

M = moment at the bottom section,

$$= \frac{P \times L}{4} = \frac{82.5 \times 75.12}{4} = 1549.35 \text{ in-kips},$$

S = section modulus

$$=\frac{b\times d^2}{6}=\frac{27.99\times 9.5^2}{6}=421 \text{ in}^3$$

The factors of safety are:

$$\varphi_y = \frac{31.9}{3.68} = 8.7 > 6$$
 (For yield strength criteria)  
 $\varphi_u = \frac{70}{3.68} = 19.0 > 10$  (For ultimate strength criteria)

#### **Door Rail Weld Evaluation**

The rail welds were evaluated by determining the reactive forces,  $F_o$  and  $F_i$ , experienced by the outer and inner welds due to applied load, W.



$$F_i = \frac{165}{2} + 20.17 = 102.67$$
 kips

Maximum stresses at the groove weld (size = 0.75 inch) are:

$$= \frac{102.67}{56.25 \times 0.75} = 2.4 \text{ ksi}$$

The factors of safety are:

$$\varphi_y = \frac{31.9}{2.4} = 13.3 > 6$$
 (For yield strength criteria)  
 $\varphi_u = \frac{70}{2.4} = 29.2 > 10$  (For ultimate strength criteria)

# Figure 3.4.3.3-1 Finite Element Model for Transfer Cask Trunnion and Shells



NAC-MPC FSAR	April 2000
Docket No. 72-1025	Revision 0

# Figure 3.4.3.3-2 Node Locations for Transfer Cask Outer Shell Adjacent To Trunnion



# Figure 3.4.3.3-3 Node Locations for Transfer Cask Inner Shell Adjacent To Trunnion



April 2000 Revision 0

# Figure 3.4.3.3-4 Stress Contours for Transfer Cask Outer Shell



# Figure 3.4.3.3-5 Stress Contours for Transfer Cask Inner Shell



## NAC-MPC FSAR Docket No. 72-1025

	Princ	ipal Stresses	s (psi)	Nodal S.I.	F.S. on Yield ²	F.S. on Ultimate ²
Node1	S1	S2	<b>S</b> 3	(psi)	(S _y /S.I.)	(S _u /S.I.)
815	18395	1593.7	-195.14	18591 ³	N/A	N/A
703	512.98	-1170.3	-16452	16965 ³	N/A	N/A
829	8895.7	4797.9	-20.588	8916.2 ³	N/A	N/A
818	7296.6	1839.7	-7.8552	7304.4 ³	N/A	N/A
862	6536.6	2745.7	-9.4784	6546	7.0	10.7
638	3595.9	-19.633	-2848.4	6444.3	7.1	10.9
776	4756.4	314.84	-1459.5	6215.9	7.3	11.3
871	18.466	-866.71	-6128.8	6147.2	7.4	11.4
709	93.408	-5325.4	-6033.2	6126.6	7.4	11.4
649	2094.5	-385.96	-3988.5	6083	7.5	11.5
827	5931.4	3499.1	-24.444	5955.9	7.7	11.8
778	5262.6	1098.1	-595.23	5857.9	7.8	11.9
864	5725.4	2444.8	-5.0196	5730.4	8.0	12.2
873	17.86	-668.51	-5669.4	5687.3	8.0	12.3
780	5345.3	2320.7	-313.59	5658.9	8.1	12.4
651	1052.3	-1275.6	-4516.8	5569.1	8.2	12.6
767	5246.3	3215.3	-188.33	5434.6	8.4	12.9
825	5317.3	3368.9	-91.577	5408.9	8.4	12.9
875	17.578	-482.15	-5250.6	5268.2	8.7	13.3
883	29.511	-789.43	-5198.3	5227.8	8.7	13.4
653	670.82	-2716.1	-4540.7	5211.5	8.7	13.4
866	5135.7	1721.4	0.66016	5135	8.9	13.6
820	5118.7	2375.9	-1.6525	5120.3	8.9	13.7
893	30.529	-567.18	-4905	4935.5	9.2	14.2
852	4898.5	2243.8	-2.5875	4901.1	9.3	14.3
769	4816.4	1032.1	-1.2375	4817.7	9.5	14.5
647	505.55	-3770.9	-4172.3	4677.9	9.7	15.0
641	2895.5	1.2697	-1776.8	4672.3	9.8	15.0
694	2169.8	4.4459	-2469.6	4639.4	9.8	15.1
903	32.924	-362.76	-4599.4	4632.4	9.8	15.1

# Table 3.4.3.3-1Top 30 Stresses for Transfer Cask Outer Shell

Notes:

1. See Figure 3.4.3.3-2 for node locations.

2.  $S_y = 45,600 \text{ psi}, S_u = 70,000 \text{ psi}.$ 

3. Local stresses that are relieved by local material yielding. Therefore, stress design factors of 6 and 10 on material yield and ultimate strength are not applicable (ANSI N14.6, Section 4.2.1.2).

NAC-MPC FSAR Docket No. 72-1025

	Princ	ipal Stresse	s (psi)	Nodal S.I.	F.S. on Yield ²	F.S. on Ultimate ²
Node ¹	S1	S2	S3	(psi)	(S _y /S.I.)	(S _u /S.I.)
1869	17689	436.58	-1999.3	19688 ³	N/A	N/A
1634	9754.8	619.16	-759.58	10514 ³	N/A	N/A
1882	7096.9	803.79	-559.18	7656 ³	N/A	N/A
1731	2629	-50.822	-4142.7	6771.7	6.7	10.3
1725	1528.8	-289.56	-5044.6	6573.4	6.9	10.6
1884	5878.7	463.76	-401.23	6279.9	7.3	11.1
1729	3436.2	79.26	-2823.1	6259.4	7.3	11.2
1742	2276.9	0.49938	-3877.4	6154.4	7.4	11.4
1782	2681.2	0.15472	-3465.3	6146.5	7.4	11.4
1797	936.56	190.95	-5161.7	6098.2	7.5	11.5
1801	499.98	-2533.2	-5548.2	6048.1	7.5	11.6
1799	674.7	-1179.9	-5362.4	6037.1	7.6	11.6
1886	5937.1	2421.9	-85.981	6023.1	7.6	11.6
1803	469.31	-3500.1	-5469.9	5939.2	7.7	11.8
1822	1911.9	5.7018	-4006.7	5918.5	7.7	11.8
1766	2791.3	-1.7689	-2991.6	5783	7.9	12.1
1727	3960.6	331.95	-1535.6	5496.2	8.3	12.7
1879	5014.2	118.74	-303.37	5317.6	8.6	13.2
1838	1675.8	22.011	-3579.3	5255.1	8.7	13.3
1646	4023.3	708.54	-1131	5154.4	8.8	13.6
1750	2409.6	-7.2699	-2506.8	4916.4	9.3	14.2
1740	2413.9	-1.0861	-2489.3	4903.3	9.3	14.3
1784	2271.4	-0.2874	-2557.6	4829	9.4	14.5
1824	2217	-0.4888	-1981.1	4198.1	10.9	16.7
1854	1463.1	39.789	-2681.4	4144.5	11.0	16.9
1806	3183.2	68.412	-943.43	4126.6	11.1	17.0
1768	1787.3	-0.4604	-2229.5	4016.7	11.4	17.4
1648	3076.7	1254.1	-849.56	3926.2	11.6	17.8
1932	3633.4	1174.9	-9.7208	3643.1	12.5	19.2
1738	2162.3	1.0241	-1473.5	3635.8	12.5	19.3

# Table 3.4.3.3-2Top 30 Stresses for Transfer Cask Inner Shell

Notes:

1. See Figure 3.4.3.3-3 for node locations.

2.  $S_y = 45,600 \text{ psi}, S_u = 70,000 \text{ psi}.$ 

3. Local stresses that are relieved by local material yielding. Therefore, stress design factors of 6 and 10 on material yield and ultimate strength are not applicable (ANSI N14.6, Section 4.2.1.2).

NAC-MPC FSAR	April 2000
Docket No. 72-1025	Revision 0

# 3.4.4 NAC-MPC Components Under Normal Operating Loads

The NAC-MPC system is evaluated using individual finite element models for the fuel basket, canister, and vertical concrete cask. Since the individual components are free to expand without interference, the structural finite element models need not be connected. Temperature dependent material properties are used in the finite element models based on calculated temperatures from the thermal analysis in Chapter 4.

## 3.4.4.1 Canister and Basket Analyses

### 3.4.4.1.1 Canister Thermal Stress Analysis

A three-dimensional finite element model of the canister was constructed using ANSYS SOLID45 elements. By taking advantage of the symmetry of the canister, the model represents one-half (180° section) of the canister, including the canister shell, bottom plate, structural lid, and shield lid. The model uses gap/spring elements to simulate contact between adjacent components. Specifically, contact between the structural and shield lids was modeled using COMBIN40 combination elements in the axial (UY) degree of freedom. Simulation of the backing ring is accomplished using a ring of COMBIN40 gap/spring elements connecting the shield lid and the canister in the axial direction at the lid lower outside radius. In addition, CONTAC52 elements were used to model the interaction between the structural lid and the canister shell and between the shield lid and canister shell, just below the respective lid weld joints. The size of the CONTAC52 gaps was determined from nominal dimensions of contacting components. The COMBIN40 elements used between the structural and shield lids and for the backing ring were assigned small gap sizes of 1E-8 inches. All gap/spring elements were assigned a stiffness of 1E+8 lbs/in. The three-dimensional ANSYS model of the canister used in the thermal stress evaluation is shown in Figure 3.4.4.1-1 through Figure 3.4.4.1-3.

The ANSYS thermal stress analysis was performed with canister temperatures that enveloped the canister temperature gradients for normal storage (100°F and -40°F ambient temperatures) and transfer conditions. Prior to performing the thermal stress analysis, the steady-state temperature distribution was determined using temperature information from the storage and transfer thermal analyses (Chapter 4). This was accomplished by converting the SOLID45 structural elements of the canister model to SOLID70 thermal elements and using the material properties from the thermal analyses. Nodal temperatures were applied at six key locations (i.e., top-center of the structural lid, top-outer diameter of the structural lid, bottom-center of the shield lid, bottom-

# NAC-MPC FSAR Docket No. 72-1025

center of the bottom plate, bottom-outer diameter of the bottom plate, and mid-elevation of the canister shell). The temperatures of the key locations used in the analysis were as follows:

Top center of the structural lid	=	140°F
Top outer diameter of the structural lid	=	100°F
Bottom center of the shield lid	=	165°F
Bottom center of the bottom plate	Ξ	250°F
Bottom outer diameter of the bottom plate	=	130°F
Mid-elevation of the canister shell	=	500°F

The temperatures for all nodes in the canister model were obtained by the solution of the steady state thermal conduction problem.

These temperatures were selected to envelope the temperature differences experienced by the canister for storage and transfer conditions as calculated in the thermal analysis presented in Chapter 4. The following table shows the temperature differences ( $\Delta$ T) of the canister in the radial and axial directions for the storage and transfer conditions and those used in the canister thermal stress analysis:

	Maximum ΔT (°F)			
Condition	Top of Structural	Bottom Plate	Canister Shell	
	Lid (Radial)	(Radial)	(Axial)	
Storage, Normal 75°F	17	113	223	
ambient				
Storage, Off-Normal	17	115	225	
100°F ambient				
Storage, Off-Normal,	15	104	213	
-40°F ambient				
Storage, Off-Normal	16	115	221	
Half Inlets Blocked				
Transfer, 75°F ambient	5	21	386	
Canister Thermal	40	120	400	
Stress Analysis		:		
Additionally, canister temperatures used for determining allowable stress values were selected to envelope the maximum temperatures experienced by the canister during storage and transfer conditions. Specifically, allowable stresses were selected at temperatures of 250°F, 550°F, and 250°F for the structural/shield lid region, the canister shell, and the bottom plate region, respectively.

After solving for the canister temperature distribution, the thermal stress analysis was then performed by converting the SOLID70 elements back to SOLID45 structural elements. A single node at the centerline of the bottom plate is restrained in the axial direction (UY) to eliminate rigid body translation in the Y-direction. The nodes along the centerline of the structural and shield lids and the bottom plate were restrained in the x-direction (UX) to prevent rigid body motion in the x-direction. The nodes on the symmetry boundary face were restrained in the direction normal to the symmetry plane (UZ). A linear solution was performed to obtain the stresses due to thermal expansion.

The resulting maximum (secondary) thermal stresses in the canister are summarized in Table 3.4.4.1-1. The sectional stresses at 15 axial locations were obtained for each angular division of the model (a total of 21 angular locations for each axial location). The locations for the stress sections are shown in Figure 3.4.4.1-4.

#### 3.4.4.1.2 Canister Dead Weight Load Analysis

The canister was structurally analyzed for dead weight load using the ANSYS model described in Section 3.4.4.1.1. The canister temperature distribution discussed in Section 3.4.4.1.1 was used in the dead load structural analysis to evaluate the material allowable stresses at temperature. The fuel and fuel basket assembly contained within the canister were not explicitly modeled, but were included in the analysis by applying a uniform pressure load representing their combined weight to the top surface of the canister bottom plate. The nodes on the bottom surface of the bottom plate were restrained in the axial direction in conjunction with the constraints described in Section 3.4.4.1.1. An acceleration of 1g was applied to the model in the axial direction (Y) to simulate the dead load.

Docket No. 72-1025	Revision 0
NAC-MPC FSAR	April 2000

The resulting maximum canister dead load stresses are summarized in Tables 3.4.4.1-2 and 3.4.4.1-3 for primary membrane and primary membrane plus bending stresses, respectively. The sectional stresses at 15 axial locations were obtained for each angular division of the model (a total of 21 angular locations for each axial location). The locations for the stress sections are shown in Figure 3.4.4.1-4.

The lid support ring is evaluated for the dead load condition using classical methods. The lid support ring is welded to the inner surface of the canister shell, under the shield lid. The lid support ring is made of ASTM A-479, Type 304 stainless steel. A temperature of 600°F is conservatively used to determine the material allowable stress. The total weight, W, imposed on the lid support ring is conservatively considered to be the weight of the structural lid (3,234 lbs), the shield lid (5,389 lbs) and the weight of the backing ring (16 lbs). The stresses on the support ring are the bearing stresses and shear stresses at its weld to the canister shell.

The bearing stress  $\sigma_{\text{bearing}}$  is calculated as follows:

$\sigma_{\text{bearing}}$	-	W area
	=	8639 108
	=	80 psi

where:

W = 3,234 + 5,389 + 16= 8,639 lbs area =  $\pi \times D \times t$  =  $108 \text{ in}^2$ D = lid support ring average diameter = 68.89 inch t = radial thickness of support ring = 0.5 inch

NAC-MPC FSAR	April 2000
Docket No. 72-1025	Revision 0

The yield strength  $(S_y)$  for A-479, Type 304 stainless steel is 18,600 psi @ 600°F. The allowable bearing stress is 1.0 S_y per ASME Section III, Subsection NB.

Margin of Safety = (18,600/80) - 1

= + Large

The weld for the lid support ring is a 3/8 inch partial penetration groove weld. The total shear force on the weld is considered to be the weight of the structural and shield lids, the backing ring, and the lid support ring (8,655 lbs). The shear stress on the weld is calculated as follows:

 $\sigma_{w} = \frac{W}{area}$  $= \frac{8655}{81.32}$ = 106 psi

where:

area =  $\pi x D x t_w$  = 81.32 in² D = shield lid diameter = 69.03 inch  $t_w$  = weld size = 0.375 inch

The yield strength  $(S_y)$  for A-479, Type 304 stainless steel is 18,600 psi @ 600°F. In accordance with ASME Section III, Subsection NB, the allowable shear stress is 0.6 x S_m.

Margin of Safety =  $(0.6 \times (2/3) \times 18,600) / 106 - 1$ 

= + Large

#### 3.4.4.1.3 Canister Maximum Internal Pressure Analysis

The canister was structurally analyzed for a design internal pressure load of 11.5 psig using the ANSYS model and temperature distribution and restraints described in Section 3.4.4.1.1. This pressure bounds the calculated pressure of 7.9 psig for normal conditions. The design internal pressure of 11.5 psi is applied as a surface force to the elements along the internal surface of the canister shell, bottom plate, and shield lid.

The resulting maximum canister stresses for maximum internal pressure load are summarized in Tables 3.4.4.1-4 and 3.4.4.1-5 for primary membrane and primary membrane plus primary bending stresses, respectively. The sectional stresses at 15 axial locations were obtained for each angular division of the model (a total of 21 angular locations for each axial location). The locations of the stress sections are shown in Figure 3.4.4.1-4.

#### 3.4.4.1.4 Canister Handling Analysis

The canister was structurally analyzed for handling loads using the ANSYS model and conditions described in Section 3.4.4.1.1. Normal handling of the canister was simulated by restraining the model at three lift points and applying a 1.1g acceleration load to the model in the axial direction, which includes a 10% dynamic load factor. The canister is lifted at six points; however, the handling analysis considers only a three-point lifting configuration. Since the model represents a one-half section of the canister, the three-point lift was simulated by restraining two nodes 120° apart (one node at the symmetry plane and a second node 120° from the first) along the bolt diameter at the top of the structural lid in the axial direction. Additionally, the nodes along the centerline of the lids and bottom plate were restrained in the radial direction, and the nodes along the symmetry face were restrained in the direction normal to the symmetry plane.

The resulting maximum stresses in the canister for the handling load are summarized in Tables 3.4.4.1-6 and 3.4.4.1-7 for primary membrane and primary membrane plus primary bending stresses, respectively. The sectional stresses at 15 axial locations were obtained for each angular division of the model (a total of 21 angular locations for each axial location). The locations for the stress sections are shown in Figure 3.4.4.1-4.

#### 3.4.4.1.5 Canister Load Combination

The canister was structurally analyzed for the combined thermal, dead, maximum internal pressure, and handling loads using the ANSYS model and conditions described in Section 3.4.4.1.1. Loads were applied to the model as discussed in Sections 3.4.4.1.1 through 3.4.4.1.4. A maximum internal pressure of 11.5 psi was used in conjunction with a positive axial acceleration of 1.1g. Two nodes 120° apart (one node at the symmetry plane and a second node 120° from the first) were restrained along the bolt diameter at the top of the structural lid in the axial direction. Additionally, the nodes along the centerline of the lids and bottom plate were restrained in the radial direction, and the nodes along the symmetry face were restrained in the direction normal to the symmetry plane.

The resulting maximum stresses in the canister for combined loads are summarized in Tables 3.4.4.1-8, 3.4.4.1-9, and 3.4.4.1-10 for primary membrane, primary membrane plus primary bending, and primary membrane plus primary bending plus secondary stresses, respectively. The sectional stresses at 15 axial locations were obtained for each angular division of the model (a total of 21 angular locations for each axial location). The locations for the stress sections are shown in Figure 3.4.4.1-4. As shown in Tables 3.4.4.1-8 through 3.4.4.1-10, the canister maintains positive margins of safety for the combined load condition.

#### 3.4.4.1.6 Canister Fatigue Evaluation

The purpose of this section is to evaluate the effects of thermal and mechanical cyclic loading conditions on the canister during storage conditions using the criteria presented in ASME Code, Section III, Subsection NB-3222.4 for the canister and Subsection NG-3222.4 for the fuel basket.

During storage conditions, the canister is housed in the vertical concrete storage cask. The storage cask is a shielded reinforced concrete overpack designed to hold a canister during long-term storage conditions. The storage cask is constructed of a thick inner steel liner surrounded by 21 inches of reinforced concrete. Because the carbon steel inner liner will be subjected to a number of temperature/stress loading cycles (1 cycle x 365 days x 50 years = 18,250 cycles) that is less than the minimum number (20,000 cycles) specified for evaluation in Table A-K4.1 of the AISC Manual of Steel Construction, no further fatigue evaluation of the inner liner is required.

Fatigue effects on the canister are addressed using the criteria presented in ASME Section III, Subsection NB-3222.4 and NG-3222.4.

In accordance with these subsections, fatigue analysis need not be performed provided the conditions of six cases are met. The six cases are as follows:

- 1. Atmospheric to Service Pressure Cycle
- 2. Normal Service Pressure Fluctuation
- 3. Temperature Difference Startup and Shutdown
- 4. Temperature Difference Normal Service
- 5. Temperature Difference Dissimilar Materials
- 6. Mechanical Loads

Evaluation of these conditions is presented in the following sections.

Condition 1 — Atmospheric to Service Pressure Cycle

The ASME code requires that the specified number of times that the pressure will be cycled from atmospheric pressure to service pressure and back to atmospheric pressure during normal service does not exceed the number of allowable cycles for the material. In the case of the canister and basket, the cycle from atmospheric to service pressure happens only twice. Since this operation occurs only twice during the 50-year life of the canister (once when the canister is sealed and once when it is opened), atmospheric to service pressure cycle loading of the canister and basket does not cause fatigue failure.

Condition 2 — Normal Service Pressure Fluctuation

To prevent fatigue failure of the canister, the specified full range of pressure fluctuations during normal service must not exceed:

$$P_{f} = \frac{1}{3} \times P_{d} \times \left(\frac{S_{a}}{S_{m}}\right) = \frac{20 \times 28.2}{3 \times 16.7} = 11.3 \text{ psi},$$

where:

- $S_a = 28.2$  ksi, the value obtained from the design fatigue curve for service cycles  $< 10^6$ ,
- $S_m = 16.7$  ksi, the allowable stress intensity,
- $P_d = 20$  psi, the design pressure (bounds maximum pressure of 17.93 psi for transfer conditions).

The maximum pressure differential for the canister occurs between transfer and storage conditions. For normal and transfer conditions the maximum pressure differential is:

 $\Delta P = 17.93 - 11.32 = 6.61 \text{ psi} < 11.3 \text{ psi}.$ 

Therefore, the effective number of cycles is zero.

Condition 3 — Temperature Difference — Startup and Shutdown

This condition is not applicable. It is only required for power plant startup and shutdown processes.

Condition 4 — Temperature Difference — Normal and Off-Normal Service

#### **Canister Evaluation**

The ASME Code specifies that temperature excursions are not significant if the temperature difference between two adjacent points does not change by more than the quantity:

$$\Delta T = \frac{S_a}{2E\alpha} = 58^{\circ} F,$$

where:

 $S_a = 28,200$  psi, the value obtained from the fatigue curve for service cycles  $< 10^6$ ,

 $E = 27 \times 10^6$  psi, modulus of elasticity at 300 °F,

 $\alpha = 9 \times 10^{-6}$  in./in./°F.

For surface temperature differences on surfaces of revolution in the meridional (axial) direction, adjacent points are defined as points that are less than the distance  $2\sqrt{Rt}$ , where R is the radius measured normal to the surface, from the axis of rotation to the midwall and t is the thickness of the part at the point under consideration. For surface temperature differences on surfaces of revolution in the circumferential direction and on flat parts, such as flanges and flat heads, adjacent points are defined as any two points on the same surface.

The greatest cyclic temperature difference will occur between the off-normal, severe hot (ambient temperature =  $100^{\circ}$ F) and the off-normal, severe cold (ambient temperature =  $-40^{\circ}$ F) conditions as evaluated in the thermal evaluation. Accident temperature conditions are not applicable.

At the hot condition, the canister bottom plate temperature varies from 237°F at its center to 123°F at its extreme radial point, a  $\Delta T$  of 114°F. At the cold condition, the canister bottom plate temperature varies from 78.3°F at its center to -24.6°F at its extreme radial point, a  $\Delta T$  of 102.9°F. Therefore, in cycling from 100°F ambient to -40°F ambient conditions, the  $\Delta T$  between adjacent points changes by 11.1°F, which is less than the 58°F  $\Delta T$  and is not considered to be a significant excursion. Heat transfer is uniform around the circumference; therefore, no cyclic  $\Delta T$  exists in adjacent points on a circumference of the shell.

At the hot condition the canister shell temperature varies from  $347.5^{\circ}F$  at its center to  $121.9^{\circ}F$  at its top, a  $\Delta T$  of  $225.6^{\circ}F$ . At the cold condition, the canister shell temperature varies from  $187.4^{\circ}F$  at its center to  $-25.6^{\circ}F$  at its top, a  $\Delta T$  of  $213^{\circ}F$ . The distance between adjacent points is:

$$d_p = 2\sqrt{Rt} = 9.35$$
 in.,

where:

R = 70.64/2 - 0.625/2 = 35.0 in, the mean radius of the canister shell,

t = 0.625 in., the wall thickness of the canister shell.

At  $T_{amb}=100$  °F, the  $\Delta T$  between center of canister and end of canister =  $(347.5^{\circ}F - 121.9^{\circ}F) = 225.6^{\circ}F$ . The  $\Delta T$  of adjacent points is  $(225.6^{\circ}F / 61.25 \text{ in.})(9.35 \text{ in.}) = 34.4^{\circ}F$ .

At  $T_{amb} = -40$  °F, the  $\Delta T$  between center of canister and end of canister =  $[187.4^{\circ}F - (-25.6^{\circ}F)] = 213^{\circ}F$ . The  $\Delta T$  of adjacent points is  $(213.0^{\circ}F / 61.25 \text{ in.})(9.35 \text{ in.}) = 32.5^{\circ}F$ .

Therefore, in cycling from 100°F ambient to -40°F ambient conditions, the  $\Delta T$  between adjacent points changes by 1.9°F, which is less than the 58°F  $\Delta T$  and is not considered to be a significant excursion.

#### **Basket Evaluation**

In storage, the basket is isolated from the influence of environmental temperature excursions by the canister. Any temperature differences within the basket structure are bounded by the evaluation of the temperature differences evaluated for the canister.

Condition 5 — Temperature Difference Between Dissimilar Materials

The canister is constructed of 304L stainless steel and does not contain dissimilar materials. The basket is constructed of several materials. However, all materials except the support disks are free to expand, thus relieving any thermal stress concentration. As noted under the Condition 4 discussion, the temperature differences within the basket are bounded by the temperature differences evaluated for the canister.

Condition 6 — Mechanical Loads

Mechanical loads are not applied to the storage cask and canister during storage conditions. Therefore, no further evaluation is required.

#### 3.4.4.1.7 Canister Pressure Test

The canister is tested using an air over water pressure test in accordance with ASME Code Section III, NB-6221 through NB-6223. The calculated pressure for normal conditions is 7.9 psig. A normal condition design-basis pressure of 11.5 psig is conservatively applied. In accordance with NB-6221, the test pressure applied is 15 psig ( $\approx 11.5 \times 1.25$ ). The stress resulting from the pressure test is evaluated in accordance with the requirements of NB-3226. The canister is not reused, and the pressure test is conducted only once. Therefore, the pressure test is not considered in the fatigue analysis.

NB-3226 requires that  $P_m$  not exceed 0.9 S_y at the test temperature. To show that the canister air over water pressure test meets this condition, the stress intensities calculated for the canister due to the normal condition design pressure of 11.5 psig (Section 3.4.4.1.3), are ratioed to account for the 15 psig test pressure. The canister material is ASME SA-240, Type 304L stainless steel, and the material temperature is conservatively taken to be 250°F. From Table 3.4.4.1-4, the maximum primary stress intensity,  $P_m$ , is 4.60 ksi. Therefore:

$$(P_m)_{test} = (15/11.5) \times (4.60 \text{ ksi}) = 6.0 \text{ ksi}$$
, which is  $< .9S_v = 18.3 \text{ ksi} (.9 \times 20.3 \text{ ksi})$ 

Thus, the criteria is met.

NB-3226 requires that for  $P_m < 0.67S$ , the primary membrane plus bending stress intensity,  $P_m + P_b$ , be  $\le 1.35S_y$ . From Table 3.4.4.1-5,  $P_m + P_b = 10.02$  ksi. Therefore:

$$(P_m + P_b)_{test} = (15/11.5) \times (10.02 \text{ ksi}) = 13.1 \text{ ksi}$$
, which is  $\le 1.35S_v = 27.4 \text{ ksi} (1.35 \times 20.3 \text{ ksi})$ 

Therefore, the criteria is met.

The exterior of the canister is at atmospheric pressure at the time the pressure test is conducted, and no external pressure is applied to the canister. Consequently, the evaluation of NB-3133 is not required.

Finally, since the test pressure is equal to the 1.25 times the design pressure, no additional stress calculations are required in accordance with NB-3226, Subparagraph e.

#### 3.4.4.1.8 Fuel Basket Support Disk Evaluation

The response of the fuel basket support disks to storage and handling conditions was evaluated using an ANSYS finite element model that represented a one-quarter section of a single support disk. These loads consist of dead load, handling, and thermal. During storage (dead load) and handling, each support disk supports its own weight and is supported at eight locations by the tierod spacers (represented as nodal point restraints in the model). Since all of the support disks experience the same loading conditions during storage and handling, only one support disk was modeled. The support disk model, shown in Figure 3.4.4.1-5 with boundary conditions, was constructed of ANSYS SHELL63 three-dimensional, six degree-of-freedom, elastic shell elements.

The structural analyses of the ANSYS support disk model were performed with temperatures that envelope those experienced by the support disk during storage and handling conditions (100°F and -40°F ambient temperatures). Prior to performing the structural analyses, the steady-state temperature distribution in the support disk model was determined using temperature information from the storage and transfer thermal analyses. This was accomplished by converting the SHELL63 structural elements to SHELL57 thermal elements. The maximum support disk temperature (450°F) was applied to the nodes at the center slot and the minimum support disk temperature (100°F) was applied to the nodes around the outer circumferential edge. All other nodal temperatures were obtained by a steady state conduction solution.

The structural analyses were performed using the SHELL63 structural elements. Since the model represents a one-quarter section of the support disk, in-plane translations and rotations were restrained at the two symmetry faces. Two nodes at the locations of the tie-rod spacers were restrained in the axial direction. The dead load stresses were then calculated by applying a 1.1g acceleration to the entire model in the axial direction, and the handling stresses were calculated by applying a 1.1g acceleration to the entire model in the axial direction. Thermal stresses were also evaluated in addition to both dead load and handling load. The results of the support disk structural analyses for dead load, handling load, and thermal load are presented in Table 3.4.4.1-11. As shown in Table 3.4.4.1-11, the support disk maintains positive margins of safety for the conditions analyzed.

#### NAC-MPC FSAR Docket No. 72-1025

#### 3.4.4.1.9 Fuel Basket Weldments Evaluation

The response of the fuel basket top and bottom weldments to storage and handling conditions was analyzed using ANSYS finite element models representing one-quarter section of a top and a bottom weldment. These loads consist of the dead weight and handling loads and thermal expansion. During storage (dead load) and handling, the top weldment plate supports its own weight, the weight of eight structural ribs, and the weight of a circumferential ring, which is welded to the plate. The top weldment plate is supported at eight locations by the tie-rod spacers (represented as nodal point restraints in the model). During storage (dead load) and handling, the bottom weldment plate supports its own weight plus the weight of 36 fuel tubes applied as sets of nodal forces around the slot locations. The bottom weldment plate is supported at eight locations by the tie-rod spacers and at 12 locations by structural ribs (represented as nodal point restraints in the model). The top and bottom weldments are both constructed of SA240, Type 304 stainless steel. The top and bottom weldments model, shown in Figures 3.4.4.1-6 and 3.4.4.1-7, respectively, with boundary conditions, were constructed of ANSYS SHELL63 three-dimensional, six degree-of-freedom, elastic shell elements.

The structural analyses of the ANSYS weldment models were performed using the methodology described in Section 3.4.4.1.8. Temperatures employed for the thermal conduction analysis of the weldments are shown below:

Weldment	Maximum Temperature (°F)	Minimum Temperature (°F)				
	(at center)	(at circumference)				
Тор	400	380				
Bottom	150	100				

Since the ANSYS finite element models represent a one-quarter section of each weldment, inplane translations and rotations were restrained at the plane of symmetry face. In each weldment model, two nodes at the locations of the tie-rod spacers were restrained in the axial direction. In addition, for the bottom weldment model, two nodes at the location of the support pads were restrained in the axial direction. The dead load stresses were then calculated by applying a 1g acceleration to the entire model in the axial direction, and the handling stresses were calculated by applying a 1.1g acceleration to the entire model in the axial direction. Thermal stresses were

NAC-MPC FSAR		April 2000
Docket No. 72-1025		Revision 0

analyses for dead load, handling load, and thermal load are presented in Table 3.4.4.1-11. To account for the hottest temperatures experienced by the weldments during storage and handling, the allowable stresses are taken at 658°F for the top weldment and 662°F for the bottom weldment. As shown in Table 3.4.4.1-11, the weldments maintain positive margins of safety for the combined load conditions.

#### 3.4.4.1.10 <u>Fuel Tube Analysis</u>

The fuel tube provides a sealed cavity to mount BORAL poison plates within the fuel basket structure but the fuel tube does not provide structural support of the fuel assembly. The fuel tube design is presented in Figure 3.4.4.1-8. The thickness of the tube wall is 0.048 inch. A structural evaluation of the tube has been performed for the dead load and handling load conditions. The thermal stress is considered to be negligible since the tube is free to expand in both axial and radial directions. The handling load is considered to be 10% of the dead load.

During storage, the fuel assemblies are in contact with the bottom weldment, which is supported by the canister bottom plate. In the vertical position, the fuel assembly load is not carried by the fuel tubes. The fuel tubes are supported by the bottom weldment. Therefore, evaluation of the fuel tube is performed considering the weight of the fuel tube, with a g-load of 1.1 (to account for both the dead load and handling load) carried by the tube cross-section. From the dimensions of the tube shown in Figure 3.4.4.1-8, the cross sectional area is:

Area =  $(7.8 + 2 \times 0.048)^2 - 7.8^2$ = 1.507 in²

The weight of a fuel tube, including the BORAL plates, is 78 pounds. Considering a g-load of 1.1, the maximum compressive and bearing stress in the fuel tube is 57 psi (78 x 1.1 / 1.507). Limiting the compressive stress level in the tube to the material yield strength ensures the tube remains in position in storage conditions. The yield strength of Type 304 stainless steel is 17,300 psi at a conservatively high temperature of 750°F.

Margin of Safety = 17,300/57 - 1

= + Large

### Figure 3.4.4.1-1 Canister ANSYS Finite Element Model



NAC-MPC FSAR Docket No. 72-1025			April 2000 Revision 0
	<u></u>		. 1 1

Figure 3.4.4.1-2 Weld Regions of Canister ANSYS Finite Element Model at Structural and Shield Lids



April 2000 Revision 0

## Figure 3.4.4.1-3 Bottom Plate of the Canister ANSYS Finite Element Model



Figure 3.4.4.1-4 Locations for Section Stresses in the Canister ANSYS Finite Element Model



### Figure 3.4.4.1-5 Fuel Basket Support Disk ANSYS Finite Element Model



NAC-MPC FSAR		April 2000
Docket No. 72-1025	 	Revision 0

### Figure 3.4.4.1-6 Fuel Basket Top Weldment ANSYS Finite Element Model



### Figure 3.4.4.1-7 Fuel Basket Bottom Weldment ANSYS Finite Element Model



April 2000 Revision 0

## Figure 3.4.4.1-8 Fuel Tube Configuration



Location No. ¹	SX	SY	sz	SXY	SYZ	SXZ	Stress Intensity
1	0.2	0.9	3.2	0.1	< 0.1	-0.2	3.08
2	-0.3	-1.8	2.0	-0.1	< 0.1	0.2	3.73
3	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	0.04
4	-0.1	0.1	0.3	< 0.1	< 0.1	< 0.1	0.34
5	< 0.1	-0.1	0.1	< 0.1	< 0.1	< 0.1	0.26
6	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	0.04
7	< 0.1	0.5	-0.8	-0.1	< 0.1	-0.1	1.37
8	1.2	-2.7	-0.7	1.1	-0.1	-0.1	4.43
9	-2.0	9.7	2.2	0.4	< 0.1	0.3	11.78
10	2.5	-10.7	-1.6	0.7	-0.1	-0.3	13.29
11	-4.7	-5.1	-2.7	-0.7	< 0.1	0.1	2.90
12	-4.3	1.6	< 0.1	-0.8	-0.1	-0.3	6.12
13	-19.5	-6.2	-18.8	< 0.1	-0.6	< 0.1	13.35
14	2.4	4.0	2.5	< 0.1	-0.2	< 0.1	1.64
15	-7.1	-5.0	-6.9	< 0.1	0.3	< 0.1	2.17

#### Table 3.4.4.1-1

1

Summary of Maximum Canister Thermal Stresses (ksi)

# Table 3.4.4.1-2Summary of Maximum Canister Dead Load Primary Membrane (Pm)Stresses (ksi)

Location No. ¹	SX	SY	SZ	SXY	SYZ	SXZ	Stress Intensity
1	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	0.04
2	< 0.1	-0.1	< 0.1	< 0.1	< 0.1	< 0.1	0.09
3	< 0.1	-0.1	< 0.1	< 0.1	< 0.1	< 0.1	0.09
4	< 0.1	-0.1	< 0.1	< 0.1	< 0.1	< 0.1	0.09
5	< 0.1	-0.1	< 0.1	< 0.1	< 0.1	< 0.1	0.08
6	< 0.1	-0.1	< 0.1	< 0.1	< 0.1	< 0.1	0.07
7	< 0.1	-0.1	< 0.1	< 0.1	< 0.1	< 0.1	0.05
8	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	0.03
9	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	0.08
10	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	0.07
11	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	0.04
12	< 0.1	-0.1	< 0.1	< 0.1	< 0.1	< 0.1	0.06
13	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	0.01
14	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	0.02
15	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	0.01

## Table 3.4.4.1-3Summary of Maximum Canister Dead Load Primary Membrane Plus<br/>Primary Bending (Pm + Pb) Stresses (ksi)

Location No. ¹	sx	SY	SZ	SXY	SYZ	SXZ	Stress Intensity
1	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	0.06
2	< 0.1	-0.1	< 0.1	< 0.1	< 0.1	< 0.1	0.12
3	< 0.1	-0.1	< 0.1	< 0.1	< 0.1	< 0.1	0.09
4	< 0.1	-0.1	< 0.1	< 0.1	< 0.1	< 0.1	0.09
5	< 0.1	-0.1	< 0.1	< 0.1	< 0.1	< 0.1	0.08
6	< 0.1	-0.1	< 0.1	< 0.1	< 0.1	< 0.1	0.07
7	< 0.1	-0.1	< 0.1	< 0.1	< 0.1	< 0.1	0.07
8	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	0.05
9	< 0.1	-0.2	-0.1	< 0.1	< 0.1	< 0.1	0.20
10	< 0.1	0.1	< 0.1	< 0.1	< 0.1	< 0.1	0.18
11	0.1	0.1	< 0.1	< 0.1	< 0.1	< 0.1	0.08
12	0.1	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	0.09
13	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	0.01
14	0.1	< 0.1	0.1	< 0.1	< 0.1	< 0.1	0.09
15	-0.1	< 0.1	-0.1	< 0.1	< 0.1	< 0.1	0.06

# Table 3.4.4.1-4Summary of Maximum Canister Internal Pressure Load Primary<br/>Membrane (Pm) Stresses (ksi)

Location No. ¹	SX	SY	SZ	SXY	SYZ	SXZ	Stress Intensity
1	-0.6	3.0	0.9	0.7	0.3	-0.1	3.92
2	1.6	-1.2	-1.0	0.6	0.1	0.1	3.02
3	< 0.1	0.3	0.6	< 0.1	< 0.1	< 0.1	0.62
4	< 0.1	0.3	0.6	< 0.1	< 0.1	< 0.1	0.64
5	< 0.1	0.3	0.6	< 0.1	< 0.1	< 0.1	0.64
6	< 0.1	0.3	0.6	< 0.1	< 0.1	< 0.1	0.64
7	< 0.1	0.3	0.3	< 0.1	< 0.1	< 0.1	0.33
8	0.1	0.2	0.2	0.1	< 0.1	< 0.1	0.18
9	-0.2	0.2	0.1	< 0.1	< 0.1	< 0.1	0.41
10	0.2	-0.1	0.1	< 0.1	< 0.1	< 0.1	0.35
11	< 0.1	-0.1	0.1	< 0.1	< 0.1	< 0.1	0.20
12	< 0.1	0.3	0.2	< 0.1	< 0.1	< 0.1	0.31
13	0.7	< 0.1	0.7	-0.7	-2.2	< 0.1	4.60
14	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	0.06
15	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	0.07

#### NAC-MPC FSAR Docket No. 72-1025

Location No. ¹	SX	SY	SZ	SXY	SYZ	SXZ	Stress Intensity
1	-4.2	0.3	0.6	1.0	0.4	-0.2	5.29
2	0.7	-9.2	-3.4	0.7	< 0.1	0.3	10.02
3	< 0.1	0.4	0.6	< 0.1	< 0.1	-0.1	0.66
4	< 0.1	0.3	0.6	< 0.1	< 0.1	< 0.1	0.65
5	< 0.1	0.3	0.6	< 0.1	< 0.1	< 0.1	0.65
6	< 0.1	0.3	0.6	< 0.1	< 0.1	< 0.1	0.65
7	< 0.1	0.4	0.3	< 0.1	< 0.1	< 0.1	0.39
8	0.1	0.1	0.2	-0.1	< 0.1	< 0.1	0.21
9	-0.1	0.9	0.4	< 0.1	< 0.1	< 0.1	1.03
10	0.2	-0.7	-0.1	< 0.1	< 0.1	< 0.1	0.92
11	-0.3	-0.4	< 0.1	0.1	< 0.1	< 0.1	0.42
12	-0.3	0.1	0.1	-0.1	< 0.1	< 0.1	0.47
13	10.3	1.8	10.0	-0.7	-2.2	< 0.1	9.40
14	-0.6	-0.1	-0.5	< 0.1	< 0.1	< 0.1	0.44
15	0.3	< 0.1	0.3	< 0.1	< 0.1	< 0.1	0.31

## Table 3.4.4.1-5Summary of Maximum Canister Internal Pressure Load Primary MembranePlus Primary Bending (Pm + Pb) Stresses (ksi)

# Table 3.4.4.1-6Summary of Maximum Canister Dead Load + Handling Load Primary<br/>Membrane (Pm) Stresses (ksi)

Location No. ¹	SX	SY	sz	SXY	SYZ	SXZ	Stress Intensity
1	-0.7	3.1	0.9	-0.7	0.3	0.1	4.11
2	1.6	-1.3	-1.1	-0.6	0.1	-0.2	3.15
3	< 0.1	0.4	< 0.1	< 0.1	< 0.1	< 0.1	0.41
4	< 0.1	0.4	< 0.1	< 0.1	< 0.1	< 0.1	0.45
5	< 0.1	0.5	< 0.1	< 0.1	< 0.1	< 0.1	0.52
6	< 0.1	0.6	< 0.1	< 0.1	< 0.1	< 0.1	0.64
7	< 0.1	0.9	< 0.1	< 0.1	0.1	< 0.1	0.94
8	< 0.1	0.9	0.2	< 0.1	0.1	< 0.1	0.90
9	-0.2	1.2	0.3	< 0.1	0.1	< 0.1	1.34
10	-0.3	0.6	0.6	-0.3	0.1	0.1	1.08
11	-0.1	0.8	0.2	< 0.1	0.1	< 0.1	0.90
12	0.2	< 0.1	0.8	-0.3	< 0.1	0.1	1.07
13	0.7	< 0.1	0.7	-0.7	-2.3	< 0.1	4.86
14	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	0.03
15	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	0.05

Table 3.4.4.1-7	Summary of Maximum Canister Dead Load + Handling Load Primary
	Membrane Plus Primary Bending $(P_m + P_b)$ Stresses (ksi)

Location No. ¹	SX	SY	SZ	SXY	SYZ	SXZ	Stress Intensity
1	-4.3	0.3	0.7	-1.0	0.4	0.2	5.50
2	0.7	-9.7	-3.6	-0.8	< 0.1	-0.3	10.51
3	< 0.1	0.5	-0.1	< 0.1	< 0.1	< 0.1	0.53
4	< 0.1	0.4	-0.1	< 0.1	< 0.1	< 0.1	0.52
5	< 0.1	0.5	-0.1	< 0.1	< 0.1	< 0.1	0.62
6	< 0.1	0.6	-0.2	< 0.1	< 0.1	< 0.1	0.74
7	< 0.1	1.0	< 0.1	< 0.1	< 0.1	< 0.1	1.01
8	< 0.1	1.1	0.3	0.1	0.1	< 0.1	1.08
9	0.2	1.5	-0.1	< 0.1	-0.1	-0.3	1.73
10	-0.4	0.9	0.6	-0.5	0.1	0.1	1.70
11	0.2	1.2	-0.1	< 0.1	< 0.1	-0.2	1.37
12	0.7	-0.4	1.0	-0.1	-0.1	0.2	1.56
13	10.9	1.9	10.5	-0.7	-2.3	< 0.1	9.91
14	-0.2	< 0.1	-0.2	< 0.1	< 0.1	< 0.1	0.18
15	0.2	< 0.1	0.2	< 0.1	< 0.1	< 0.1	0.18

April 2000 Revision 0

## Table 3.4.4.1-8Summary of Maximum Canister Combined Load Primary Membrane (Pm)Stresses (ksi)

Location No. ¹	SX	SY	SZ	SXY	SYZ	SXZ	Stress Intensity	Allowable Stress ²	Margin of Safety
1	-1.3	6.1	1.9	-1.4	0.6	0.2	8.02	16.70	1.08
2	3.2	-2.5	-2.1	-1.2	0.2	-0.3	6.16	16.70	1.71
3	< 0.1	0.7	0.6	< 0.1	< 0.1	< 0.1	0.71	14.50	19.42
4	< 0.1	0.8	0.6	< 0.1	< 0.1	< 0.1	0.76	14.50	18.08
5	< 0.1	0.8	0.6	< 0.1	< 0.1	< 0.1	0.84	14.50	16.26
6	< 0.1	0.9	0.6	< 0.1	< 0.1	0.1	0.94	14.50	14.43
7	< 0.1	1.2	0.3	< 0.1	0.1	< 0.1	1.23	16.70	12.63
8	0.1	1.1	0.4	0.1	0.1	< 0.1	1.06	16.70	14.77
9	-0.3	1.3	0.4	< 0.1	0.1	0.1	1.67	16.70	9.00
10	< 0.1	0.5	0.8	-0.2	< 0.1	0.1	0.86	16.70	18.31
11	-0.1	0.9	0.3	< 0.1	0.1	0.1	0.98	16.70	15.97
12	0.1	0.3	1.1	-0.3	< 0.1	0.1	1.19	10.69 ⁴	7.98
13	1.4	0.1	1.3	-1.5	-4.5	< 0.1	9.47	16.70	0.76
14 ³	-0.1	< 0.1	-0.1	< 0.1	< 0.1	< 0.1	0.09	20.00	221.22
15	< 0.1	< 0.1	< 0.1	< 0.1	0.1	< 0.1	0.14	16.70	121.34

1 See Figure 3.4.4.1-4 for definition of locations of stress sections.

2 Allowable stresses for bottom plate region (location no. 1-2, 13) taken at 250°F; allowable stresses for canister shell region between shield lid and bottom plate (location no. 3-6) taken at 550°F; allowable stresses for structural/shield lid region (location no. 7-12, 14-15) taken at 250°F.

3 The allowable stress for SA240, Type 304 stainless steel was used for location No. 14. The allowable stress for SA240, Type 304L stainless steel was used for all other locations.

4 Includes two stress reduction factors for weld =  $0.8 \times 0.8 = 0.64$  (See Section 3.6.1).

April 2000 Revision 0

## Table 3.4.4.1-9Summary of Maximum Canister Combined Load Primary Membrane Plus<br/>Primary Bending (Pm + Pb) Stresses (ksi)

Location No. ¹	SX	SY	sz	SXY	SYZ	SXZ	Stress Intensity	Allowable Stress ²	Margin of Safety
1	-8.5	0.7	1.3	-2.0	0.8	0.4	10.77	25.05	1.33
2	1.4	-18.8	-7.0	-1.5	< 0.1	-0.6	20.50	25.05	0.22
3	< 0.1	0.9	0.6	< 0.1	< 0.1	< 0.1	0.92	21.75	22.64
4	< 0.1	0.8	0.7	< 0.1	< 0.1	0.1	0.79	21.75	26.53
5	< 0.1	0.9	0.8	< 0.1	< 0.1	0.1	0.88	21.75	23.72
6	< 0.1	1.0	0.8	< 0.1	< 0.1	0.1	1.00	21.75	20.75
7	< 0.1	1.3	0.3	< 0.1	< 0.1	< 0.1	1.31	25.05	18.14
8	0.1	1.2	0.4	0.1	0.1	< 0.1	1.17	25.05	20.45
9	-0.4	1.4	0.4	< 0.1	0.1	0.1	1.85	25.05	12.53
10	-0.1	1.7	1.1	-0.5	0.1	0.1	2.06	25.05	11.14
11	-0.4	1.0	0.3	< 0.1	0.1	0.1	1.45	25.05	16.26
12	0.3	-0.3	1.3	-0.1	-0.1	0.2	1.66	16.03 ⁴	8.66
13	21.2	3.7	20.5	-1.4	-4.5	< 0.1	19.33	25.05	0.30
14 ³	-0.5	-0.1	-0.5	< 0.1	< 0.1	< 0.1	0.43	30.00	68.77
15	0.8	0.1	0.8	< 0.1	0.1	< 0.1	0.67	25.05	36.12

1 See Figure 3.4.4.1-4 for definition of locations of stress sections.

2 Allowable stresses for bottom plate region (location no. 1-2, 13) taken at 250°F; allowable stresses for canister shell region between shield lid and bottom plate (location no. 3-6) taken at 550°F; allowable stresses for structural/shield lid region (location no. 7-12, 14-15) taken at 250°F.

3 The allowable stress for SA240, Type 304 stainless steel was used for location no. 14. The allowable stress for SA240, Type 304L stainless steel was used for all other locations.

4 Includes two stress reduction factors for weld =  $0.8 \times 0.8 = 0.64$  (See Section 3.6.1).

Location No. ¹	SX	SY	SZ	SXY	SYZ	SXZ	Stress Intensity	Allowable Stress ²	Margin of Safety
1	-9.1	0.6	3.6	2.1	0.8	-0.6	13.38	50.10	2.74
2	2.1	-22.9	-5.4	-1.5	0.1	-0.5	25.20	50.10	0.99
3	< 0.1	0.9	0.6	< 0.1	< 0.1	< 0.1	0.90	43.50	47.33
4	-0.1	0.7	1.0	< 0.1	< 0.1	-0.1	1.11	43.50	38.19
5	-0.1	0.7	0.9	< 0.1	< 0.1	-0.1	1.01	43.50	42.07
6	< 0.1	1.0	0.8	< 0.1	< 0.1	0.1	1.02	43.50	41.65
7	< 0.1	1.8	-0.6	-0.1	0.1	-0.1	2.43	50.10	19.59
8	1.2	-2.3	-0.5	-1.2	-0.1	0.1	4.22	50.10	10.86
9	-2.3	10.8	2.6	0.3	0.1	0.3	13.14	50.10	2.81
10	2.8	-12.4	-1.8	-0.8	-0.2	0.7	15.32	50.10	2.27
11	-5.0	-5.5	-2.8	0.9	< 0.1	-0.1	3.34	50.10	14.01
12	-4.9	1.8	< 0.1	-0.9	-0.1	-0.3	6.99	32.06 ⁴	3.59
13	-26.4	0.5	-25.2	-1.4	-4.0	< 0.1	27.74	50.10	0.81
14 ³	1.7	3.9	1.8	< 0.1	-0.1	0.1	2.21	60.00	26.15
15	1.7	3.9	1.8	< 0.1	-0.1	0.1	2.21	50.10	21.68

Table 3.4.4.1-10Summary of Maximum Canister Combined Load Primary Membrane PlusPrimary Bending Plus Secondary  $(P_m + P_b + Q)$  Stresses (ksi)

1 See Figure 3.4.4.1-4 for definition of locations of stress sections.

2 Allowable stresses for bottom plate region (location no. 1-2, 13) taken at 250°F; allowable stresses for canister shell region between shield lid and bottom plate (location no. 3-6) taken at 550°F; allowable stresses for structural/shield lid region (location no. 7-12, 14-15) taken at 250°F.

3 The allowable stress for SA240, Type 304 stainless steel was used for location no. 14. The allowable stress for SA240, Type 304L stainless steel was used for all other locations.

4 Includes two stress reduction factors for weld =  $0.8 \times 0.8 = 0.64$  (See Section 3.6.1).

# Table 3.4.4.1-11Summary of Maximum Stresses for the Fuel Basket Weldments and<br/>Support Disks

Component	Load	Stress I	ntensity	Allowal	Allowable Stress			
-	Condition	Reported	Value (psi)	Criteria	Value (psi)	Safety		
	Dead Load	Pm	0	Sm	16,299			
		$P_m + P_b$	3,297	1.5S _m	24,449	6.42		
Top	Dead Load +	$P_m + P_b + Q$	32,364	3.0S _m	48,898	0.51		
WELDMENT	Thermal							
	Dead Load +	Pm	0	S _m	16,299			
	Handling		2 (2)	1.50	24.440	5.74		
		$P_m + P_b$	3,626	1.55 _m	24,449	5.74		
	Dead Load +	$P_m + P_b + Q$	32,521	3.0S _m	48,898	0.50		
	Handling +							
	Dead Load	D			16 260			
	Dead Load		057	3m	24 402	27.47		
<b>D</b>	DUIT	$P_m + P_b$	637	1.55 _m	40 002	0.11		
BOTTOM WELDMENT	Dead Load + Thermal	$P_m + P_b + Q$	44,094	5.03 _m	40,000	0.11		
W ELDWENT	Dead Load +	P	0	S	16.269			
	Handling	- 10	-					
		$P_m + P_b$	942	1.5S _m	24,403	24.89		
	Dead Load +	$P_m + P_b + Q$	44,119	3.0S _m	48,806	0.11		
	Handling +							
	Thermal							
	Dead Load	P _m	0	S _m	41,528			
		$P_m + P_b$	870	1.5S _m	62,292	70.60		
SUPPORT	Dead Load +	$P_m + P_b + Q$	35,427	3.0S _m	124,584	2.52		
Disks	Thermal							
	Dead Load +	P _m	0	Sm	41,528			
	Handling							
		$P_m + P_b$	958	1.5S _m	62,292	64.02		
	Dead Load +	$P_m + P_b + Q$	35,495	3.0S _m	124,584	2.51		
	Handling +							
	Thermal							

#### 3.4.4.2 Vertical Concrete Storage Cask - Concrete Stress Analysis

ŝ

This section evaluates the stresses in the storage cask concrete for normal conditions of storage. The evaluation for the steel pedestal at the bottom of the cask is presented in Section 3.4.3.1. The stresses in the concrete due to dead load, live load, and thermal load are calculated below. The evaluations for off-normal and accident loading conditions are presented in Chapter 11. Summary of calculated stresses for the load combinations defined in Table 2.2-2 is presented in Table 3.4.4.2-1. The maximum stress in the concrete and the maximum force in the reinforcing bars and the comparison to their allowable limits are summarized in Table 3.4.4.2-2. As shown in Table 3.4.4.2-2, the storage cask meets the structural requirements of ACI-349-85.

#### 3.4.4.2.1 Dead Load

The dead load of the storage cask concrete is reacted by the lower concrete surface only. The concrete compression stress due to the self-weight of the storage cask is:

 $\sigma_v = -W/A = -21.44$  psi (compression)

where:

W = 151,364 lbs concrete cask dead weight D = 128 in. concrete exterior diameter ID = 86 in. concrete interior diameter A =  $\pi$  (D² - ID²)/4 = 7,059.2 in.²

Stress evaluation at the base of the concrete conservatively considers the weight of the empty concrete cask, rather than the concrete alone. The weight of the canister is not supported by the concrete.

#### 3.4.4.2.2 <u>Live Load</u>

The storage cask is subjected to two live loads: (1) the snow load and (2) the weight of the fully loaded transfer cask resting atop the storage cask. These loads are conservatively assumed to be applied to the concrete portion of the storage cask. No loads are assumed to be taken by the steel liner. The loads from the canister and its contents are transferred to the steel support inside the

#### NAC-MPC FSAR Docket No. 72-1025

storage cask and are not applied to the concrete. The stress in the steel support is evaluated in Section 3.4.3.1. Under these conditions, the only stress component is the vertical compression stress.

#### Snow Load

The snow load on the storage cask is determined in accordance with ANSI/ASCE 7-93 as follows:

The uniformly distributed snow load on the top of the storage cask, P_f, is:

$$P_f = 0.70 C_e C_t I P_g = 100.8 lbf/ft^2$$
 (Section 2.2.4)

The storage cask top area:

$$A_{top} = \pi (D/2)^2 = 12,868 \text{ in.}^2 = 89.36 \text{ ft}^2$$

The maximum snow load, F_s, is:

$$F_s = P_f \times A_{top} = 100.8 (89.36) = 9,007 \text{ lbf.}$$

The snow load is uniformly distributed over the top surface of the concrete.

The live load of the transfer cask is 135,473 lbs, which is much greater than the weight of the snow. Consequently, the stress due to the snow load is bounded by the weight of the transfer cask.

W	= 135,473 lbs-transfer cask weight (fully loaded)
D	= 128 inconcrete exterior diameter
D	= 86 inconcrete interior diameter
Α	$=\pi (D^2 - ID^2)/4$
	$= 7,059.2 \text{ in.}^2$

Compression stress at the base of the concrete is:

$$\sigma_v = W/A = -19.2 \text{ psi (compressive)}$$

#### 3.4.4.2.3 <u>Thermal Load</u>

An axisymmetric finite element model consisting of ANSYS PLANE42 elements for the steel liner and concrete shell was developed to calculate the thermal stresses in the concrete (see Fig. 3.4.4.2-1). The nodes at the steel liner/concrete interface are coincident and are connected analytically by coupling the degrees of freedom. Overall shell height is 160 in., the inner radius of the 3.5-in. thick carbon steel liner is 39.5 in., and the outer concrete radius is 64.0 in. The model obtains first a thermal solution (temperatures) and then a structural solution (stresses).

The steady-state, two-dimensional heat transfer conduction solution uses the surface temperature boundary conditions as calculated by the thermal analysis for normal conditions as presented in Section 4.4.1.1. These temperatures were applied without load factor along the steel liner interior and concrete exterior. The coincident nodes located along the steel and concrete interface were coupled with the temperature degree of freedom.

After the thermal solution, the thermal model is converted to a structural model. The nodal temperatures developed from the heat transfer analysis become the thermal load boundary conditions for the structural model.

Analysis with these boundary conditions provides the magnitude of three stress states  $\sigma_r$ ,  $\sigma_y$ ,  $\sigma_{\theta}$ , which are denoted radial, vertical, and circumferential stresses, respectively (these designations correspond to the x, y, and z axes, respectively, in the model). Stress magnitudes are calculated at various points along the concrete wall mid-span to determine the critical bounding cross sections.

The radial stress  $(\sigma_r)$  varies through the concrete wall as shown in the following diagram.



The maximum interior stress of -142.2 psi is a bearing (compressive) stress due to thermal expansion of the steel liner.

Applying the ACI 349-85 load reduction factor, the allowable bearing stress on the concrete is:

 $\phi = 0.70$   $f_c' = 4,000 \text{ psi}$  $\sigma_{\text{bearing}} = \phi f_c' = (0.70) (4,000) = 2,800 \text{ psi}$ 

The maximum 75°F normal operating thermally induced stress of -142.2 psi, when factored by the 1.275 load factor (see Table 2.2-2 in Chapter 2), represents a peak potential stress of -181.3 psi at the inner concrete shell surface. As shown in the diagram above, the radial compressive stress decreases through the wall thickness. This stress is considered to be insignificant.

Vertical membrane and bending stress varies through the concrete wall as shown in the diagram.


A linear equation describes the stress as a function of wall thickness.

 $\sigma_v$  = 46.98 t - 525.4, where t = r - 43 and r is the radius from the centerline of the storage cask to the external surface of the concrete. Substituting for t:

$$= 46.98r - 2,545.54$$

Integration of vertical tensile stress over the area in the plane r-  $\theta$  gives the tensile loads acting in the vertical direction.

 $F_v' = \int \sigma_v \, da = \int \int (46.98 \, r - 2,545.54) \, r \, dr \, d\theta$ = 863,706 lbf

56 outer vertical reinforcing bars are equally spaced at a 60.63 in. radius, which is close to the 60.73 in. radius tensile load center. The maximum tensile load applied to the reinforcing bar is:

 $F_{v \text{ applied}} = F_v' / 56 = 15,423$  lbf per vertical reinforcing bar.

Using a 1.275 load factor for normal operating loads:

$$1.275 F_{v \text{ applied}} = 1.275(15,423) = 19,664 \text{ lbf}$$

Calculating the allowable load for the reinforcing bar:

$$\sigma_{\theta \text{ allowable}} = U_c = \phi S_v = 0.90 (60) = 54 \text{ ksi}$$

where:

S_y = Reinforcing bar Yield Strength = 60 ksi
 φ = 0.9 for axial and bending tension loading (strength reduction factor in accordance with ACI 349-85, Section 9.2)

The reinforcing bar tensile load capacity is:

 $F_{\theta} = (\sigma_{\theta \text{ allowable}})(A_{\#6}) = (54,000) (.44) = 23,760 \text{ lbs}$ where,  $A_{\#6} = \#6$  reinforcing bar area = 0.44 in.²

The calculated load of 19,664 lbs is less than the allowable reinforcing bar load of 23,760 pounds. Therefore, the design is adequate with a margin of safety equal to:

M.S. =  $\frac{23,760}{19,664} - 1 = +0.21$ 

The circumferential membrane and bending stress varies through the concrete wall. The maximum values occur at 92.6 in. from the concrete cask lower surface as shown in the following diagram.



A linear equation describes the stress as a function of wall thickness:

$$\sigma_{\theta}$$
 = 30.457 t - 177.7

Integration of circumferential stress over the area in plane r-y gives the load acting in the circumferential ( $\theta$ ) direction. Integration of this stress over the concrete wall thickness provides the distributed load per unit height.

Circumferential tensile loads are found by integrating the stress function (using integration limits for wall thickness of 5.83 to 21 in).

$$F'_{t} = \int \sigma_{\theta} dt = \int (30.457 t - 177.7) dt$$
$$= (15.23 t^{2} - 177.7t) \Big|_{t=5.83}^{t=21}$$
$$= 3.503.5 \text{ lbs/in}$$

Outer hoop reinforcing bars are spaced on 4-in. centers at a 60.63 in. radius. The maximum circumferential tensile load acting on an outer hoop reinforcing bar in this spacing is:

$$F_{t\theta \text{ applied}} = F_t' \times 4 = 14,014 \text{ lbs}$$

Using a 1.275 load factor for normal operating loads

 $1.275 F_{t\theta applied} = 1.275(14,014) = 17,868 lbf$ 

The calculated load of 17,868 lbf is less than the reinforcing bar allowable of 23,760 lbf. Therefore, the design is adequate with a margin of safety equal to:

M.S. = 
$$\frac{23,760}{17,868} - 1 = +0.33$$

## Figure 3.4.4.2-1 Concrete Cask Axisymmetric Thermal Stress Model



## NAC-MPC FSAR Docket No. 72-1025

April 2000 Revision 0

Load	Load Stress Stress ² (psi)								
Comb ¹	Direction	Dead	Live	Wind ³	Thermal ⁴	Seismic ⁵	Tornado ⁶	Flood ⁷	Total
	<u> </u>	<u> </u>	CON	CRETE OU	TSIDE SURFA	CE:			
1	Vertical	-30.0	-32.6						-45.2
2	Vertical	-22.5	-24.5						-47.0
3	Vertical	-22.5	-24.5	-14.7					-61.7
4	Vertical	-21.4	-19.2						-40.6
5	Vertical	-21.4	-19.2			-42.9			-83.5
7	Vertical	-21.4	-19.2					-10.6	-51.2
8	Vertical	-21.4	-19.2				-11.5		-52.1
	I <u></u>	_1	Со	NCRETE IN	SIDE SURFAC	E:			
1	Vertical	-30.0	-32.6						-62.2
	Circumferential								
2	Vertical	-22.5	-24.5		-669.9				-716.9
	Circumferential				-226.6				-226.6
3	Vertical	-22.5	-24.5	-9.9	-669.9				-726.8
	Circumferential				-226.6				-226.6
4	Vertical	-21.4	-19.2		-660.1				-700.7
	Circumferential				-127.5				-127.5
5	Vertical	-21.4	-19.2		-525.4	-31.2			-597.2
	Circumferential				-177.7				-177.7
7·	Vertical	-21.4	-19.2		-525.4			-7.1	-573.1
	Circumferential				-177.7				-177.7
8	Vertical	-21.4	-19.2		-525.4		-7.7		-573.7
	Circumferential				-177.7				-177.7

## Table 3.4.4.2-1 Stress Summary for Concrete Cask Load Combinations

¹ Load Combinations are defined in Table 2.2-2. See Section 11.2.11 and 11.2.12 for Evaluations of Drop/Impact Conditions for Load combination No. 6.

² Positive stress values indicate tensile stresses and negative values indicate compressive stresses.

³ Stress results from Section 11.2.13 (Tornado) are conservatively used with a load factor of 1.275.

⁴ Tensile stresses (at concrete outside surface) are taken by the steel reinforcing bars and therefore are not shown in this Table. Stress Results for T_a (Load Comb. #4) are obtained from Section 11.2.10.

⁵ Stress results are obtained from Section 11.2.2.

⁶ Stress results are obtained from Section 11.2.13 (Tornado Wind).

⁷ Stress results are obtained from Section 11.2.6.

## Table 3.4.4.2-2 Maximum Concrete Stress and Reinforcing Bar Forces

	Calculated	Allowable ¹	Margin of Safety
Concrete	727 psi	2,800 psi	+2.85
Reinforcing Bar			
Normal - vertical	19,664 lbs	23,760 lbs	+0.21
- hoop	17,868 lbs	23,760 lbs	+0.33
Accident ² - vertical	19,380 lbs	23,760 lbs	+0.22
- hoop	23,196 lbs	23,760 lbs	+0.02

¹ Allowable stress for concrete is (0.7)(4,000 psi) = 2,800 psi, where 0.7 is the strength reduction factor per ACI 349-85, Section 9.3; 4,000 psi is the concrete strength.
 Allowable for Reinforcing Bar is determined based on No. 6 Reinforcing Bar as shown in the calculation in this Section.

² Results are obtained from Section 11.2.10.

3.4.5 <u>Cold</u>

Severe cold environments are analyzed and reported in Section 11.1.4. As shown in that section, the temperature of the structures with a full heat load will not fall to levels where brittle fracture would become an issue. Furthermore, an analysis has been performed for the cask in severe cold conditions after 50 years of storage. The required material toughness is 12.6 ft-lbs at -30°F. For conservatism 15 ft-lbs at -50°F is stated in the fabrication specification so that the NAC-MPC system can be handled, even during extreme temperatures.

THIS PAGE INTENTIONALLY LEFT BLANK

## 3.5 Fuel Rods

The NAC-MPC system is designed to limit fuel cladding temperatures to levels below those where zircaloy degradation is expected to lead to fuel clad failure. As shown in Chapter 4, fuel cladding temperature limits have been established to be 380°C for 5-year cooled fuel and 340°C for 10-year cooled fuel for normal conditions of storage and 570°C for short term off-normal and accident conditions. As shown in Table 4.1-3, the calculated maximum fuel cladding temperatures are well below the temperature limits for all design conditions of storage.

THIS PAGE INTENTIONALLY LEFT BLANK

## 3.6 Canister Closure Weld Evaluation – Normal Conditions

### 3.6.1 Stress Evaluation for the Canister Closure Weld

The closure weld for the canister is a partial penetration weld with a thickness of 0.9 inches. The evaluation of this weld, in accordance with NRC guidance, is to incorporate two separate weld stress reduction factors: a 0.8 factor per NRC ISG-4, Item 5, and an additional 0.8 factor to provide conservative consideration of the weld configuration. These two stress reduction factors are incorporated by applying a factor of 0.64 (0.8 x 0.8) to the stress allowable for this weld.

The stresses for the canister are evaluated using sectional stresses as permitted by Subsection NB. The location of the section for the canister weld evaluation is shown in Figure 3.4.4.1-4 and corresponds to Section 12. The governing  $P_m$ ,  $P_m$ +  $P_b$ , and P + Q stress intensity for Section 12 and the associated allowables are listed in Table 3.4.4.1-8, Table 3.4.4.1-9, and Table 3.4.4.1-10, respectively. The factored allowables, incorporating a 0.64 stress reduction factor, and the resulting controlling Margin of Safeties are:

Stress Type	Analysis Stress Intensity (ksi)	0.64 x Allowable Stress (ksi)	Margin of Safety
P _m	1.19	10.69	7.98
$P_m + P_b$	1.66	16.03	8.66
P + Q	6.99	32.06	3.59

This confirms that the canister closure weld is acceptable for normal operation conditions.

### 3.6.2 Critical Flaw Size for the Canister Closure Weld

The closure weld for the canister is comprised of multiple weld beads using a compatible weld material for Type 304L stainless steel. An allowable (critical) flaw evaluation has been performed to determine the critical flaw size in the weld region. The result of the flaw evaluation is used to define the minimum flaw size, which must be identifiable in the nondestructive examination of the weld. Due to the inherent toughness associated with Type 304L stainless steel, a limit load analysis is used in conjunction with a J-integral/tearing modulus approach.

The safety margins used in this evaluation correspond to the stress limits contained in Section XI of the ASME Code.

One of the stress components used in the evaluation for the critical flaw size is the radial stress component in the weld region of the structural lid. For the normal operation condition, in accordance with ASME Code Section XI, a safety factor of 3 is required. For the purpose of identifying the stress for the flaw evaluation, the weld region corresponds to Section 12 Figure 3.4.4.1-4 is considered. The radial stress corresponds to SX in Tables 3.4.4.1-1 through 3.4.4.1-10 for the normal operating condition. The maximum reported radial tensile stress is 0.7 ksi. To perform the flaw evaluation, a 10 ksi stress is conservatively used, resulting in a significantly larger safety factor than the required safety factor of 3. Using 10 ksi as the basis for the evaluation, the critical flaw size is 0.52 inch for a flaw that extends 360 degrees around the circumference of the canister. Stress components for the circumferential and axial directions are also reported in the above tables, which would be associated with flaws oriented in the radial or horizontal directions, respectively. The maximum stress reported for these components is 1.8 ksi, which is also enveloped by the stress value of 10 ksi for radial stresses. The 360-degree flaw employed for the circumferential direction is considered to be bounding with respect to any partial flaw in the weld, which could occur in the radial and horizontal directions. Therefore, using a minimum detectable flaw size of 0.375 inch is acceptable, since it is less than the very conservatively determined 0.52-inch critical flaw size.

## 3.7 <u>References</u>

ACI-349-85, Code Requirements for Nuclear Safety Related Concrete Structures, American Concrete Institute, March 1986.

ANSI N 14.6-1993, American National Standard for Radioactive Material - Special Lifting Devices for Shipping Containers Weighing 10,000 Pounds (4,500 kg) or More, American National Standards Institute, Inc., 1993.

ARMCO 17-4 PH Stainless Steel, Product Data Bulletin No. S-22, Armco Inc., 1988.

ASME Boiler and Pressure Vessel Code, Section II, Materials, Part D, Properties, 1995.

ASME Boiler and Pressure Vessel Code, Section III, Rules for Construction of Nuclear Power Plant Components, Division 1, Subsection NB, 1995.

ASME Boiler and Pressure Vessel Code, Section III, Rules for Construction of Nuclear Power Plant Components, Division 1, Subsection NG, Article NG-3222.4, 1995.

ASME Boiler and Pressure Vessel Code, Section III, Rules for Construction of Nuclear Power Plant Components, Division 1, Subsection NF, 1995.

ASME Boiler and Pressure Vessel Code, Section III, Rules for Construction of Nuclear Power Plant Components, Division 1, Appendices, 1995.

Baumeister and Marks, "Standard Handbook for Mechanical Engineers," 1966.

Fintel, M., Handbook of Concrete Engineering, 2nd Edition, Van Nosttrand Reinhold Co., New York.

Hanford Engineering Development Laboratory, Nuclear Systems Materials Handbook, Richland, WA, 1976.

Machinery's Handbook, 23rd Edition, 1990.

Manual of Steel Construction, Allowable Stress Design, Ninth Edition, American Institute of Steel Construction, Inc., 1 East Wacker Drive, Chicago, Illinois, 1991.

Military Handbook, MIL-HDBK-5G, November 1994.

NUREG-0612, Control of Heavy Loads at Nuclear Power Plants, U.S. Nuclear Regulatory Commission, 1980.

NUREG/CR-6322, "Buckling Analysis of Spent Fuel Basket," Lawrence Livermore National Laboratory, Livermore, California, May 1995.

Ross, R.B., Metallic Material Specification Handbook, 4th Edition, Ross Material Technology, Glasgow, Chapman And Hall, 1992.

TRW, Nelson Division, "Embedment Properties of Headed Studs," Design Data 10, 1975.

Young, W. C., Roark's formulas for Stress & Strain, 6th Edition, McGraw-Hill Book Company, New York, 1989.

NAC-MPC FSAR	April 2000
Docket No. 72-1025	 Revision 0
	<u> </u>

## 3.8 <u>Coating Specifications</u>

This Section presents the technical data sheets for Keeler & Long E-Series epoxy enamel and Ameron PSX 738. These coatings are applied to protect exposed carbon steel surfaces of the transfer cask and vertical concrete cask.

THIS PAGE INTENTIONALLY LEFT BLANK

____

#### 3.8.1

Keeler and Long E-Series Epoxy Enamel



#### INTRODUCTION

In the 1960's Keeler & Long made the commitment to develop Protective Coating Systems for Nuclear Power Plants. Coating Systems were developed and qualified in accordance with accepted standards, with emphasis upon their usage and specification for NEW construction projects. These systems were applied directly to either concrete or carbon steel substrates utilizing ideal surface preparation.

Presently, there is a necessity to apply these same coating systems or newly formulated systems over the original systems or over substrates which cannot be ideally prepared. Several years ago, Keeler & Long initiated a test program In order to test and qualify systems in conjunction with competitors products and/or with methods of preparation which are considered less than ideal. This test program provides OPERATING Nuclear Plants with qualified methods of preparation and a variety of qualified mixed coating systems.

#### HISTORY

In 1967, we embarked upon a testing program in order to comply with standards being prepared by the experts in the field and under the jurisdiction of The American National Standards Institute (ANSI). Earlier testing had involved research in order to determine the radiation tolerance and the decontamination properties of a variety of generic coating types including zinc rich, alkyds, chlorinated rubbers, vinyls, latex emulsions, and epoxies. This testing was conducted by various independent laboratories, such as Oak Ridge National Laboratory, Idaho Nuclear, and The Western New York Nuclear Research Center. It was concluded from these tests that almost any generic coating type would produce satisfactory radiation resistance and decontaminability.

Upon completion of the first ANSI Standards, however, it became evident that only Epoxy Coatings would meet the specific minimum acceptance criteria set forth in these standards. The single most important change from the earlier testing was the inclusion of a test which simulates the operation of the emergency core cooling system. This test is referred to as the Loss of Coolant Accident (LOCA) or the Design Basis Accident Condition (DBA). The test involves a high pressure, high temperature, alkaline, immersion environment. Upon completion of the first ANSI Standards, howeve

Simultaneous with the preparation of these standards, we prepared to test Epoxy Systems in order to comply with the requirements. First hand knowledge of these standards was available since our personnel assisted in the development of these documents. Equipment was designed and built by our laboratory in order to conduct in-house DBA tests. The required physical and chemical tests were either conducted by us or by unpersities through tresserb crants. us or by universities through research grants.

In 1972, the testing program was taken a step further in order

The Franklin Institute of to establish more credibility. to establish more creationiny. The Franklin institute of Philadelphia constructed an apparatus in order to simulate various Design Basis Accident Conditions and we prepared blocks and panels for an independent evaluation. The test results were among the "First" from an independent source, and these tests substantiated more than two years of in-house testing. testing

The Franklin Institute tests, along with our in-house testing program, were used as a basis for qualification until 1976. During this period also the following ANSI standards were revised and/or developed:

ANSI N5.9-1967 "Protective Coatings (Paints) for the Nuclear Industry" (Rev. ANSI N512-1974)

ANSI N101.2-1972 "Protective Coatings (Paints) for Light Water Nuclear Reactor Containment Facilities" Coatings

ANSI N101.4-1972 "Quality Assurance for Protective Coatings Applied to Nuclear Facilities*

Simultaneously, we developed a written Quality Assurance Program in compliance with ANSI N101.4 - 1972, Appendix B 10CFR50 of the Federal Register, and ANSI N45.2-1971 "Quality Assurance Program Requirements For Nuclear Power Plantet

In 1976, Oak Ridge national Laboratory (ORNL) established a testing program in order to conduct Rediation, Decontamination, and DBA tests under one roof. Keeler & Decontermination, and DBA tests under one roof. Keeler & Long, under contract with ORNL, conducted a series of tests in compliance with the parameters established by a major engineering firm and the ANSI standards. These tests, and similar series of tests conducted two years later in 1978, became the basis for the qualification of several of our concrete and carbon steel coating systems. From 1978 to the present day we have continued to qualify through ORNL and several other independent testing agencies any modifications to existing formulas and any changes in surface preparation or application requirements. We have also maintained an in-house testing program used to screen new products as well as modifications of existing systems. Furthermore, progress has continued in the revision of the ANSI standards during this time frame. Revision of these documents is presently under the continued in the revision of the Ansa staticates during this time frame. Revision of these documents is presently under the jurisdiction of the American Society for Testing and Materials (ASTM) as outlined in D3842-80 "Standard Guide for Selection of Test Methods for Coatings Used in Light-Water Nuclear Power Plants".

The future dictates significantly less construction of new Nuclear Plants and much more emphasis upon the repair and maintenance of existing facilities. Our commitment remains the same as it was in 1965; that is, to meet the coating requirements of Nuclear Power Plants.

SSU-1

# Level One Coating Systems

The following Coating Systems are qualified for Coating Service Level One of a Nuclear Power Plant. "Coating Service Level One pertains to those systems applied to structures, systems and other safety related components which are essential to the prevention of, or the mitigation of the consequences of postulated accidents that could cause undue risk to the health and safety of the public."

SYSTEM IDENTIFICATION	COATING SYSTEMS	DRY FILM THICKNESS RANGE
CARBON STEEL COATING SYSTEMS		
System 5-1		
Primer	No. 6548/7107 EPOXY WHITE PRIMER	3.0 - 14.0 mits DFI
Finish	No. E-1 SERIES EPOXY ENAMEL	2.5 - 6.0 milis DF I
System 8-10		E.O. 10.0 mile DET
Primer	No. 6548//10/ EPOXY WHITE PRIMEH	5.0 - 12.0 mills DF1
Finish	No. D-1 SEHIES EPOXY HI-BUILD ENAMEL	3.0 - 6.0 mils DF1
System S-11		
Primer/Finish	No. 6548/7107 EPOXY WHITE PRIMEH	8.0 - 16.0 mils Dr i
System 8-12		
Primer/Finish	No. 4500 EPOXY SELF-PRIMING SURFACING ENAMEL	5.0 · 16.0 mils DF1
System S-14 (FLOORS ONLY)		10.0 OF 0 mile DET
Finish	No. 5000 EPOXY SELF-LEVELING FLOOR COATING	10.0 - 25.0 mils Dr I
System S-15		
Primer	No. 6548/7107 EPOXY WHITE PHIMER	2.5 - 6.0 mils DFT
Finlah	No. 9600 N KEELOCK	5.0 + 8.0 mils DFT
CONCRETE COATING SYSTEMS		
System KL-2		
Curing Compound/Sealer	No. 4129 EPOXY CLEAR CURING COMPOUND	0.5 - 1.75 mils DFT
Surfacer	No. 6548-S EPOXY SURFACER	Flush - 50.0 mils DFT
Finish	No. E-1 SERIES EPOXY ENAMEL	2.5 - 6.0 mils DFT
System KL-8		
Curing Compound/Sealer	No. 4129 EPOXY CLEAR CURING COMPOUND	0.5 - 1.75 mils DFT
Surfacet	No. 6548-S EPOXY SURFACER	Flush - 50.0 mils DFT
Finish	No. D-1 SERIES EPOXY HI-BUILD ENAMEL	4.0 - 8.0 mils DFT
System XL-9		
Curing Compound/Sealer	No. 4129 EPOXY CLEAR CURING COMPOUND	0.5 - 1.75 mils DFT
Surfacer	No. 6548/7107 EPOXY WHITE PRIMER	5.0 - 10.0 mila DFT
Finish	No. D-1 SERIES EPOXY HI-BUILD ENAMEL	3.0 - 8.0 mils DFT
System KL-10		
Curing Compound/Sealer	No. 4129 EPOXY CLEAR CURING COMPOUND	0.5 - 1.75 mile DFT
Surfacer	No. 4000 EPOXY SURFACER	Flush - 50.0 mils DFT
Finish	No. D-1 SERIES EPOXY HI-BUILD ENAMEL	3.0 - 6.0 mils DFT
System KL-12		
Curing Compound/Sealer	No. 4129 EPOXY CLEAR CURING COMPOUND	0.5 - 1.75 mils DFT
Surfacer/Finish	No. 4500 EPOXY SELF-PRIMING SURFACING ENAMEL	10.0 - 50.0 mils DFT
System KL-14 (FLOORS ONLY)		
Primer/Sealer	No. 6129 EPOXY CLEAR PRIMER/SEALER	1.5 - 2.5 mils DFT

#### SUMMARY OF QUALIFICATION TEST RESULTS

KEELER & LONG maintains a complete file of Nuclear Test Reports which substantiate the specification of the carbon steel and concrete coating systems listed in this bulletin. This file was initiated in the early 1970's and provides complete qualification in accordance with ANSI Standards N512 and N101.2. Results for radiation tolerance, decontamination, and the Design Basis Accident Condition are reported as performed by independent Laboratories. Also reported are the chemical and physical tests which were conducted by the Keeler & Long Laboratory in compliance with the ANSI Standards.

KAL COATING			KEELI	H & LONG	TEST REPOR	IT NO.		
SYSTEM	SUBSTRATE	76-0728-1	78-0810-1	85-0404	85-0524	90-0227	93-0818	93-0601
\$-1	Steel	•	• 1					
\$-10	Steel							
· \$-11	Steel		•	•				
S-12	Steel	i		-		*		
5-14	Steel		· ·				•	1 1
KL-2	Concrete	•						1
KL-8	Concrete	•						
KL-9	Concrete	•						1
KL-10	Concrete		- 1		•			
KL-14	Concrete					•		•
This information is presented as	accurate and correct, in goo	d faith, to assist	the user in appl	ication. No wa	uranty is expres	used or Implied	I. No liability i	s assumed.
								A ALCO
	0712121	ותלו						_
	ורורוזיו	עיוריי						

NAC-MPC FSAR Docket No. 72-1025



NAC-MPC FSAR Docket No. 72-1025

E-SERIES		F 340
TEC	CHNICAL DA	
PHYSICAL DATA:	Weight per gallon: Flash Point (Pensky-Martens): Shelf Life: Pot Life @ 72°F: Temperature Resistance: Viscosity @ 77°F: Gloss (60° meter): Storage Temperature: Mixing Ratio (Approx. by Volume):	10.2 ± 0.5 (pounds) 85°F± 2° 1 Year 8 Hours 350°F 85 ± 5 (Krebs Units) 95 ± 5 (E-1) 55 - 95°F 4:1
APPLICATION DATA:	Application Procedure Guide: Wet Film Thickness Range: Dry Film Thickness Range: Temperature Range: Relative Humidity: Substrate Temperature: Minimum Surface Preparation: Induction Time @ 72°F: Recommended Solvent @ 50 - 85°F: @ 86 - 120°F:	APG-2 4.0 - 5.0 mils 2.0 - 2.5 mils 55 - 120°F 80% Maximum Dew Point + 5°F Primed 1 Hour No. 4093 No. 2200
	Application Methods	
	Air Spray Tip Size: Pressure: Thin:	.055" 30 - 60 PSIG 1.0 - 2.0 Pts/Gal
	Airless Spray Tip Size: Pressure: Thin:	.011"017" 2500 - 3000 PSIG 0.5 - 1.5 Pts/Gal
	Brush or Roller Thin:	1.0 - 2.0 Pts/Gal
()		
Koloks Tel:	P. O. Box 460, 856 Echo Lake Road Watertown, CT 06795 (860) 274-6701 Fax: (860) 274-5857	SSPC
This Information is p No warranty is exp notice. Data listed i	presented as accurate and correct, in good faith, to assist the user in aperased or implied. No liability is assumed. Product specifications are above is for white or base color of the product. Data for other colors may	scification and application. SUSTAINING MEMBER subject to change without r differ.

#### Ameron PSX 738 Siloxane Coating 3.8.2



PSX Advantage: PSX 738 is a patented siloxane product that can withstand twice as much heat as conventional heat-resistant coatings, without the need for heat curing before service.

## **Product Data**

- Protects stainless steel from chloride attack under insulation
- Heat resistance to 2000°F
- Self-priming, single-coat efficiency—significantly lower applied cost
- · Resistant to severe acid conditions and moisture
- Chemical resistant
- Excellent adhesion
- Suitable to protect steel under thermal insulation at high
- temperatures • Cures at room temperature, no bake required before service
- Applied direct to metal
- High solids and low VOC

PSX 738 is resistant to a wide range of mineral and organic acids, including glacial acetic acid. (See Chemical Resistance Guide - below).

#### **Typical Uses**

- Power plants
- Exhaust ducts and stacks
- Refineries
- · Exteriors of reactors
- Chemical facilities
- Under insulation, on equipment and pipe exteriors

### **PSX 738 Chemical Resistance Guide**

PSX 738 may be exposed to the following chemicals under splash and spillage conditions:

Chemical	Temperature
Salt water (3%)	up to 200°F
DI Water	սր <b>ւ</b> օ 200°F
Sodium silicate	Ambient
Ethviene glycol	Ambient
Acetone	Ambient
Methviene chloride	Ambient
Xviene	Ambient
Mineral spirits	Ambient
JP-4 fuel	up to 140°F
Petroleum ether	Ambient
Kerosene	up to 140°F
Transmission fluid	Ambient
Accuir acid (all concentrations)	Ambient
Tall oil fatty acid	up to 120°F
Sulfuric acid (50%)	Ambient
Nitric acid (5%)	Ambient
Hydrochloric acid(20%)	Ambient
Note: Not recommended for caustic solution the lining of mineral acid storage tanks.	is. PSX 738 is not recommended for

## **PSX® 738**

Engineered Siloxane® coating Patent No. 4, 113, 665

r.)_4

### **Physical Data**

Finish	Flat	
Color*	Deep gray (appr	ox ANSI-33).
	gray (approx Gr	{-3}
Components	2	
Curing mechanism	Chemical reacti	on and
	solvent release	
Volume solids	0.40/ . 20/	
(ASTM D2697 modified)	64%±3%	150 microne)
Dry film thickness per coat	4.0 mus (100.	130 1110 0113)
Refer to maximum DFT table for service	e temperature limii	ations.
Coats	1- for high heat.	over Dimetcore-
	and not suger	ive cervice
<b></b>	2- mgmy corros	m?/I
Theoretical coverage	1247	330
1 mil (25 microns)	135	3.3
TO HILS (250 HILL OILS)	lb/mel	σ/l.
VUC	0.8	<b>9</b> 6
mixed/thinned (1/2 pt/gal)	1.2	144
	•F	•C
continuous	2000	1093
over Dimetcote 9 or 21-9	750	400
Elach point (SETA)	۰F	°C
limid	55	13
Amercoat 935	208	98
Amercoat 101	145	63
Amercoat 15	105	- <u>91</u> -18
Amercoat 12	U	-10
A II alta - Dala		
Application Data		
Applied over	Prepared or pri	med steel.
	stainless steel t	ypes 304 & 310,
	concrete or Diff	lecote
Surface preparation		
types 304 & 316	Abrasive blast S	SSPC-SP10
concrete	ASTM D4258	
Dimetcote 9 or 21-9	Clean, dry surfa	100
Method	Conventional sp	pray
Mixing ratio (by weight)	1 part liquid to	1.9 parts
2 2 2	powder	
Pot life @ 70°F and 50% R.H.	5 hours	
Environmental conditions		
Temperature	"F	°C
air	45 to 120	/ 10 49
surface	See unmer rec	ommendations
Relative numidity during		
application and initial orying	40%	
manim	Avoid condens	ation
Surface temperatures must be at least !	5"F (3"C) above dew p	oint to prevent
condensation during application and in	itial dry through.	
*Note- color will lighten at high ter	nperatures	

Formerly Amercoat 738

#### Surface Preparation

Coating performance is, in general, proportional to the degree of surface preparation. Prior to coating all surfaces must be clean. dry, undamaged and free of all contaminants, including salt deposits.

#### **Application Data Summary**

See Application Instructions for complete information on surface preparation, equipment, environmental conditions and application procedures. For conditions outside the requirements or limitations described, contact your Ameron representative.

#### Maximum DFT (mils)

		No. of		Service Te			
	Thinner	Coats	2000/1093	1500/816	1000/538	750/399	500/260
Steel	15/101	1	8	10	10	10	15
	15/101	2	NR	10	10	14	20
	935	1	NR	8	8	12	15
Dimetcote	15/101	1		-	_	10	15
	935	NR	NR	NR	NR	NR	NR
NR≈ Not re	comment	ied					

Important: Exceeding maximum DFT may result in PSX 738 surface cracking

#### **Safety Precautions**

Read each component's material safety data sheet before use. Mixed material has hazards of each component. Safety precautions must be strictly followed during storage, handling and use

This product is for professional use only. Not for residential use in California.

#### Warranty

Ameron warrants its products to be free from defects in material and workmanship. Ameron's sole obligation and Buyer's exclusive remedy in connection with the products shall be limited, at Ameron's option, to either replacement of products not conforming to this Warranty or credit to Buyer's account in the invoiced amount of the nonconforming products. Any claim under this Warranty must be made by Buyer to Ameron in writing within five (5) days of Buyer's discovery of the claimed defect, but in no event later than the expiration of the centre of the bland defect. of the applicable shelf life, or one year from the delivery date, whichever is earlier. Buyer's failure to notify Ameron of such nonconformance as required herein shall bar Buyer from recovery under this Warranty.

Ameron makes no other warranties concerning the product. No other warranties, whether express, implied, or statutory, such as warranties of merchantability or fluenes for a particular purpose, shall apply, in no event shall Ameron be liable for consequential or incidental damages.

Any recommendation or suggestion relating to the use of the products made by Ameron, whether in its technical literature, or in response to specific inquiry, or otherwise, it is based on data believed to be reliable; however, the products and information are intended for use by Buyers having requisite skill and kniw-how in the industry, and therefore it is for Buyer to satisfy itself of the suitability of the products for its own particular use and it shall be demed that Buyer has done so, at its sole discretion and nsk. Variation in environment, changes in prordures of Use on extraolation of data may easis unpatified procession. use, or extrapolation of data may cause unsatisfactory results.

#### Limitation of Liability

Ameron's lability on any claim of any kind, including claims based upon Ameron's negligence or strict lability, for any loss or damage ansing out of. connected with, or resulting from the use of the products, shall in no case exceed the purchase price allocable to the products or part thereof which give rise to the claim. In no event shall Ameron be liable for consequential or incidental damages damayes.

Desing time** (ASTM D1	640)/h	nurs)	°F/	°C	
Diying and (abinibi	20/49	90/32	70/21	50/10	40/4
Touch @ 50% R.H. unthinned thinned w/15 or 101***	1/2 1/2	1 2	1 ¹ /2 3	2 4	3 6
Through @ 50% R.H. unthinned thinned w/15 or 101***	14 14	20 26	28 36	<b>48</b> 60	72 80.
Through @ 75% R.H. unthinned thinned w/15 or 101***	7 7	10 15	14 18	24 30	36 40
Recoat @ 50% R.H. unthinned thinned***	1 2	2 4	3 6	5 10	7 14
Recoat @ 75% R.H. unthinned thinned	1 1 ¹ /2	1 3	2 5	3 8	4 10
Cure** (days) Service temp. up to 1000'	'F-san	ie as Dry	/ Throu	gh time	
Service temp. above 1000 (thinned or unthinned) @ 50% R.H. @ 75% R.H.	9°F 4 2 ¹ /2	8 4	13 7	30 19	50 32
Chemical resistance (day @50% R.H.	ys)				
unthinned thinned	6 10	10 15	15 25	40 60	80 NR
@ 75% R.H. unthinned thinned	3	6 10	10 15	21 36	40 72

NR = Not Re

thinned***

"Note - at 15 mils total DFT Drying and Cure times will increase by 30 %. ***Note – Thinned material at 2 coats will increase cure times.

Thinners

Amercoat 15 (up to 100°F surface temperature) Amercoat 101 (up to 140°F surface temperature) Amercoat 935 (140° - 200°F surface temperature)

DO NOT use Amercoat 935 below 140°F surface temperatures.

Equipment cleaner Thinner or Amercoat 12

#### Shipping Data

1-gai	5-gal
lb	kg
	•
6.1	2.8
11.1	5.1
30.4	13.8
52.9	24.1
	6.1 11.1 30.4 52.9

liquid 8 months from manufacture date 2 years from shipment date powder Numerical values are subject to normal manufacturing tolerances, color and

testing variances. Allow for application losses and surface irregularitie The mixed product is nonphotochemically reactive as defined by the South Coast Air Quality Management District's Rule 102 or equivalent regulations.



Systems

738 005

201 North Berry Street, Brea, California 92622-1020 • J. F. Kennedylaan 7, 4191 MZ Geldermalsen, The Netherlands

©1983 Ameron + Printed in U.S.A. + R4/94 s.



## **PSX® 738**

Engineered Siloxane® coating

Patent No. 4, 113, 665

## **Application Instructions**

Refer to the Product Data sheet for properties and uses

Adhere to all application instructions, precautions, conditions, and limitations to obtain the maximum performance. For conditions outside the requirements or limitations described. contact your Ameron representative.

#### **Surface Preparation**

Coating performance is, in general, proportional to the degree of surface preparation. Prior to coating all surfaces must be clean, dry, undamaged and free of all contaminants, including salt deposits.

Steel or Stainless Steel Types 304 & 316 - Dry abrasive blast according to SSPC-SP10.

Concrete - Clean according to ASTM D4258.

Dimetcote® 9 or 21-9 - Clean dry surface. See Dimetcote 9 or 21-9 Application Instructions for complete information and safety precautions.

#### Application Equipment

The following is a guide: suitable equipment from other manufacturers may be used. Changes in pressure, hose and up size may be needed for proper spray characteristics.

Conventional spray - Industrial equipment such as DeVilbiss MBC or JGA spray gun. An oil and moisture trap in the main air supply line are essential. Separate regulators for air and fluid pressure and mechanical pot agitator are recommended.

#### **Environmental Conditions**

Temperature	°F	°C
air	45 to 120	7 to 49
surface	See thinner:	recommendations
Relative humidity during application and initial drying		
minimum	40%	
maximum	Avoid conde	nsation

Surface temperatures must be at least 5°F (3°C) above dew point to prevent condensation during application and initial dry through.

#### **Application Procedure**

1. Flush equipment with thinner or Amercoat 12 before use to remove moisture. Moisture can harden Amercoat 738 in spray equipment.

Slowly stir powder into liquid and continue to mix until 2 powder is well dispersed. Do not reverse order.

3. Keep mixture in covered container until ready to use, as moisture from the atmosphere can cause skinning or gelling. Do not mix more material than can be used within the expected pot life, 5 hours at 70°F and 50% R.H.

4. Agitate slowly and continuously during spray application to maintain uniform suspension.

5. Apply a wet coat in multiple passes, if necessary, to obtain desired dry film thickness. Maintain 18" maximum distance between spray gun and surface to avoid dry spray.

Maximum DFT (mils)

Maxim	011 D1 1	No of	No. of Service Temperature *F/*C					
	Thinner	Coats	2000/1093	1500/816	1000/538	750/349	500/260	
Steel	15/101	1	8	10	10	10	15	
	15/101	2	NR	10	10	14	20	
	935	1	NB	8	8	12	15	
Dimetcote	15/101	1		-	-	10	15	
	935	NR	NR	NR	NR	NH	NR	
NR= Not re	comment	ied						

Important: Exceeding maximum DFT may result in PSX 738 surface cracking Drving time* (ASTM D1640)(hours) °E/°C

bijing unic (Abitabi	120/40	00/32	70/21	50/10	40/4
Toursh @ 50% P U	120/47	90/32	10/21	30/10	10/1
ioucil @ 30% n.m.	1/2	1	11/2	2	3
this and w/15 or 101	· 1/-	2	12	4	ĕ
thinned w/ 15 or 101-	-'Z	2	5	7	U
Through @ 50% R.H.		•••	•••		-
unthinned	14	20	28	48	12
thinned w/15 or 101**	14	26	36	60	80
Through @ 75% R.H.					
unthinned	7	10	14	24	36
thinned w/15 or 101**	7	15	18	30	4
Personat @ 50% P U					
necoal @ 50 /s n.m.	1	2	2	5	7
	2	2	5	10	14
ininnea	2	4	U	10	14
Recoat @ 75% R.H.					
unthinned	1	1	2	3	4
thinned**	1 ¹ /2	3	5	8	10
Cure* (days)					
Service temp, up to 1000	°F – san	ne as Dr	y Throu	gh time	
Service temp. above 100	ሳ°F			0	
(thinned or unthinned)	01				
		Q	12	30	50
@ 30% R.H.	21/2	4	13	10	32
<i>штэ</i> % к.н.	2.72	-	1	17	26
Chemical resistance (da	iys)				
@50% R.H.					~~
unthinned	6	10	15	40	80
thinned**	10	15	25	60	NR

unthinned thinned** NR = Not Recommended

@75% R.H.

*Note - at 15 mils total DFT Drying and Cure times will increase by 30 %.
*Note - Thinned material at 2 coats will increase cure times.

6

10

3

6

10

15

21

36

40

72

Thin with recommended thinner as required for 6.

workability, see below: Thinners

Amercoat 15 (up to 100°F surface temperature)

Amercoat 101 (up to 140°F surface temperature) Amercoat 935 (140° - 200°F surface temperature)

DO NOT use Amercoat 935 below 140°F surface temperatures.

7. Clean all equipment with thinner or Amercoat 12 cleaner immediately after use.

Formerly Amercoat 738

#### Safety Precautions

Read each component's material safety data sheet before use. Mixed material has hazards of each component. Safety precautions must be strictly followed during storing, handling and use.

CAUTION – Improper use and handling of this product can be hazardous to health and cause fire or explosion.

Do not use this product without first taking all appropriate safety measures to prevent property damage and injuries. These measures may include, without limitation: implementation of proper ventilation, use of proper lamps, wearing of proper protective clothing and masks, tenting and proper separation of application areas. Consult your supervisor. Proper ventilation and protective measures must be provided during application and drying to keep spray mists and vapor concentrations within safe limits and to protect against toxic hazards. Necessary safety equipment must be used and ventilation requirements carefully observed, especially in confined or enclosed spaces, such as tank interiors and buildings.

This product is to be used by those knowledgeable about proper application methods. Ameron makes no recommendation about the types of safety measures that may need to be adopted because these depend on application environment and space. of which Ameron is unaware and over which it has no control.

If you do not fully understand these warnings and instructions or if you cannot strictly comply with them, do not use the product.

Note: Consult Code of Federal Regulations Title 29, Labor. parts 1910 and 1915 concerning occupational safety and health standards and regulations. as well as any other applicable federal, state and local regulations on safe practices in coating operations.

This product is for professional use only. Not for residential use in California.

#### Warranty

Ameron warrants its products to be free from defects in material and workmanship. Ameron's sole obligation and Buyer's exclusive remedy in connection with the products shall be limited, at Ameron's option, to either replacement of products not conforming to this Warranty or credit to Buyer's account in the invoiced amount of the nonconforming products. Any claim under this Warranty must be made by Buyer to Ameron in writing within five (5) days of Buyer's discovery of the claimed defect, but in no event later than the expiration of the applicable shelf life, or one year from the delivery date, whichever is earlier. Buyer's failure to notify Ameron of such nonconformance as required herein shall bar Buyer from recovery under this Warranty.

Ameron makes no other warranties concerning the product. No other warranties, whether express, implied, or statutory, such as warranties of merchantability or fitness for a particular purpose, shall apply. In no event shall Ameron be liable for consequential or incidental damages.

Any recommendation or suggestion relating to the use of the products made by Ameron, whether in its technical literature, or in response to specific inquiry, or otherwise, is based on data believed to be reliable: however, the products and information are intended for use by Buyers having requisite skill and knowhow in the industry, and therefore it is for Buyer to satisfy itself of the suitability of the products for its own particular use and it shall be deemed that Buyer has done so, at its sole discretion and risk. Variation in environment, changes in procedures of use, or extrapolation of data may cause unsatisfactory results.

#### Limitation of Liability

Ameron's liability on any claim of any kind, including claims based upon Ameron's negligence or strict liability, for any loss or damage arising out of, connected with, or resulting from the use of the products, shall in no case exceed the purchase price allocable to the products or part thereof which give rise to the claim. In no event shall Ameron be liable for consequential or incidental damages.



738.AJ

201 North Berry Street, Brea, California 92622-1020 + J. F. Kennedylaan 7, 4191 MZ Geldermalsen, The Netherlands

## **Table of Contents**

.

4.0	THE	RMAL E	VALUATION	4.1-1
4.1	Discu	ssion		4.1-1
4.2	Sumn	nary of Th	ermal Properties of Materials	4.2-1
4.3	Speci	fication of	f Components	4.3-1
4.4	Thern	nal Evalua	ation for Normal Conditions of Storage	4.4-1
	4.4.1	Thermal	Models	4.4-1
		4.4.1.1	Two-Dimensional Axisymmetric Air Flow and	
			Concrete Cask Model	4.4-2
		4.4.1.2	Three-Dimensional Canister Model	4.4-13
		4.4.1.3	Two-Dimensional Fuel Model	4.4-19
		4.4.1.4	Two-Dimensional Fuel Tube Model	4.4-22
		4.4.1.5	Three-Dimensional Transfer Cask and Canister Model	4.4-25
		4.4.1.6	Two-Dimensional Reconfigured Fuel Assembly Model	4.4-29
	4.4.2	Test Mo	del	4.4-32
	4.4.3	Maximu	m Temperatures for Normal Conditions	4.4-32
		4.4.3.1	Maximum Temperatures in Long-Term Storage	4.4-32
		4.4.3.2	Maximum Temperatures in Transient Operations	4.4-33
	4.4.4	Minimu	m Temperatures	4.4-44
	4.4.5	Maximu	m Internal Pressure for Normal Conditions	4.4-44
	4.4.6	Maximu	m Thermal Stresses for Normal Conditions	4.4-48
	4.4.7	Evaluati	on of Cask Performance for Normal Conditions of Storage	4.4-48
4.5	Refere	ences		4.5-1

-

## List of Figures

Figure 4.4.1.1-1	Two-Dimensional Axisymmetric Air Flow and Concrete Cask Model 4.4-8
Figure 4.4.1.1-2	Two-Dimensional Axisymmetric Air Flow and Concrete Cask
	Finite Element Model 4.4-9
Figure 4.4.1.1-3	Axial Power Distribution for Yankee Class Fuel 4.4-10
Figure 4.4.1.1-4	Convergence Process of Mass Flow Rate 4.4-11
Figure 4.4.1.2-1	Three-Dimensional Canister Model 4.4-17
Figure 4.4.1.2-2	Three-Dimensional Canister Model - Cross-Section
Figure 4.4.1.3-1	Two-Dimensional Fuel Model 4.4-21
Figure 4.4.1.4-1	Two-Dimensional Fuel Tube Model 4.4-24
Figure 4.4.1.5-1	Three-Dimensional Transfer Cask and Canister Model 4.4-28
Figure 4.4.1.6-1	Two-Dimensional Reconfigured Fuel Assembly Model 4.4-31
Figure 4.4.3-1	Temperature Distribution (°F) for Normal Storage
Figure 4.4.3-2	Air Flow Pattern in the Storage Cask in the Normal Storage
Figure 4.4.3-3	Air Temperature Field in Storage Cask During the Normal
	Storage Condition
Figure 4.4.3-4	Concrete Temperature Field During the Normal Storage Condition 4.4-38
Figure 4.4.3-5	History of Maximum Component Temperatures for the Nominal
	Transfer Conditions

NAC-MPC FSAR Docket No. 72-1025 . . . . . .

## List of Tables

Table 4.1-1	Summary of Thermal Design Conditions for Storage	4.1-2
Table 4.1-2	Summary of Thermal Design Conditions for Transfer	4.1-3
Table 4.1-3	Maximum Allowable Temperature Limits (°F)	4.1-4
Table 4.1-4	Summary of Thermal Evaluation for NAC-MPC Storage System	4.1-5
Table 4.2-1	Thermal Properties of Solid Neutron Shield (NS-4-FR)	4.2-2
Table 4.2-2	Thermal Properties of Stainless Steels	4.2-3
Table 4.2-3	Thermal Properties of Chemical Lead	4.2-4
Table 4.2-4	Thermal Properties of Type 6061-T6 Aluminum Alloy	4.2-5
Table 4.2-5	Thermal Properties of Helium	4.2-6
Table 4.2-6	Thermal Properties of Dry Air	4.2-7
Table 4.2-7	Thermal Properties of Concrete	4.2-8
Table 4.2-8	Thermal Properties of ASTM A 36, ASTM A 588 and ASTM A350	
	Carbon Steel	4.2-9
Table 4.2-9	Thermal Properties of Zircaloy and Zircaloy-4 Cladding	4.2-10
Table 4.2-10	Thermal Properties of Fuel (UO ₂ )	4.2-11
Table 4.2-11	Thermal Properties of BORAL Composite Sheet	4.2-12
Table 4.4.1.1-1	Comparison of Numerical Results Using Different Element Sizes	5
	and Number of Elements	4.4-12
Table 4.4.3-1	Maximum Component Temperatures for the Normal Condition	
	of Storage	4.4-40
Table 4.4.3-2	Maximum Component Temperatures for the Nominal	
	Helium Transfer Condition	4.4-41
Table 4.4.3-3	Maximum Component Temperatures for the	
	Vacuum Transfer Condition	4.4-42
Table 4.4.3-4	Maximum Component Temperatures for the	
	Reconfigured Fuel Assembly	4.4-43

THIS PAGE INTENTIONALLY LEFT BLANK

_

### 4.0 THERMAL EVALUATION

### 4.1 Discussion

This section presents the thermal analysis of the NAC-MPC system for normal conditions of storage. The significant thermal design feature of the NAC-MPC system is the passive convective air flow up along the side of the canister. Cool (ambient) air enters at the bottom of the storage cask through four inlet vents. Heated air exits through the four outlets at the top of the storage cask. Radiant heat transfer also occurs from the canister shell to the concrete cask liner. Consequently, the liner also heats the convective air flow. Conduction does not play a substantial role in heat removal from the canister surface. This natural circulation of air inside the vertical concrete cask (storage cask), in conjunction with radiation from the canister surface, maintains the fuel cladding temperature and all of the storage cask component temperatures below their design limits.

The thermal evaluation considers normal, off-normal, and accident conditions of storage. Each of these conditions can be described in terms of the environmental temperature, use of solar insolance, and the condition of the air inlet and outlet vents, as shown in Table 4.1-1. The design conditions for transfer are defined in Table 4.1-2.

This evaluation applies different component temperature limits and different material stress limits for long-term (steady-state) conditions and for short-term (transient) conditions. Normal storage is considered to be a steady-state condition. Off-normal and accident events, as well as the vacuum condition that temporarily occurs during the preparation of the canister while it is in the transfer cask and the time the canister, is in the transfer cask while filled with helium, are evaluated as transient conditions. The maximum allowable material temperatures for long-term and for transient conditions are provided in Table 4.1-3. The maximum component temperatures are provided in Table 4.1-4.

The NAC-MPC system is designed to store Yankee class spent fuel with a maximum heat load of 12.5 kW and reconfigured fuel assemblies with a maximum heat load of 0.102 kW per assembly. The temperature effects on the NAC-MPC storage cask and the canister, due to the reconfigured fuel assemblies, are bounded by the temperatures produced in the cask by the design basis fuel. Table 4.1-4 summarizes the results of the thermal evaluation. As shown in this table, the calculated temperatures are well below the allowable component temperatures for normal (long-term) storage conditions and for short-term events.

4.1-1

Table 4.1-1	Summary of	Thermal Design	Conditions for	Storage
	<b>4</b>	0		0-

	ENVIRONMENTAL		<b>CONDITION OF</b>
	TEMPERATURE (°F)	SOLAR	STORAGE
CONDITION		INSOLANCE (1)	CASK VENTS
Normal	75	Yes	All vents open
Off-Normal	75	Yes	Two inlets
- Half Air Inlets Blocked			blocked
Off-Normal	100	Yes	All vents open
- Severe Heat			
Off-Normal	-40	No	All vents open
- Severe Cold			
Accident	125	Yes	All vents open
- Extreme Heat			
Accident	75	Yes	All vents blocked
- All Air Inlets and Outlets			
Blocked ⁽²⁾			
Accident	75	No	All vents blocked
- Cask Burial Under			
Debris ⁽³⁾			

 ⁽¹⁾ Solar Insolance per 10CFR71: Curved Surface: 400 g cal/cm² (1475 Btu/ft²) for a 12-hour period.

Flat Horizontal Surface: 800 g cal/cm² (2950 Btu/ft²) for a 12-hour period.

- ⁽²⁾ This condition bounds the case in which all inlets are blocked, with all outlets open.
- ⁽³⁾ In the burial under debris condition, the inlets/outlets are blocked and, in addition, the debris is considered not to permit any heat transfer from the surface of the concrete. This is a highly conservative assumption.

## Table 4.1-2Summary of Thermal Design Conditions for Transfer

CONDITION ⁽¹⁾	<b>DURATION</b> (Hours)
Water Filled	20
Vacuum Drying ⁽²⁾	10
Canister filled with Helium	36

⁽¹⁾ The canister is inside the Transfer Cask, with an ambient temperature of  $75^{\circ}$ F.

⁽²⁾ The canister is filled with water for a maximum of 20 hours before the start of the vacuum drying process. The initial water temperature is considered to be 100°F.

MATERIAL ¹	LONG-TERM	SHORT-TERM	REFERENCE
Concrete	$150(B)/200(L)^2$	350	ACI 349
Zircaloy Fuel Cladding	644 ³	806 ⁴	PNL-6189
			PNL-4835
Stainless Steel Fuel Cladding	644 ³	806	EPRI TR-106440
Aluminum Disk	650	700	MIL-HDBK-5G
NS-4-FR	300	300	Genden
Lead	600	600	Baumeister
SA693 Type 630 Stainless Steel	650	800	ASME B & PV Armco
SA240 Type 304 Stainless Steel	800	800	ASME B & PV
SA240 Type 304L Stainless Steel	800	800	ASME B & PV
ASTM A588 Carbon Steel	700	700	ASME Code Case
			N-71-17
ASTM A36 Carbon Steel	700	700	ASME Code Case N-71-17
BORAL Composite Sheet	850	1,000	AAR Advanced Structures

## Table 4.1-3Maximum Allowable Temperature Limits (°F)

- 1. The minimum allowable temperature limit for all materials is -40°F.
- 2. B and L refer to bulk temperatures and local temperatures, respectively. The local temperature allowable applies to a restricted region where the bulk temperature allowable may be exceeded.
- 3. The temperature limits for 5-year and 10-year-cooled fuel are 380°C and 340°C, respectively. The lower value (340°C) is used.
- 4. The temperature limit for stainless steel-cladding is conservatively used for Zircaloy cladding.

## NAC-MPC FSAR Docket No. 72-1025

April 2000 Revision 0

<b>F</b>										
		Material Temperature (°F)								
	1	Fuel	6061-T6			SA693	SA240	SA240	A36	A588/A350LF2
<b>Design Conditions</b>	Concrete ⁴	Clad	Al Alloy ²	NS-4-FR ³	Lead'	$630 \text{ SS}^2$	$304 \text{ SS}^4$	304L SS ⁴	Steel ¹	Steel ³
Allowable		Zr/SS								
Long-Term	150(Bulk)	644/644	650	300	600	650	800	800	700	700
	200(Local)									
Short-Term	350	806/806	700	300	600	800	800	800	700	700
Long-Term Conditions										
Normal (75°F Ambient)	133(Bulk)	563	527	N/A	N/A	529	183	319	165	N/A
	165(Local)									
Short-Term Conditions							-			
Off-Normal	168	565	529	N/A	N/A	531	192	318 ⁵	169	N/A
-Half Inlets Blocked										
(75°F Ambient)										
Off-Normal	196	587	552	N/A	N/A	554	213	347	196	N/A
-Severe Heat										
(100°F Ambient)										
Off-Normal	5	453	411	N/A	N/A	412	44	187	5	N/A
-Severe Cold										
(-40°F Ambient)										
Accident	228	607	574	N/A	N/A	575	241	372	229	N/A
-Extreme Heat										
(125°F Ambient)										
Transfer	N/A	424	339	98	98	339	100	274	N/A	103
-Vacuum Drying										
Transfer	N/A	597	569	188	191	570	140	430	N/A	237
-Backfilled with Helium						ł				

## Table 4.1-4 Summary of Thermal Evaluation for NAC-MPC Storage System

1. Concrete cask components: Concrete cask and steel liner (A36).

2. Fuel basket components: Heat transfer disks (6061-T6) and support disks (SA 693, Type 630 stainless steel).

3. Transfer cask components: Shells (A588); bottom doors (A350LF2), neutron shield (NS-4-FR); and, gamma shield (lead).

4. Canister components: Shield lid (SA 240, Type 304 stainless steel) and shell, bottom plate and structural lid (SA 240 Type 304L stainless steel).

5. Although the maximum canister temperature is 1 degree lower than that of the normal condition, the overall canister temperature for the half inlet blocked condition is higher than that of the normal condition, which results in higher temperatures inside of the canister (fuel clad and disks).

THIS PAGE INTENTIONALLY LEFT BLANK

____

-----

## 4.2 <u>Summary of Thermal Properties of Materials</u>

The thermal properties used in the thermal analyses are shown in Tables 4.2-1 through 4.2-11. The derivation of the effective conductivities is described in Sections 4.4.1.3 and 4.4.1.4.

## Table 4.2-1Thermal Properties of Solid Neutron Shield (NS-4-FR)

Property ¹ (units)	Value
Conductivity (Btu/hr-in-°F)	0.0311
Density (lbm/in ³ ) (borated)	0.0589
Density (lbm/in ³ ) (nonborated)	0.0607
Specific Heat (Btu/lbm-°F)	0.39

1. Data developed by BISCO Products (NS-4-FR is now supplied by Genden Engineering Services and Construction Company. Genden has established a maximum continuous service temperature of 300°F.).
| Type 304 and Type 304L                      |                         |                             |                 |        |  |  |
|---------------------------------------------|-------------------------|-----------------------------|-----------------|--------|--|--|
|                                             | Te                      | emperature (°F              | ') ⁵ |        |  |  |
| Property (units)                            | 212                     | 392                         | 572             | 752    |  |  |
| Conductivity ¹ (Btu/hr-in-°F)    | 0.7800                  | 0.8592                      | 0.9333          | 1.0042 |  |  |
| Density ¹ (lbm/in ³ ) | 0.2888                  | 0.2888 0.2872 0.2855 0.2839 |                 |        |  |  |
| Specific Heat ¹ (Btu/lbm-°F)     | 0.1207                  | 0.1272                      | 0.1320          | 0.135  |  |  |
| Emissivity ²                     |                         | 0.36 at 300°F               |                 |        |  |  |
|                                             | 17                      | -4РН, Туре 63               | 0               |        |  |  |
|                                             | Те                      | emperature (°F              | ⁽⁾⁶  |        |  |  |
| Property (units)                            | 100                     | 200                         | 500             | 700    |  |  |
| Conductivity ³ (Btu/hr-in-°F)    | 0.8417                  | 0.8833                      | 1.0167          | 1.1000 |  |  |
| Density ⁴ (lbm/in ³ ) | 0.284 0.284 0.284 0.284 |                             |                 |        |  |  |
| Specific Heat ⁴ (Btu/lbm-°F)     | 0.11                    | 0.11                        | 0.11            | 0.11   |  |  |
|                                             |                         |                             |                 |        |  |  |

1. Hanford.

2. Bucholz.

3. "ASME Code Section II, Part D."

4. ARMCO.

5. Maximum Service Temperature: 800°F.

6. Maximum Service Temperature: 650°F long-term and 800°F short-term.

	Temperature (°F) ³				
Property (units)	209	400	581	630	
Conductivity ¹ (Btu/hr-in-°F)	1.6308	1.5260	1.2095	1.0079	
Density ¹ (lbm/in ³ )	0.411	0.411	0.411	0.411	
Specific Heat ¹ (Btu/lbm-°F)	0.03	0.03	0.03	0.03	
Emissivity ²	0.28 at 75°F				

## Table 4.2-3Thermal Properties of Chemical Lead

- 1 Edwards.
- 2 Baumeister.
- 3 Maximum service temperature: 600°F (based on preventing the lead from reaching its melting point of 620°F).

## Table 4.2-4Thermal Properties of Type 6061-T6 Aluminum Alloy

	Temperature (°F) ³					
Property (units)	200	300	400	500	600	700
Conductivity ³ (Btu/hr-in-°F)	8.25	8.38	8.49	8.49	8.49	8.49
Emissivity ²	0.22	0.22	0.22	0.22	0.22	0.22

1. ASME Code, Section II, Part D:

The maximum temperature tabulated is 400°F. Since the conductivity increases as the temperature increases, using the value of 8.49 for higher temperatures is conservative (MIL-HDBK-5G, Figure 3.6.2.0).

- 2. Recommended value for aluminum in SCOPE (Bucholz) Version 1.2 Abbreviated Input Data Guide.
- 3. Maximum Service Temperature: 700°F, short-term service 650°F, long-term service.

	Temperature (°F)				
Property (units)	200	400	600	800	
Conductivity ¹ (Btu/hr-in-°F)	0.00808	0.00942	0.01075	0.01150	
Density ¹ (lbm/in ³ )	4.83E-6	3.70E-6	3.01E-6	2.52E-6	
Specific Heat ¹ (Btu/lbm-°F)	1.24	1.24	1.24	1.24	

# Table 4.2-5Thermal Properties of Helium

¹ Kreith.

T-1-1- 10 C	The	Deconsting	of Dery Aire
1 apre 4, 2-n	Inermai	Probenues	
I GOIO II DO		1.00000000	

	Temperature (°F)				
Property (units)	100	300	500	700	
Conductivity ¹ (Btu/hr-in-°F)	0.00128	0.00161	0.00193	0.0023	
Density ¹ (lbm/in ³ )	4.11E-5	3.23E-5	2.38E-5	1.97E-5	
Specific Heat ¹ (Btu/lbm-°F)	0.240	0.244	0.247	0.253	

¹ Kreith.

## Table 4.2-7Thermal Properties of Concrete

Property (units)	Temperature Range (°F) ⁵ 32 - 400
Conductivity ¹ (Btu/hr-in-°F)	0.059
Specific Heat ¹ (Btu/lbm-°F)	0.20
Density ² (lbm/ft ³ )	140
Emissivity ^{3,4}	0.90

(Emissivity = 0.93 for masonry, 0.94 for rough concrete; 0.9 is used)^{3,4}

1 Fintel (Fig. 6-31, 0.71/12=0.059 Btu/hr. in.-°F).

- 2 ASTM-C150.
- 3 Kreith.
- 4 Siegel.
- 5 Maximum service temperature long-term: 150°F bulk and 200°F local; short-term: 350°F. The local temperature allowable applies to a restricted region where the bulk temperature allowable may be exceeded.

# Table 4.2-8 Thermal Properties of ASTM A 36, ASTM A 588 and ASTM A350 Carbon Steel

	Temperature (°F) ⁵					
Property (units)	100	200	400	500	700	
Conductivity ¹ (Btu/hr-in-°F)	1.992	2.033	2.017	1.975	1.867	
Density ² (lbm/in ³ )	0.284	0.284	0.284	0.284	0.284	
Specific Heat ³ (Btu/lbm-°F)	0.113	0.113	0.113	0.113	0.113	
Emissivity ⁴	0.80	0.80	0.80	0.80	0.80	

1 ASME Code, Section II, Part D, Table TCD.

2 Ross.

3 Kreith.

4 Baumeister.

5 Maximum service temperature: 700°F.

## Table 4.2-9 Thermal Properties of Zircaloy and Zircaloy-4 Cladding

	Temperature (°F)				
Property (units)	392	572	752	932	
Conductivity ¹ (Btu/hr-in-°F)	0.69	0.73	0.80	0.87	
Emissivity ¹	0.75	0.75	0.75	0.75	

¹ NUREG/CR-0497 (minimum value of emissivity for a cladding surface).

# Table 4.2-10 Thermal Properties of Fuel (UO₂)

	Temperature (°F)				
Property (units)	100	440	570	793	
Conductivity ¹ (Btu/hr-in-°F)	0.29	0.29	0.27	0.19	

¹ NUREG/CR-0497 (The lower boundary of temperatures tabulated is 500°K [440°F]. Since the conductivity decreases as the temperature increases, using the value of 0.29 for the 100°F entry is conservative.).

Table 4.2-11 Thermal Properties of BORAL Composite Sheet

	Temperature (°F) ³		
Property (units)	100	500	
Conductivity ¹ (Btu/hr-in-°F)			
Aluminum Clad ¹	7.805	8.976	
Core Matrix ¹	4.136	3.698	
Emissivity ^{1,2}	0.15	0.15	

- 1 AAR Advanced Structures, standard specification for BORAL composite BRJREVO-940107.
- 2 The emissivity of the aluminum clad of the BORAL sheet ranges from 0.10 to 0.19 based on the BORAL specification. An averaged value of 0.15 is used.
- 3 Maximum service temperature: 850°F.

#### 4.3 <u>Specification of Components</u>

There are three major components that must be maintained within their safe operating temperature ranges: the lead gamma shield and the NS-4-FR solid neutron shield in the transfer cask, and the aluminum heat transfer disk in the canister basket.

The safe operating ranges for the lead gamma shield, solid neutron shield and aluminum heat transfer disk are as follows:

Component	Safe Operating Range
Lead gamma shield	-40°F to +600°F
NS-4-FR solid neutron shield	-40°F to +300°F
Aluminum heat transfer disk	-40°F to +650°F (long-term);
	+700°F (short-term)

The safe operating range of the lead gamma shield is based on preventing the lead from reaching its melting point of 620°F (Baumeister).

The maximum operating temperature limit of the NS-4-FR solid neutron shield material to ensure sufficient neutron shielding capability is specified by the product supplier to be 300°F.

The safe operating range of the aluminum heat transfer disk is based on the integrity of the aluminum being maintained. The aluminum heat transfer disk is not a structural component to transfer load within the basket. Based on the MIL-HDBK-5F, aluminum at 700°F retains component performance. The maximum long-term and short-term operating temperatures for the aluminum heat transfer disk are taken to be 650°F and 700°F, respectively.

The maximum operating temperatures for other materials are shown in Table 4.1-3.

As shown in Tables 4.4.3-1 and 4.4.3-2, the maximum temperatures for these materials in the normal (long-term) conditions of storage are well below the allowable maximum temperatures. Maximum temperatures for off-normal and accident conditions are addressed in Chapter 11.

THIS PAGE INTENTIONALLY LEFT BLANK

_____

#### 4.4 Thermal Evaluation for Normal Conditions of Storage

## 4.4.1 <u>Thermal Models</u>

As listed below, six finite element models are utilized for the thermal evaluation of the NAC-MPC system for normal conditions of storage. The ANSYS program generates all models.

- 1. Two-Dimensional Axisymmetric Air Flow and Concrete Cask Model
- 2. Three-Dimensional Canister Model
- 3. Two-Dimensional Fuel Model
- 4. Two-Dimensional Fuel Tube Model
- 5. Three-Dimensional Transfer Cask and Canister Model
- 6. Two-Dimensional Reconfigured Fuel Assembly Model

The two-dimensional axisymmetric air flow and concrete cask model includes: the concrete storage cask, air in the air inlets, annulus and the air outlets, and the canister shell. It is used to perform computational fluid dynamic analyses to determine the mass flow rate, velocity and temperatures of the air flow, as well as the temperature distribution of the concrete, concrete cask steel liner and the canister shell.

The three-dimensional canister model comprises the fuel assemblies, fuel tubes, stainless steel support disks, aluminum heat transfer disks, the canister shell, lids and bottom plate. The canister model is employed to evaluate the temperature distribution of the fuel cladding and components of the canister and basket. The fuel regions and the fuel tubes with BORAL plates in the three-dimensional canister model are modeled using effective conductivities.

The effective conductivity of the fuel is determined using the two-dimensional fuel model, which is a detailed two-dimensional thermal model of the fuel assembly. The model includes the fuel pellets, cladding and gas (considered to be helium) occupying the space between the fuel rods and in the gap between the fuel pellets and the fuel rod cladding. The two-dimensional fuel tube model is used to determine the effective conductivities of the tube wall and BORAL plate.

The three-dimensional transfer cask and canister model comprises the transfer cask, the canister and the canister internals—i.e., the three-dimensional canister model with the transfer cask added. This model is used to perform transient analysis for the transfer condition when the canister is in the vacuum condition during drying.

The two-dimensional reconfigured fuel assembly model comprises the fuel rods, fuel tubes, the shell casing (the square tube with the same external dimensions as an intact fuel assembly) and the gas (helium) occupying the gap between fuel rod and tube, the space between fuel tubes and the gap between shell casing and the fuel assembly tube. This model is used to determine the temperature distribution of the reconfigured fuel assembly.

These models are described in Section 4.4.1.1 through 4.4.1.6.

#### 4.4.1.1 <u>Two-Dimensional Axisymmetric Air Flow and Concrete Cask Model</u>

The storage cask consists of the fuel canister, concrete, steel liner, air gap between the fuel canister shell and steel liner, and air inlets/outlets through the concrete region. The fuel canister with a design heat load of 12.5 kW will be cooled by (1) natural/free convection of air through the lower vents, the vertical air annulus, and the upper vents; and (2) radiant heat transfer between the surfaces of the canister shell and the steel liner. The heat transferred to the liner will be rejected by air convection in the annulus and by conduction through the concrete. The heat flow through the concrete will be dissipated to the surroundings by natural convection and radiant heat transfer. The temperature in the concrete region is controlled by radiant heat transfer between the vertical annulus surfaces (canister shell outer surface and steel liner inner surface), natural convection of air in the annulus, and boundary conditions applicable to the storage cask outer surfaces—e.g., natural convection and radiant heat transfer between the outer surfaces and the environment, including consideration of incident solar energy. These heat transfer modes are combined in the air flow and concrete cask model. The entire thermal system, including mass, momentum, and energy, is analyzed using ANSYS FLOTRAN.

## 4.4.1.1.1 Finite Element Model for Air Flow in Storage Cask

The storage cask has four air inlets at the bottom and four air outlets at the top that extend through the concrete. Since the configuration is symmetrical, it can be simplified into a twodimensional axisymmetric model, as shown schematically in Figure 4.4.1.1-1, by using equivalent dimensions for the air inlets and outlets, which are assumed to extend around the storage cask periphery. This model consists of the following regions: canister shell, steel liner, concrete, air inlet, transitional region, vertical annulus, and air outlet. The canister shell is included in this model in order to apply the heat flux for the design fuel heat. The canister model is described in Section 4.4.1.2.

The two-dimensional axisymmetric air flow and concrete cask finite element model is shown in Figure 4.4.1.1-2. The model has the following features. In the air region, only quadrilateral elements are used and the element sizes are nonuniform with much smaller element sizes close to the walls. In other regions to simulate conduction, a mix of quadrilateral elements and triangular elements are used.

In this model, the radiation heat transfer across the vertical air annulus is also included.

## 4.4.1.1.2 Loads and Boundary Conditions

In the normal storage condition, the concrete cask has the following loads and boundary conditions:

1. Heat flux from the active fuel region.

Since only the canister shell is included in this air flow model, an equivalent nonuniform heat flux from the active fuel region is applied to the inner surface of the canister shell. This heat flux corresponds to 12.5kW, which is the heat generated by the spent fuel. The distribution of the heat flux is based on the axial power distribution, as described in Section 5.2.3, for the design-basis fuel (Figure 4.4.1.1-3).

2. Solar insolance to the outer surfaces of the storage cask.

The solar insolance to the storage cask outer surface is considered in the model. The incident solar energy is applied based on 24-hour averages as shown below.

Side surface:  $\frac{1475Btu / ft^2}{24 hrs} = 61.46Btu / hr \cdot ft^2$ 

Top surface:  $\frac{2950Btu / ft^2}{24hrs} = 122.92Btu / hr \cdot ft^2$ 

3. Natural convection heat transfer at the outer surfaces of the storage cask.

Natural convection heat transfer at the outer surfaces of the storage cask is evaluated using the heat transfer correlation for vertical and horizontal plates (Kreith, Incropera). This method assumes a surface temperature and then estimates Grashof (Gr) or Rayleigh (Ra) numbers to determine whether correlation for a laminar model or for a turbulence model should be used. Since Grashof or Rayleigh numbers are much higher than the critical values, correlation for a turbulence model is used as shown in the following.

Side surface (Kreith):

$$Nu = 0.13(Gr \cdot Pr)^{1/3}$$
  
$$h_c = Nu \cdot k_f / H_{VCC}$$
for Gr > 10⁹

Top surface (Incropera):

$$Nu = 0.15 Ra^{1/3}$$
  
h_c = Nu · k_f / L for Ra > 10⁷

where:

- h_c Average natural convection heat transfer coefficient
- H_{vcc} Height of the storage cask
- Gr Grashof number
- k_f Conductivity
- L Top surface characteristic length, L = area / perimeter
- Nu Average Nusselt number
- Pr Prandtl number
- Ra Rayleigh number

All material properties required in the above equations are evaluated based on the film temperature—that is, the average value of the surface temperature and ambient temperature.

4. Radiation heat transfer at the storage cask outer surfaces.

The radiation heat transfer between the outer surfaces and ambient is evaluated in the model by calculating an equivalent radiation heat transfer coefficient.

$$h_{rad} = \frac{\sigma(T_1^2 + T_2^2)(T_1 + T_2)}{\frac{1}{\epsilon_1} + \frac{1}{\epsilon_2} + \frac{1}{F_{12}} - 2}$$

where:

h _{rad}	Equivalent radiation heat transfer coefficient
F ₁₂	View factor
$T_1$ and $T_2$	Surface (T1) and ambient (T2) temperatures (°K)
$\varepsilon_1$ and $\varepsilon_2$	Surface ( $\varepsilon_1$ ) and ambient ( $\varepsilon_2=1$ ) emissivities
σ	Stefan-Boltzmann Constant

At the storage cask side surface, an emissivity for concrete surface of  $\varepsilon_1 = 0.9$  is used and a calculated view factor (F₁₂) = 0.197 (Chapter 13 of Incropera) is applied. Since the center cask is surrounded by other casks, its view factor to the ambient is reduced.

At the top, an emissivity of  $\varepsilon_1 = 0.8$  for a carbon steel surface is used, and a view factor (F₁₂) = 1 is applied.

5. Other flow and thermal boundary conditions.

At all walls bounding the air flow region, a zero velocity, vx=vy=0, is applied. At the inlet and outlet surfaces, a relative zero pressure,  $p_{in} = p_{out} = 0$ , is applied. The inlet temperature of the air is reasonably assumed to be the same as the ambient temperature. The effect of surrounding loaded casks on air inlet temperature is considered to be negligible because of the 15-foot center to center spacing of the casks (approximately 4.5 feet, clear opening). The natural air flow patterns between the casks negate potential stagnation effects.

## 4.4.1.1.3 Accuracy Check of the Numerical Simulation

To ensure the accuracy of the numerical simulation on the air flow in the storage cask and to ensure reliable numerical results, the following checks and confirmations have been performed.

1. Global convergence of the iteration process for the nonlinear system.

The system controlling air flow through the cask and, therefore, the temperature field is nonlinear and is solved iteratively. The global iteration process is monitored by checking the variation of parameters with the global iteration—e.g., the maximum air temperature, the mass flow rate, and the heat carried out of the storage cask by air convection. The mass flow rate varying with the global iteration is shown in Figure 4.4.1.1-4. As shown, after 10,000 global iterations, the mass flow rate is constant, indicating that the global iteration process has converged. At the converged state, the mass flow rate at the air inlets agrees with that at the air outlets. All of the results presented are at the converged state.

2. Finite element mesh adequacy study.

Element size or number of elements used in the simulation will affect the accuracy and reliability of the numerical prediction, even though converged results can be obtained.

Two tests, using different element sizes, have been performed for the normal case. Results are shown in Table 4.4.1.1-1, together with the number and size of elements close to the vertical annulus surfaces. As shown, results for the two test cases are in agreement, with the maximum difference being less than 3.3 percent in net heat carried out of the cask by air ( $Q_{air}$ ).

3. Overall energy balance and mass balance.

This step validates the overall energy balance and mass balance. The mass balance is shown in Figure 4.4.1.1-4. At the converged state, the mass flow rate at the air inlets matches the mass flow rate at the air outlets, showing that an excellent mass balance has been obtained.

The overall energy balance is checked by computing the total heat input  $(Q_{in})$  and total heat output  $(Q_{out})$ . The total heat input includes the total heat from the fuel  $(Q_{fuel})$  and the total solar energy  $(Q_{sun})$  incident on the storage cask outer surfaces. The total heat output is the sum of heat carried out of the cask by air  $(Q_{air})$  and by convection and radiation heat loss at the storage cask outer surfaces  $(Q_{con})$ . During the normal storage condition:

 $Q_{in}=Q_{fuel}+Q_{sun}=12.5kW+9.85kW=22.35kW$  $Q_{out}=Q_{air}+Q_{con}=12.43kW+10.25kW=22.68kW$  $Q_{out}/Q_{in}=1.015$ 

Therefore, the overall energy balance is demonstrated to be within 1.5 percent.

April 2000 Revision 0

## Figure 4.4.1.1-1 Two-Dimensional Axisymmetric Air Flow and Concrete Cask Model



4.4-8

NAC-MPC FSAR	April 2000
Docket No. 72-1025	Revision 0

Figure 4.4.1.1-2 Two-Dimensional Axisymmetric Air Flow and Concrete Cask Finite Element Model



## Figure 4.4.1.1-3 Axial Power Distribution for Yankee Class Fuel



Top of Active Fuel

## Figure 4.4.1.1-4 Convergence Process of Mass Flow Rate

Mass in: mass flow rate at the air inlets, kg/s

Mass out: mass flow rate at the air outlets, kg/s



# Table 4.4.1.1-1Comparison of Numerical Results Using Different Element Sizes<br/>and Number of Elements

Test	Element	Element	Tmax _{air}	Tmaxcon	Mass	Vx _{out}	Tout	Qair
Case	Number	Size(m)	(K)	(K)	(kg/s)	(m/s)	(K)	(kW)
ES1	12167	0.00184	433.25	345.33	.3582	.7780	332.84	12.84
ES2	15631	0.00136	430.61	346.89	.3543	.7673	332.45	12.43
ES1/ES2	.7784	1.353	1.0061	.9955	1.011	1.014	1.0012	1.033

where:

Tmax _{air}	Maximum air temperature
Tmax _{con}	Maximum concrete temperature
Element size	Refer to the element adjacent to the wall of the vertical gap
Mass	Average mass flow rate of air between air inlet and outlet
Vx _{out}	Average velocity component in x-direction
T _{out}	Average air temperature at the air outlet
Q _{air}	Net heat carried out of the cask by air

#### 4.4.1.2 Three-Dimensional Canister Model

The three-dimensional canister model is shown in Figures 4.4.1.2-1 and 4.4.1.2-2. ANSYS SOLID70 three-dimensional conduction elements and LINK31 radiation elements are used to construct the model. The model includes: the fuel assemblies, fuel tubes, support disks, heat transfer disks, the canister shell, lids, bottom plate and helium. Based on symmetry, only half of the canister is modeled. As shown in Figure 4.4.1.2-1, the interior of the canister contains the active fuel region. No conduction elements are defined outside of this region in the axial direction; i.e., top and bottom fittings of the fuel assemblies, fuel tubes enclosing the top and bottom fittings, the first support disk (counted from the top end), and the top and bottom Conduction through these components is weldments are not included in the model. conservatively ignored. The canister shell temperatures obtained from the two-dimensional axisymmetric air flow and concrete cask model (Section 4.4.1.1) are applied at the canister shell surface in the model as boundary conditions. The top surface of the canister lid and the bottom surface of the canister bottom plate in the model are considered to be adiabatic. In the model, the fuel assemblies are considered to be centered in the fuel tubes. The fuel tubes are centered in the slots of the support disks and heat transfer disks. The basket is centered in the canister. These assumptions are conservative, since any contact between components will provide a more efficient path to reject the heat.

This model includes the following gaps:

LOCATION	<u>GAP SIZE</u>
1. Gap between the support disk and the canister shell	0.205 inch
2. Gap between the heat transfer disk and the	0.345 inch
canister shell	
3. Gap between the exterior surface of the fuel tube	0.079 inch
and the inside surface of the slots in the disks	

4. Gap between the BORAL sheet and tube/cladding0.003 inch

Gas inside the canister is modeled as helium. Note that the nominal gap (room temperature) size between the support disk and the canister shell is 0.12 inch, and the nominal gap size between the heat transfer disk and the canister shell is 0.26 inch. As shown above, the thermal model conservatively considers these gaps to be 0.205 inch and 0.345 inch (at room temperature) for the support disk and heat transfer disk, respectively. Considering the differential thermal expansion

of the disks and the canister shell, these gap sizes are adjusted for the solution of the thermal analysis. In the model, the gap between the support disk and the canister shell is increased to 0.22 inch, and the gap between the heat transfer disk and the canister shell is reduced to 0.295 inch. These gaps are conservative and remain bounding even if the worst case fabrication tolerances are considered. The structural lid and the shield lid are expected to be in full contact due to the weight of the structural lid. The thermal resistance across the contact surface is considered to be negligible and, therefore, no gap is modeled between the lids.

All material properties used in the model, except the effective properties discussed below, are shown in Tables 4.2-1 through 4.2-11.

The fuel regions (inside tubes) are modeled as homogenous regions with effective conductivities, determined by the two-dimensional fuel model as described in Section 4.4.1.3. The center slot of the basket contains no fuel and is modeled as helium. The fuel tube and the BORAL plate, including helium gaps on both sides of the BORAL sheet and the gap between the stainless steel cladding and the disks, are modeled as one-element thick with effective conductivities, established using the two-dimensional tube model described in Section 4.4.1.4.

In this model, radiation heat transfer is taken into account in the following locations:

- 1. From the top of the fuel region to the bottom surface of the shield lid.
- 2. From the bottom of the fuel region to the top surface of the canister bottom plate.
- 3. From the exterior surfaces of the fuel tubes (between SS and AL disks) to the inner surface of the canister shell.
- 4. Across all four gaps, described above.

Radiation elements (LINK31) are used to model the radiation effect for the first three locations. Radiation across gaps is accounted for by establishing effective conductivities for the gas in the gap, as shown below. Since the gaps represented in the model are small compared to the surfaces separated by the gaps, the view factor is taken to be unity.

Radiation heat transfer between two nodes i (hotter node) and j (colder node) is accounted for by the expression:

$$q_{r} = \sigma \varepsilon_{eff} AF \left( T_{i}^{4} - T_{j}^{4} \right)$$

where:

- $\sigma$  = the Stefan-Boltzman constant = 1.19 x 10⁻¹¹ Btu/hr-in²-°R⁴
- $\varepsilon_{\rm eff}$  = effective gray body emissivity
- A = surface area
- F = the gray body shape factor for the surfaces
- $T_i$  = temperature of the i th node
- $T_i$  = temperature of the j th node

The total heat transfer can be expressed as the sum of the radiation and the conduction processes.

 $Q_t = q_r + q_k$ 

where  $q_r$  is specified above for the radiation heat transfer and  $q_k$ , which is the heat transfer by conduction is expressed as:

$$q_k = \frac{KA}{g} 1(T_i - T_j)$$

where:

T_i = temperature of the i th node
 T_j = temperature of the j th node
 g = gap distance (between the two surfaces defined by node i and node j)
 K = conductivity of the gas in the gap
 A = area of gap surface

By combining the two expressions (for  $q_k$  and  $q_r$ ) and factoring out the term  $A(T_i - T_j)/g$ ,

$$Q_{t} = [g\sigma \varepsilon_{eff} F(T_{i}^{2} + T_{j}^{2})(T_{i} + T_{j}) + K][A(T_{i} - T_{j})/g]$$

or

 $Q_t = K_{eff}A(T_i - T_j)/g$ 

where:

$$K_{eff} = g\sigma\varepsilon_{eff}F(T_i^2 + T_j^2)(T_i + T_j) + K$$

The material conductivity used in the analysis for the elements comprising the gap includes the heat transfer by both conduction and radiation.

Effective emissivities are used for all radiation calculations, based on the formula below (Kreith).

 $\varepsilon_{eff} = 1/(1/\varepsilon_1 + 1/\varepsilon_2 - 1)$  where  $\varepsilon_1$  and  $\varepsilon_2$  are the emissivities of two parallel plates

Radiation between the exterior surfaces of the fuel tubes and the radiation between the stainless steel support disk and the aluminum disk are conservatively ignored in this model.

Volumetric heat generation (Btu/hr-in³) is applied to the active fuel region based on a total heat load of 12.5 kW, active fuel length of 91 inches and an axial power distribution as shown in Figure 4.4.1.1-3.

# Figure 4.4.1.2-1 Three-D

Three-Dimensional Canister Model



## Figure 4.4.1.2-2 Three-Dimensional Canister Model - Cross-Section



#### 4.4.1.3 <u>Two-Dimensional Fuel Model</u>

The effective conductivity of the fuel is determined by a two-dimensional finite element model of the fuel assembly. The model includes: the fuel pellets, cladding, gas between fuel rods and gas (considered to be helium) occupying the gap between the fuel pellets and cladding. The configuration of the design basis fuel (CE Type A) is used in the model. Note that the fuel cladding material for this fuel is zircaloy, which is less conductive than stainless steel. Therefore, the fuel model bounds the fuel with zircaloy cladding, as well as the fuel with stainless steel cladding. Modes of heat transfer modeled include conduction and radiation between individual fuel rods for the steady-state condition. The model is shown in Figure 4.4.1.3-1.

ANSYS PLANE55 conduction elements and LINK31 radiation elements are used in the model, which includes a total of 240 fuel rods. Each fuel rod consists of the pellet, zircaloy cladding, and a gap between the pellet and cladding. The gas in the gap between the pellet and cladding, as well as the gas between fuel rods, is considered to be helium. Radiation elements are defined between fuel rods and from the fuel rods to the boundary of the model (inside surface of the fuel tube). Radiation effects at the gap between the pellets and the cladding are conservatively ignored. Effective emissivities are determined using the formula shown in Section 4.4.1.2.

The effective conductivity for the fuel is determined using the equation (SANDIA) to determine the maximum temperature of a square cross-section of an isotropic homogeneous fuel with a uniform volumetric heat generation. At the boundary of the square cross-section, the temperature is constrained to be uniform. The expression for the maximum temperature is given by:

 $T_c = T_e + 0.29468 (Q a^2 / K_{eff})$ 

where:

- $T_c$  = the temperature at the center of the fuel (°F)
- $T_e$  = the temperature applied to the exterior of the fuel (°F)
- Q = volumetric heat generation rate (Btu/hr-in³)

a = half length of the square cross-section of the fuel (inch)

K_{eff} = effective thermal conductivity for the isotropic homogeneous fuel material (Btu/hr-in-°F)

NAC-MPC FSAR	April 2000
Docket No. 72-1025	Revision 0

Volumetric heat generation (Btu/hr-in³) based on the design heat load of 12.5 kW is applied to the pellets. The effective conductivity is determined based on the heat generated and the temperature difference from the center of the model to the edge of the model. The temperature-dependent effective properties, as shown below, are established using different boundary temperatures. The effective conductivity in the axial direction of the fuel assembly is calculated based on the material area ratio.

#### Conductivity (Btu/hr-in-°F)

<u>remperature (°F)</u>	<u>kxx</u>	<u>kyy</u>	<u>kzz</u>
128	0.0167	0.0167	0.163
322	0.0204	0.0204	0.152
518	0.0262	0.0262	0.141
714	0.0330	0.0330	0.138
911	0.0405	0.0405	0.140

Note:

- 1. x and y axes are in-plane of the model, z is in the canister axial direction.
- 2. The temperature associated with each row is the average temperature of the fuel assembly.

## Figure 4.4.1.3-1 Two-Dimensional Fuel Model



#### 4.4.1.4 <u>Two-Dimensional Fuel Tube Model</u>

The purpose of the two-dimensional fuel tube model is to determine the effective conductivity of the fuel tube and BORAL plate, which is used in the three-dimensional canister model. As shown in Figure 4.4.1.4-1, this model includes the fuel tube, the BORAL plate (including the core matrix sandwiched by aluminum claddings), helium gaps on both sides of the BORAL plate, and the helium gap between the stainless steel cladding on the outside of the BORAL plate and the support disk or heat transfer disk.

ANSYS PLANE55 conduction elements and LINK31 radiation elements are used to construct the model. The model consists of eight layers of conduction elements and six radiation elements that are defined at the helium gaps (two for each gap). The thickness of the model (x-direction) is the distance measured from the inside face of the fuel tube to the inside face of the slot in the support disk (assuming the fuel tube is centered in the hole in the disk). The tolerance of the BORAL plate thickness, 0.003 inch, is used as the gap size for both sides of the BORAL plate. The height of the model is defined as equal to the width of the model.

Heat flux is applied at the left side of the model (fuel tube), and the temperature at the right boundary of the model is constrained. The heat flux is determined based on the design heat load of 12.5 kW. The maximum temperature of the model (at the left boundary) and the temperature difference ( $\Delta$ T) across the model are calculated by ANSYS. The effective conductivity is determined using the following formula:

 $q = k (A/L) \Delta T$ or  $k = q L / (A \Delta T)$ 

where:

q = heat rate (Btu/hr) A = area (in²) L = length of model (in)  $\Delta T = temperature difference across the model (°F)$ 

k = effective conductivity (Btu/hr-in-°F)

NAC-MPC FSAR	April 2000
Docket No. 72-1025	Revision 0

The temperature-dependent conductivity (k) is determined by varying the temperature constraints at one boundary of the model and resolving for the heat rate (q) and temperature difference. The effective conductivity for the parallel path (the Y direction in Figure 4.4.1.4-1) is calculated by:

$$Kyy = \frac{\sum K_i t_i}{T}$$

where:

 $K_i$  = thermal conductivity of each layer (fuel tube, helium aluminum clad)

 $t_i$  = thickness of each layer

T = total thickness of fuel tube, gaps (in Figure 4.4.1.4-1)

## Figure 4.4.1.4-1 Two-Dimensional Fuel Tube Model


#### 4.4.1.5 Three-Dimensional Transfer Cask and Canister Model

The three-dimensional transfer cask and canister model comprises the transfer cask, the canister and the canister internals. This model is one half (90°) of the three-dimensional canister model (Section 4.4.1.2) with the transfer cask added. The region below the active fuel is modeled with solid elements. The radiation from the bottom of the fuel to the top of the canister bottom plate is considered using the method described in Section 4.4.1.2. The trunnions and the retaining ring at the top of the transfer cask are not included in the model. This is conservative, since it reduces the surface area for radiation and convection. The model is used to calculate the transient temperature distribution for the fuel cladding and the components of the transfer cask, canister and basket for the transfer condition when the canister is back-filled with helium. The model is shown in Figure 4.4.1.5-1. ANSYS SOLID70 elements and LINK31 elements are used to construct the model. Convection and radiation heat transfer are considered at the surfaces of the transfer cask and on the top of the canister lid. The bottom of the transfer cask is conservatively considered to be adiabatic. An ambient temperature of 75°F is assumed and no solar insolance is considered, since the transfer operation occurs inside a building. The canister is considered to be centered in the cavity of the transfer cask. In addition to the gaps inside the canister as described in Section 4.4.1.2, the following two gaps are considered in the model:

### LOCATION GAP SIZE

1.	Gap between transfer cask inner shell and lead	0.063 inch
2.	Gap between the canister outer surface and the	
	transfer cask inner shell	0.43 inch

These two gaps are considered to be filled with air. The 0.063-inch gap size between the transfer cask inner shell and the lead is used based on the dimensional tolerances of the transfer cask design. The gap size of 0.43 inch is based on the nominal dimensions of canister shell and the transfer cask. Radiation heat transfer across the gaps is considered using the same method described in Section 4.4.1.2. Radiation at the transfer cask outer surface and canister lid top surface is accounted for by the expression:

 $q_r = \sigma \epsilon_{eff} AF \left(T_i^4 - T_j^4\right)$ 

where:

 $\sigma = \text{the Stefan-Boltzman constant}$ = 1.19 x 10⁻¹¹ Btu/hr-in²-°R⁴  $\varepsilon_{eff} = \text{emissivity}$ A = surface area F = the gray body shape factor for the surfaces T_i = temperature of the hotter node T_j = temperature of the colder node

Radiation heat transfer from the surface can be incorporated in the model by modifying the convection coefficient as shown below:

 $Q_t = q_r + q_c$ 

where  $q_r$  is specified above for the radiation heat transfer and  $q_c$ , which is the heat transfer by convection, is expressed as:

$$q_c = h_c A(T_i - T_i)$$

where:

 $h_c = film \text{ coefficient (Btu/hr-in²)}$ 

The  $q_r$  can be rewritten as:

$$q_r = \sigma \varepsilon_{eff} AF(T_i^2 + T_j^2)(T_i + T_j)(T_i - T_j)$$

By combining both expressions:

$$Q_{t} = (\sigma \varepsilon_{eff} F(T_{i}^{2} + T_{j}^{2})(T_{i} + T_{j}) + h_{c})A(T_{i} - T_{j})$$

or

$$Q_t = h_{eff} A(T_i - T_i)$$

where:

 $h_{eff} = \sigma \varepsilon_{eff} F(T_i^2 + T_j^2)(T_i + T_j) + h_c$ 

The convection coefficient,  $h_{eff}$ , used for the cask surface now includes the radiation heat transfer. The form factor (F) is taken to be unity.

The convection heat transfer coefficient is determined by the following expression, which is established by empirical correlation for vertical plate and cylinders and horizontal plates (Kreith):

 $h_c = 0.0015 \Delta T^{1/3} (Btu/hr-in^2-{}^{\circ}F)$ 

where  $\Delta T$  is the temperature difference between the cask surface and ambient.

Volumetric heat generation (Btu/hr-in³) is applied to the active fuel region based on a total heat load of 12.5 kW, active fuel length of 91 inches and an axial power distribution, as shown in Figure 4.4.1.1-3.

## Figure 4.4.1.5-1 Three-Dimensional Transfer Cask and Canister Model



### 4.4.1.6 Two-Dimensional Reconfigured Fuel Assembly Model

The two-dimensional reconfigured fuel assembly model is generated to calculate the temperature distribution of the hottest cross-section (1 inch long in the cask axial direction) of the Reconfigured Fuel Assembly (RFA). Because of symmetry, the model considers one-fourth of a cross-section of the RFA. The model is shown in Figure 4.4.1.6-1. ANSYS 'PLANE55' conduction elements and "LINK31" radiation elements are used in the model. The model includes a total of 16 fuel rods, 16 fuel tubes, the shell casing (the square tube with the same external dimensions as an intact fuel assembly) and the cover gas (considered to be Helium). Each fuel rod is located inside a stainless steel fuel tube. The fuel rod, which consists of the zircaloy clad, the fuel pellet (UO₂) and a small gap between the clad and fuel pellet, is modeled as a solid rod with the thermal conductivity of the UO₂. This is conservative, since the conductivity of UO₂ is less than that of the zircaloy and the main interest of the fuel rod is the cladding temperature. The gas between the fuel rod and the fuel tube, the gas between fuel tubes and the gas outside of the shell casing are considered to be helium.

As shown in Figure 4.4.1.6-1, radiation elements are defined between tubes and from tubes to the inner surface of the shell casing. Form factor of 1 is used for the radiation elements. Effective emissivity is computed using the following formula (Kreith) based on corresponding material emissivities:

 $\varepsilon_{\rm eff} = 1/(1/\varepsilon_1 + 1/\varepsilon_2 - 1)$ 

where  $\in_1$  and  $\in_2$  are the emissivities of two parallel plates.

Radiation between the fuel rod and the fuel tube is conservatively ignored. Radiation between the shell casing and the inner surface of the fuel assembly tube is accounted for by establishing effective conductivities for the gas in the gap using the method described in Section 4.4.1.2.

Volumetric heat generation (Btu/hr-in³) based on the design heat load of 0.0016 kW/pin is applied to the fuel rod elements. An active fuel length of 91 inches and a peaking factor of 1.15 are used.

Heat generation rate = Q / V= 0.6595 Btu/hr-in³

where:

Q = heat rate per pin (unit height) = (0.0016) (3413) (1.15) /(91) = 0.069 Btu/hr V = volume of pin (unit height) =  $\pi 0.365^2$  /4 = 0.1046 inch³

Boundaries of the model at planes of symmetry (at X=0 and at Y=0) are considered to be adiabatic. The temperature at the right and top boundaries (at X=3.9 inch and at Y=3.9 inch) of the model is constrained to be uniform based on the maximum temperatures of the fuel tube as shown below. These temperatures envelope the calculated maximum tube temperatures for the design basis Yankee Class fuel assembly and are conservative, since the heat load for the RFA (0.102 kW) is less than one-third of the heat load for the design basis fuel assembly (0.347 kW).

	Maximum Fuel	
Design Condition	<u>Tube¹ Temperature (°F)</u>	
Normal	540	
Off-Normal and Accident	580	
Transfer	715	

1. The Fuel Tube is shown in Drawing 455-881.





### 4.4.2 <u>Test Model</u>

The NAC-MPC system is conservatively designed by analysis so that testing is not required.

#### 4.4.3 <u>Maximum Temperatures for Normal Conditions</u>

Normal conditions consider, primarily, the temperature distribution in the spent fuel, the canister and the concrete cask in long-term storage. The routine operations of loading, closing and transfer of the canister to the concrete cask are transient conditions that result in different temperature conditions, depending on the configuration of the canister. In the transient conditions, the maximum temperature of the fuel is maintained below the maximum allowable short-term temperature limit (806°F) by implementing specific actions described in the procedures (Section 8.1.1) and in the Technical Specifications provided in Appendix 12A of Chapter 12. Temperature distributions for the off-normal and accident conditions are presented in Sections 11.1 and 11.2.

#### 4.4.3.1 <u>Maximum Temperatures in Long-Term Storage</u>

Figure 4.4.3-1 shows the temperature distribution of the concrete cask and the canister for the normal, long-term storage condition. The air flow field and air temperatures in the annulus between the canister and the storage cask liner for the normal condition of storage are shown in Figures 4.4.3-2 and 4.4.3-3, respectively. The temperature distribution in the concrete portion of the storage cask is shown in Figure 4.4.3-4. The maximum component temperatures for the reconfigured fuel assembly are shown in Table 4.4.3-4.

As shown in Figure 4.4.3-3, a high-temperature gradient exists near the wall of the canister and at the liner of the concrete cask. The air in the center of the annulus exhibits a much lower temperature gradient, indicating cooler air. The higher temperature at the concrete cask steel liner surface indicates that radiation heat transfer occurs across the annulus. As shown in Figure 4.4.3-4, the local temperature in the concrete, directly affected by the radiation heat transfer across the annulus, can reach 165°F (347°K), which is less than the 200°F allowable temperature. The bulk temperature in the concrete, as determined by averaging the temperatures at the midradius of the concrete region, is less than the allowable value of 150°F.

#### 4.4.3.2 Maximum Temperatures in Transient Operations

The time history of the maximum temperature of the fuel region in the canister for the transient conditions (the canister containing water for 20 hours, vacuum for 10 hours and helium for 36 hours while in the transfer cask) is shown in Figure 4.4.3-5.

Using the results for the temperature history evaluation presented in Figure 4.4.3-5, the heat up rate of the spent fuel is:

- 5.3°F per hour in the water condition
- 21.9°F per hour in the vacuum condition
- 4.8°F per hour in the helium condition

These heat up rates are used to develop the allowable time limits for completing required procedural steps for loading, closing and removing the canister from the transfer cask that are incorporated in the Technical Specifications provided in Chapter 12. The Technical Specifications specify the remedial actions required to ensure that the fuel cladding temperature does not exceed the maximum allowable short-term temperature limit of 806°F, if these time frames are not met. The actual loading, closing and transfer operations are expected to be completed in less time.

The fuel temperature at the start of the vacuum drying operation will be below 200°F based on the temperature of the water in the canister. The drying process is allowed to continue for 16 hours (LCO 3.1.5), which could result in a maximum fuel temperature of 550°F. The fuel is then placed in a helium condition for 26 hours (LCO 3.1.6), which could result in a fuel temperature of 675°F. The time limits imposed by the LCO thus ensure that the fuel cladding temperature does not exceed the maximum allowable short-term temperature of 806°F.

If the time limits of LCO 3.1.5 for the vacuum drying process are not met, the Technical Specifications require that the transfer cask and canister be returned to the spent fuel pool or that forced air cooling of the canister in the transfer cask be initiated. After 24 hours in either of these conditions, the fuel cladding temperature is expected to be below 466°F. This value is based on the maximum differential temperature between the fuel and the canister shell, as presented in Section 11.1.4.2, and a canister water temperature of 200°F. The vacuum drying process is then allowed to continue for 10 hours (LCO 3.1.5), which could result in a fuel temperature of 685°F. The canister may then be placed in a helium condition for up to 15

4.4-33

NAC-MPC FSAR	April 2000
Docket No. 72-1025	Revision 0

additional hours (LCO 3.1.6), which could result in a fuel temperature of 757°F. The time limits imposed by LCO 3.1.5 and LCO 3.1.6, thus ensure that the fuel cladding temperature does not exceed the maximum allowable short-term temperature limit of 806°F.

Consequently, for all of the transient conditions, the maximum fuel temperature, as controlled by the LCO time limits, is significantly below the maximum short-term temperature limit for the fuel of 806°F.





## Figure 4.4.3-2 Air Flow Pattern in the Storage Cask in Normal Storage





<u>°F</u>

I= 187 J= 201

79

#### Air Temperature Field in the Storage Cask During the Normal Storage Figure 4.4.3-3 Condition



.

# Figure 4.4.3-4 Concrete Temperature Field During the Normal Storage Condition



<u>°F</u>
MX= 165
*A= 77
*B= 82
C= 87
D= 92
E= 97
F= 102
G= 107
H= 112
I= 117
<b>J</b> = 122
K= 127
L= 132
M= 137
N= 142
O= 147
P= 152
Q= 157
R= 162

MX = Maximum temperature

*These temperatures occur in the lower edge of the concrete above the inlet vent. They are not shown due to the scale of the figure.

4.4-38

## Figure 4.4.3-5 History of Maximum Component Temperatures for the Nominal Transfer Conditions



- 1. Temperature of Support Disks
- 2. Temperature of Aluminum Disks
- 3. Temperature of the Fuel
- 4. Temperature of the Canister
- 5. Temperature of the Inner Shell
- 6. Temperature of the Lead
- 7. Temperature of the Neutron Shield
- 8. Temperature of Transfer Cask Doors

Component	Maximum Temperature (°F)	Allowable Temperature (°F)
Fuel Cladding	563	644
Aluminum Disk	527	650
Support Disk	529	650
Canister	319	800
Concrete Liner (steel)	165	700
Concrete	165 (local)	200 (local)
	133 (bulk*)	150 (bulk)

## Table 4.4.3-1 Maximum Component Temperatures for the Normal Condition of Storage

* The average temperature of the concrete region is used as the bulk concrete temperature.

1

Component	Maximum Temperature (°F) ¹	Allowable Temperature (°F)
Fuel	597	806
Lead	191	600
Neutron Shield	188	300
Aluminum Disk	569	700
Support Disk	570	800
Canister	430	800
Transfer Cask Shells	237	700

# Table 4.4.3-2 Maximum Component Temperatures for the Nominal Helium Transfer Condition

1. Maximum temperatures calculated for the nominal transient condition that considers the canister inside of the transfer cask containing water for 20 hours, a vacuum for 10 hours, and helium for 36 hours.

Component	Maximum Temperature (°F) ¹	Allowable Temperature (°F)
Fuel	424	806
Lead	98	600
Neutron Shield	98	300
Aluminum Disk	278	700
Support Disk	279	800
Canister	151	800
Transfer Cask Shells	105	700

## Table 4.4.3-3 Maximum Component Temperatures for the Vacuum Transfer Condition

1. Maximum temperatures calculated for the transient condition that considers the canister inside of the transfer cask containing water for 20 hours and a vacuum for 10 hours.

### Table 4.4.3-4 Maximum Component Temperatures for the Reconfigured Fuel Assembly

	Maximum Temperatures (°F)			
Design Condition	PWR Fuel Tube ^{1,2}	Shell Casing ^{3,4}	Reconfigured Fuel Assembly Tube ^{5,6}	Fuel Rod Cladding ⁷
Normal	540	543	563	563
Off-Normal and Accident Conditions	580	583	602	602
Transfer	715	718	734	734

1. Fuel Tube as shown in Drawing 455-881.

- 2. Bounding fuel tube temperatures as described in Section 4.4.1.6.
- 3. Reconfigured Fuel Assembly Shell Casing as shown in Figure 4.4.1.6-1.
- 4. Material allowable temperature: 800°F.
- 5. Reconfigured Fuel Assembly (Fuel) Tube as shown in Figure 4.4.1.6-1.
- 6. Material allowable temperature: 800°F.
- 7. Fuel cladding allowable temperatures:

Normal conditions: 644°F

Off-normal, Accident and Transfer conditions: 806°F.

### 4.4.4 <u>Minimum Temperatures</u>

Section 11.1 provides the temperature distribution for the off-normal severe cold environmental conditions of  $-40^{\circ}$ F. At this extreme condition, the components are above their minimum material limits.

### 4.4.5 <u>Maximum Internal Pressure for Normal Conditions</u>

The NAC-MPC canister is backfilled with helium to atmospheric pressure (0.0 psig) and closed by welding. Normal condition pressure comprises the pressure, due to the heating of the backfilled helium, plus the pressure due to the postulated failure of 3 percent of the stored fuel rods with the subsequent release of 30 percent of the fission gas and all of the rod charge gas to the canister cavity, at temperature, from those failed rods. All of the gases except the fission gases are assumed to be helium. The total pressure for each volume is found by calculating the molar quantity of each gas and summing those directly. The calculated average temperature of the helium gas is 442°F based on the thermal analysis results using the three-dimensional canister model described in Section 4.4.1.2. The pressure is calculated using the Ideal Gas Law and applying a conservative average temperature of 450°F. The gas constant, R, is 0.0821 (atm x liters)/(Mole °K) (Lamarsh). The design basis fuel assembly for the internal pressure calculation is the CE Type A assembly. This assembly has the highest rod backfill pressure (315 psig) and received the highest burnup (36,000 MWd/MTU).

The number of moles of the backfill gases is calculated using the Ideal Gas Law, PV = NRT. Backfill gas for the canister is assumed to be initially at 1 atmosphere absolute. The quantity of fission gas is derived from the SAS2H generated isotopics of the CE Type A fuel assembly. The release of fission gas is as assumed for directly loaded fuel. For normal operating conditions, 3 percent of the fuel rods are assumed to fail, releasing 30 percent of their total fission gas and all of the backfill helium.

The fuel rod plenum volume is:

$$V_1 = \pi r^2 L - \frac{M_{Spring}}{r}$$

NAC-MPC FSAR Docket No. 72-1025

$$V_{1} = \pi \left\{ \left( \frac{0.317 \text{ inches}}{2} \right)^{2} \times 1.942 \text{ inches} \right\} - \frac{\left( 3.3 \text{ g} \times 2.2046 \times 10^{-3} \frac{\text{lb}}{\text{g}} \right)}{0.288 \frac{\text{lb}}{\text{inch}^{3}}} = 0.1280 \text{ inches}^{3}$$

The pellet clad gap volume is:

$$V_{2} = \pi L \left( r_{\text{Clad ID}}^{2} - r_{\text{Pellet OD}}^{2} \right)$$

$$V_{3} = \pi \times r_{\text{Clad ID}}^{2} \times L$$

$$V_{3} = \pi \times \frac{\left( 0.317 \text{ inches} \right)^{2}}{4} \times 2.458 \text{ inches} = 0.1940 \text{ inches}^{3}$$

The total fuel rod backfill volume is:

 $V_{Rod Back-Fill} = V_1 + V_2 + V_3$ 

 $V_{\text{Rod Back-Fill}} = 0.1280 \text{ inches}^3 + 0.2915 \text{ inches}^3 + 0.1940 \text{ inches}^3 = 0.6135 \text{ inches}^3$ 

For the loaded canister, the total backfill gas volume is:

V = Total Back - Fill 0.6135 inches³ × 231 
$$\frac{\text{Rods}}{\text{Assembly}}$$
 × 36  $\frac{\text{Assemblies}}{\text{Cask}}$  ×  $\left(2.54 \frac{\text{cm}}{\text{inch}}\right)^3$  ×  $\frac{0.001 \ell}{\text{cm}^3}$   
= 83.605  $\frac{\ell}{\text{Cask}}$ 

From the rod backfill volume and pressure, the quantity of rod backfill gas is calculated using the ideal gas law:

$$N = \frac{Pv}{RT}$$

$$N = \frac{\left\{ (315 \text{ psig} + 14.7) \times \frac{1 \text{ atm}}{14.7 \text{ psia}} \right\} \times 83.605 \frac{\ell}{Cask}}{0.0821 \frac{\text{atm} \ell}{\text{Mole K}} \times 293 \text{ K}} = 77.95 \frac{\text{Moles of Rod Fill Gas}}{Cask}$$

The number of moles of fission gas per assembly is:

	Atomic Weight	Mass (gram)	Number of
<u>Isotope</u>	(gram/mole)		Moles
KR	83	10.9	0.131
KR	84	29.9	0.356
KR	85	3.54	0.042
KR	86	48.4	0.563
Ι	127	11	0.087
Ι	129	47.2	0.366
XE	130	1.78	0.014
XE	131	112	0.855
XE	132	285	2.159
XE	134	395	2.948
XE	136	577	4.243
Total			11.76

There is a maximum of 36 assemblies per cask. Therefore the number of moles of fission gas per cask is 36 times that of the single assembly.

$$N = 36 \frac{\text{Assemblies}}{\text{Cask}} \times 11.76 \frac{\text{Moles}}{\text{Assembly}} = 423.44 \frac{\text{Moles of Fission Gas}}{\text{Cask}}$$

The canister is backfilled to 1.5 atmosphere with helium at room temperature for leak testing, after which the pressure is reduced to 1 atmosphere. During leak testing the temperature of the helium may rise above the 150°F assumed in the pressure evaluation.

NAC-MPC FSAR Docket No. 72-1025

н

$$V_{\text{Free Gas Volume}}^{\text{TSC}} = V_{\text{Canister}} - \left(\frac{\left(M_{\text{TSC Shield Lid}} + M_{\text{TSC Structural Lid}}\right)}{\rho_{\text{Steel}}} + V_{\text{Basket}}^{\text{TSC}} + V_{\text{Fuel}}\right)$$

$$V_{\text{Canister}} = \pi \frac{d^2}{4} \left( L_{\text{Canister}} - L_{\text{TSC Bottom Plate}} \right) = \pi \times \frac{(69.39 \text{ inches})^2}{4} \times (122.50 \text{ inches} - 1.0 \text{ inch})$$

$$V_{Canister} = 459,472.93 \text{ inches}^3$$

$$V_{\text{Basket}}^{\text{TSC}} = \frac{\left(M_{\text{BORAL}}^{\text{TSC}} - \left(M_{\text{BORAL}} + M_{\text{Aluminum}} + M_{\text{Support Disks}}\right)\right)}{\rho_{\text{Steel}}} + \frac{M_{\text{BORAL}}}{\rho_{\text{BORAL}}} + \frac{M_{\text{Aluminum}}}{\rho_{\text{Aluminum}}} + \frac{M_{\text{Support Disk}}}{\rho_{\text{17-4-PH}}}$$

$$V_{\text{Basket}}^{\text{TSC}} = \frac{(9,530 \text{ lb} - (694.81 \text{ lb} + 810 \text{ lb} + 3,720 \text{ lb}))}{0.288 \frac{\text{lb}}{\text{in}^3}} + \frac{694.81 \text{ lb}}{0.095 \frac{\text{lb}}{\text{in}^3}} + \frac{810 \text{ lb}}{0.098 \frac{\text{lb}}{\text{in}^3}} + \frac{3,720 \text{ lb}}{0.282 \frac{\text{lb}}{\text{in}^3}}$$

$$V_{Basket}^{TSC}$$
 = 43,719.16 inches³

$$V_{\text{Free Gas Volume}}^{\text{TSC}} = 459,472.93 - \left(\frac{(5,390 \text{ lb} + 3,230 \text{ lb})}{0.288 \frac{\text{lb}}{\text{in}^3}} + 43,719.16 \text{ inches}^3 + 88,171.78 \text{ inches}^3\right)$$

$$V_{\text{Free Gas Volume}}^{\text{TSC}} = 297,651.43 \frac{\text{inches}^3}{\text{Cask}}$$

$$V_{\text{Free Gas Volume}}^{\text{TSC}} = 297,651.43 \ \frac{\text{inches}^3}{\text{Cask}} \times \frac{1 \ \ell}{61.02 \ \text{inches}^3} = 4,877.93 \ \frac{\ell}{\text{Cask}}$$

$$N = \frac{1 \text{ atm} \times 4,877.93 \quad \frac{\ell}{Cask}}{0.0821 \frac{\text{ atm} \ \ell}{Mole \ K} \times 339 \ K} = 175.26 \quad \frac{\text{Moles of TSC Backfill Gas}}{Cask}$$

The maximum normal operating pressure (MNOP) in the canister is calculated using the ideal gas law where:

$$N = N_{TSC Back-Fill} + 0.03 (N_{Rod Back-Fill}) + 0.3(0.03) (N_{Fission Gas})$$

$$N = 175.26 \frac{Moles}{Cask} + 0.03 (77.95 \frac{Moles}{Cask}) + 0.3(0.03) (423.44 \frac{Moles}{Cask})$$

$$N = 181.41 \frac{Moles}{Cask}$$

Therefore, the maximum normal operating condition canister internal pressure is:

$$P = \frac{\left(181.41 \frac{\text{Moles}}{\text{Cask}}\right) \times \left(0.0821 \frac{\text{atm}\,\ell}{\text{mole}\,\text{K}}\right) \times 505.37\,\text{K}}{\left(4,877.93 \frac{\ell}{\text{Cask}}\right)} = 1.54\,\text{atm} \approx 22.6\,\text{psia} \approx 7.9\,\text{psig}$$

## 4.4.6 <u>Maximum Thermal Stresses for Normal Conditions</u>

The canister and concrete storage cask thermal stresses are evaluated in Section 3.4.4.

## 4.4.7 Evaluation of Cask Performance for Normal Conditions of Storage

As shown in the preceding sections, the NAC-MPC system operates within the thermal design limits. Therefore, no degradation due to temperature effects on material or components is expected over the lifetime of the cask.

4.5 <u>References</u>

AAR Advanced Structures, Manufacturer's Data, BORAL-Product Performance Report 624, Standard Specification for BORAL Composite Sheet, Rev. 0, 1994.

ACI 349-85, American Concrete Institute, "Code Requirement for Nuclear Safety Related Concrete Structures and Commentary."

ARMCO, Product Data Bulletin No. S-22, "17-4PH, Precipitation-Hardening Stainless Steel," ARMCO, Inc., 1988.

"ASME Boiler and Pressure Vessel Code," Section II, Part D, The American Society of Mechanical Engineers, 1995.

"ASME Boiler and Pressure Vessel Code," Section III, Appendix I, The American Society of Mechanical Engineers, 1992.

"ASME Boiler and Pressure Vessel Code, Code Cases - Boilers and Pressure Vessels," The American Society of Mechanical Engineers, 1989.

ASTM-C 150-95a, American Society for Testing and Materials, "Standard Specification for Portland Cement."

Baumeister, T., and L.S. Marks, <u>Standard Handbook for Mechanical Engineers</u>, Seventh Edition, McGraw-Hill Book Company, New York, 1967.

BISCO Products Data (NS-4-FR is now supplied by Genden Engineering Services and Construction Company).

Bucholz, J. A., "Scoping Design Analyses for Optimized Shipping Casks Containing 1-, 2-, 3-, 5-, 7-, or 10-Year-Old PWR Spent Fuel," Oak Ridge National Laboratory, ORNL/CSD/TM-149, 1983.

Edwards, A. L., "Compilation of Thermal Property Data for Computer Heat-Conduction Calculations," Lawrence Livermore Radiation Laboratory, February 1969.

EPRI TR-106440, "Evaluation of Expected Behavior of LWR Stainless Steel-Clad Fuel in Long-Term Dry Storage," Electric Power Research Institute, Palo Alto, CA, April 1996.

Fintel, M., <u>Handbook of Concrete Engineering</u>, Second Edition, Van Nostrand Reinhold Co., New York.

Genden Engineering Services and Construction Company, Tokyo, Japan.

Hanford Engineering Development Laboratory, Richland, Washington, 1976. "Nuclear Systems Materials Handbook," Volume 1, Design Data.

Incropera, E. P., and D.P. DeWitt, Fundamentals of Heat and Mass Transfer, Fourth Edition.

Kreith and Bohn, <u>Principles of Heat Transfer</u>, Fifth Edition, International Textbook Company, Scranton, Pennsylvania, 1993.

Lamarsh, John R., "Introduction to Nuclear Engineering," Second Edition, Addison-Wesley Publishing Company, 1983.

MIL-HDBK-5G, Military Handbook, "Metallic Materials and Elements for Aerospace Vehicle Structures," November 1994.

NUREG-1536, "Standard Review Plan for Dry Cask Storage Systems," January 1997.

NUREG/CR-0497, "A Handbook of Materials Properties for Use in the Analysis of LWR Fuel Rod Behavior," compiled and edited by D. L. Hagrman and G. A. Reymann, EG&G Idaho, Inc., 1979.

Olander, Donald R., "Fundamental Aspects of Nuclear Reactors Fuel Elements," Technical Information Center (U. S. Department of Energy), 1985.

PNL-4835, "Technical Basis for Storage of Zircaloy-Clad Spent Fuel in Inert Gases," 1983.

PNL-6189, "Recommended Temperature Limits for Dry Storage of Spent Light Water Reactor Zircaloy Clad Fuel Rods in Inert Gas," 1987.

Ross, R. B., "Metallic Material Specification Handbook," Fourth Edition, Ross Material Technology, Glasgow, Chapman & Hall, 1992, (Rev. 1).

Sandia 90-2406, Sanders, T. L., et al., "A Method for Determining the Spent-Fuel Contribution to Transport Cask Containment Requirements," TTC-1019, UC-820, November 1992.

Siegel, R. and Howell, J. R., Thermal Radiation Heat Transfer, Third Edition, Hemisphere Publishing Company, 1992.

1

THIS PAGE INTENTIONALLY LEFT BLANK

NAC-MPC FSAR Docket No. 72-1025

### **Table of Contents**

5.0	SHIE	LDING EVALUATION
5.1	Discu	ssion and Results
5.2	Sourc	e Specification
	5.2.1	Gamma Source
	5.2.2	Neutron Source
	5.2.3	Source Axial Profile
5.3	Mode	l Specification
	5.3.1	Description of the Radial and Axial Shielding Configurations5.3-2
		5.3.1.1 One-Dimensional Radial and Axial Shielding Models
		5.3.1.2 Three-Dimensional Top and Bottom Shielding Models
	5.3.2	Shield Regional Densities
5.4	Shield	ling Evaluation
	5.4.1	Calculational Methods
	5.4.2	Flux-to-Dose Rate Conversion Factors
	5.4.3	Dose Rates
		5.4.3.1 One-Dimensional Storage Cask Dose Rates
		5.4.3.2 Three-Dimensional Storage Cask Dose Rates
		5.4.3.3 One-Dimensional Transfer Cask Dose Rates
		5.4.3.4 Three-Dimensional Transfer Cask Dose Rates
	5.4.4	Storage Cask Shielded Source Terms5.4-7
5.5	Refere	ences

# List of Figures

Figure 5.2-1	Yankee Class Combustion Engineering Design Basis Fuel Assembly Array 5.2-4
Figure 5.2-2	Yankee Class Combustion Engineering Design Basis Fuel Assembly
	Source Regions and Elevations
Figure 5.2-3	NAC-MPC Design Basis Burnup Profile
Figure 5.2-4	NAC-MPC Design Basis Neutron and Gamma Source Axial Profiles
Figure 5.3-1	NAC-MPC Storage Cask Three-Dimensional Top Model5.3-9
Figure 5.3-2	NAC-MPC Storage Cask Three-Dimensional Bottom Model
Figure 5.3-3	NAC-MPC Transfer Cask Three-Dimensional Top Model Including
	Shield and Structural Lid
Figure 5.3-4	NAC-MPC Transfer Cask Three -Dimensional Bottom Model
Figure 5.4-1	Storage Cask Radial Dose Rate Profile
Figure 5.4-2	Storage Cask Three-Dimensional Model Surface Dose Rate Profile5.4-9
Figure 5.4-3	Storage Cask Top Outlet Vent Dose Rate Profile
Figure 5.4-4	Storage Cask Bottom Inlet Vent Dose Rate Profile
Figure 5.4-5	Storage Cask Dose Rate Profile at Stand Cutout Elevation
Figure 5.4-6	Storage Cask Top Dose Rate Profile as a Function of Radius from the
	Centerline and Distance from Surface
Figure 5.4-7	Storage Cask Top Surface Dose Rate Profile by Source Component
Figure 5.4-8	Transfer Cask Side Dose Rate Profile as a Function of Elevation
	and Distance, Wet Cavity
Figure 5.4-9	Transfer Cask Side Dose Rate Profile by Source Component, Wet Cavity5.4-16
Figure 5.4-10	Transfer Cask Side Dose Rate Profile as a Function of Elevation
	and Distance, Dry Cavity
Figure 5.4-11	Transfer Cask Side Dose Rate Profile by Source Component, Dry Cavity5.4-18
Figure 5.4-12	Transfer Cask Top Surface Dose Rate as a Function of Radius and
	Distance from Surface, Shield Lid, Structural Lid, and Temporary
	Shield On, Dry Cavity
Figure 5.4-13	Transfer Cask Top Surface Dose Rate by Source Component,
	Shield Lid, Structural Lid, and Temporary Shield On, Dry Cavity
Figure 5.4-14	Transfer Cask Bottom Surface Dose Rate, Wet Cavity
Figure 5.4-15	Transfer Cask Bottom Surface Dose Rate, Dry Cavity

NAC-MPC FSAR Docket No. 72-1025

## List of Tables

Table 5.1-1	Summary of NAC-MPC Storage Cask Maximum Dose Rates	
	with Design Basis Fuel	5.1-5
Table 5.1-2	Summary of NAC-MPC Transfer Cask Maximum Dose Rates	
	with Design Basis Fuel	5.1-5
Table 5.2-1	Yankee Class Fuel Assembly Physical Parameters	5.2-8
Table 5.2-2	Yankee Class Design Basis Fuel Characteristics for Shielding Eva	luations 5.2-9
Table 5.2-3	Yankee Class Design Basis Fuel Reactor Operating Conditions	5.2-10
Table 5.2-4	NAC-MPC Design Basis Fuel Source Terms	5.2-10
Table 5.2-5	NAC-MPC Design Basis Fuel Gamma Source Spectra	5.2-11
Table 5.2-6	NAC-MPC Design Basis Fuel Neutron Source Spectra	5.2-12
Table 5.3-1	Fuel/Basket Regional Volumes (cm ³ )	5.3-13
Table 5.3-2	Fuel/Basket Regional Masses (kg)	5.3-13
Table 5.3-3	Regional Homogenized Densities and Shield Densities	5.3-14
Table 5.4-1	ANSI Standard Neutron Flux-to-Dose Rate Factors	5.4-23
Table 5.4-2	ANSI Standard Gamma Flux-to-Dose Rate Factors	5.4-24
Table 5.4-3	NAC-MPC Storage Cask One-Dimensional Projectile Accident R	adial Dose
	Rates	5.4-24
Table 5.4-4	NAC-MPC Shielded Gamma Flux	5.4-25
Table 5.4-5	NAC-MPC Shielded Neutron Flux	5.4-26

THIS PAGE INTENTIONALLY LEFT BLANK

•

_____

#### 5.0

# SHIELDING EVALUATION

#### 5.1 Discussion and Results

The regulation governing spent fuel storage, 10 CFR 72, does not establish specific cask dose rate limits. However, 10 CFR 72.104 and 10 CFR 72.106 specify that for an array of casks in an Independent Spent Fuel Storage Installation (ISFSI), the annual dose to an individual outside the controlled area boundary must not exceed 25 mrem to the whole body, 75 mrem to the thyroid and 25 mrem to any other organ during normal operations. In the case of a design basis accident, the dose to an individual outside the area boundary must not exceed 5 rem to the whole body or any organ. The ISFSI must be at least 100 meters from the owner controlled area boundary. In addition, the occupational dose limits and radiation dose limits for individual members of the public in 10 CFR Part 20 (Subparts C and D) must be met. Chapter 10, Section 10.3, demonstrates NAC-MPC compliance with the requirements of 10 CFR 72 with regard to annual and occupational doses at the owner controlled area boundary. This chapter presents the shielding evaluations of the NAC-MPC storage system. Dose rate profiles are calculated as a function of distance from the side, top and bottom of the NAC-MPC storage and transfer casks. Shielded source terms from the NAC-MPC storage cask are calculated to establish owner controlled area boundary dose estimates due to the presence of the ISFSI.

The NAC-MPC accommodates up to 36 CE Yankee Class fuel assemblies with a maximum of 36,000 MWD/MTU burnup and with a minimum of 8 years cool time. While 8.1 years cooling is required to meet cask total heat load requirements, 8.0 years is conservatively used as the shielding design basis. CE fuel with this burnup and cool time is defined as the design basis fuel. CE, UN and Westinghouse Yankee Class fuel assemblies with a maximum burnup of 32,000 MWD/MTU at minimum cool times of 7.0, 7.1 and 21.0 years, respectively, may also be loaded in the NAC-MPC. Exxon fuel at 36,000 MWD/MTU requires a minimum cool time of 9 years and 16 years for assemblies containing Zircaloy and stainless steel fuel region hardware, respectively. For shielding evaluation purposes the Exxon assembly type is identical to the CE fuel. The physical parameters of the Yankee Class fuel assemblies are presented in Table 5.2-1.

The NAC-MPC storage system is comprised of a transportable storage canister, a transfer cask, and a vertical concrete storage cask. License drawings for these items are provided in Section 1.5. The transfer cask containing the canister and the basket is loaded under water in the spent fuel pool. Once filled with fuel, the shield lid is placed on top of the canister and transfer cask is removed from the pool. After draining about 12 inches of water (approximately 50 gallons) from the cavity, the shield lid is welded in place, and the canister is drained and dried. Finally, the

structural lid is welded in place. The transfer cask is then used to transfer the canister to the storage cask where it is stored dry until transport. Shielding evaluations are performed for the transfer cask with both a wet and dry canister cavity as would occur during the welding of the shield lid and during the welding of the structural lid, respectively. Shielding evaluations are performed for the storage cask with the cavity dry.

A canister may contain one or more reconfigured fuel assemblies. The reconfigured fuel assembly is designed to confine Yankee Class spent fuel rods, or portions thereof, which have been classified as failed. Each assembly can accommodate up to a total of 64 fuel pins, which is significantly less than other Yankee Class fuel assemblies. A depiction of the assembly is provided in Figure 1.2-5. Because the source term (neutron and gamma) is directly proportional to fuel mass, for a given burnup and enrichment, the reconfigured assembly source term is bounded by that of a design basis fuel assembly. Consequently, a separate shielding analysis is not required for the reconfigured fuel assembly.

The transfer cask has a multiwall radial shield comprised of 0.75 inches of carbon steel, 3.5 inches of lead, 2 inches of solid borated polymer (NS-4-FR), and 1.25 inches of carbon steel. An additional 0.625 inch of stainless steel shielding is provided, radially, by the canister shell. Gamma shielding is provided primarily by the steel and lead layers, and neutron shielding is provided primarily by the NS-4-FR. The transfer cask bottom shield design is a solid section of 9.50 inches of carbon steel. The top shielding is provided by the stainless steel canister shield and structural lids, which are 5 inches and 3 inches thick, respectively. In addition, 5 inches of carbon steel is used as temporary shielding during welding, draining, drying and helium backfill operations. This temporary shielding is removed prior to storage.

The storage cask radial shield design is comprised of a 3.5-inch thick carbon steel inner liner surrounded by 21 inches of concrete. Gamma shielding is provided by both the carbon steel and concrete, and neutron shielding is provided primarily by the concrete. As in the transfer cask, an additional 0.625 inch thickness of stainless steel radial gamma shielding is provided by the canister shell. The storage cask top shielding design is comprised of 8 inches of stainless steel from the canister lids, a shield plug containing a 1 inch thickness of NS-4-FR and 4.125 inches of carbon steel, and a 1.5 inch thick carbon steel lid. Since the bottom of the storage cask sits on a concrete pad, the storage cask bottom shielding is comprised of 1 inch of stainless steel from the canister bottom plate, 2 inches of carbon steel (pedestal plate) and 1 inch of carbon steel cask base plate. The base plate and pedestal base are structural components that position the canister above the air inlets. The cask base plate supports the storage cask during lifting, and forms the cooling air inlet channels at the cask bottom.

Shielding evaluations of the NAC-MPC transfer and storage casks are performed with SCALE 4.3 for the PC (ORNL). In particular, the SCALE shielding analysis sequence SAS2H (Herman) is used to generate source terms for the design basis fuel, using the 27 group ENDF/B-IV (Jordan) library, 27GROUPNDF4. SAS1 (Knight) is used to perform one-dimensional radial and axial shielding analysis, and a modified version of SAS4 (Tang) is used to perform threedimensional shielding analysis. The SCALE 4/3 SAS4 code sequence has been modified to allow multiple surface detectors; the new code sequence is entitled SAS4A. The use of the surface subdetectors enables the user to obtain surface profiles of the detector response and the surface tallies on the cask surfaces other than the cask shield. Dose tally routines were modified to accept user-defined surface detectors instead of the fixed surfaces, 1m, 2m, and 4m detectors in SCALE 4.3 SAS4. Each surface can be defined by specifying the cylindrical or disk surface tally location and extent. Each surface may be broken into multiple subdetectors. Code modifications were tested against the SCALE 4.3 manual and NAC test cases. Reliability of the subdetectors was verified by comparison to point detector results. The 27 group neutron, 18 group gamma, coupled cross section library (27N-18COUPLE) based on ENDF/B-IV is used in all shielding evaluations. Source terms include: fuel neutron, fuel gamma, and activated hardware gamma. Dose rate evaluations include the effect of fuel burnup peaking on fuel neutron and gamma source terms.

Dose rate profiles are shown for the storage and transfer casks in Section 5.4.3. Maximum dose rates for the storage cask under normal and accident conditions are shown in Table 5.1-1 for design basis fuel. These dose rates are based on three-dimensional Monte Carlo and one-dimensional discrete ordinates calculations. Monte Carlo error  $(1\sigma)$  is indicated in parenthesis. In normal conditions with design basis fuel, the storage cask maximum side dose rate is 47.3 (0.4%) mrem/hr at the bottom endfitting elevation and 54.0 (4.9%) mrem/hr on the top lid surface just above the heat transfer annulus. Since the storage cask is vertical during normal storage operation, the bottom is inaccessible. The dose rates at the inlet and outlet vents are 99.0 (5.4%) mrem/hr and 23.8 mrem/hr (5.0%) due to radiation streaming. Under accident conditions involving a projectile impact and a loss of 6 inches of concrete, the surface dose rate increases to 314 mrem/hr with design basis fuel. There are no design basis accidents that result in a tip-over of the NAC-MPC storage cask.

Maximum dose rates for the transfer cask with design basis fuel and with a wet and dry canister cavity are shown in Table 5.1-2. The maximum dose rates with design basis fuel and the canister cavity wet during shield lid welding operations are 210.2 (0.8%), 188.7 (1.1%) and 77.2 (0.7%)

NAC-MPC FSAR	April 2000
Docket No. 72-1025	Revision 0

mrem/hr on the side, top, and bottom, respectively. The maximum dose rates with design basis fuel and the canister cavity dry during structural lid welding operations are 413.4 (1.5%), 358.9 (2.6%) and 398.0 (3.9%) mrem/hr on the side, top, and bottom, respectively. These values include the addition of 5 inches of carbon steel operational shielding installed on the shield lid during its closure and on the structural lid during its handling and closure. In normal operations during welding of the canister lids, the bottom of the transfer cask is generally inaccessible.
		Cas	sk Surface (mrem/)	hr)	1 Meter	r From Surface (r	n Surface (mrem/hr)	
Condition	Source	Side	Тор	Bottom ¹	Side	Тор	Bottom ¹	
	neutron	0.3	42.5	32.0	0.3	11.4	2.3	
NORMAL	gamma	47.0	11.5	67.0	21.5	4.0	2.8	
	Total	47.3 (0.4%)	54.0 (4.9%)	99.0 ( 5.4%)	21.8 (1.3%)	15.4 (2.5%)	5.1 ( 6.0%)	
POSTULATED	neutron	4.0	na	na	1.5	na	na	
ACCIDENT ²	gamma	310.0	na	na	137.5	na	na	
	Total	314.0	na	na	139.0	na	na	

#### Table 5.1-1 Summary of NAC-MPC Storage Cask Maximum Dose Rates with Design Basis Fuel

Table 5.1-2 Summary of NAC-MPC Transfer Cask Maximum Dose Rates with Design Basis Fuel

		Casl	k Surface (mrem/l	hr)	1 Meter	From Surface (n	nrem/hr)
Condition	Source	Side	Тор	Bottom	Side	Тор	Bottom
NORMAL	neutron	0.0	1.2	0.2	0.7	0.3	0.0
WET ³	gamma	210.2	187.5	77.0	39.8	388.8	19.0
	Total	210.2 (0.8%)	188.7 (1.1%)	77.2 (0.7%)	40.5 (0.7%)	389.1 (5.1%)	19.0 (2.1%)
Normal	neutron	77.5	116.6	276.2	44.8	13.5	33.4
Dry ⁴	gamma	335.9	242.3	121.8	58.6	26.1	33.3
	Total	413.4 (1.5%)	358.9 (2.6%)	398.0 (3.9%)	103.4 (0.6%)	39.6 (4.5%)	66.7 (3.6 %)

¹ Bottom surface is inaccessible. Dose rates adjacent to bottom inlet indicated.
² Projectile impact, 6 inches loss of concrete.
³ 5 inches of carbon steel temporary shielding, shield lid in position.
⁴ 5 inches of carbon steel temporary shielding, shield lid and structural lid in position.

THIS PAGE INTENTIONALLY LEFT BLANK

### 5.2 <u>Source Specification</u>

The NAC-MPC storage system can safely transfer and store Yankee Class fuel from four vendors in two fuel rod configurations. These are: Combustion Engineering (CE) 16 x 16 Type A and Type B, Exxon 16 x 16 Type A and Type B, United Nuclear (UN) 16 x 16 Type A and Type B, and Westinghouse (WE) 18 x 18 stainless steel clad Type A and Type B. The geometry of the Type A and Type B fuel assemblies allows a cruciform control rod to be inserted between assemblies during reactor operation. The cross-section of a typical Yankee Class fuel assembly is shown in Figure 5.2-1.

The NAC-MPC accommodates up to 36 CE Yankee Class fuel assemblies with a maximum of 36,000 MWD/MTU burnup and with a minimum of 8 years cool time. While 8.1 years cooling is required to meet cask total heat load requirements, 8.0 years is conservatively used as the shielding design basis. CE fuel with this burnup and cool time is defined as the design basis fuel. CE, UN and Westinghouse Yankee Class fuel assemblies with a maximum burnup of 32,000 MWD/MTU at minimum cool times of 7.0, 7.1 and 21.0 years, respectively, may also be loaded in the NAC-MPC. Exxon fuel at 36,000 MWD/MTU requires a minimum cool time of 9 years and 16 years for assemblies containing Zircaloy and stainless steel fuel region hardware, respectively. For shielding evaluation purposes, the Exxon assembly type is identical to the CE fuel. The physical parameters of the Yankee Class fuel assemblies are presented in Table 5.2-1.

The SAS2H code sequence (Herman) is used to generate source terms. This code sequence is part of the SCALE 4.3 code package for the PC (ORNL). SAS2H includes an XSDRNPM (Greene) neutronics model of the fuel assembly and ORIGEN-S (Herman) fuel depletion/source terms calculations. The 27 energy group ENDF/B-IV neutron cross section library, 27GROUPNDF4, is used in the source terms calculations. Source terms are generated for both  $UO_2$  fuel and fuel assembly hardware. The hardware activation is calculated by light element transmutation using the incore neutron flux spectrum produced by the SAS2H neutronics model. The hardware is assumed to be Type 304 stainless steel with 1.2 g/kg of ⁵⁹Co impurity. The effects of axial flux spectrum and magnitude variation on hardware activation are estimated by flux ratios based on empirical data (Luskic).

An evaluation of the Yankee Class fuel types established the Combustion Engineering (CE) 16 x 16 Type A fuel assembly (Table 5.2-2) at 36,000 MWD/MTU burnup and 8 years cool time as the Yankee Class design basis fuel assembly for the shielding evaluations. A minimum fuel

enrichment of 3.7 wt % ²³⁵U is assumed to maximize the fuel neutron source. Reactor operating conditions assumed for the analysis are shown in Table 5.2-3.

The NAC-MPC design basis fuel source terms are shown Table 5.2-4. Source strengths are defined for five source regions: active fuel, upper end fitting, upper plenum, lower end fitting and lower plenum. The fuel assembly length, active fuel region length and fuel assembly hardware elevations are shown for the design basis fuel assembly in Figure 5.2-2.

## 5.2.1 <u>Gamma Source</u>

The design basis fuel and hardware gamma source spectra are shown in Table 5.2-5. The fuel gamma source contains contributions from both fission products and actinides. The spectra are presented in the 18 group structure consistent with the SCALE 4.3 27N-18COUPLE cross section library. The hardware gamma spectra contains contributions primarily from ⁶⁰Co due to the activation of Type 304 stainless steel with 1.2 g/kg ⁵⁹Co impurity and with some minor contributions from ⁵⁹Ni and ⁵⁸Fe. The magnitude of this spectra is based on the irradiation of 1 kg of stainless steel in the incore flux spectrum produced by the SAS2H neutronics calculation.

The activated fuel assembly hardware source terms are found by multiplying the source strength from 1 kilogram by the number of kilograms of steel or inconel material in the plenum, upper end fitting and lower end fitting regions, and by multiplying by a regional flux ratio. The regional flux ratio accounts for the effects of both magnitude and spectrum variation on hardware activation. These ratios are based on empirical data (Luskic). A flux ratio of 0.2 is applied to hardware regions directly adjacent to the active core region, i.e., upper and lower plenum. A flux ratio of 0.1 is applied to hardware regions once removed from the active core region, i.e., upper and lower end fitting region.

## 5.2.2 <u>Neutron Source</u>

The design basis fuel neutron spectrum is shown in Table 5.2-6. The neutron source results from actinide spontaneous fission and from  $(\alpha,n)$  reactions with the oxygen in UO₂. The isotopes ²⁴²Cm and ²⁴⁴Cm characteristically produce all, but a few percent of the spontaneous fission neutrons and  $(\alpha,n)$  source in light water reactor fuel. The next largest contribution is from  $(\alpha,n)$  reactions from ²³⁸Pu. The neutron spectra from spontaneous fission are based on fission spectrum

measurements of ²³⁵U and ²⁵²Cf. Neutron spectra from  $(\alpha,n)$  reactions is based on Po- $\alpha$ -O source measurements. These spectra are included in the ORIGEN-S nuclear data libraries of the SCALE 4.3 code package. The spectra are automatically collapsed from the energy group structure of the data library into that of the SCALE 27 group neutron cross section library (Herman).

### 5.2.3 <u>Source Axial Profile</u>

An enveloping burnup shape for three-dimensional shielding and thermal evaluations is created based on core depletion calculations for the Yankee Class fuel. The normalized burnup profile, averaged over the range from 30,000 to 36,000 MWD/MTU, and the corresponding enveloping shape, is shown in Figure 5.2-3. A burnup peak of 1.15 is found to envelope the design basis fuel axial burnup distribution. The corresponding gamma and neutron source distribution is shown in Figure 5.2-4. The gamma source distribution follows the burnup shape directly and has a 1.15 peaking factor. However, the neutron source distribution peaks to a higher level. Based on SAS2H calculations of the neutron source magnitude as a function of burnup, a 4.2 power dependence of the neutron source on burnup, i.e., neutron source  $\sim B^{4.2}$ , is exhibited. This yields a 1.80 peaking factor for neutrons.

NAC-MPC FSAR	April 2000
Docket No. 72-1025	Revision 0





NAC-MPC FSAR	April 2000
Docket No. 72-1025	Revision 0



Figure 5.2-2 Yankee Class Combustion Engineering Design Basis Fuel Assembly Source Regions and Elevations

<u>SOURCE</u>	<u>REGION</u>
FUEL	Active fuel
UPLM	Upper Plenum
UEF	Upper End Fitting
LPLM	Lower Plenum
LEF	Lower End Fitting

-





Condition: 30,000 - 36,000 MWD/MTU





April 2000 Revision 0 

## Table 5.2-1 Yankee Class Fuel Assembly Physical Parameters

	CE	CE	Exxon	Exxon	Exxon	Exxon	Westinghouse	Westinghouse	United Nuclear	United Nuclear
Parameter	Type A	Туре В	Туре А	Type B	Туре А	Type B	Туре А	Type B	Туре А	Type B
Assembly Configuration	-	-	-	-	-	-	-	-	-	-
Assembly Array	16x16	16x16	16x16	16x16	16x16	16x16	18x18	18x18	16x16	16x16
Max. Enrichment (wt % ²³⁵ U)	3.90	3.90	4.00	4.00	4.00	4.00	4.94	4.94	4.00	4 00
Min. Enrichment (wt % ²³⁵ U)	3.50	3.50	3.50	3.50	3.50	3.50	4.94	4.94	4.00	4.00
Max. MTU	0.2394	0.2384	0.2394	0.2384	0.2394	0.2384	0.2869	0.2860	0.2456	0.2446
Fuel Rod Configuration	-	-	-	-	-	-	-	_	-	
Fuel Rod Pitch (cm)	1.1989	1.1989	1.1989	1.1989	1.1989	1.1989	1.0719	1.0719	1 1887	1 1887
Active Fuel Length (cm)	231.1400	231.1400	231.1400	231.1400	231.1400	231.1400	233.9975	233.9975	231,1400	231 1400
Rod OD (cm)	0.9271	0.9271	0.9271	0.9271	0.9271	0.9271	0.8636	0.8636	0.9271	0.9271
Clad ID (cm)	0.8052	0.8052	0.8052	0.8052	0.8052	0.8052	0.7569	0.7569	0.8052	0.8052
Pellet OD (cm)	0.7887	0.7887	0.7887	0.7887	0.7887	0.7887	0.7468	0.7468	0.7887	0.7887
Diametral Gap (cm)	0.0165	0.0165	0.0165	0.0165	0.0165	0.0165	0.0102	0.0102	0.0165	0.0165
Rods per Assembly	231	230	231	230	231	230	305	304	237	236
Fuel Material	UO ₂	<u> </u>								
Clad Material	Zircaloy	Zircaloy	Zircaloy	Zircaloy	Zircaloy	Zircaloy	SS 348	SS 348	Zircalov	Zircalov
Displacement Rod Configuration	-	-	-	-		-	-	-	-	-
Displacement Rod Material	N/A	Zircalov - 4	Zircalov - 4							
Displacement Rod Diameter (cm)	N/A	0.9271	0.9271							
Number Per Assembly	N/A	2	2							
Guide Bar Configuration	-	-	-	-	-	-	-	-	-	
Guide Bar Material	Zircaloy - 4	Zircaloy - 4	SS 304L	SS 304L	Zircaloy	Zircalov	N/A	N/A	N/A	N/A
Guide Bar Width (cm)	1.0973	1.0973	1.0566	1.0566	1.0566	1.0566	N/A	N/A	N/A	N/A
Guide Bar Shape (cm)	Square	Square	Square	Square	Square	Square	N/A	N/A	N/A	N/A
Number Per Assembly	8	8	8	8	8	8	N/A	N/A	N/A	N/A
Instrument Tube Configuration	-	-	-	-	-	-	-		-	11/A
Instrument Tube ID (cm)	0.9970	0.9970	0.9970	0.9970	0.9970	0.9970	0.9995	0.9995	0.0005	0.0005
Instrument Tube OD (cm)	1.1481	1.1481	1.0884	1.0884	1.0884	1.0884	1 0884	1 0884	1 0994	1 0004
Number Per Assembly	1	1	1	1	1	1	1.0004	1.0004	1.0004	1.0804
Instrument Tube Material	Zircaloy - 4	Zircaloy - 4	SS 304	SS 304	Zircaloy	Zircaloy	SS 304	SS 304	SS 304	

NAC-MPC FSAR Docket No. 72-1025

Parameter	CE Type A
Assembly Configuration	-
Assembly Length (inches)	111.79
Assembly Array	16x16
Minimum Enrichment (wt % ²³⁵ U)	3.70 ³
UO ₂ Fuel Mass (kg) at 95% TD	271.6
Fuel Rod Configuration	-
Fuel Rod Pitch (inches)	0.472
Overall Rod Length (inches)	95.4
Active Fuel Length (inches)	91
Rod OD (inches)	0.365
Clad ID (inches)	0.317
Pellet OD (inches)	0.3105
Diametral Gap (inches)	0.0065
Rods per Assembly	231
Clad Material	Zircaloy
Guide Bar Configuration	-
Guide Bar Material	Zircaloy - 4
Guide Bar Width (inches)	0.432
Guide Bar Length (inches)	96.52
Guide Bar Shape	Square
Number Per Assembly	8
Instrument Tube Configuration	-
Instrument Tube ID (inches)	0.3925
Instrument Tube OD (inches)	0.452
Instrument Tube Length (inches)	97.35
Number Per Assembly	1
Instrument Tube Material	Zircaloy - 4
Hardware Configuration	-
Top Nozzle Material	SS 304
Bottom Nozzle Material	SS 304
Upper Plenum Spring Material	SS 302 ¹
Top Nozzle Length (inches)	7.98
Bottom Nozzle Length (inches)	7.19
Upper Plenum Length (inches)	1.942
Top Nozzle Mass (kg)	5.5
Bottom Nozzle Mass (kg)	5.18
Upper Plenum Spring Mass (kg)	0.762
Upper Plenum Grid Mass (kg)	0.590
Lower Plenum Material	Stainless Steel and Inconel ^{1,2}
Lower Plenum Mass (kg)	1.73 ²
Incore Grid Spacers	Zircaloy -4

Table 5.2-2 Yankee Class Design Basis Fuel Characteristics for Shielding Evaluat	Table 5 2-2	Tahl	le 5 2-2	Yankee Class	Design Bas	is Fuel	Characteristics	for	Shielding	Evalua	tions
----------------------------------------------------------------------------------	-------------	------	----------	--------------	------------	---------	-----------------	-----	-----------	--------	-------

Notes: 1. For simplicity, all Inconel and steel are modeled as Type 304 stainless steel.

- 2. Includes Inconel grid and lower plenum spacer.
- 3. Combustion Engineering fuel may be loaded at a cool time of 8.0 years with a maximum burnup of 32,000 MWD/MTU and a minimum enrichment of 3.5 wt%.

NAC-MPC FSAR	April 2000
Docket No. 72-1025	Revision 0

Table 5.2-3 Yankee Class Design Basis Fuel Reactor Operating Condition
------------------------------------------------------------------------

Assembly Power, MW	8.486
Temperature _{fuel} , °K	797
Temperature _{clad} , °K	600
Temperature _{mod} °K	551
Density _{mod} , g/cc	0.766
Boron, ppm	800
Fuel Burnup, MWD/MTU	36,000
Burnup Cycle, days	2 Cycles of
	496.22 days
Down Time, days	60

## Table 5.2-4 NAC-MPC Design Basis Fuel Source Terms

Decay Heat, kW	12.5
Active Fuel, photons/sec	6.423+16
Active Fuel, neutrons/sec	2.415+9
Upper End fitting, photons/sec	8.330+13
Upper Plenum, photons/sec	2.309+13
Lower End fitting, photons/sec	7.876+13
Lower Plenum, photons/sec	5.242+13

Condition: 36 Combustion Engineering Yankee Class Fuel Assemblies, 8 Years Cooled, and 36,000 MWD/MTU Burnup.

	Gamma Source Spectra ¹			
	Fuel Fuel Hardware			
Group	Photons/sec	Photons/sec-kg		
1	3.7701E+04	0.0000E+00		
2	1.7759E+05	0.0000E+00		
3	9.0547E+05	0.0000E+00		
4	2.2566E+06	0.0000E+00		
5	6.2676E+08	1.0141E-15		
6	5.1211E+09	3.3511E+04		
7	1.0789E+11	2.1611E+07		
8	9.9933E+10	9.5163E-03		
9	4.8070E+12	9.1066E+11		
10	3.4718E+13	3.2247E+12		
11	6.3503E+13	4.3841E+09		
12	8.2333E+14	3.8100E+06		
13	1.1897E+14	1.0971E+07		
14	1.7831E+13	1.7359E+08		
15	2.8386E+13	1.3230E+08		
16	1.0201E+14	2.6645E+09		
17	1.3136E+14	1.1044E+10		
18	4.5899E+14	5.5673E+10		
Total	1.7842E+15	4.2095E+12		

## Table 5.2-5 NAC-MPC Design Basis Fuel Gamma Source Spectra

1. 36,000 MWD/MTU and 8 years cool time.

April 2000 Revision 0

Group	Neutrons/sec ¹		
1	1.2290E+06		
2	1.4080E+07		
3	1.5760E+07		
4	8.7930E+06		
5	1.1840E+07		
6	1.2870E+07		
7	2.5190E+06		
8	0.0000E+00		
9	0.0000E+00		
10	0.0000E+00		
11	0.0000E+00		
12	0.0000E+00		
13	0.0000E+00		
14	0.0000E+00		
15	0.0000E+00		
16	0.0000E+00		
17	0.0000E+00		
18	0.0000E+00		
19	0.0000E+00		
20	0.0000E+00		
21	0.0000E+00		
22	0.0000E+00		
23	0.0000E+00		
24	0.0000E+00		
25	0.0000E+00		
26	0.0000E+00		
<u>27</u>	<u>0.0000E+00</u>		
TOTAL	6.7090E+07		

## Table 5.2-6 NAC-MPC Design Basis Fuel Neutron Source Spectra

1. 36,000 MWD/MTU and 8 year cool time.

#### 5.3 <u>Model Specification</u>

Both one-dimensional SAS1 and three-dimensional SAS4 models are used in the shielding evaluations of the NAC-MPC storage system. The SAS1 radial and axial models are used to estimate the peak and average dose rates on the sides, top and bottom of the storage and transfer casks, and to determine the minimum cool time for the 36,000 MWD/MTU fuel. The one-dimensional models represent the casks as either semi-infinite cylinders or slabs. The method of solution uses the XSDRNPM (Greene) discrete ordinates code and the XSDOSE (Buckholz) flux at a point estimation code. Bucklings are applied to the SAS1 models to account for transverse leakage. One-dimensional analysis also serves as a cross check to the more complex three-dimensional model results.

The SAS4 three-dimensional shielding models are used to estimate the dose profiles at the surfaces of the cask and at streaming paths such as the storage cask inlets and outlets, or the canister vent and drain ports. The method of solution is adjoint discrete ordinates and Monte Carlo (Tang) using the XSDRNPM and MORSE codes, respectively. Since SAS4 requires model symmetry at the fuel midplane, two models are created for each cask, a top and a bottom model. Radial biasing is performed to estimate dose rates on the sides of the cask, and axial biasing is performed to estimate dose rates on the top and bottom surfaces of the cask. Modifications are made to SAS4 to tally dose rates all along the radial, top and bottom surfaces of the cask as well as any cylindrical surface surrounding the cask. Thus, detailed dose rate profiles are determined that explicitly show peaks due to the fuel burnup profile, activated hardware gamma emission and streaming paths.

In both SAS1 and SAS4 models, the fuel and hardware source regions are homogenized within the volumes defined by the periphery of the basket tubes and by the elevations of the basket heat transfer zone, the active fuel, the plenum and the end fittings (Table 5.3-1). Within these volumes, the material masses of the fuel assembly and basket are homogenized (Table 5.3-2). The design basis fuel assembly and the NAC-MPC basket materials are obtained from Table 5.2-2 and from drawings 455-881, 455-891, 455-892, 455-893, 455-894 and 455-895 (Section 1.5). The SCALE 4.3 standard composition library (Landers) default compositions and isotopic distributions are used except for BORAL, NS-4-FR and concrete (Section 5.3.2). The resultant material zones and nuclide densities are summarized in Table 5.3-3. In all models, the cask and canister shield thicknesses are explicitly represented.

5.3-1

Both the SAS1 and SAS4 models utilize fuel midplane symmetry. Thus, all shielding models are developed with respect to the fuel midplane as the origin. This symmetry is required in the SAS4 models due to the automated biasing techniques employed and because the dose rate tallies from the symmetric halves are averaged together for computational efficiency.

### 5.3.1 Description of Radial and Axial Shielding Configurations

The NAC-MPC storage cask has an interior cavity with a radius of 39.5 inches (100.33 cm). Radial shielding consists of a 3.5-inch (8.89 cm) carbon steel shell surrounded by 21 inches (53.34 cm) of concrete. Gamma shielding is provided by both the carbon steel and concrete, and neutron shielding is provided primarily by the concrete. An additional 0.625 inch (1.59 cm) of stainless steel is provided by the canister shell for radial gamma shielding. The storage cask top shielding comprises 8 inches (20.32 cm) of stainless steel from the canister lids, 4.125 inches (10.48 cm) of carbon steel from the shield plug which encloses 1 inch (2.54 cm) of NS-4-FR and finally, 1.5 inches (3.81 cm) of carbon steel from the storage cask lid. The bottom of the storage cask rests on the concrete pad and is inaccessible. In the case of the storage cask inlets, some shielding is provided by the storage cask structural components. These are 2 inches (5.08 cm) of carbon steel from the pedestal plate, 1 inch (2.54 cm) of carbon steel from the cask base plate and 1 inch (2.54 cm) of stainless steel from the canister bottom plate.

The NAC-MPC transfer cask has an inside radius of 35.75 inches (90.81 cm) and has a multiwall radial shield design consisting of 0.75 inch (1.91 cm) of carbon steel, 3.5 inches (8.89 cm) of lead, 2 inches (5.08 cm) of a solid borated polymer (NS-4-FR), and 1.25 inches (3.18 cm) of carbon steel. Gamma shielding is provided by the steel and lead layers, and neutron shielding is provided primarily by the NS-4-FR. An additional 0.625 inch (1.59 cm) of stainless steel gamma shielding is provided by the canister shell. The transfer cask bottom shield design comprises carbon steel doors 9.50 inches (24.13 cm) thick. The top shielding of the transfer cask is provided by the 5-inch (12.70 cm) stainless steel shield lid and the 3-inch (7.62 cm) stainless steel structural lid. In addition, a 5-inch (12.70 cm) carbon steel temporary shield is used during welding, draining, drying and transfer operations. This temporary shielding is removed prior to storage.

#### 5.3.1.1 One-Dimensional Radial and Axial Shielding Models

Since the fuel assembly and basket features are not explicitly modeled in one-dimensional analysis, the fuel/basket interior is modeled as a set of homogenized material volumes based on equivalent cylindrical volumes. These volumes are defined by the areas created by: the central basket hole; the periphery of the basket tubes and the edge of the steel support disks; and by the elevations created by the basket heat transfer zone, the active fuel, the fuel assembly plenums and the fuel assembly end fittings.

The NAC-MPC fuel basket is divided into three radial regions: a central hole (void), fuel/basket region, and a basket/disk region. These regions have equivalent radii of 4.66, 30.63 and 34.51 inches (11.83, 77.80, and 87.66 cm), respectively. Axially, the basket is divided into seven regions: the top, middle and bottom fuel/basket regions; the upper and lower plenum/basket regions; and the upper and lower end fittings/basket regions. For the top models, the top fuel, top plenum and top end fitting regions have elevations with respect to the fuel midplane of 45.50, 49.01, and 56.99 inches (115.57, 124.49 and 144.75 cm), respectively. For the bottom models, the bottom fuel, bottom plenum and bottom end fitting regions have elevations with respect to the fuel midplane of 45.50, the fuel midplane of 45.50, 47.61, and 54.80 inches (115.57, 120.93 and 139.19 cm), respectively.

In each of these regions, the relevant fuel assembly material and any basket material present are homogenized. Basket materials include the steel support disks, aluminum heat transfer disks, top and bottom weldments, fuel tubes, BORAL sheets, and BORAL cover sheets. Fuel assembly materials include:  $UO_2$ , cladding, grids, plenum springs and spacers, and end-fittings. The resultant material and nuclide densities are described in Section 5.3.2.

The one-dimensional radial models of the storage cask and the transfer cask are based on the cylindrical representation of the fuel/basket source regions (previously described) surrounded by the explicit canister and cask radial shield dimensions. An axial buckling equal to the active fuel height of 91 inches (231.14 cm) is assumed for all radial models.

The one-dimensional top and bottom axial models of the storage and transfer casks are based on a slab representation of the fuel/basket axial regions covered by the explicitly modeled canister and storage cask axial shield regions. As previously stated, the one-dimensional axial model elevations are specified from the active fuel centerline, which SAS1 automatically establishes as

NAC-MPC FSAR	April 2000
Docket No. 72-1025	Revision 0

the reflecting boundary. Two models are utilized for each cask: one from the active fuel centerline to the top of the cask; and one from the active fuel centerline to the base plate of the cask. Two transverse bucklings equal to the fuel/basket zone equivalent diameter of 61.26 inches (155.6 cm) are assumed for both axial models.

### 5.3.1.2 <u>Three-Dimensional Top and Bottom Shielding Models</u>

SAS4 three-dimensional shielding analysis allows detailed modeling of the fuel assemblies, basket and cask shield configuration including streaming paths. Some fuel assembly and basket detail is homogenized to simplify model input and improve computational efficiency. Thus, the three-dimensional models maintain the equivalent fuel/basket source volumes developed for the one-dimensional models, but explicitly model the radial and axial extent of the source regions and the cask body details. As in the SAS1 models, the fuel and hardware source regions are homogenized within the volumes defined by the periphery of the basket tubes and by the elevations of the basket heat transfer zone, the active fuel, the plenum and the end fittings. Cask body details include the true axial extent of the cask shields as described by the drawings in Section 1.5, as well as radiation streaming paths such as the storage cask inlets and outlets and the canister vent and drain ports.

SAS4 requires cask model symmetry at the fuel midplane due to the nature of the automated biasing techniques employed and because dose rate tallies from the symmetric halves of the model are averaged together for computational efficiency. Thus, two models are created for each cask, a top and a bottom model. As in the SAS1 models, all three-dimensional shielding models are developed with respect to the fuel midplane as the origin.

The geometry of SAS4 is based on MARS (West) combinatorial geometry embedded in the MORSE code (Emmett). In this geometry, bodies such as cylinders and rectangular parallelepipeds are used to describe the extent of zones of material. Zones are volumes of constant material (cross sections) and are defined by logical operations on geometric bodies.

SAS4 employs an automated biasing technique for the MORSE Monte Carlo calculations based on either a radial or an axial XSDRNPM adjoint calculation. In the case of radial biasing, the adjoint calculation is performed for the radial shields and corresponding fuel/basket regions. In the case of axial biasing, the adjoint calculation is performed for the top or bottom shields and corresponding axial fuel/basket regions. Radial biasing is employed to improve the Monte Carlo computational efficiency and dose rate statistics on the sides of the cask. Axial biasing is

NAC-MPC FSAR	April 2000
Docket No. 72-1025	Revision 0

employed to improve Monte Carlo computational efficiency and dose rate statistics on the top or bottom surfaces of the cask. The dose rate profiles resulting from both radial and axial biasing calculations yield a complete dose profile of the entire cask with design basis fuel.

MORSE Monte Carlo calculations are performed for each type of source in each source region. In the case of the NAC-MPC basket and design basis fuel assembly configuration, this leads to eight source terms: middle fuel neutron, gamma and n-gamma; top or bottom fuel neutron, gamma and n-gamma; activated plenum hardware gamma and activated end fitting gamma. Twenty to thirty million histories (gamma and neutron combined) are tracked to yield a dose rate surface profile for each surface.

## 5.3.1.2.1 NAC-MPC Storage Cask Three-Dimensional Models

The three-dimensional top model of the NAC-MPC storage cask containing 36 design basis Yankee Class fuel assemblies is based on the homogenized cylindrical representation of the basket, and the following top features of the storage cask:

- Heat transfer annulus
- Carbon steel shell with four cutouts for outlet vents
- Concrete shield with four cutouts for outlet vents
- Four outlet vents including carbon steel lining
- Carbon steel shield plug
- Shield plug neutron shield
- Carbon steel top lid

Details on the elevations and radii used in creating the three-dimensional top model are taken directly from the drawings in Section 1.5. Elevations associated with the storage cask three-dimensional features are established with respect to the active fuel midplane of the Yankee Class fuel assembly (Figure 5.2-2) for the combinatorial model. The three-dimensional storage cask top model is shown in Figure 5.3-1. The MARS geometry requires 71 bodies (23 right circular cylinders and 48 rectangular parallelepipeds) to define 39 model zones with combinatorial logic.

NAC-MPC FSAR	April 2000
Docket No. 72-1025	Revision 0

The three-dimensional bottom model of the NAC-MPC storage cask is based on the homogenized cylindrical representation of the fuel/basket and the following bottom features of the storage cask:

- Heat transfer annulus
- Carbon steel shell with four cutouts for inlet vents
- Concrete shield with four cutouts for inlet vents
- Four inlet vents with carbon steel linings
- Carbon steel bottom base plate
- Carbon steel support stand with four cutouts for air flow
- Carbon steel shield ring
- Carbon steel storage cask bottom
- Concrete pad below base plate

The three-dimensional storage cask bottom model is shown in Figure 5.3-2. The MARS geometry requires 55 bodies (30 right circular cylinders and 25 rectangular parallelepipeds) to define 50 model zones with combinatorial logic.

## 5.3.1.2.2 NAC-MPC Transfer Cask Three-Dimensional Models

Several different three-dimensional models of the top portion of the transfer cask are used in the shielding evaluations. These include wet and dry cavity conditions as well as the corresponding shield lid and structural lid placement. The top configuration of the transfer cask is evaluated in detail for the welding, draining and drying operations. As with the storage cask models, top models of the NAC-MPC transfer cask containing 36 design basis Yankee fuel assemblies are based on a homogenized representation of the basket, but the rectangular periphery of the basket source region is modeled in order to more accurately estimate vent and drain port dose rates.

The basket disks outside the fuel/basket region are explicitly modeled to more accurately account for basket streaming. The following features of the transfer cask are considered:

- Vent and drain port openings in the shield lid
- Upper weldment shield ring
- Edge tapering and port cutouts in the temporary shielding
- Two lifting trunnions cut through the radial shield to the inner shell
- Lead and neutron shielding overlap at the top as shown on the transfer cask drawings

Details of the elevations and radii used in creating the three-dimensional top model are taken directly from the drawings in Section 1.5. As with the other three-dimensional models, elevations associated with the transfer cask three-dimensional features are established with respect to the active fuel midplane of the Yankee Class fuel assembly for the combinatorial geometry model. The three-dimensional transfer cask top model including shield and structural lid installation is shown in Figure 5.3-3. The MARS geometry requires 108 bodies (94 right circular cylinders and 14 rectangular parallelepipeds) to define 68 model zones.

The three-dimensional bottom model of the NAC-MPC transfer cask is based on the same homogenized representation of the fuel/basket as the top model. As with the top model of the transfer cask, the evaluations include both wet and dry canister cavity. The following bottom features of the transfer cask are considered:

- Termination of the radial shields at the bottom doors
- 9.5 inches of carbon steel shielding representing the bottom doors

The transfer cask bottom model is shown in Figure 5.3-4. The MARS geometry requires 69 bodies (56 right circular cylinders and 13 rectangular parallelepipeds) to define 59 model zones with combinatorial logic.

## 5.3.2 Shield Regional Densities

The SCALE 4.3 standard composition library (Landers) default compositions and isotopic distributions are used unless otherwise indicated. The composition densities before homogenization are:

<u>Material</u>	Density (g/cc)
UO ₂ at 95% TD	- 10.412
Zircaloy	- 6.56
H ₂ O	- 0.9982
Stainless Steel 304	- 7.92
Lead	- 11.344
Aluminum	- 2.702
BORAL (core)	- 2.623 (non-standard)
NS-4-FR	- 1.629 (non-standard)
Concrete	- 2.243 (based on 140 lb/ft ³ design spec. minimum)
Carbon Steel	- 7.821

Reinforcing steel is conservatively ignored in the concrete density. Basket and fuel assembly regional volumes are shown in Table 5.3-1 and the regional masses are shown in Table 5.3-2. The resultant regional homogenized densities for the design basis CE fuel assembly and shield densities are provided in Table 5.3-3.

April 2000 Revision 0

"Figure 5.3-1 NAC-MPC Storage Cask Three-Dimensional Top Model

Figure 5.3-2 NAC-MPC Storage Cask Three-Dimensional Bottom Model

Figure 5.3-3 NAC-MPC Transfer Cask Three-Dimensional Top Model Including Shield and Structural Lid

Figure 5.3-4 NAC-MPC Transfer Cask Three-Dimensional Bottom Model

Axial Zone ¹	Fuel/Basket Region	<b>Basket/Disk Region</b>
LEF	3.3927E+05	9.3029E+04
LPLM	9.9562E+04	2.7301E+04
FUELhot	5.8652E+05	1.6083E+05
FUELmid	2.9368E+06	8.0530E+05
FUELtop	7.7054E+05	2.1129E+05
UPLM	1.6562E+05	4.5415E+04
UEF	3.7654E+05	1.0325E+05

Table 5.3-1Fuel/Basket Regional Volumes (cm³)

1. See Figure 5.2-2 for zone locations.

Table 5.3-2	Fuel/Basket Regional Masses	(kg)	)
-------------	-----------------------------	------	---

	Fuel/Basket Region					Basket/Disk		
	Fi C	uel Assen Contribut	nbly ion	<b>Basket Contribution</b>			Region	
Axial Zone ¹	UO ₂	Zirc	SS 304	SS 304	Al	B ₄ C	SS 304	Al
LEF		0.00	186.48	141.37			51.24	
LPLM		61.01	62.28	42.55			51.24	
FUEL _{bot}	1335.47	359.43	0.00	173.01	36.46	5.90	102.47	
<b>FUEL</b> _{mid}	6687.02	1884.71	0.00	1004.41	320.40	29.55	768.52	235.15
FUELtop	1754.48	472.20	0.00	237.61	47.89	7.75	153.70	
UPLM		94.69	27.43	52.41			51.24	
UEF		0.0000	198.00	56.15			51.24	

1. See Figure 5.2-2 for zone locations.

## Table 5.3-3 Regional Homogenized Densities and Shield Densities

Zone/Material	Density (g/cc)	Nuclides	Density (atom/b-cm)
Middle Fuel Zone			
UO ₂	2.2769	²³⁴ U	2.793E-07
		²³⁵ U	3.656E-05
	1	²³⁸ U	5.041E-03
		0	1.016E-02
Zircaloy	0.6417	Zr	4.236E-03
SS 304	0.3420	Cr	7.526E-04
		Mn	7.498E-05
		Fe	2.563E-03
		Ni	3 334E-04
Aluminum	0.1091	Al	2 435E-03
B₄C	0.0101	¹⁰ B	8 764E-05
		¹¹ B	3.528E-04
	<u> </u>	C	1.101E-04
H ₂ O Wet Transfer Cask	0.6000	<u>с</u> н	1.101E-04
1120 thet Hunster Cusk	0.0000	<u>n</u>	3.021E.02
		- <u> </u>	5.02115-02
Middle Basket/Disk			
Zone			
SS 304	0.9543	Cr	2.100E-03
		Mn	2.092E-04
		Fe	7.152E-03
		Ni	9.303E-04
Aluminum	0.2920	Al	6.517E-03
H ₂ O Wet Transfer Cask	0.7700	Н	5.151E-02
·····		0	2.575E-02
Steel Shielding		······································	
SS 304	7.9200	Cr	1 743E-02
		Mn	1.745E-02
		Fe	5.9336-02
		Ni	7 721E-03
······································			1.1211-05
Carbon Steel Shielding			
Carbon Steel	7.8212	Fe	8.350E-02
		C	3 925E-03
Concrete Shielding			5.5251-05
Concrete	2.2430	Fe	3 386F-04
		<del>H</del>	1 340F-02
		A1	1.340E-02
		Ca	1.702E-05
		0	1.403E-03
		Na	1 70/E 02
······································		N	1.704E-03
		11	1.0212-02

Material/Zone	Density (g/cc)	Nuclides	Density (atom/b-cm)
Top Fuel/Basket Zone			
UO ₂	2.2769	²³⁴ U	2.793E-07
		²³⁵ U	3.656E-05
		²³⁸ U	5.041E-03
		0	1.016E-02
Zircaloy	0.6128	Zr	4.046E-03
SS 304	0.3084	Cr	6.787E-04
		Mn	6.761E-05
		Fe	2.311E-03
		Ni	3.006E-04
Aluminum	0.0622	Al	1.388E-03
B₄C	0.0101	¹⁰ B	8.764E-05
	1	¹¹ B	3.528E-04
		С	1.101E-04
H ₂ O Transfer Cask Wet	0.6303	Н	4.214E-02
		0	3.123E-02
Top Plenum Zone			
Zircalov	0.5718	Zr	3.775E-03
SS 304	0.4821	Cr	1.485E-03
		Mn	1.057E-04
		Fe	3.613E-03
		Ni	4.700E-04
H ₂ O Transfer Cask Wet	0.7019	Н	4.695E-02
		0	2.348E-02
Top End Fitting Zone			·····
SS 304	0.6749	Cr	1.743E-02
		Mn	1.480E-04
		Fe	5.058E-03
		Ni	6.579E-04
H ₂ O Transfer Cask Wet	0.9132	Н	6.108E-02
	-	0	3.054E-02
Neutron Shield			
NS-4-FR	1.6291	¹⁰ B	8.553E-05
		¹¹ B	3.422E-04
		Al	7.763E-03
		Н	5.854E-02
		0	2.609E-02
		N	1.394E-03
*********		С	2.264E-02

## Table 5.3-3 Regional Homogenized Densities and Shield Densities (Continued)

Material/Zone	Density (g/cc)	Nuclides	Density (atom/b-cm)
Bottom Fuel/Basket			
Zone			
UO ₂	2.2769	²³⁴ U	2.793E-07
		²³⁵ U	3.656E-05
		²³⁸ U	5.041E-03
		0	1.016E-02
Zircaloy	0.6128	Zr	4.046E-03
SS 304	0.2350	Cr	6.492E-04
		Mn	6.467E-05
		Fe	2.211E-03
		Ni	2.876E-04
Aluminum	0.0622	Al	1.388E-03
B ₄ C	0.0101	¹⁰ B	8.764E-05
		¹¹ B	3.528E-04
		С	1.101E-04
H ₂ O Transfer Cask Wet	0.6322	Н	4.228E-02
		0	3.130E-02
<b>Bottom Plenum Zone</b>			
Zircaloy	0.6128	Zr	4.046E-03
SS 304	1.0529	Cr	2.317E-03
		Mn	2.308E-03
		Fe	7.891E-03
		Ni	1.026E-03
H ₂ O Transfer Cask Wet	0.5447	Н	3.644E-02
		0	1.822E-02
Bottom End fitting			
Zone			
<u>SS 304</u>	0.9664	Cr	2.127E-03
		Mn	2.119E-04
· · · · · · · · · · · · · · · · · · ·		Fe	7.243E-03
		Ni	9.421E-04
H ₂ O Transfer Cask Wet	0.8764	Н	5.862E-02
		0	2.931E-02

## Table 5.3-3 Regional Homogenized Densities and Shield Densities (Continued)

### 5.4 <u>Shielding Evaluation</u>

This section evaluates the shielding design of NAC-MPC transfer and storage casks. The calculational methods are described. Shielding calculations are performed with design basis Yankee Class fuel source terms at 36,000 MWD/MTU and 8 years cooling time. Dose rate profiles are reported as a function of distance from the sides, top, and bottom of the NAC-MPC storage and transfer casks. Storage cask shielded source terms (neutron and gamma fluxes at the cask surface) are provided for ISFSI controlled area boundary dose evaluations.

### 5.4.1 <u>Calculational Methods</u>

Shielding evaluations of the transfer and storage casks are performed with SCALE 4.3 for the PC. In particular, SCALE shielding analysis sequence SAS2H is used to generate source terms for the design basis fuel. SAS1 is used to perform one-dimensional radial and axial shielding analysis, and a modified version of SAS4 is used to perform three-dimensional shielding analysis. The coupled 27 group neutron, 18 group gamma ENDF/B-IV (27N-18COUPLE) cross section library is used in all shielding evaluations. Source terms include fuel neutron, fuel gamma, and gamma contributions from activated hardware. Dose rate evaluations include the effect of fuel burnup peaking on fuel neutron and gamma source terms. The SCALE shielding analysis sequences and cross section libraries recently have been benchmarked to measurements of light water reactor fuel source terms, shielding material dose rate attenuation, and spent fuel storage and transport cask dose rates (Broadhead).

The SAS2H code sequence is used to generate source terms for the Yankee Class design basis fuel. SAS2H includes an XSDRNPM neutronics model of the fuel assembly and ORIGEN-S fuel depletion/source terms calculations. The 27 energy group ENDF/B-IV neutron cross section library, 27GROUPNDF4, is used in the source terms calculations. Source terms are generated for both  $UO_2$  fuel and fuel assembly hardware. The hardware activation is calculated by ORIGEN-S using the incore neutron flux spectrum produced by the SAS2H neutronics model. The hardware is assumed to be Type 304 stainless steel with 1.2 g/kg ⁵⁹Co impurity. The effects of axial flux spectrum and magnitude variation on hardware activation are estimated by flux ratios based on empirical data (Luskic).

Both the one-dimensional SAS1 and the three-dimensional SAS4 shielding models are used in the evaluations of the NAC-MPC storage system. The SAS1 radial and axial models are used to estimate the peak and average dose rates on the sides, top, and bottom of the storage and transfer casks. The models represent the cask as either semi-infinite cylinders or slabs. The method of solution is XSDRNPM discrete ordinates. Bucklings are applied to the one-dimensional models to account for transverse leakage. One-dimensional analysis also serves as a cross check of the three-dimensional model results and is employed to establish the minimum cool time for the loading of 32,000 MWD/MTU burnup assemblies and the Exxon 36,000 MWD/MTU assemblies.

The SAS4 shielding models are used to estimate the dose profiles along the surfaces of the transfer and storage casks and to estimate doses in and around streaming paths such as the storage cask inlets and outlets, and the canister vent and drain ports. The SAS4 models represent the cask body and any streaming paths with combinatorial logic. The method of solution is adjoint discrete ordinates and Monte Carlo using the XSDRNPM and MORSE codes, respectively. Since SAS4 requires model symmetry at the fuel midplane, two models are created for each cask, a top and a bottom model. Radial biasing is performed to estimate dose rate on the sides of the cask. Modifications are made to SAS4 to determine dose rates all along the radial, top and bottom surfaces of the cask as well as any cylindrical surface surrounding the cask. Thus, detailed dose profiles are determined that explicitly show peaks due to the fuel burnup profile, activated hardware gamma emission and any streaming paths.

In both the SAS1 and SAS4 models, the fuel and hardware source regions are homogenized within the volumes described by the periphery of the basket tubes, and defined by the fuel assembly active fuel, plenum, and end fitting elevations. Within these volumes, the material masses of the fuel assembly and basket are preserved.

## 5.4.2 <u>Flux-to-Dose Rate Conversion Factors</u>

The ANSI/ANS 6.1.1-1977 flux-to-dose rate conversion factors are used in all NAC-MPC shielding evaluations. These factors are default for SCALE 4.3. Tables 5.4-1 and 5.4-2 show the group flux-to-dose rate factors associated with the coupled 27 group neutron and 18 group gamma cross section library used in the shielding evaluations.

### 5.4.3 Dose Rates

This section provides detailed dose rate profiles for the NAC-MPC storage and transfer cask based on the source terms presented in Section 5.2. Design basis fuel source terms include contributions from fuel neutron, fuel gamma and activated hardware gamma. The fuel assembly activated hardware gamma source terms include: steel and inconel in the upper and lower fuel assembly end fittings, and upper and lower fuel rod plenum hardware. Peaking factors of 1.15 and 1.80 are applied to the one-dimensional radial fuel gamma and neutron dose rates, respectively. The three-dimensional model dose rates include the axial profiles for neutron and gamma source distributions shown in Figure 5.2-4.

#### 5.4.3.1 <u>One-Dimensional Storage Cask Dose Rates</u>

One-dimensional radial dose rates with design basis fuel were found to be in good agreement with the three-dimensional models at the radial midplane. However, the peaks in the radial dose rates, due to activated endfittings, cannot be captured by one-dimensional analysis. One-dimensional dose rates at the top of the storage cask are significantly lower than those calculated using three-dimensional analysis. This is primarily due to the neutron component of the dose rates and the transverse bucklings applied in the one-dimensional axial model as well as streaming effects caused by the heat transfer annulus and top vents. Except for the neutron component of the top axial model and obvious limitations in geometry, one-dimensional analysis is found to support the results of the more complicated three-dimensional models. Except for the storage cask loss of concrete accident radial dose rates shown in Table 5.4-3, the dose rate results from three-dimensional analysis are reported.

One-dimensional radial surface dose rates are also used to determine the minimum cool times, based on the design basis fuel values, for CE, UN, and WE fuel types with a maximum burnup of 32,000 MWD/MTU and Exxon assemblies at 36,000 MWD/MTU. The calculated minimum cool times for CE, UN, and WE fuel at 32,000 MWD/MTU are 8, 13 and 21 years, respectively. Exxon assemblies with steel hardware require 16 years cooling while the Exxon assemblies with Zircaloy hardware require only 9 years cooling.

### 5.4.3.2 <u>Three-Dimensional Storage Cask Dose Rates</u>

The NAC-MPC storage cask three-dimensional model dose rates are presented in Figures 5.4-1 through 5.4-7. Approximately 50 million particle histories (neutron and gamma) are tracked to yield the dose rate profiles presented in theses figures. The average standard deviation for the side total dose rate shown in Figures 5.4-1 and 5.4-2 is less than  $\pm 2\%$ , and the average standard deviation for the top total dose rate shown in Figures 5.4-6 and 5.4-7 is less than  $\pm 5\%$ . The average standard deviation for the inlet and outlet vent total dose rates shown in Figures 5.4-3, 5.4-4 and 5.4-5 is less than  $\pm 5\%$ , and the standard deviation for the peak dose rates at the vent opening are less than  $\pm 10\%$ .

The vertical profile along the radial surface of the storage cask, as well as at distances of 30.48 cm (1 foot), 1 meter, and 2 meters from it, are plotted in Figure 5.4-1 as a function of elevation. Each datum represents the circumferentially average dose rate at the corresponding distance and elevations. The negative elevations are the dose rates from the bottom model computations, while the positive elevations are the dose rates from the top model computations. The discontinuity observed at zero elevation (midplane of the fuel) is a modeling artifact due to the decoupling of the upper and lower portions of the cask. In the vertical dose profile, peaking is observed at the upper and lower end fitting locations, as well as at the locations of the lower intake and upper outlet vents. The average and maximum side surface dose rate for the storage cask are 37 (0.3%) and 47.3 (0.4%) mrem/hr, respectively.

The radial surface dose profile is further described by source component in Figure 5.4-2. The source components in both models contribute to the radial doses largely as one would expect, i.e., at the elevations where they are located. Since these doses are circumferential averages, the detailed circumferential dose rate profile at the top vent elevation and the bottom vent inlet are shown radially in Figure 5.4-3 and Figure 5.4-4, respectively. The dose rates shown in Figures 5.4-3 and 5.4-4 were computed using a variance-weighted average of the dose rates in the four symmetric quadrants at the vent elevation. A maximum dose rate of 24 mrem/hr (5%) is calculated at the surface of the outlet vent and a maximum dose rate of 99 (5.4%) mrem/hr was calculated at the entrance of the inlet vent.

In Figure 5.4-5, the circumferential dose rate profile at the support ring cutout elevation is shown on the storage surface and at distances of 30.48 cm (1 foot), 1 meter, and 2 meters from the

surface. The peak in the circumferential dose rate is not at the location of the cutout, but above the inlet vent location. Note that these peak dose rates are higher at 1 foot from the storage cask than they are on the surface of the storage cask. This is due to photon scattering off the storage cask concrete base through the inlet vent opening and up to the cutout elevation.

The dose rate profiles on the top surface of the storage cask and at distances of 1 foot and 1 meter above the lid are shown in Figure 5.4-6. The dose rates are plotted radially out from the centerline of the storage cask. Two dose rate peaks are observed on the storage cask top surface: one in the vicinity of 90 to 100 cm, which corresponds to the location of the heat transfer annular gap and another at approximately 130 cm. The dose rate profile on the top surface of the storage cask is shown by source component in Figure 5.4-7 and indicates that the peak dose rate above the annular gap is caused by neutrons streaming up the gap. The component profile also indicates that the second peak is created by gammas from the end fitting, top fuel, and top plenum source regions. This peak occurs at approximately the same radial location as the vertical leg in the upper outlet vent. Thus, it is a result of a decrease in effective shield thickness caused by the void in the concrete due to the outlet vents. The average dose rate over the top of the storage cask is 54 mrem/hr (4.9%).

### 5.4.3.3 One-Dimensional Transfer Cask Dose Rates

One-dimensional radial dose rates with design basis fuel are in good agreement with the threedimensional models at the radial midplane. As with the storage cask one-dimensional radial model, the peaks in the radial dose rates due to activated endfittings cannot be captured by onedimensional analysis. One-dimensional top dose rates at the top and bottom of the transfer cask were significantly lower than three-dimensional analysis. This was primarily due to the neutron component of the dose rates and the transverse bucklings applied in the one-dimensional axial model as well as streaming effects around the temporary shielding. Except for the neutron component of the top axial model, one-dimensional analysis supports the results of the more complicated three-dimensional models.

## NAC-MPC FSAR Docket No. 72-1025

## 5.4.3.4 <u>Three-Dimensional Transfer Cask Dose Rates</u>

The transfer cask three-dimensional model dose rates are presented in Figures 5.4-8 through 5.4-15. Approximately 100 million particle histories (neutron and gamma) are tracked to yield the dose rate profiles presented in theses figures. The average standard deviation for the side total dose rates shown in Figures 5.4-8 through 5.4-11 is less than  $\pm 2\%$ , and the average standard deviation for the top total dose rates shown in Figures 5.4-12 through 5.4-15 is less than  $\pm 2\%$ .

The transfer cask side dose rate profiles with a wet cavity are shown in Figure 5.4-8 as a function of distance and in Figure 5.4-9 as a function of source component. In this condition, the majority of the dose rate is from fuel gamma and activated end fitting gamma. It is assumed in the model that the water level in the canister is lowered for welding operations, thus, the top end fitting is uncovered and causes a large peak in dose rate at the top of the transfer cask due to the gamma source from the activated top end fitting. In this condition, the peak and average dose rates on the side of the transfer cask are 210.2 (0.8%) and 79.5 (0.3%) mrem/hr, respectively, and the peak and average dose rates at 1 meter are 40.5 (0.7%) and 26.4 (0.2 %) mrem/hr, respectively.

The transfer cask side dose rate profiles with a dry cavity are shown in Figure 5.4-10 as a function of distance and in Figure 5.4-11 as a function of source component. In this condition, the majority of the dose rate is from fuel neutron and gamma source, but significant peaks are shown from the activated end fittings. The peak and average dose rates on the side of the transfer cask are 413.4 (1.5%) and 226.3 (0.2%) mrem/hr, respectively, and the peak and average dose rates at 1 meter are 103.4 (0.6%) and 72.2 (0.2%) mrem/hr, respectively.

The transfer cask peak and average dose rate on the temporary shield surface are 188.7 (1.1%) and 172.0 (0.3%) mrem/hr, respectively. In this condition, the majority of the dose rate is from the activated top end fitting. The peak and average dose rate at 1 meter are 389.1 (5.1%) and 263.7 (1.5%) mrem/hr, respectively.

In the final configuration, the canister cavity is dry, the shield lid and structural lid are in place, and 5" of temporary steel shielding is installed. In this condition, the transfer cask top dose rate are shown in Figure 5.4-12 as a function of distance and in Figure 5.4-13 as a function of component. The majority of the dose rate is from the fuel neutron. The dose rate peaks at the lid edge due to gamma streaming around the tapered edge of the temporary shield. The peak and average dose rates on the top of the transfer cask are 358.9 (2.6%) and 224.6 (0.9%) mrem/hr,
respectively, and the peak and average dose rates at 1 meter are 39.6 (4.5%) and 34.2 (1.3 %) mrem/hr, respectively.

The transfer cask bottom dose rate profiles with the cavity wet and dry are shown in Figures 5.4-14 and 5.4-15, respectively. In the wet cavity situation, the peak and average dose rates on the bottom of the transfer cask are 77.2 (0.7%) and 55.9 (0.2%) mrem/hr, respectively, and the peak and average dose rates at 1 meter are 19.0 (2.1%) and 12.2 (0.3%) mrem/hr, respectively. In the dry cavity situation, the peak and average dose rates on the bottom of the transfer cask are 398.0 (3.9%) and 194.7 (0.3%) mrem/hr, respectively, and the peak and average dose rates at 1 meter are 66.7 (3.6%) and 28.2 (0.4%) mrem/hr, respectively.

#### 5.4.4 Storage Cask Shielded Source Terms

The storage cask shielded source terms are provided in this section for use in the ISFSI controlled area boundary dose evaluations. These shielded source terms are the neutron and gamma fluxes at the surface of the NAC-MPC storage cask due to the neutron and gamma sources specified in Section 5.2. The cask surface fluxes are obtained from one-dimensional SAS1 radial and axial shielding evaluations. These fluxes are in the 27 group neutron and 18 group gamma energy group structure consistent with the SCALE 4.3 27N-28COUPLE cross section library. The group wise fluxes are listed in Tables 5.4-11 through 5.4-13 for the side and the top of the storage cask. At the bottom of each column is the total source strength for use in SKYSHINE-III. This source strength, in the case of the radial component, is based on the surface area of the surface area of the top lid. The total source strengths and spectra are used in the SKYSHINE-III direct and air-scatter dose evaluations presented in Chapter 10 for an array of NAC-MPC storage casks at an ISFSI.



Figure 5.4-1 Storage Cask Radial Dose Rate Profile

NAC-MPC FSAR	April 2000
Docket No. 72-1025	Revision 0











## Figure 5.4-4 Storage Cask Bottom Inlet Vent Dose Rate Profile

# Figure 5.4-5 Storage Cask Dose Rate Profile at Stand Cutout Elevation



NAC-MPC FSAR		April 2000
Docket No. 72-1025		 Revision 0

Figure 5.4-6 Storage Cask Top Dose Rate Profile as a Function of Radius from the Centerline and Distance from Surface





# Figure 5.4-7 Storage Cask Top Surface Dose Rate Profile by Source Component

NAC-MPC FSAR	April 2000
Docket No. 72-1025	Revision 0

Figure 5.4-8 Transfer Cask Side Dose Rate Profile as a Function of Elevation and Distance, Wet Cavity







Note: Peak in nozzle dose at 135 cm is due to draining 50 gallons of water which uncovers the nozzle.









NAC-MPC FSAR	April 2000
Docket No. 72-1025	Revision 0

Figure 5.4-12 Transfer Cask Top Surface Dose Rate as a Function of Radius and Distance from Surface, Shield Lid, Structural Lid, and Temporary Shield On, Dry Cavity



NAC-MPC FSAR	April 2000
Docket No. 72-1025	Revision 0

Figure 5.4-13 Transfer Cask Top Surface Dose Rate by Source Component, Shield Lid, Structural Lid, and Temporary Shield On, Dry Cavity











Table 5.4-1	ANSI Standard Neutron Flux-To-Dose Rate Factors
-------------	-------------------------------------------------

Group	(Rem/Hr)/(N/cm ² /Sec)
1	1.49160E-04
2	1.44640E-04
3	1.27010E-04
4	1.28110E-04
5	1.29770E-04
6	1.02810E-04
7	5.11830E-05
8	1.23189E-05
9	3.83650E-06
10	3.72469E-06
11	4.01500E-06
12	4.29259E-06
13	4.47439E-06
14	4.56760E-06
15	4.55809E-06
16	4.51850E-06
17	4.48790E-06
18	4.46649E-06
19	4.43450E-06
20	4.32709E-06
21	4.19750E-06
22	4.09759E-06
23	3.83900E-06
24	3.67480E-06
25	3.67480E-06
26	3.67480E-06
27	3.67480E-06

Group	(Rem/Hr)/(N/cm ² /Sec)
1	8.77160E-06
2	7.47849E-06
3	6.37479E-06
4	5.41360E-06
5	4.62209E-06
6	3.95960E-06
7	3.46860E-06
8	3.01920E-06
9	2.62759E-06
10	2.20510E-06
11	1.83260E-06
12	1.52280E-06
13	1.17250E-06
14	8.75940E-07
15	6.30610E-07
16	3.83380E-07
17	2.66930E-07
18	9.34720E-07

## Table 5.4-2 ANSI Standard Gamma Flux-To-Dose Rate Factors

 Table 5.4-3
 NAC-MPC Storage Cask One-Dimensional Projectile Accident Radial Dose

 Rates

Source	Surface (mrem/hr)	1 meter (mrem/hr)
Fuel Neutron	4.1	1.5
Fuel Gamma	306.0	136.0
Fuel N-Gamma	4.5	1.8
Total	315	139

1

·····			
	Radial Gamma-Ray	Axial Hardware	Axial Fuel Gamma-
Energy Group	Group Flux	Gamma-Ray Group	Ray Group Flux
Boundary (MeV)	(y/sec/cm ² )	Flux (γ/sec/cm ² )	$(\gamma/\text{sec/cm}^2)$
8.0-10.0	1.76E+00	0.00E+00	9.26E-01
6.5-8.0	1.36E+01	0.00E+00	8.63E+00
5.0-6.5	1.71E+01	0.00E+00	4.04E+00
4.0-5.0	1.98E+01	0.00E+00	3.09E+00
3.0-4.0	2.72E+01	4.52E-24	3.74E+00
2.5-3.0	1.77E+01	5.27E-05	2.59E+00
2.0-2.5	9.40E+01	1.73E-02	5.37E+00
1.66-2.0	8.15E+01	1.39E-02	4.90E+00
1.33-1.66	4.65E+02	8.23E+01	1.45E+01
1.0-1.33	1.17E+03	2.05E+02	3.48E+01
0.8-1.0	1.56E+03	1.81E+02	3.26E+01
0.6-0.8	3.10E+03	2.48E+02	4.89E+01
0.4-0.6	5.53E+03	3.42E+02	7.96E+01
0.3-0.4	4.01E+03	2.06E+02	4.83E+01
0.2-0.3	5.63E+03	2.18E+02	5.20E+01
0.1-0.2	1.38E+04	1.37E+02	3.19E+01
0.05-0.1	4.62E+03	3.48E+00	8.14E-01
0.01-0.05	1.75E+01	5.24E-03	5.39E-03
Total Group Flux	4.02E+04	1.62E+03	3.77E+02
Total Source Strength			
(y/sec)	$1.18E+10^{1}$	$1.34E+08^{2}$	$3.12E+07^2$

Table 5.4-4 INAC-IMPC Shielded Galillia Flux	Table 5.4-4	NAC-MPC Shielded Gamma	Flux
----------------------------------------------	-------------	------------------------	------

1. Total radial source is total flux multiplied by the radial cask area (2.905 x  $10^5$  cm²).

2. Total axial source is total flux multiplied by cask axial surface area (8.301 x  $10^4$  cm²).

### Table 5.4-5 NAC-MPC Shielded Neutron Flux

	Total Radial	<b>Total Axial Neutron</b>
Energy Group	Neutron Group Flux	<b>Group Flux</b>
Boundary (MeV)	(n/sec/cm ² )	(n/sec/cm ² )
6.43 - 20.0	3.82E-02	4.25E-02
3.0 - 6.43	1.42E-01	1.79E-01
1.85 - 3.0	3.42E-01	6.21E-01
1.4 - 1.85	1.71E-01	8.13E-01
0.9 - 1.4	1.56E-01	5.60E+00
0.4 - 0.9	3.51E-01	3.94E+01
0.1 -0.4	3.98E-01	7.47E+01
1.7E-02 - 0.1	3.62E-01	5.15E+01
3.0E-03 - 1.7E-02	2.85E-01	2.78E+01
5.5E-04 - 3.0E-03	3.25E-01	1.44E+01
1.0E-04 - 5.5E-04	4.04E-01	1.46E+01
3.0E-05 -1.0E-04	3.23E-01	9.36E+00
1.0E-05 - 3.0E-05	3.43E-01	7.76E+00
3.05E-06 - 1.0E-05	4.09E-01	7.06E+00
1.77E-06 - 3.05E-06	2.08E-01	3.00E+00
1.3E-06 -1.77E-06	1.28E-01	1.55E+00
1.13E-06 - 1.3E-06	6.01E-02	6.65E-01
1.0E-06 - 1.13E-06	5.37E-02	5.49E-01
8.0E-07 - 1.0E-06	1.00E-01	9.59E-01
4.0E-07 - 8.0E-07	3.62E-01	2.69E+00
3.25E-07 - 4.0E-07	1.35E-01	6.64E-01
2.25E-07 - 3.25E-07	5.56E-01	1.00E+00
1.0E-07 - 2.25E-07	6.37E+00	1.67E+00
5.0E-08 - 1.0E-07	1.37E+01	1.05E+00
3.0E-08 - 5.0E-08	9.49E+00	4.35E-01
1.0E-08 - 3.0E-08	8.35E+00	2.26E-01
1.0E-10 - 1.0E-08	1.65E+00	3.11E-02
Total Group Flux	4.52E+01	2.68E+02
Total Source Strength		
(n/sec)	$1.31E+07^{1}$	2.223E+07 ²

Total radial source is total flux multiplied by radial cask area of 2.905 x 10⁵ cm².
 Total axial source is total flux multiplied by cask axial surface area of 8.301 x 10⁴ cm².

#### 5.5 <u>References</u>

Broadhead, B.L., Tang, J.S., Childs, R.L., Parks, C.V., and Taniuchi, H., "Evaluation of Shielding Analysis Methods in Spent Fuel Cask Environment," EPRI TR-104329, May 1995.

Buckholz, J. A., "XSDOSE: A Module for Calculating Fluxes and Dose Rates at Points Outside A Shield," ORNL/NUREG/CSD-2/V3/R5, September 1995.

Emmett, M.E., "PICTURE A Printer Plot Package for Making 2-D Pictures of MARS Geometries," ORNL/NUREG/CSD-2/V3/R5, September 1995.

Emmett, M.B., J.T. West, T.J. Hoffman, and W.C. Jordan, "MORSE-SGC for the SCALE System," "Scale Cross-Section Libraries," ORNL/NUREG/CSD-2/V3/R5, Section M4.A2, September 1995.

Greene, N. M. and L. M. Petrie, "XSDRNPM: A One-Dimensional Discrete Ordinates Code for Transport Analysis," ORNL/NUREG/CSD-2/V2/R5, September 1995.

Herman, O. W. and C. V. Parks, "SAS2H: A Coupled One-Dimensional Depletion and Shielding Analysis Module," ORNL/NUREG/CSD-2/V2/R5, September 1995.

Jordan, W. C., "Scale Cross-Section Libraries," ORNL/NUREG/CSD-2/V3/R5, Section M4.A2, September 1995.

Knight, J. R., et al., "SAS1: A One-Dimensional Shielding Analysis Module," ORNL/NUREG/ CSD-2/V2/R5, Volume 1, Section S1, September 1995.

Landers, N. F., J. A. Bucholz, L. M. Petrie, "The Material Information Processor For Scale," ORNL/NUREG/CSD-2/R5, Section M7, September 1995.

Luskic, A., et al., "Spent Fuel Hardware Characterization and 10CFR61 Classification for Waste Disposal," PNL-6906, Volume 1, June 1989.

ORNL, "SCALE 4.3: Modular Code System for Performing Standardized Computer Analyses for Licensing Evaluation for Workstations and Personal Computers," CCC-545, September. 1995.

Tang, J. S., "SAS4: A Monte Carlo Cask Shielding Analysis Module Using an Automated Biasing Procedure," ORNL/NUREG/CSD-2/V2/R5, Volume 1, Section S4, September 1995.

West, J. T., and M. B. Emmett, "MARS: A Multiple-Array System Using Combinatorial Geometry," ORNL/NUREG/CSD-2/V3/R5, Volume 3, Section M9, September 1995.

Westfall, R. M. and O. W. Herman, "ORIGEN-S: SCALE System Module to Calculate Fuel Depletion, Actinide Transmutation, Fission Product Buildup and Decay, and Associated Radiation Source Terms," ORNL/NUREG/CSD-2/V2/R5, September 1995.